

Monitoring and management of mother and fetus at risk

Citation for published version (APA): Moors, S. (2020). Monitoring and management of mother and fetus at risk. [Phd Thesis 1 (Research TU/e / Graduation TU/e), Electrical Engineering]. Technische Universiteit Eindhoven.

Document status and date: Published: 15/12/2020

Document Version:

Publisher's PDF, also known as Version of Record (includes final page, issue and volume numbers)

Please check the document version of this publication:

• A submitted manuscript is the version of the article upon submission and before peer-review. There can be important differences between the submitted version and the official published version of record. People interested in the research are advised to contact the author for the final version of the publication, or visit the DOI to the publisher's website.

• The final author version and the galley proof are versions of the publication after peer review.

• The final published version features the final layout of the paper including the volume, issue and page numbers.

Link to publication

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- · Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
- You may not further distribute the material or use it for any profit-making activity or commercial gain
 You may freely distribute the URL identifying the publication in the public portal.

If the publication is distributed under the terms of Article 25fa of the Dutch Copyright Act, indicated by the "Taverne" license above, please follow below link for the End User Agreement:

www.tue.nl/taverne

Take down policy

If you believe that this document breaches copyright please contact us at:

openaccess@tue.nl

providing details and we will investigate your claim.

MONITORING AND MANAGEMENT OF MOTHER AND FETUS AT RISK

Suzanne Moors

Monitoring and management of mother and fetus at risk

PROEFSCHRIFT

ter verkrijging van de graad van doctor aan de Technische Universiteit Eindhoven, op gezag van de rector magnificus prof.dr.ir. F.P.T. Baaijens, voor een commissie aangewezen door het College voor Promoties, in het openbaar te verdedigen op dinsdag 15 december 2020 om 11:00 uur

© Copyright 2020, Suzanne Moors All rights reserved. No part of this book may be reproduced in any form by any means, without prior permission of the author.

A catalogue record is available from the Eindhoven University of Technology Library.

ISBN: 978-94-6416-233-2 Cover image: Image from Etsy.com (LotusArt), edited by Hildegard Hick, Hick Fotografie Layout: Daniëlle Balk, persoonlijkproefschrift.nl Printed by: Ridderprint BV



Financial support for this thesis has been kindly provided by: BrideaMedical, Chipsoft BV, Gedeon Richter Benelux, GrafiMedics BV, Huisartsenpraktijk Lidwina, ICT Healthcare Technology Solutions, MedSim, MáximaMC, Nemo Healthcare, Stöpler Medical BV, Technische Universiteit Eindhoven. door

Suzanne Moors

geboren te Geldrop

Dit proefschrift is goedgekeurd door de promotoren en de samenstelling van de promotiecommissie is als volgt:

Voorzitter	prof.dr.ir. P.M.J. van den Hof
Promotor Copromotoren	prof.dr. S.G. Oei dr.ir. M.B. van der Hout-van der Jagt dr. J.O.E.H. van Laar
Leden	prof.dr. R.M. Aarts prof.dr. W.P. de Boode (<i>Radboud Universiteit</i>) prof.dr. L.R.C. Dekker prof.dr. M.E.A. Spaanderman (<i>Radboud Universiteit en</i> <i>Universiteit Maastricht</i>)

Het onderzoek dat in dit proefschrift wordt beschreven is uitgevoerd in overeenstemming met de TU/e Gedragscode Wetenschapsbeoefening.

TABLE OF CONTENTS

Chapter 1	General introduction	7
Chapter 2	Speckle tracking echocardiography in hypertensive pregnancy disorders: a systematic review. Obstetrical & Gynecological Survey 2020;75(8):497-509.	29
Chapter 3	Heart rate variability in hypertensive pregnancy disorders: a systematic review. Pregnancy Hypertension: An International Journal of Women's Cardiovascular Health. 20C (2020): 56-68.	57
Chapter 4	Changes in maternal heart rate variability in response to the ad- ministration of routine obstetric medications in hospitalized patients; study protocol for a cohort study (MAMA-heart study). <i>Submitted</i> .	95
Chapter 5	Management of intrapartum fetal distress in The Netherlands: a clinical practice survey. European Journal of Obstetrics and Gynecology and Reproductive Biology. 2016;205:48-53.	115
Chapter 6	Intrauterine resuscitation during term labor by maternal hyperoxy- genation: a randomized controlled trial .(INTEREST O2). <i>Trials</i> . 2018;19:195.	131
Chapter 7	The effect of intrauterine resuscitation by maternal hyperoxygenation; a randomized controlled trial. American Journal of Obstetrics & Gynecology MFM. 2020;2:100102.	153
Chapter 8	A randomized controlled trial studying the effect of maternal hyper- oxygenation on fetal heart rate in case of fetal distress. Physiol Meas. 2020 Oct 13. doi: 10.1088/1361-6579/abc0b6. Online ahead of print.	175
Chapter 9	General discussion and future perspectives	203
Chapter 10	English summary	221
Chapter 11	Nederlandse samenvatting	233
Chapter 12	Appendices List of abbreviations List of publications Dankwoord Curriculum Vitae	247 248 250 252 255





General introduction and outline of this thesis

GENERAL INTRODUCTION

There is no moment in the human life cycle when two individuals are more closely physiologically intertwined than during pregnancy. This period thus represents a unique time window where the health of the mother and her baby are very closely related.¹

Maternal hemodynamics undergo several physiological adaptations during pregnancy to provide for the increased metabolic demands of the mother, as well as to ensure adequate uteroplacental blood supply, nutrient delivery and oxygenation to the fetus for fetal development and growth.²⁻⁴ These maternal hemodynamic changes are mediated through the autonomic nervous system (ANS) and the renin-angiotensin-aldosterone system, and begin as early as four to five weeks of gestation.⁴

Early in pregnancy, the total peripheral vascular resistance reduces.⁵ This triggers several compensatory mechanisms.⁵ The plasma volume is expended progressively throughout pregnancy by approximately 1100-1500 ml,^{6,7} most of this increase has occurred by 34 weeks' gestation.^{3,7} Moreover, the maternal cardiac output increases by both an increase in stroke volume and heart rate.^{2,3}

These alterations are important to maintain a relatively uniform blood pressure during pregnancy. Besides a mild reduction in blood pressure in the second trimester,⁴ the blood pressure can remain stable because the decrease in systemic vascular resistance is matched in magnitude by the increase in cardiac output.^{2,8} A stable blood pressure is important for the perfusion of organs.

Along with these hemodynamic changes, the pregnancy also causes anatomical changes through remodeling of the heart and blood vessels.^{2,8} Pregnancy-mediated changes in the heart musculature include an increase in left atrial diameter, left ventricular (LV) wall mass, and LV thickness.⁸ As early hemodynamic and cardiac changes are very important for a healthy pregnancy, consequently, impairment of these adaptation mechanisms can cause serious fetal and maternal complications.⁹

Hypertensive pregnancy disorders (HPD) are an example of pregnancy complications attributed to inadequate hemodynamic adaptations during pregnancy, probably due to an inability to cope with the vascular load of pregnancy.⁶ Moreover, the placenta, which forms the connection between the mother and her baby, also appears to have an important role in the origin of HPD.¹⁰⁻¹³ HPD are a major cause of maternal morbidity and mortality and the placenta insufficiency associated with HPD also imposes an important risk for a deprived fetal condition.¹³⁻¹⁷

Hence, although luckily most pregnancies and deliveries are uncomplicated, complications during pregnancy can have extensive consequences for the wellbeing of both the mother

and her baby. Thus it is very important to adequately monitor and manage mothers and their fetuses who are at risk of such complications.

This thesis is subdivided into two parts. In part I, we will discuss the monitoring and management of mothers at risk of pregnancy complications, specifically HPD. Subsequently, in part II we will discuss the monitoring and management of fetal distress during labor.

PART I – MONITORING AND MANAGEMENT OF THE MOTHER AT RISK OF HYPERTENSIVE PREGNANCY DISORDERS

Etiology and consequences of hypertensive pregnancy disorders

HPD are a group of conditions including chronic hypertension during pregnancy, gestational hypertension, preeclampsia (PE), and PE superimposed on chronic hypertension.¹⁷⁻¹⁹ The separate HPD are named based on the moment of onset, either prior to, or developed during pregnancy. Chronic hypertension, also referred to as pre-existing hypertension, consists of hypertension discovered preconception or prior to 20 weeks of gestation.²⁰ Gestational hypertension is defined as *de novo* hypertension occurring after 20 weeks' gestation with a systolic blood pressure \geq 140 mmHg and/ or a diastolic blood pressure \geq 90 mmHg.¹⁷ Gestational hypertension can evolve into PE, when this new-onset hypertension is accompanied by either proteinuria (>300 mg/ day), uteroplacental dysfunction, and/or maternal organ dysfunction including renal insufficiency, impaired liver function, thrombocytopenia, and pulmonary edema.^{17-19.21} However, PE can also be superimposed on chronic hypertension. Approximately 5-10% of pregnancies are affected by HPD worldwide.^{6.10,14,22} The most common HPD is gestational hypertension, respectively.¹¹

Although the exact etiology of HPD remains elusive, failure of the maternal cardiovascular system to adapt to pregnancy is hypothesized to be one of the primary mechanisms leading to HPD. Abnormal placentation appears to have an important role in the origin of HPD, as well as increased vascular peripheral resistance.¹⁰⁻¹² It is unknown why these processes occur in women who develop HPD.

HPD are a major cause of maternal and fetal complications during pregnancy, including an increased risk of fetal growth restriction, preterm birth, renal and liver failure, placental abruption, and cardiovascular morbidity including cardiomyopathy.²³⁻²⁶ Furthermore, HPD are major contributors to maternal death in developing countries.¹⁵

Moreover, the consequences of HPD are not limited to pregnancy itself. Women who suffered from HPD have a lifelong elevated risk of developing chronic hypertension,

ischemic heart disease, heart failure, stroke, and end-stage renal disease.²⁷⁻²⁹ PE has also been identified as an independent risk factor for cardiovascular death, especially earlyonset PE.^{27,28} Whether HPD unmask an already existing risk of developing cardiovascular disease (CVD) or whether HPD itself is causally related to the maternal risk of CVD, remains uncertain.^{30,31} The first hypothesis is supported by shared risk factors between HPD and CVD, such as smoking, obesity, diabetes mellitus, and high cholesterol.^{25,31} Furthermore, endothelial dysfunction plays a central role in the development of both disorders.³¹ On the other hand, both the pregnancy itself and the stress on the vessels that occurs during pregnancy could contribute to arterial wall inflammation. Furthermore, products of the dysfunctional placenta in HPD could permanently compromise the maternal cardiovascular system, leading to a (further) increased risk of CVD.^{25,31} The two hypotheses are not mutually exclusive.

Complications following HPD do not only affect the health of mothers, but also extend to the health of their offspring. Children born after pregnancies complicated by HPD have an increased risk of developing CVD, stroke, elevated blood pressure, higher body mass index, and more subclinical cardiac changes during childhood or adolescence.³²⁻³⁶

Early detection of hypertensive pregnancy disorders

Accurate screening for HPD, followed by appropriate treatment, can have major benefits for maternal, fetal, and neonatal health. Identifying women at high risk for HPD provides opportunities to reduce their chance of developing HPD. For example, the daily use of low-dose aspirin and calcium between 12 to 36 weeks' gestation in women at moderate to high risk of HPD can reduce the risk of developing HPD significantly.^{37,38}

Currently, risk prediction is mainly based on maternal history, which has limited predictive ability.^{39,40} Townsend et al. recently published a review of reviews on the prediction of preeclampsia.⁴¹ A total of 126 reviews were included, reporting on over 90 predictors and 52 prediction models. No single marker in this meta-analysis showed a test performance suitable for clinical practice. Prediction models that combined markers are promising to accurately predict HPD. However, none of those models including multiple markers have undergone external validation. Therefore, there is still a need for adequate strategies to identify women at risk of HPD.

One of the methods that has not been included by Townsend et al. is the measurement of changes in cardiac function.⁴¹ Pregnancy is associated with several physiological changes to adapt to the increased plasma volume, increased cardiac output, and decreased peripheral vascular resistance.^{2,3,5} In HPD, the maternal cardiovascular system fails to adequately adapt to pregnancy.^{6,42} This failure to adapt causes subtle changes of the myocardial tissue shape, size, and function.^{6,43-45}

Unfortunately, conventional echocardiography is shown to be unsuitable for the detection of early subclinical myocardial changes because measures such as left ventricular ejection fraction (LVEF) and diastolic function provide only an indirect estimate of myocardial contractile function. Moreover, in case of subclinical myocardial dysfunction, these parameters remain within normal range due to compensatory mechanisms activated in the heart.^{9,46}

Speckle tracking echocardiography (STE) might be considered as a diagnostic tool to overcome these conventional echocardiography problems. This relatively new echocardiographic method analyzes the motion of tissues by tracking the position of acoustic reflections called speckles, which appear in the myocardium as a result of scattering of the ultrasound beam by the tissue (see Figure 1).^{46,47} By directly quantifying the extent of myocardial contraction function, it appears to be suitable for detecting subclinical cardiac changes.^{9,42,48,49} Although the use of STE is not yet widespread during pregnancy, its use for various other clinical settings has increased remarkably over recent years.⁴⁷

One of the advantages STE has over conventionally used technique is its relative independence of loading conditions.^{9,46} During pregnancy, there is a continuous variation of loading conditions of the heart, which affects conventional chamber function parameters such as LVEF and stroke volume.^{9,46} Thus, STE seems to be a promising method to study changes in myocardial function during pregnancy.

Previously, STE abnormalities were found to have prognostic value in patients with hypertension, heart failure, and hypertensive heart disease.^{50–54} Furthermore, Shahul et al. showed that in women with chronic hypertension a decrease in LV global longitudinal strain, has a predictive value for developing superimposed PE.⁵⁵ Therefore, differences in STE parameters between HPD and normotensive pregnancies might help in risk stratification of HPD.

Besides evaluation of cardiac function using STE, the search for methods to aid in the early detection and prediction of HPD may also focus on advanced analytics of physiological measurements. The ANS has a prominent role in several of the adaptive changes that occur during pregnancy, especially in the cardiovascular, gastrointestinal, and urinary system control.^{29,56} Therefore, the inadequate adaptation of the cardiovascular system to pregnancy, which may ultimately lead to HPD, might be associated with a dysfunction of the ANS.^{29,57-59} Measuring this dysfunction of the ANS can possibly be used as a precursor of developing HPD.



Figure 1. Speckle tracking echocardiography. The cardiac deformation is measured in three directions, namely the longitudinal, radial, and circumferential strain. Adapted from Visentin et al.⁹

Various tests and techniques can be used to assess the function of the ANS. However, not all tests are suitable for routine use due to their invasive character.⁶⁰ In contrast, heart rate variability (HRV) is a non-invasive, widely used clinical method to assess cardiac autonomic functions.⁶¹ The assessment of the oscillations in the heart rate signal provides an estimate of the sympathetic and parasympathetic activity, as well as the balance between those two. Several studies state that HPD are associated with a sympathetic overdrive, that might already occur before the onset of clinical symptoms.^{62,63} When measuring HRV, it is important to take possible confounders into account, such as the circadian rhythm, physical exercise.⁶¹ The use of long-acting medication, such as antenatally administered corticosteroids, could also have a confounding effect on HRV. However, the effect of such medication on HRV is not fully known.

In addition to (early) detection of HPD, techniques like STE and HRV might also help in better differentiating regarding the risk of CVD in the high-risk group of women with HPD. Indeed, the American Heart Association already recommends that pregnancy history should be part of the evaluation of the risk of CVD in women.⁶⁴ However, women with HPD and alterations in HRV or STE might need a more strict follow-up, as this could possibly indicate an even higher chance of developing CVD in later life. Thereby, these techniques might enable better screening and treatment based on each women's personal risk profile.

In conclusion, new methods for prediction and early detection of HPD are needed, as currently used methods are limited in their predictive abilities. As HPD are associated with dysfunction of the ANS and changes in cardiac function, HRV and STE might be able to detect early, subclinical changes. Both of these methods are non-invasive and their use can be easily integrated into the clinical workflow. Furthermore, STE and HRV might also play a role in the risk stratification of developing CVD after a pregnancy complicated by HPD.

PART II – MONITORING AND MANAGEMENT OF THE FETUS AT RISK FOR FETAL DISTRESS

Fetal oxygenation

During pregnancy, the fetus depends on its mother for nutrients and oxygenation. The main source of fetal energy production is cellular aerobic metabolism, for which oxygen and glucose are needed. The oxygen flow towards the fetus is obtained via the maternal respiration and circulation, through placental perfusion, placental gas exchange, and the umbilical cord.⁶⁵ The oxygen exchange via the placenta is visualized in Figure 2.

The fetal oxygen tension (or partial pressure of oxygen, pO_2) is lower than the maternal oxygen tension. This difference in pO_2 between the maternal and fetal circulation facilitates the maternal-to-fetal transfer of oxygen via diffusion over the placental membranes.^{65,67}



Figure 2 Placental structure and circulation. Maternal blood enters the intervillous space via the spiral arteries. Transport of oxygen and nutrients from the maternal to the fetal blood occurs from the intervillous space towards the villous capillaries, respectively. Oxygenated blood reaches the fetus through the umbilical vein. The deoxygenated blood from the fetus is carried back to the placenta via the umbilical arteries. Adapted from Lofthouse.⁶⁶

Several adaptive mechanisms allow the fetus to function in a low-oxygen environment, like the higher affinity for oxygen in fetal hemoglobin at the same pO₂ a higher cardiac output, and the delivery of blood with the highest oxygenation levels to the most vital organs.^{65,67} Despite these adaptive mechanisms, continuous oxygen supply is still very important for the fetus, as oxygen cannot be stored and later mobilized. Hence, even a short interruption of a few minutes can put the fetus at risk of oxygen deficiency.

Birth asphyxia

There are three categories of oxygen deficit, depending on the severity of the oxygen deficiency⁶⁵;

- 1. Hypoxemia, in which the reduced oxygen supply causes a reduction in arterial oxygen concentration without affecting cell and organ function.
- 2. Hypoxia, in which reduction in oxygen causes anaerobic metabolism and subsequently leading to a decrease in pH, accumulation of lactate, and an increased base excess, mainly in the peripheral tissues.
- 3. Asphyxia, the last phase of oxygen deprivation in which hypoxia extends to the central organs (i.e. the brain, heart, and adrenal glands). Asphyxia can potentially lead to metabolic acidosis and death.

The risk of compromised fetal oxygenation is increased during labor,⁶⁵ which can have lifelong consequences due to neuronal damage. During labor, uterine contractions temporarily decrease the perfusion of the placental bed. Furthermore, these contractions cause compression of the umbilical cord thus reducing umbilical blood flow.^{65,68} Both mechanisms may lead to decreased oxygen delivery to the fetus, and when lasting too long may ultimately lead to perinatal hypoxia and eventually to asphyxia.^{65,68}

Worldwide, birth asphyxia is one of the main causes of neonatal mortality,^{69,70} causing approximately 0.69-0.92 million neonatal deaths each year.^{69,71} Asphyxia can cause severe morbidities, such as brain damage, renal and hepatic failure and cardiac dysfunction.⁷²⁻⁷⁸

Each year, approximately four million babies are born worldwide showing signs of perinatal asphyxia.⁷⁸ Annually, in the Netherlands, approximately 400 term babies are born with asphyxia (0.3%).⁷⁹ Asphyxia imposes a huge burden on patients, their families, and society, and presents an important challenge to clinical and research professionals to establish effective diagnosis and treatment methods.

Assessment and treatment of deprived fetal oxygenation

To reduce the occurrence of perinatal asphyxia, adequate recognition of fetal distress is needed, allowing timely intervention to manage the effects of hypoxemia and hypoxia before asphyxia occurs. However, it is not yet possible to directly measure fetal oxygenation during labor, and methods for continuous intrapartum monitoring of fetal oxygen saturation (SpO₂), pO2, partial carbon dioxide pressure (pCO₂), and pH are either not effective or not yet suitable for clinical practice.^{80–82} Therefore, evaluation of the fetal heart rate (FHR) pattern, with additional fetal scalp blood sampling on indication, is still one of the most used methods of intrapartum fetal monitoring.^{83,84}

FHR is regulated by the ANS. When oxygen delivery towards the fetus is impaired, the fetal ANS is activated as a compensation mechanism.^{65,85} This ANS activation results in changes in FHR and blood flow distribution to maintain blood pressure and economize oxygen consumption.^{68,83,86} Several studies show that a non-reassuring FHR pattern is considered to be indicative of potential fetal hypoxia.^{83,87-90}

Unfortunately, evaluation of FHR has a high false-positive rate in the prediction of fetal hypoxia.^{83,91} This implies that a non-reassuring FHR pattern is not always reflecting a compromised fetus. This limitation in the prediction of fetal hypoxia makes it challenging for clinicians to decide when interventions are indicated. Monitoring the fetal condition using continuous FHR could lead to many unnecessary emergency operative deliveries or instrumental vaginal deliveries,⁸⁸ which are associated with an increased risk of adverse maternal and neonatal outcome.⁹²⁻⁹⁴ Instead of aiming for immediate delivery in case of suspected fetal distress, interventions can be used to restore fetal oxygenation while keeping the fetus inside the uterus. When successful, these interventions, i.e. intrauterine resuscitation, can avoid unnecessary invasive interventions.

Depending on the presumable cause of fetal distress, the intrauterine resuscitation technique should be focusing on increasing oxygen delivery to the fetus, improvement of uteroplacental blood flow and/or alleviation of umbilical cord compression. For example, adjustment or discontinuation of oxytocin infusion or the use of tocolytic drugs can alleviate umbilical cord compression caused by prolonged or frequent uterine contractions, whereas intermittent pushing and maternal repositioning improve the uteroplacental blood flow, while maternal hyperoxygenation focusses on increasing the oxygen delivery towards the fetus.^{95,96}

Even though these interventions are widely used in clinical practice, solid evidence regarding their beneficial effect on neonatal outcome is limited and sometimes contradictory.^{95,96} As a result, there is no consensus on the use of these intrauterine resuscitation techniques during labor. Consequently, recommendations on the management of fetal distress may vary in international guidelines. This may lead to variations in clinical practice between hospitals.

Intrauterine resuscitation through maternal hyperoxygenation

One intrauterine resuscitation technique that raises discussion is maternal hyperoxygenation, in which the mother receives additional oxygen to treat fetal distress. The rationale behind this intervention is that, even though the oxygen saturation of

healthy women in labor is nearly 100%, the additional oxygen causes an increase in maternal pO₂, which results in an increase in oxygen transfer towards the fetus.^{97,98} Vasicka et al. showed that five minutes of maternal hyperoxygenation with 100% oxygen results in an increase in pO₂ in maternal arterial blood, as well as a 52% increase in pO₂ in the umbilical artery.⁹⁷ Maternal hyperoxygenation may thus contribute to improved fetal oxygenation and prevention of fetal hypoxia and acidosis.

Measurements with non-invasive blood oxygen level-dependent magnetic resonance imaging (MRI) demonstrated increased oxygenation in a number of fetal organs during maternal hyperoxygenation, specifically the fetal liver, spleen, and kidneys.⁹⁹ In addition, several small, non-randomized studies showed an improvement of FHR pattern and/or fetal scalp pH when additional oxygen was applied in the event of suspected fetal distress.¹⁰⁰⁻¹⁰⁵

Despite these positive effects of maternal hyperoxygenation, there have been posed counter-arguments, especially concerning the potentially harmful effects of this therapy. One of the arguments to plead against maternal hyperoxygenation is a possible lowering of umbilical cord arterial pH. Thorp et al. performed a randomized controlled trial (RCT) to investigate the effect of prophylactic oxygen administration during the second stage of uncomplicated labor.¹⁰⁶ In this study, women with reassuring fetal heart rate tracings were randomized between breathing room air or maternal hyperoxygenation. The mean umbilical cord arterial pH did not differ between the groups, however, significantly more umbilical cord arterial pH levels below 7.20 were found in the oxygenation group, compared to controls.¹⁰⁶ Two other RCTs on the use of prophylactic oxygen administration during the second stage of uncomplicated labor did not find any significant differences in arterial cord blood pH.^{107,108} Furthermore, a recent noninferiority RCT among patients with non-reassuring FHR during the active phase of labor compared maternal hyperoxygenation to breathing room air.¹⁰⁹ This study showed no difference in umbilical artery lactate.

Another possible detrimental effect of maternal oxygen supplementation is the potential increase in free radical activity in both the mother and fetus.¹¹⁰⁻¹¹² Free oxygen radicals are reactive atoms with one more unpaired electron in their outer orbit. To a certain degree, free oxygen radicals are formed under physiologic conditions.¹¹³ During labor, uterine contractions might be regarded as a small series of ischemia-reperfusion injuries causing free oxygen radicals, especially in case of non-reassuring FHR.¹¹⁴An increase in free radical activity due to oxidative stress may lead to cell damage and altered cellular function.¹¹⁵ Unfortunately, the brief lifespan of free oxygen radicals makes it extremely difficult to measure them directly. To measure free oxygen radical activity, more stable derivatives can be used as proxy measurements.¹¹⁶ For instance, malondialdehyde (MDA), a by-product of lipid peroxidation, appears to be an accurate non-invasive biomarker for free radical damage.^{114,116}

In an RCT on the prophylactic use of maternal hyperoxygenation during elective cesarean delivery, the levels of MDA in maternal blood and in arterial umbilical cord blood were found to be significantly higher in the oxygen group compared to the group of women breathing room-air.¹¹² This raises concerns about a possible increase in free oxygen radical activity.

Yet, the effect and clinical implication of maternal hyperoxygenation in case of fetal distress on free radical activity is not known. What we do know is that after birth, neonatal resuscitation using 100% oxygen may lead to higher morbidity and mortality compared with resuscitation with room-air consisting 21% oxygen.^{117,118} However, oxygenation of the fetus is facilitated by the large gradient over the placenta between the pO_2 in the maternal blood and the umbilical blood.¹¹⁹ Therefore, the increase in fetal pO2 due to maternal hyperoxygenation will never reach the levels obtained by the direct application of 100% oxygen directly to either the fetus or neonate.¹²⁰

The conflicting literature has resulted in different recommendations on the use of maternal hyperoxygenation between international guidelines. The American College of Obstetricians and the Gynecologists (ACOG) and the Society of Obstetricians and Gynaecologists of Canada (SOGC) support the use of maternal hyperoxygenation as an intrauterine resuscitation technique during labor.^{121,122} In contrast, the British guideline, provided by the Royal College of Obstetricians and Gynaecologists (RCOG) advises against the use of maternal hyperoxygenation in clinical practice.¹²³ The Dutch Society of Obstetricians and Gynecologists (NVOG) has updated its recommendation in July 2019 and also advises against the routine use of maternal hyperoxygenation to treat suspected fetal distress due to knowledge gap regarding this intervention.¹²⁴

To our knowledge, no RCT has studied the effect of maternal hyperoxygenation in case of fetal distress on FHR pattern or neonatal outcome. The lack of robust evidence regarding the effect of maternal hyperoxygenation is also addressed in a Cochrane review, which concludes there is insufficient evidence to support the use of maternal hyperoxygenation for the treatment of fetal distress and emphasizes the need for a randomized controlled trial to study the effect of maternal hyperoxygenation in the compromised fetus during the second stage of labor.¹²⁵

In conclusion, the fetus depends on oxygen delivery from the placenta. During labor, the risk of compromised fetal oxygenation is increased, and when lasting too long may ultimately lead to perinatal hypoxia and eventually to asphyxia. Therefore, it is important to monitor the fetal condition and, if needed, optimize the fetal condition before hypoxia and asphyxia occur. To do so, several intrauterine resuscitation techniques can be used. However, solid evidence regarding their beneficial effect on neonatal outcome is limited and sometimes contradictory. One intrauterine resuscitation technique that raises discussion is maternal hyperoxygenation, of which the effect during the second stage of labor has only been studied in small, non-randomized studies.

OUTLINE OF THIS THESIS

This thesis aims to answer the following questions:

- Is STE a suitable method to detect differences in cardiac function in pregnant women with HPD or women with a history of HPD compared to normotensive women?
 Chapter 2 aims to answer this research question by systematically reviewing and summarizing the literature on STE in HPD.
- 2. Does HRV detect differences in the function of the autonomic nervous system in pregnant women with HPD or women with a history of HPD compared to normotensive women?

Chapter 3 gives a systematic overview of currently available literature regarding HRV in HPD in order to answer this research question.

3. What is the effect of routine obstetric medication on maternal HRV?

Chapter 4 presents the study protocol of an observational cohort study to investigate the effect of routine obstetric medication on maternal HRV, to provide an answer to this research question.

- 4. Which methods are recommended in international guidelines regarding the monitoring and management of fetal distress during labor?
- 5. Which fetal monitoring methods are used in Dutch clinical practice, and which intrauterine resuscitation techniques are utilized in case of suspected fetal distress?

Chapter 5 reports on the practice variation in fetal monitoring and management of fetal distress during labor in Dutch hospitals. Furthermore, it provides a comparison of recommendations of the national guidelines of various Western countries regarding fetal monitoring and management of fetal distress during labor. This study was set up to answer research questions 4 and 5.

- 6. What is the effect of intrauterine resuscitation by maternal hyperoxygenation during the second stage of term labor on neonatal and maternal outcome?
- 7. Does maternal hyperoxygenation, applied in case of suspected fetal distress during the second stage of term labor, affect FHR?

Chapter 6 describes the study protocol of an RCT, the INTEREST O2 study, to investigate the effect of maternal hyperoxygenation as an intrauterine resuscitation technique during the second stage of labor. This RCT is set-up to answer research questions 6 and 7.

Chapter 7 presents the results of the INTEREST O2 study on perinatal and maternal outcome as a result of maternal hyperoxygenation, and this chapter thus contributes to answering research question 6.

Chapter 8 presents the detailed FHR analyses of the INTEREST O2 study, using specific technical FHR features related to poor neonatal outcome. The goal of this study was to contribute to answering research question 7.

Chapters 2 to 8 have been published, or are submitted for publication. These chapters are written to be self-contained which might cause some overlap between these chapters.

Chapter 9 contains a general discussion of the results presented in this thesis and provides suggestions for future research.

Chapters 10 and 11 provide an English and Dutch summary of the data presented in this thesis, respectively.

REFERENCES

- 1. Van Leeuwen P, Geue D, Lange S, Cysarz D, Bettermann H, Gronemeyer DHW. Is there evidence of fetal-maternal heart rate synchronization? *BMC Physiol.* 2003;3:2. doi:10.1186/1472-6793-3-2
- 2. Tan EK, Tan EL. Alterations in physiology and anatomy during pregnancy. *Best Pract Res Clin Obstet Gynaecol.* 2013;27(6):791-802. doi:10.1016/j.bpobgyn.2013.08.001
- 3. Soma-Pillay P, Nelson-Piercy C, Tolppanen H, Mebazaa A. Physiological changes in pregnancy. *Cardiovasc J Afr.* 2016;27(2):89-94. doi:10.5830/CVJA-2016-021
- 4. Fu Q. Hemodynamic and Electrocardiographic Aspects of Uncomplicated Singleton Pregnancy. *Adv Exp Med Biol.* 2018;1065:413-431. doi:10.1007/978-3-319-77932-4_26
- 5. Duvekot JJ, Cheriex EC, Pieters FA, Menheere PP, Peeters LH. Early pregnancy changes in hemodynamics and volume homeostasis are consecutive adjustments triggered by a primary fall in systemic vascular tone. *Am J Obstet Gynecol.* 1993;169(6):1382-1392. doi:10.1016/0002-9378(93)90405-8
- 6. Kalafat E, Thilaganathan B. Cardiovascular origins of preeclampsia. *Curr Opin Obstet Gynecol.* 2017;29(6):383-389. doi:10.1097/GCO.00000000000419 [doi]
- 7. de Haas S, Ghossein-Doha C, van Kuijk SMJ, van Drongelen J, Spaanderman MEA. Physiological adaptation of maternal plasma volume during pregnancy: a systematic review and meta-analysis. *Ultrasound Obstet Gynecol Off J Int Soc Ultrasound Obstet Gynecol.* 2017;49(2):177-187. doi:10.1002/uog.17360
- 8. Melchiorre K, Sharma R, Thilaganathan B. Cardiac structure and function in normal pregnancy. *Curr Opin Obstet Gynecol*. 2012;24(6):413-421. doi:10.1097/GCO.0b013e328359826f
- 9. Visentin S, Palermo C, Camerin M, et al. Echocardiographic Techniques of Deformation Imaging in the Evaluation of Maternal Cardiovascular System in Patients with Complicated Pregnancies. *Biomed Res Int.* 2017;2017:4139635. doi://dx.doi.org/10.1155/2017/4139635
- 10. Folk DM. Hypertensive Disorders of Pregnancy: Overview and Current Recommendations. *J Midwifery Womens Health*. 2018;63(3):289-300. doi:10.1111/jmwh.12725
- 11. Shah S, Gupta A. Hypertensive Disorders of Pregnancy. *Cardiol Clin.* 2019;37(3):345-354. doi:10.1016/j.ccl.2019.04.008
- 12. Spradley FT. Sympathetic nervous system control of vascular function and blood pressure during pregnancy and preeclampsia. *J Hypertens*. 2019;37(3):476-487. doi:10.1097/HJH.000000000001901
- 13. Apicella C, Ruano CSM, Méhats C, Miralles F, Vaiman D. The Role of Epigenetics in Placental Development and the Etiology of Preeclampsia. *Int J Mol Sci.* 2019;20(11). doi:10.3390/ijms20112837
- 14. Umesawa M, Kobashi G. Epidemiology of hypertensive disorders in pregnancy: prevalence, risk factors, predictors and prognosis. *Hypertens Res.* 2017;40(3):213-220. doi:10.1038/hr.2016.126 [doi]
- Khan KS, Wojdyla D, Say L, Gulmezoglu AM, Look PF Van. WHO analysis of causes of maternal death: a systematic review. *Lancet (London, England)*. 2006;367(9516):1066-1074. doi:S0140-6736(06)68397-9 [pii]

- 16. Saleem S, McClure EM, Goudar SS, et al. A prospective study of maternal, fetal and neonatal deaths in low- and middle-income countries. *Bull World Health Organ*. 2014;92(8):605-612. doi:10.2471/BLT.13.127464
- 17. ACOG Practice Bulletin No. 202: Gestational Hypertension and Preeclampsia. *Obstet Gynecol.* 2019;133(1):e-e25. doi:10.1097/AOG.000000000003018 [doi]
- 18. NICE project Team. NICE Guideline Hypertension in Pregnancy: Diagnosis and Management.; 2019.
- 19. Brown MA, Magee LA, Kenny LC, et al. The hypertensive disorders of pregnancy: ISSHP classification, diagnosis & management recommendations for international practice. *Pregnancy Hypertens*. 2018;13:291-310. doi:10.1016/j.preghy.2018.05.004
- 20. ACOG Practice Bulletin No. 203: Chronic Hypertension in Pregnancy. *Obstet Gynecol.* 2019;133(1):e26-e50. doi:10.1097/AOG.0000000000003020
- 21. Braunthal S, Brateanu A. Hypertension in pregnancy: Pathophysiology and treatment. SAGE open Med. 2019;7:2050312119843700. doi:10.1177/2050312119843700
- 22. Hutcheon JA, Lisonkova S, Joseph KS. Epidemiology of pre-eclampsia and the other hypertensive disorders of pregnancy. *Best Pract Res Obstet Gynaecol.* 2011;25(4):391-403. doi:10.1016/j.bpobgyn.2011.01.006 [doi]
- 23. Lisonkova S, Sabr Y, Mayer C, Young C, Skoll A, Joseph KS. Maternal morbidity associated with early-onset and late-onset preeclampsia. *Obstet Gynecol*. 2014;124(4):771-781. doi:10.1097/AOG.00000000000472 [doi]
- 24. Seely EW, Ecker J. Chronic hypertension in pregnancy. *Circulation*. 2014;129(11):1254-1261. doi:10.1161/CIRCULATIONAHA.113.003904
- 25. Bergen NE, Schalekamp-Timmermans S, Roos-Hesselink J, Roeters van Lennep JE, Jaddoe VVW, Steegers EAP. Hypertensive disorders of pregnancy and subsequent maternal cardiovascular health. *Eur J Epidemiol.* 2018;33(8):763-771. doi:10.1007/s10654-018-0400-1
- 26. Langenveld J, Ravelli ACJ, van Kaam AH, et al. Neonatal outcome of pregnancies complicated by hypertensive disorders between 34 and 37 weeks of gestation: a 7 year retrospective analysis of a national registry. *Am J Obstet Gynecol.* 2011;205(6):540.e1-7. doi:10.1016/j.ajog.2011.07.003
- 27. Wu P, Haththotuwa R, Kwok CS, et al. Preeclampsia and Future Cardiovascular Health: A Systematic Review and Meta-Analysis. *Circ Qual outcomes*. 2017;10(2):10.1161/ CIRCOUTCOMES.116.003497. Epub 2017 Feb 22. doi:e003497 [pii]
- Bellamy L, Casas JP, Hingorani AD, Williams DJ. Pre-eclampsia and risk of cardiovascular disease and cancer in later life: systematic review and meta-analysis. *BMJ*. 2007;335(7627):974. doi:bmj.39335.385301.BE [pii]
- 29. Reyes LM, Usselman CW, Davenport MH, Steinback CD. Sympathetic Nervous System Regulation in Human Normotensive and Hypertensive Pregnancies. *Hypertens (Dallas, Tex 1979)*. 2018;71(5):793-803. doi:10.1161/HYPERTENSIONAHA.117.10766
- 30. Karumanchi SA, Granger JP. Preeclampsia and Pregnancy-Related Hypertensive Disorders. *Hypertens (Dallas, Tex 1979).* 2016;67(2):238-242. doi:10.1161/ HYPERTENSIONAHA.115.05024
- 31. Riise HKR, Sulo G, Tell GS, et al. Hypertensive pregnancy disorders increase the risk of maternal cardiovascular disease after adjustment for cardiovascular risk factors. *Int J Cardiol*. 2019;282:81-87. doi:10.1016/j.ijcard.2019.01.097

- 32. Davis EF, Lewandowski AJ, Aye C, et al. Clinical cardiovascular risk during young adulthood in offspring of hypertensive pregnancies: insights from a 20-year prospective follow-up birth cohort. *BMJ Open.* 2015;5(6):e008136. doi:10.1136/ bmjopen-2015-008136
- 33. Sacks KN, Friger M, Shoham-Vardi I, et al. Prenatal exposure to preeclampsia as an independent risk factor for long-term cardiovascular morbidity of the offspring. *Pregnancy Hypertens An Int J Women's Cardiovasc Heal*. 2018;13:181-186. doi:10.1016/j. preghy.2018.06.013
- 34. Tenhola S, Rahiala E, Martikainen A, Halonen P, Voutilainen R. Blood Pressure, Serum Lipids, Fasting Insulin, and Adrenal Hormones in 12-Year-Old Children Born with Maternal Preeclampsia. *J Clin Endocrinol Metab.* 2003;88(3):1217-1222. doi:10.1210/jc.2002-020903
- 35. Timpka S, Macdonald-Wallis C, Hughes AD, et al. Hypertensive Disorders of Pregnancy and Offspring Cardiac Structure and Function in Adolescence. *J Am Heart Assoc.* 2016;5(11). doi:10.1161/JAHA.116.003906
- 36. Kajantie E, Eriksson JG, Osmond C, Thornburg K, Barker DJP. Pre-eclampsia is associated with increased risk of stroke in the adult offspring: the Helsinki birth cohort study. *Stroke*. 2009;40(4):1176-1180. doi:10.1161/STROKEAHA.108.538025
- 37. Rolnik DL, Wright D, Poon LC, et al. Aspirin versus Placebo in Pregnancies at High Risk for Preterm Preeclampsia. *N Engl J Med.* 2017;377(7):613-622. doi:10.1056/ NEJMoa1704559
- 38. Hofmeyr GJ, Lawrie TA, Atallah AN, Torloni MR. Calcium supplementation during pregnancy for preventing hypertensive disorders and related problems. *Cochrane database Syst Rev.* 2018;10:CD001059. doi:10.1002/14651858.CD001059.pub5
- 39. Poon LCY, Kametas NA, Chelemen T, Leal A, Nicolaides KH. Maternal risk factors for hypertensive disorders in pregnancy: a multivariate approach. *J Hum Hypertens*. 2010;24(2):104-110. doi:10.1038/jhh.2009.45
- 40. Verghese L, Alam S, Beski S, Thuraisingham R, Barnes I, MacCallum P. Antenatal screening for pre-eclampsia: evaluation of the NICE and pre-eclampsia community guidelines. *J Obstet Gynaecol*. 2012;32(2):128-131. doi:10.3109/01443615.2011.63 5224
- 41. Townsend R, Khalil A, Premakumar Y, et al. Prediction of pre-eclampsia: review of reviews. *Ultrasound Obstet Gynecol*. 2019;54(1):16-27. doi:10.1002/uog.20117 [doi]
- 42. Buddeberg BS, Sharma R, O'Driscoll JM, Agten AK, Khalil A, Thilaganathan B. Cardiac maladaptation in term pregnancies with preeclampsia. *Pregnancy Hypertens*. 2018;13:198-203. doi:S2210-7789(18)30089-8 [pii]
- 43. Melchiorre K, Sutherland GR, Baltabaeva A, Liberati M, Thilaganathan B. Maternal cardiac dysfunction and remodeling in women with preeclampsia at term. *Hypertension*. 2011;57(1):85-93. doi://dx.doi.org/10.1161/HYPERTENSIONAHA.110.162321
- 44. Melchiorre K, Sutherland G, Sharma R, Nanni M, Thilaganathan B. Mid-gestational maternal cardiovascular profile in preterm and term pre-eclampsia: a prospective study. *BJOG.* 2013;120(4):496-504. doi:10.1111/1471-0528.12068 [doi]
- 45. Melchiorre K, Sutherland GR, Watt-Coote I, Liberati M, Thilaganathan B. Severe myocardial impairment and chamber dysfunction in preterm preeclampsia. *Hypertens Pregnancy*. 2012;31(4):454-471. doi://dx.doi.org/10.3109/10641955.2012.697951
- 46. Bansal M, Kasliwal RR. How do I do it? Speckle-tracking echocardiography. *Indian Heart J.* 2013;65(1):117-123. doi:10.1016/j.ihj.2012.12.004 [doi]

- 47. Cameli M, Mandoli GE, Sciaccaluga C, Mondillo S. More than 10 years of speckle tracking echocardiography: Still a novel technique or a definite tool for clinical practice? *Echocardiography*. Published online April 2019. doi:10.1111/echo.14339 [doi]
- D'Andrea A, Radmilovic J, Ballo P, et al. Left ventricular hypertrophy or storage disease? the incremental value of speckle tracking strain bull's-eye. *Echocardiography*. 2017;34(5):746-759. doi:10.1111/echo.13506 [doi]
- 49. Davis EF, Lewandowski AJ, Leeson P. Cardiac dysfunction and preeclampsia can imaging give clues to mechanism? *Circ Cardiovasc Imaging*. 2012;5(6):691-692. doi://dx.doi. org/10.1161/CIRCIMAGING.112.979831
- 50. Saito M, Khan F, Stoklosa T, Iannaccone A, Negishi K, Marwick TH. Prognostic Implications of LV Strain Risk Score in Asymptomatic Patients With Hypertensive Heart Disease. JACC Cardiovasc Imaging. 2016;9(8):911-921. doi:10.1016/j.jcmg.2015.09.027
- 51. Lee W-H, Liu Y-W, Yang L-T, Tsai W-C. Prognostic value of longitudinal strain of subepicardial myocardium in patients with hypertension. *J Hypertens*. 2016;34(6):1195-1200. doi:10.1097/HJH.000000000000903
- 52. Ersboll M, Valeur N, Mogensen UM, et al. Relationship between left ventricular longitudinal deformation and clinical heart failure during admission for acute myocardial infarction: a two-dimensional speckle-tracking study. J Am Soc Echocardiogr. 2012;25(12):1280-1289. doi:10.1016/j.echo.2012.09.006 [doi]
- 53. Sengelov M, Jorgensen PG, Jensen JS, et al. Global Longitudinal Strain Is a Superior Predictor of All-Cause Mortality in Heart Failure With Reduced Ejection Fraction. *JACCCardiovascular imaging.* 2015;8(12):1351-1359. doi:S1936-878X(15)00719-6[pii]
- 54. Shah AM, Claggett B, Sweitzer NK, et al. Prognostic Importance of Impaired Systolic Function in Heart Failure With Preserved Ejection Fraction and the Impact of Spironolactone. *Circulation*. 2015;132(5):402-414. doi:10.1161/ CIRCULATIONAHA.115.015884 [doi]
- Shahul S, Ramadan H, Mueller A, et al. Abnormal mid-trimester cardiac strain in women with chronic hypertension predates superimposed preeclampsia. *Pregnancy Hypertens*. 2017;10:251-255. doi:S2210-7789(17)30144-7 [pii]
- 56. M Balajewicz-Nowak, A Furgala, K Pitynski, P Thor, H Huras, K Rytlewski. The dynamics of autonomic nervous system activity and hemodynamic changes in pregnant women. *Neuroendocrinol Lett.* 2016;37(1):70-77.
- 57. Guyenet PG. The sympathetic control of blood pressure. *Nat Rev Neurosci.* 2006;7(5):335-346. doi:10.1038/nrn1902
- 58. Julius S, Majahalme S. The changing face of sympathetic overactivity in hypertension. Ann Med. 2000;32(5):365-370. doi:10.3109/07853890008995939
- 59. Brown CA, Lee CT, Hains SM, Kisilevsky BS. Maternal heart rate variability and fetal behavior in hypertensive and normotensive pregnancies. *Biol Res Nurs*. 2008;10(2):134-144. doi:10.1177/1099800408322942 [doi]
- 60. Yousif D, Bellos I, Penzlin AI, et al. Autonomic Dysfunction in Preeclampsia: A Systematic Review. *Front Neurol*. 2019;10:816. doi:10.3389/fneur.2019.00816
- 61. Task Force of The European Society of Cardiology and The North American Society of Pacing and Electrophysiology. Heart rate variability. Standards of measurement, physiological interpretation, and clinical use. *Eur Heart J.* 1996;17(3):354-381.
- 62. Pal GK, Pal P, Nanda N, Amudharaj D, Karthik S. Spectral analysis of heart rate variability (HRV) may predict the future development of essential hypertension. *Med Hypotheses*. 2009;72(2):183-185. doi:10.1016/j.mehy.2008.07.060

- 63. Pal GK, Shyma P, Habeebullah S, Pal P, Nanda N, Shyjus P. Vagal withdrawal and sympathetic overactivity contribute to the genesis of early-onset pregnancy-induced hypertension. *Int J Hypertens*. 2011;2011:361417. doi:10.4061/2011/361417 [doi]
- 64. Mosca L, Benjamin EJ, Berra K, et al. Effectiveness-based guidelines for the prevention of cardiovascular disease in women--2011 update: a guideline from the American Heart Association. J Am Coll Cardiol. 2011;57(12):1404-1423. doi:10.1016/j.jacc.2011.02.005
- 65. Yli BM, Kjellmer I. Pathophysiology of foetal oxygenation and cell damage during labour. Best Pract Res Clin Obstet Gynaecol. 2016;30:9-21. doi:10.1016/j.bpobgyn.2015.05.004
- 66. Lofthouse EM. The accumulation of glutamate in the placental syncytiotrophoblast as a driver of membrane transport. *Thesis, Univ Southampt.* Published online 2014:9.
- 67. Meschia G. Fetal oxygenation and maternal ventilation. *Clin Chest Med.* 2011;32(1):15-19. doi:10.1016/j.ccm.2010.11.007
- 68. Westgate JA, Wibbens B, Bennet L, Wassink G, Parer JT, Gunn AJ. The intrapartum deceleration in center stage: a physiologic approach to the interpretation of fetal heart rate changes in labor. *Am J Obstet Gynecol.* 2007;197(3):236.e-236.11. doi:S0002-9378(07)00429-2 [pii]
- 69. Lawn JE, Cousens S, Zupan J. 4 million neonatal deaths: when? Where? Why? Lancet (London, England). 2005;365(9462):891-900. doi:10.1016/S0140-6736(05)71048-5
- 70. Liu L, Johnson HL, Cousens S, et al. Global, regional, and national causes of child mortality: an updated systematic analysis for 2010 with time trends since 2000. *Lancet* (*London, England*). 2012;379(9832):2151-2161. doi:10.1016/S0140-6736(12)60560-1
- 71. Liu L, Oza S, Hogan D, et al. Global, regional, and national causes of under-5 mortality in 2000-15: an updated systematic analysis with implications for the Sustainable Development Goals. *Lancet (London, England)*. 2016;388(10063):3027-3035. doi:10.1016/S0140-6736(16)31593-8
- 72. Selewski DT, Charlton JR, Jetton JG, et al. Neonatal Acute Kidney Injury. *Pediatrics*. 2015;136(2):e463-73. doi:10.1542/peds.2014-3819
- 73. Al Yazidi G, Boudes E, Tan X, Saint-Martin C, Shevell M, Wintermark P. Intraventricular hemorrhage in asphyxiated newborns treated with hypothermia: a look into incidence, timing and risk factors. *BMC Pediatr.* 2015;15:106. doi:10.1186/s12887-015-0415-7
- 74. Szpecht D, Frydryszak D, Miszczyk N, Szymankiewicz M, Gadzinowski J. The incidence of severe intraventricular hemorrhage based on retrospective analysis of 35939 full-term newborns-report of two cases and review of literature. *Childs Nerv Syst.* 2016;32(12):2447-2451. doi:10.1007/s00381-016-3164-5
- 75. Gunn AJ, Thoresen M. Neonatal encephalopathy and hypoxic-ischemic encephalopathy. *Handb Clin Neurol.* 2019;162:217-237. doi:10.1016/B978-0-444-64029-1.00010-2
- 76. De Haan TR, Langeslag J, van der Lee JH, van Kaam AH. A systematic review comparing neurodevelopmental outcome in term infants with hypoxic and vascular brain injury with and without seizures. *BMC Pediatr.* 2018;18(1):147. doi:10.1186/s12887-018-1116-9
- 77. Marret S, Vanhulle C, Laquerriere A. Pathophysiology of cerebral palsy. *Handb Clin Neurol*. 2013;111:169-176. doi:10.1016/B978-0-444-52891-9.00016-6
- LaRosa DA, Ellery SJ, Walker DW, Dickinson H. Understanding the Full Spectrum of Organ Injury Following Intrapartum Asphyxia. *Front Pediatr.* 2017;5:16. doi:10.3389/ fped.2017.00016
- 79. Perined. Perinatale Zorg in Nederland Anno 2018. Landelijke Perinatale Cijfers En Duiding.; 2019.

- 80. McNamara HM, Dildy GA 3rd. Continuous intrapartum pH, pO2, pCO2, and SpO2 monitoring. *Obstet Gynecol Clin North Am.* 1999;26(4):671-693. doi:10.1016/s0889-8545(05)70106-6
- 81. East CE, Begg L, Colditz PB, Lau R. Fetal pulse oximetry for fetal assessment in labour. *Cochrane database Syst Rev.* 2014;(10):CD004(10):CD004075. doi:10.1002/14651858. CD004075.pub4 [doi]
- 82. Dildy GA, van den Berg PP, Katz M, et al. Intrapartum fetal pulse oximetry: fetal oxygen saturation trends during labor and relation to delivery outcome. *Am J Obstet Gynecol.* 1994;171(3):679-684. doi:10.1016/0002-9378(94)90081-7
- 83. Ayres-de-Campos D, Spong CY, Chandraharan E, Panel FIFMEC. FIGO consensus guidelines on intrapartum fetal monitoring: Cardiotocography. *Int J Gynaecol Obstet*. 2015;131(1):13-24. doi:10.1016/j.ijgo.2015.06.020 [doi]
- 84. Neilson JP. Fetal electrocardiogram (ECG) for fetal monitoring during labour. *Cochrane database Syst Rev.* 2003;(2):CD0001(2):CD000116. doi:10.1002/14651858. CD000116 [doi]
- 85. Weyrich J, Ortiz JU, Muller A, et al. Intrapartum PRSA: a new method to predict fetal acidosis?-a case-control study. *Arch Gynecol Obstet*. 2020;301(1):137-142. doi:10.1007/s00404-019-05419-y
- 86. Fletcher AJW, Gardner DS, Edwards CMB, Fowden AL, Giussani DA. Development of the ovine fetal cardiovascular defense to hypoxemia towards full term. *Am J Physiol Heart Circ Physiol*. 2006;291(6):H3023-34. doi:10.1152/ajpheart.00504.2006
- 87. Devane D, Lalor JG, Daly S, McGuire W, Cuthbert A, Smith V. Cardiotocography versus intermittent auscultation of fetal heart on admission to labour ward for assessment of fetal wellbeing. *Cochrane database Syst Rev.* 2017;1:CD005122. doi:10.1002/14651858. CD005122.pub5 [doi]
- 88. Alfirevic Z, Devane D, Gyte GM, Cuthbert A. Continuous cardiotocography (CTG) as a form of electronic fetal monitoring (EFM) for fetal assessment during labour. *Cochrane database Syst Rev.* 2017;2:CD006066. doi:10.1002/14651858.CD006066.pub3 [doi]
- 89. Elliott C, Warrick PA, Graham E, Hamilton EF. Graded classification of fetal heart rate tracings: association with neonatal metabolic acidosis and neurologic morbidity. *Am J Obstet Gynecol.* 2010;202(3):258.e1-8. doi:10.1016/j.ajog.2009.06.026
- 90. American College of Obstetricians and Gynecologists. ACOG Practice Bulletin No. 106: Intrapartum fetal heart rate monitoring: nomenclature, interpretation, and general management principles. *Obstet Gynecol.* 2009;114(1):192-202. doi:10.1097/ AOG.0b013e3181aef106
- 91. Schiermeier S, Pildner von Steinburg S, Thieme A, et al. Sensitivity and specificity of intrapartum computerised FIGO criteria for cardiotocography and fetal scalp pH during labour: multicentre, observational study. *BJOG*. 2008;115(12):1557-1563. doi:10.1111/j.1471-0528.2008.01857.x
- 92. O'Mahony F, Hofmeyr GJ, Menon V. Choice of instruments for assisted vaginal delivery. *Cochrane database Syst Rev.* 2010;(11):CD005455. doi:10.1002/14651858.CD005455. pub2
- Simpson KR. Intrauterine resuscitation during labor: review of current methods and supportive evidence. J Midwifery Womens Health. 2007;52(3):229-237. doi:S1526-9523(06)00628-3 [pii]

- 94. Ekeus C, Hogberg U, Norman M. Vacuum assisted birth and risk for cerebral complications in term newborn infants: a population-based cohort study. *BMC Pregnancy Childbirth*. 2014;14:36. doi:10.1186/1471-2393-14-36
- 95. Bullens LM, van Runnard Heimel PJ, van der Hout-van der Jagt MB, Oei SG. Interventions for Intrauterine Resuscitation in Suspected Fetal Distress During Term Labor: A Systematic Review. *Obstet Gynecol Surv.* 2015;70(8):524-539. doi:10.1097/ OGX.00000000000215 [doi]
- Simpson KR, James DC. Efficacy of intrauterine resuscitation techniques in improving fetal oxygen status during labor. *Obstet Gynecol.* 2005;105(6):1362-1368. doi:105/6/1362 [pii]
- 97. Vasicka A, Quilligan EJ, Aznar R, Lipsitz PJ, Bloor BM. Oxygen tension in maternal and fetal blood, amniotic fluid, and cerebrospinal fluid of the mother and the baby. *Am J Obstet Gynecol*. 1960;79:1041-1047. doi:0002-9378(60)90508-1 [pii]
- 98. McNamara H, Johnson N, Lilford R. The effect on fetal arteriolar oxygen saturation resulting from giving oxygen to the mother measured by pulse oximetry. *Br J Obstet Gynaecol*. 1993;100(5):446-449.
- Sorensen A, Peters D, Simonsen C, et al. Changes in human fetal oxygenation during maternal hyperoxia as estimated by BOLD MRI. *Prenat Diagn*. 2013;33(2):141-145. doi:10.1002/pd.4025
- 100. Althabe Jr O, Schwarcz RL, Pose S V, Escarcena L, Caldeyro-Barcia R. Effects on fetal heart rate and fetal pO2 of oxygen administration to the mother. *Am J Obstet Gynecol.* 1967;98(6):858-870. doi:0002-9378(67)90205-0 [pii]
- 101. Haydon ML, Gorenberg DM, Nageotte MP, et al. The effect of maternal oxygen administration on fetal pulse oximetry during labor in fetuses with nonreassuring fetal heart rate patterns. *Am J Obstet Gynecol*. 2006;195(3):735-738. doi:S0002-9378(06)00867-2 [pii]
- 102. Gare DJ, Shime J, Paul WM, Hoskins M. Oxygen administration during labor. *Am J Obstet Gynecol.* 1969;105(6):954-961. doi:0002-9378(69)90104-5 [pii]
- 103. Dildy GA, Clark SL, Loucks CA. Intrapartum fetal pulse oximetry: the effects of maternal hyperoxia on fetal arterial oxygen saturation. *Am J Obstet Gynecol*. 1994;171(4):1120-1124. doi:0002-9378(94)90048-5 [pii]
- 104. Hidaka A, Komatani M, Ikeda H, Kitanaka T, Okada K, Sugawa T. A comparative study of intrauterine fetal resuscitation by beta-stimulant and O2 inhalation. *Asia-Oceania J Obstet Gynaecol.* 1987;13(2):195-200.
- 105. Khazin AF, Hon EH, Hehre FW. Effects of maternal hyperoxia on the fetus. I. Oxygen tension. *Am J Obstet Gynecol*. 1971;109(4):628-637. doi:0002-9378(71)90639-9 [pii]
- 106. Thorp JA, Trobough T, Evans R, Hedrick J, Yeast JD. The effect of maternal oxygen administration during the second stage of labor on umbilical cord blood gas values: a randomized controlled prospective trial. *Am J Obstet Gynecol*. 1995;172(2 Pt 1):465-474. doi:0002-9378(95)90558-8 [pii]
- 107. Qian G, Xu X, Chen L, et al. The effect of maternal low flow oxygen administration during the second stage of labour on umbilical cord artery pH: a randomised controlled trial. *BJOG*. 2017;124(4):678-685. doi:10.1111/1471-0528.14418 [doi]
- 108. Sirimai K Boriboonhirunsarn D. AR. The correlation of intrapartum maternal oxygen administration and umbilical cord blood gas values. *Acta Obstet Gynecol Scand Suppl.* 1997;76(167:2):

- Raghuraman N, Wan L, Temming LA, et al. Effect of Oxygen vs Room Air on Intrauterine Fetal Resuscitation: A Randomized Noninferiority Clinical Trial. JAMA Pediatr. 2018;179(9):818-823. doi:10.1001/jamapediatrics.2018.1208 [doi]
- 110. Nesterenko TH, Acun C, Mohamed MA, et al. Is it a safe practice to administer oxygen during uncomplicated delivery: a randomized controlled trial? *Early Hum Dev.* 2012;88(8):677-681. doi:10.1016/j.earlhumdev.2012.02.007 [doi]
- 111. Hamel MS, Anderson BL, Rouse DJ. Oxygen for intrauterine resuscitation: of unproved benefit and potentially harmful. *Am J Obstet Gynecol*. 2014;211(2):124-127. doi:10.1016/j.ajog.2014.01.004 [doi]
- 112. Khaw KS, Wang CC, Ngan Kee WD, Pang CP, Rogers MS. Effects of high inspired oxygen fraction during elective caesarean section under spinal anaesthesia on maternal and fetal oxygenation and lipid peroxidation. *Br J Anaesth*. 2002;88(1):18-23. doi:S0007-0912(17)36557-1 [pii]
- 113. Torres-Cuevas I, Parra-Llorca A, Sanchez-Illana A, et al. Oxygen and oxidative stress in the perinatal period. *Redox Biol.* 2017;12:674-681. doi:S2213-2317(17)30057-5 [pii]
- 114. Dede FS, Guney Y, Dede H, Koca C, Dilbaz B, Bilgihan A. Lipid peroxidation and antioxidant activity in patients in labor with nonreassuring fetal status. *Eur J Obstet Gynecol Reprod Biol.* 2006;124(1):27-31. doi:S0301-2115(05)00210-1 [pii]
- 115. Blackburn S. Free radicals in perinatal and neonatal care, part 2: oxidative stress during the perinatal and neonatal period. *J Perinat Neonatal Nurs.* 2006;20(2):125-127. doi:00005237-200604000-00005 [pii]
- 116. Longini M, Belvisi E, Proietti F, Bazzini F, Buonocore G, Perrone S. Oxidative Stress Biomarkers: Establishment of Reference Values for Isoprostanes, AOPP, and NPBI in Cord Blood. *Mediators Inflamm*. 2017;2017:1758432. doi:10.1155/2017/1758432 [doi]
- 117. Vento M, Sastre J, Asensi MA, Vina J. Room-air resuscitation causes less damage to heart and kidney than 100% oxygen. *Am J Respir Crit Care Med.* 2005;172(11):1393-1398. doi:200412-1740OC [pii]
- 118. Saugstad OD, Ramji S, Soll RF, Vento M. Resuscitation of newborn infants with 21% or 100% oxygen: an updated systematic review and meta-analysis. *Neonatology*. 2008;94(3):176-182. doi:10.1159/000143397 [doi]
- 119. Khaw KS, Ngan Kee WD. Fetal effects of maternal supplementary oxygen during Caesarean section. *Curr Opin Anaesthesiol.* 2004;17(4):309-313. doi:10.1097/01. aco.0000137089.37484.5e [doi]
- 120. Garite TJ, Nageotte MP, Parer JT. Should we really avoid giving oxygen to mothers with concerning fetal heart rate patterns? *Am J Obstet Gynecol*. 2015;212(4):45-60, 459.e1. doi:10.1016/j.ajog.2015.01.058 [doi]
- 121. The American College of Obstetricians and Gynecologists. Practice bulletin no. 116: Management of intrapartum fetal heart rate tracings. *Obstet Gynecol*. 2010;116(5):1232-1240. doi:10.1097/AOG.0b013e3182004fa9
- 122. Liston R, Crane J, Hughes O, et al. Fetal health surveillance in labour. J Obstet Gynaecol Can. 2002;24(4):342-355. doi:S1701-2163(16)30628-4 [pii]
- 123. National Collaborating Centre for Women's and Children's Health (UK). *Intrapartum Care: Care of Healthy Women and Their Babies During Childbirth.*; 2014.
- 124. Nederlandse Vereniging voor Obstetrie en Gynaecologie. NVOG-Richtlijn Intrapartum Foetale Bewaking à Terme.; 2019.
- 125. Fawole B, Hofmeyr GJ. Maternal oxygen administration for fetal distress. *Cochrane database Syst Rev.* 2012;12:CD000136. doi:10.1002/14651858.CD000136.pub2 [doi]





Speckle tracking echocardiography in hypertensive pregnancy disorders: a systematic review

S Moors NHM van Oostrum C Rabotti X Long MEMH Westerhuis HMC Kemps SG Oei JOEH van Laar

Obstetrical & Gynecological Survey 2020 Aug;75(8):497-509.

ABSTRACT

Importance

Hypertensive pregnancy disorders (HPD) are associated with an increased risk of longterm cardiovascular disease. Speckle tracking echocardiography (STE) might be useful in the early detection of pre-clinical cardiac changes in women with hypertensive pregnancy disorders.

Objective

To study whether STE is a suitable method to detect differences in cardiac function in pregnant women with HPD compared to normotensive pregnant women or between women with a history of a pregnancy complicated by HPD compared to women with a history of an uncomplicated pregnancy.

Evidence Acquisition

The databases Medline, EMBASE, and Central were systematically searched for studies comparing cardiac function measured with STE in pregnant women with HPD or women with a history of HPD and women with (a history of) normotensive pregnancies.

Results

The search identified 16 studies, including 870 women with (a history of) HPD and 693 normotensive controls. Most studies during pregnancy (n=12/13) found a decreased left ventricle (LV) global longitudinal strain (GLS) in HPD compared to normotensive pregnant controls. LV global radial strain (GRS) and LV global circumferential strain (GCS) are decreased in women with early-onset and severe preeclampsia. Women with a history of early-onset preeclampsia show lasting myocardial changes, with significantly decreased LV-GLS, LV-GCS, and LV-GRS.

Conclusions and relevance

LV-GLS is significantly decreased in pregnant women with HPD compared to normotensive pregnant women. Other deformation values show a significant decrease in women with severe or early-onset preeclampsia, with lasting myocardial changes after early-onset preeclampsia.

INTRODUCTION

Hypertensive pregnancy disorders (HPD) are one of the most common medical problems encountered during pregnancy, affecting 5-10% of pregnancies. ¹⁻⁴

HPD, defined as gestational hypertension (GH), chronic hypertension (CH), preeclampsia (PE), or superimposed PE.⁵ provoke serious maternal and fetal complications and are the leading cause of maternal death worldwide. 5-7 Besides increased maternal cardiovascular complications in pregnancy. HPD are also associated with an increased risk of longterm cardiovascular disease (CVD).⁸⁻¹⁰ Failure of the maternal cardiovascular system to adapt to pregnancy is hypothesized to be the primary mechanism leading to HPD. ^{11,12} Due to this inability to adapt, the maternal myocardium changes subtly in shape, size, and function.^{11,13-15} Early detection of these subtle changes, with the institution of appropriate screening and treatment, may reduce the risk of future CVD.^{16,17} Myocardial changes can be identified by ultrasound.^{10,12,18} However, conventional echocardiography, measuring left ventricular ejection fraction (LVEF) and diastolic function, was shown to be unsuitable for the detection of early subclinical myocardial changes, as these measures provide an indirect estimate of myocardial contractile function and change late in the cascade of myocardial dysfunction due to compensatory mechanisms.^{17,19} Speckle tracking echocardiography (STE), a relatively new echocardiographic method, could overcome this problem.

STE is a grey-scale based ultrasound technique, based on frame-by-frame tracking of acoustic reflections, speckles. STE is increasingly used in the assessment of left ventricular cardiac function,^{20,21} as it has advantages over conventionally used techniques.^{17,19,22,23} It directly quantifies the extent of myocardial contraction function and STE has equal to superior reproducibility.^{24,25} STE is proven to be suitable for the detection of subclinical cardiac changes.^{12,17,20,22} STE abnormalities are also shown to have prognostic value in patients with hypertension and hypertensive heart disease.^{26,27} Therefore, STE might also be useful in the detection of cardiac changes in HPD affected mothers. The aim of this systematic review is to study whether STE detects differences in cardiac function between pregnant women with HPD compared to normotensive pregnant women and between women with a history of a pregnancy complicated by HPD compared to women with a history of an uncomplicated pregnancy.

MATERIALS AND METHODS

Inclusion and exclusion criteria

All published studies that compared cardiac STE features between human females with (a history of) HPD and a normotensive control group were included in this review. We included studies that compared pregnant women with HPD to normotensive pregnant

women and we included studies that compared women with a history of a pregnancy complicated by HPD to women with a history of an uncomplicated pregnancy.

Studies that compared pregnant women with HPD with normotensive non-pregnant women were excluded. Studies that used other ultrasound techniques than STE to measure maternal heart function were excluded. Furthermore, review studies, guidelines, editorials, comments and conference abstracts were also excluded. No language restrictions or restrictions imposed on year of publication were applied. In the event an article in a language other than English or Dutch was found eligible based on title or abstract, the full text was retrieved and translated by a native speaker.

Search strategy

The databases of Medline (Pubmed), EMBASE, and CENTRAL were systematically searched from inception up to and including September 2019. The search consisted of the following terms and a wide variety of their synonyms: preeclampsia, hypertensive pregnancy disorders, speckle tracking echocardiography, and strain. A professional medical research librarian assisted to set the search strategy. The search strings are shown in Appendix S1. The review process was conducted according to the Preferred Reporting Items for Systematic Reviews and Meta-Analyses checklist (PRISMA),²⁸ and prospectively registered in the international prospective register of systematic reviews, PROSPERO (*CRD42019124031, Available from:*

http://www.crd.york.ac.uk/PROSPERO/display_record.php?ID=CRD42019124031)

Study selection and data extraction

Two reviewers, SM and NvO, independently reviewed the titles and abstracts to judge their eligibility for inclusion. The full text of these potentially eligible studies was retrieved and independently assessed for eligibility. Disagreement was solved by consensus; if needed, a third reviewer, MW, was consulted. After, we hand-searched all references and related articles of the selected articles to identify additional relevant publications.

Relevant information was extracted from each paper by two researchers (SM and NvO) independently, using a pre-determined form. The retrieved data comprised study characteristics such as year of publication, number of included women, in- and exclusion criteria and methods including study design, timing of echocardiography, and ultrasound device and STE software used.

Outcome measures

STE is an imaging technique that analyzes the motion of tissues in the heart by using the naturally occurring speckle pattern in the myocardium resulting from scattering of the ultrasound beam by the tissue.^{19,29} The STE software measures the change in distance between speckles over time, which is expressed as strain.^{19,21} Strain rate (SR) is the rate at which the myocardial deformation occurs in 1/sec. In cardiac STE,

strain represents the myocardial shortening and lengthening during a cardiac cycle of contraction and relaxation and can be measured in different directions where it is expressed as a percentage.^{19,21,29} The outcome measures of interest were the following parameters measured by STE; left ventricular (LV) global longitudinal strain (GLS), LV global radial strain (GRS), LV global circumferential strain (GCS), and SR. The GLS is measured in the apical long-axis images and the short-axis images are used to measure GRS and GCS.^{19,21,29}

Quality assessment and data analysis

The methodological quality of the eligible studies was assessed using the Newcastle-Ottawa Quality Assessment Form for Case-Control Studies (NOS) according to the guidelines of the Dutch Cochrane Center. ³⁰⁻³² The NOS is a qualitative assessment tool for observational studies. Assessment is done using the following three dimensions; selection, comparability, and exposure. A maximum of nine stars can be awarded to a single study.³⁰ The studies were classified in categories as suggested by Losilla et al.,³³ by which studies with 0-3, 4-6, or 7-9 stars were classified as low, moderate, and high quality, respectively. Two authors (SM and NvO) completed the quality and risk of bias assessment independently. Disagreement was solved by consensus; if needed, a third reviewer, MW, was consulted. Data from the included studies were aggregated to provide a narrative synthesis of the findings.

RESULTS

Search results

A total of 200 studies were identified through database searching. The process of study inclusion according to the PRISMA statement is shown in figure 1. After removing duplicates, 158 studies remained. From these articles, 63 articles were found eligible for full-text assessment. When multiple articles from the same authors were found eligible, authors were contacted to ensure no duplicate study populations would be included. One study was excluded for this reason.³⁴ After full-text assessment, 16 articles were included in this review.^{12,16,18,35-47} No additional articles were identified by hand searching the references of the included articles. One article was published in Chinese⁴² and translated to English by a native Chinese speaker.

The study characteristics of the included studies are summarized in Table 1. All included studies had an observational case-control design, of which three were longitudinal studies. HPD were defined according to the definition by the International Society for the study of Hypertension in Pregnancy, ^{12,38-40,43,47} American College of Obstetricians and Gynecologists, ^{16,18,35-37,46} the National High Blood Pressure Education Program Working Group on High Blood Pressure in Pregnancy, ^{41,45} the definition as stated in Obstetrics & Gynecology by Xie and Gou, ⁴² or by the classification by Davey et al.^{44,48}

The 16 selected studies included a total number of 1563 patients. Ten studies evaluated women during their pregnancy, including 601 women with a pregnancy complicated by HPD and 523 controls with a normotensive pregnancy. In three studies cardiac function was measured postpartum only, including 128 women with a history of HPD and 86 controls with a history of normotensive pregnancy. Three studies analyzed women both during their pregnancy and as well as postpartum. Of these, 141 women had HPD and 84 had a normotensive pregnancy. The characteristics and quality of the included studies are presented in Table 1.

The type of ultrasound machine and STE software used to perform STE varied among the studies. The patients in the studies performed during pregnancy varied with regard to the gestational age included (Table 1). The included gestational ages varied from 23 to 42 weeks. Most studies included both preterm and term women,^{16,18,35-39,41,42,45-47} whereas one study included term women only.¹² The population in the studies concerning postpartum women showed a wide variation with regard to the period between delivery and the timing of the study measurement, ranging from 6 weeks to 13 years postpartum.^{35,40,43-46}

Three studies differentiated between early-onset PE (EO-PE) and late-onset PE (LO-PE), i.e. occurring before or after 34 weeks of gestation, respectively.^{39,40,43}



Figure 1. PRISMA Flow diagram demonstrating an overview of the selection process.

lable 1. Char	acteristics ar	nd Quality of Includ€	ed Studies.				
Author Year	Total number of women	Population (n=)	HPD definition Inclusion and exclusion criteria	Methods	Mean GA (weeks) or period postpartum at	Quality assessment and risk of bias (NOS)	Losilla quality categories
Studies exami	ining women c	during their pregnanc					
			 HPD definition: ACOG Inclusion: Age 18-42, GA 28-38 weeks 				
			- Exclusion: preexisting hypertension, cardiac,	Study design: case control			
			pulmonary, renal disease or other pathology that may	Machine: Vivide TM			
Ajmi et al. ¹⁸ 2018	07	Cases: HPD (30)	influence ultrasound data. EF <55%, difficult interpretation or noor subitived shotsorrable	E9, GE STE software: not described	Cases: 32	(0/0/2/2	High audity
0107			- HPD definition: ISSHP	Study design: case		1 /01 21 21	
			 Inclusion: GA at term, before start of any antihypertensive 	control			
Buddeberg et	±.)		medication	Machine: Vivd Q, GE.			
al.** 2018	70	Control: NTP (40)	 Exclusion: any caralovascular co-morbidity 	S I E SOTTWARE: EchoPac	Control: 38.3±1.0	5 (3/0/2)	ivioderate quality
			- HPD definition: NHBPEPWG	Study design: case			
			on HBPP - Inclusion: nregnancy	CONTROL			
			- Exclusion: DM, essential	Machine: Vivid 7, GE			
Cho et al. ⁴¹		Cases: GH (106)	hypertension, or symptomatic	STE software:	Cases: 33.3±3.6		Moderate
2011	199	Control: NTP (93)	coronary artery disease	EchoPac	Control: 35.1±3.4	4 (2/0/2)	quality
Author Year	Total number of women	Population (n=)	HPD definition Inclusion and exclusion criteria	Methods	Mean GA (weeks) or period postpartum at assessment	Quality assessment and risk of bias (NOS)	Losilla quality categories
		Cases: - FO-DF (43)		Study design: case	Cases: EO-PE: 28.9±2.7 LO-PE: 36.4±1.3		
Cong et al. ³⁹ 2015	165	- LO- PE (41) - LO- PE (41) Control: - NTP < 34weeks (41) - NTP > 34weeks (40)	 HPD definition: ISSHP Inclusion: singleton pregnancy Exclusion: preexisting medical conditions 	Machine : Vivid E9, GE. STE software : Echopae.	Control <34: 28.2±2.9 Control ≥34 : 36.4±1.3	7 (3/2/2)	High quality
				Study design: case			
			 HPD definition: ISSHP Inclusion: singleton pregnancy Exclusion: poor quality images, 	Machine: Vivid E9, GE.			
Cong et al. ³⁸ 2018	86	Cases: PE (45) Control: NTP (41)	smoking history or any previous medical condition	STE software : EchoPAC.	Cases: 31.8±4.2 Control: 32.7±3.7	8 (3/2/3)	High quality
				Study design: Case control			

Cases: 34.7±5.0 **Control:** 34.7±3.4 7 (3/2/2) High quality

Machine: Vivid E9, GE. STE software: Not described

> - HPD definition: ISSHP Cases: s-PE (33) - Inclusion: nonsmoking Control: NTP (20) - Exclusion: comorbidities

> > Pan et al.⁴⁷ 2019 53

36

Author Year	Total number of women	Population (n=)	HPD definition Inclusion and exclusion criteria	Methods	Mean GA (weeks) or period postpartum at assessment	Quality assessment and risk of bias (NOS)	Losilla quality categories
Shahul et al. ³⁷ 2012	80	Cases: PE (11) Control: NTP (17)	 HPD definition: ACOG HPD definition: ACOG Inclusion: Age ≥18, singleton pregnancy, GA 24-41weeks Exclusion: pre-existing CVD, pulmonary disease, DM, or poor image quality 	Study design: case control Machine: Siemens X-300. STE software: Tomtec.	Cases: 36.6 (32.7- 37.4) Control: 38.0 (35.6-39.6)	5 (3/0/2)	Moderate quality
Shahul et al. ³⁶ 2016	167	Cases: PE (62) Control: NTP (105)	 HPD definition: ACOG Inclusion: Age ≥18, singleton pregnancy, GA <41weeks Exclusion: preexisting cardiomyopathy, ischemic or valvular heart disease, pulmonary disease, DM, or labor. 	Study design: case control Machine: Philips CX-50. STE software: Autostrain, Tomtec.	Cases: 32.8±3.7 Control: 30.7±4.3	5 (3/0/2)	Moderate quality
Vaught et al ¹⁶ 2018	6 6	Cases: s- PE (63) Control: NTP (36)	 HPD definition: ACOG Inclusion: Singleton pregnancy, GA >23 weeks Exclusion: SLE, congenital or valvular heart disease, APS, cardiomyopathy, pulmonary hypertension or embolism, cardiac surgery, connective tissue or interstitial lung disease, poor image quality 	Study design: case control Machine: GE or Philips ultrasound machine STE software: Epsilon software.	Cases: 33.1±3.6 Control: 31.8±4.9	6 (3/1/2)	Moderate quality
Table 1. Conti	nued.	• • •					
Author Year	Total number of women	Population (n=)	HPD definition Inclusion and exclusion criteria	Methods	Mean GA (weeks) or period postpartum at assessment	Quality assessment and risk of bias (NOS)	Losilla quality categories
Xia et al. ⁴² 2017	197	Cases: - m-PE (73) - s-PE (64) Control: NTP (60)	 HPD definition: as in Obstetrics & Gynecology by Xie and Gou. Inclusion: gestational age ≥32 weeks Exclusion: congenital heart disease, other serious comorbidities 	Study design: case control Machine: Vivid E9, GE Software: EchoPac	Mild PE: 35.9±3.3 Severe PE: 35.2±3.1 Control: 34.9±2.7	5 (3/0/2)	Moderate quality
Studies examir	ling women p	oostpartum			Mean period after index pregnancy 11.2±0.6 year		
Al-Nashietal ⁴⁴ 2016	31	History of: Cases: PE (15) Control: NTP (16)	 HPD definition: ACOG Inclusion: primiparous women during index pregnancy Exclusion: smoking, cardiovascular risk factors 	Study design: case control Machine: Vivid 7, GE. STE software: EchoPac.	Years after last pregnancy: Cases: 7,9±3,3 year Control:6.6±2,4 year	8 (3/2/3)	High quality

39

Table 1. Conti	inued.						
Author Year	Total number of women	Population (n=)	HPD definition Inclusion and exclusion criteria	Methods	Mean GA (weeks) or period postpartum at assessment	Quality assessment and risk of bias (NOS)	Losilla quality categories
Clemmensen et al. ⁴⁰ 2018	63	History of: Cases: - EO-PE (31) - LO-PE (22) Control: - NTP (40)	 HPD definition: ISSHP HPD definition: ISSHP Inclusion: women who gave birth between 1998-2008 Exclusion: new pregnancy, breastfeeding, natural menopause, or address >100km 	Study design: case control Machine: Vivid E9, GE. STE software: EchoPac	Time since index pregnancy in years EO-PE cases: 13±4 LO-PE cases: 12±3 Control: 12±3		High quality
Orabona et al. ⁴³ 2017	06	History of: Cases: - EO-PE (30) - LO-PE (30) Control: NT (30)	 HPD definition: ISSHP Inclusion: women who had normal blood pressure and no pathological proteinuria 6 months after delivery Exclusion: smoking, dyslipidemia, obesity, DM, chronic hypertension, cardiopathy, nephropathy, immune disorders, PE superimposed on CH, multiple pregnancy, chromosomopathy, or fetal malformation. 	Study design: case control Machine: Vivid 7, GE . STE software: not described	Time since delivery in years: 6 months – 4 years Mean time since delivery: EO-PE: 2.3±0.7 LO-PE: 2.2±0.8 Control: 2.2±0.6	9 (4/2/3)	High quality
Table 1. Cont Author Year	inued. Total number of women	Population (n=)	HPD definition Inclusion and exclusion criteria	Methods	Mean GA (weeks) or period postpartum at assessment	Quality assessment and risk of bias (NOS)	Losilla quality categories
Levine et al ⁴⁶ 2019	58 58	Cases: s-PE (29) Control: NTP (29)	 Brancy and postpartum HPD definition: ACOG HPD definition: ACOG Inclusion: Ac ≥ 18 Exclusion: Ac ≥ 18 Exclusion: preexisting CVD, chronic hypertension without superimposed PE, or multiple gestations 	Study design: Longitudinal case control Machine: Vivid E9 and E95, GE. STE software: Tomtec	GA at assessment during pregnancy Cases: 31.3±3.9 Control: 31.7±3.6 Time since delivery at assessment postpartum Cases: 6 (5-6) Control: 7 (6-9) GA at assessment during pregnancy PE: 32.2 (29.1- 36.1) GH or CH: 37.9	7 (3/2/2)	High quality
Shahul et al. ³⁵ 2018	85	Cases: PE (32) Control: - GH or CH (28) - NTP (25)	 HPD definition: ACOG Inclusion: Age ≥18, singleton pregnancy, GA <41weeks Exclusion: preexisting cardiomyopathy, ischemic or valvular heart disease, pulmonary disease, DM, or women in labor. 	Study design: Longitudinal case control Machine: Philips CX-50. STE software: Autostrain, Tomtec.	(33.3-39.2) Control : 30.7 (26.7-32.3) Time since delivery at assessment postpartum: 12 months	5 (3/0/2)	Moderate quality

40

4	2

Table 1. Co	ntinued.		
Author	Total	Population (n=)	HPD definit
Year	number of		Inclusion ar
	women		

Author	Total	Population (n=)	HPD definition	Methods	Mean GA	Quality	Losilla
Year	number o	f	Inclusion and exclusion criteria		(weeks) or period	assessment	quality
	women				postpartum at assessment	and risk of bias (NOS)	categories
					GA at assessment		
					during pregnancy		
			- HPD definition: NHBPEPWG	Study design:	PE: 32±6		
			on HBPP	Longitudinal case	GH: 34±3		
			 Inclusion: women aged ≥18, 	control	Control: 33±4		
			singleton pregnancy, and				
			nonsmoking	Machine: Siemens	Time since		
		Cases:	- Exclusion: gestational	S2000 (axius,	delivery at		
		- GH (27)	diabetes, previous history	Siemens)	assessment		
Yu et al. ⁴⁵		- PE (25)	of hypertension, CVD, poor	STE software: Axius,	postpartum:		Moderate
2018	82	Control: NTP (30)	images quality.	Siemens medical	3 months	5 (3/0/2)	quality

1= Diabetes Mellitus. GA= Gestational age. GH= Gestational hypertension. HPD= Hypertensive pregnancy disorders. LVD= Cardiovascular disease.
1= Diabetes Mellitus. GA= Gestational age. GH= Gestational hypertension. HPD= Hypertensive pregnancy disorders. ISSHP= International Society the study of Hypertension in Pregnancy. NHBPEPWG on HBPP= National High Blood Pressure Education Program Working Group on High Blood issure in Pregnancy. NTP= Normotensive pregnancy. PE= Preeclampsia. EO-PE= Early-onset PE. LO-PE= Late onset- PE. m-PE= Mild PE. s-PE= Severe SLE= Systemic lupus erythematosus. Data are mean+SD or median (international context). Pressure i PE. SLE= 9 DM= for the AC

Quality assessment

The risk of bias of the included studies is reported in Table 1. The risk of bias score varied between four and nine stars on the Newcastle-Ottawa Scale, which corresponds with moderate to high quality.

Echocardiographic measurements

An overview of STE results during pregnancy is presented in Table 2. Table 3 presents an overview of STE results postpartum.

Left ventricle global longitudinal strain (LV-GLS)

Results during pregnancy

All included studies reported results on LV-GLS. Twelve studies showed a significant decrease in LV-GLS in pregnant participants with HDP compared to normotensive participants.^{12,18,35-39,41,42,45-47} In one study, the same trend of decreased LV-GLS was observed in women with PE compared to normotensive pregnant women, however, this was not significant.¹⁶

Results postpartum

Six studies measured LV-GLS postpartum.^{35,40,43-46} The measurement timing ranged from 6 weeks⁴⁶ to 13 years⁴⁰ postpartum. Three studies, evaluating women up to one year postpartum, showed a significant decreased LV-GLS after a pregnancy complicated by PE compared to women after a normotensive pregnancy.^{35,45,46} One study, comparing women 11 years after a pregnancy complicated by PE with women with a history of a normotensive pregnancy, did not demonstrate a significant difference in LV-GLS.⁴⁴ Two studies differentiated between EO-PE and LO-PE. ^{40,43} Both studies showed a decreased LV-GLS after EO-PE, whereas LO-PE was shown not significantly different from normotensive controls. 40,43

Comparing women with a history of a pregnancy complicated by GH or CH to women with a history of a normotensive pregnancy, one study found a decreased LV-GLS in women with either GH or CH,³⁵ whereas another study showed no significant difference between GH and normotensive controls.⁴⁵

Left ventricle global radial strain (LV-GRS)

Results during pregnancy

LV-GRS was measured in five studies.^{18,37,39,42,45} LV-GRS was shown significantly lower in women with PE in two studies.^{37,45} whereas one study showed no significant difference.¹⁸ One study differentiated between EO-PE and LO-PE.³⁹ Compared to normotensive pregnant women, LV-GRS was significantly decreased in EO-PE but comparable in LO-PE.³⁹ In the fifth study, patients suffering from either severe or mild PE were compared with normotensive pregnant women. LV-GRS was only shown significantly lower in patients with severe PE.42

Comparable LV-GRS was shown in two studies in patients with GH compared to normotensive pregnant controls. ^{18,45}

Results postpartum

LV-GRS was measured postpartum in two studies. ^{43,45} The measurement timing ranged between 3 months⁴⁵ to 4 years⁴³ postpartum. Women suffering from PE or EO-PE showed decreased LV-GRS in two studies.^{43,45} After a pregnancy complicated by LO-PE or CH, however, LV-GRS was shown comparable to normotensive women.^{43,45}

Left ventricle global circumferential strain (LV-GCS)

Results during pregnancy

Six studies report results concerning LV-GCS during pregnancy.^{18,37,39,42,45,46} Comparing women with a pregnancy complicated by PE to normotensive pregnant women, three studies showed a significantly decreased LV-GCS in PE,^{37,39,45} whereas two studies showed no significant difference.^{18,46} The fifth study compared women with either mild or severe PE to normotensive pregnant controls.⁴² LV-GCS was only significantly decreased in severe PE.⁴²

In GH, LV-GCS was increased in one study 45 and comparable to normotensive controls in another study. 18

Results postpartum

LV-GCS was measured in three studies postpartum.^{43,45,46} The measurement timing ranged between 6 weeks⁴⁶ to 4 years⁴³ postpartum. Compared to a history of a normotensive pregnancy, LV-GCS was significantly decreased after PE in one study,⁴⁵ and comparable to a history of PE in one study.⁴⁶ One study differentiated between a history of EO-PE or LO-PE.⁴³ Compared to a history of a normotensive pregnancy, LV-GCS was significantly decreased after LO-PE.⁴³

LV-GCS was not shown significantly different between women with a history of GH and women with a history of a normotensive pregnancy.⁴⁵

Left ventricle strain rate (LV-SR)

Only one study reported on LV-SR differences between pregnant women with HPD and normotensive pregnant controls.¹² LV-SR was significantly lower in patients with PE compared to normotensive pregnant women. None of the studies including women postpartum reported results of LV-SR measurements.

Author Year	Ajmi 2018	Buddeberg 2018	Cho 2011	Cong 2015	Cong 2018	Levine 2019	Pan 2019	Shahul 2012	Shahul 2016	Shahul 2018	Vaught 2018	Xia 2017	Yu 2018
Number of participants	60	70	199	165	86	58	53	28	167	85	66	197	82
GA Cases	32	39±1	33±4	29±3/36±1	32±4	31±4	35±5	37	33±4	32	33±4	m-PE 36±3 pE 35+3	PE 32±6 GH 34+3
GA Control	33	38±2	35±3	28±3/36±1	33±4	31±4	35±3	38	31±4	31	32±5	35±3	33±4
LV-GLS	PE ↓ GH ↓	PE <	↑*O9H	PE 🔶	→ JA	S-PE↓	S-PE ↓	$\vdash FE \leftarrow$	→ JA	→ JA	S-PE =	PE↓	PE↓ GH ↓
LV-GRS	PE = GH =			EO-PE ↓, LO-PE =				→ FE ←				s-PE:m-PE: =	PE↓ GH =
LV-GCS	PE = GH =			PE 🔶		s-PE=		ÞE ←				s-PE:m-PE: =	PE↓ GH↑
SR		PE 🕹											

Table 2. Echocardiography features during pregnancy

↑: significant increase, ↓: significant decrease, =: no significant difference compared to the control group of normotensive pregnant women. Empty cell: parameter not analyzed in the study.GA: gestational age in weeks, PE: pre-eclampsia, GH: gestational hypertension, EO: early-onset, LO: late-onset, s-PE: severe PE, m-PE: mild PE, HPD: hypertensive pregnancy disorders. LV-GLS: left ventricular global longitudinal strain, LV-GRS: left ventricular global radial 10 developed PE. GH of whom et al. included patients with by Cho study k strain rate. SR: strain. ential : S: global circumfer strain, LV-GC

Author Year	Al-Nashi 2016	Clemmensen 2018	Levine 2019	Orabona 2017	Shahul 2018	Yu 2018
Number of participants	31	93	58	60	58	82
Mean period postpartum	11.2±0.6	EO-PE 13±4 year	6 weeks	EO-PE 2.3±0.7 year	1 year	3 months
Lases	11.2±0.6	LO-PE 12±3 Control 12±3	7 weeks	LU-PE 2.5±0.8 Control 2.2±0.6	1 year	3 months
Control						
LV-GLS	PE =	EO-PE &, LO-PE =	s-PE ↓	EO-PE &, LO-PE =	PE 4	PE+
SQU //I						
				LO-FL		GH =
LV-GCS			s-PE =	EO-PE 4, LO-PE =		ΡEΨ
						GH =
SR						

onset, LO: late-onset. s-PE: severe global radial strain, LV-GCS: global clampsia. GH; gestational hypertension, EO: early-global longitudinal strain, LV-GRS: left ventricular eclampsia. ase, = no significant airtere ed in the study. PE: pre-ecl .. LV-GLS: left ventricular g' analyzed ↑ significant increase, ↓ significant decreas pregnancy. Empty cell: parameter not analyze PE. HPD: hypertensive pregnancy disorders. I circumferential strain, SR: strain rate.

STE in HPD: a systematic review

DISCUSSION

Main findings

A systematic review concerning STE in women with (a history of) HPD compared to normotensive controls was performed. Our major finding was a significantly decreased LV-GLS in HPD during pregnancy in all studies but one. That study showed a trend towards decreased LV-GLS in PE, however, because multiple-testing correction was used in that study, this was not significant.¹⁶ The use of multiple-testing correction could explain why the results of that study were inconsistent with all other studies.

LV-GLS represents LV myocardial shortening in the longitudinal axis and is an important index of global LV function. LV-GLS is capable of early and accurate detection of cardiac alterations that may affect subendocardial longitudinal fibers.²⁹ These fibers are involved in the first, subclinical stages of several diseases such as ischemic injury and arterial hypertension, showing a reduced LV-GLS.

Furthermore, LV-GLS is associated with major adverse cardiac events in patients with asymptomatic hypertensive heart disease.²⁶ Therefore, a decreased LV-GLS in women with HPD can also be an indicator of subclinical deterioration of the myocardium and may be a useful tool for early detection of women at risk for cardiac dysfunction later in life.^{26,29} As the currently used cardiac function parameter LVEF is typically still normal in the subclinical phase of these diseases, LV-GLS could be a more useful tool in early detection.29

This review showed a difference between FO-PE and LO-PE postpartum, LV-GLS, LV-GRS, and LV-GCS were decreased in women with a history of a pregnancy complicated by EO-PE compared to normotensive controls, whereas LV-GLS, LV-GRS, and LV-GCS were shown comparable in women with a history of LO-PE and women with a history of a healthy pregnancy.^{40,43} This suggests lasting myocardial changes in women suffering from EO-PE in pregnancy, up to 13 years postpartum.

Growing evidence suggests that EO-PE and LO-PE have different etiologies and should be regarded as two different types of the disease.^{10,11,14,49} EO-PE and LO-PE have a different maternal cardiovascular adaptation to pregnancy.^{39,50} Also, EO-PE is associated with a higher risk of morbidity and mortality during pregnancy than LO-PE, which is probably due to differences in the severity of the disease. ^{9,51}

When distinguishing between the three different HPD (i.e. GH, CH, and PE), STE parameters seem to be altered more in PE than in GH or CH, both during pregnancy as postpartum. This is in line with the findings of the review by Castleman et al. about conventional echocardiography in HPD, which showed that if echocardiographic changes are seen in HPD, these changes are more severe in case of PE compared with GH.⁵² This

increased impact on the heart in PE is in line with the increased CVD incidence after a pregnancy complicated by PE compared to after GH or CH. 53

Although it is not well established whether cardiovascular derangement in HPD is a primary etiological factor or a secondary effect, the differences in STE results between the separate HPD attributes to our understanding of the differences between these disorders.

Strengths and limitations

One of the strengths of this systematic review is our rigorous search without restrictions on language or year of publication. It is conceivable that studies are more likely to be published in an international English-language journal if results are significant, whereas non-significant findings are more likely to be published in a local, non-English journal.³² Another strength of this review is the moderate to high quality of the included studies.^{30,33}

A limitation of this review is that the generalizability of the results remains unclear. Due to heterogeneity of the studies, a meta-analysis was not performed as suggested by the Cochrane handbook for systematic reviews of interventions.³²

Clinical implications and future research

The use of STE for various clinical settings has increased remarkably over recent years.²⁹ In the acute phase of myocardial infarction, reduced LV-GLS has been proven to be the single most powerful marker of manifest LV hemodynamic deterioration.⁵⁴ Moreover, in heart failure, LV-GLS is an independent predictor of mortality and a superior compared to other echocardiographic parameters. ^{55,56}

However, the use of STE in pregnancy is not very widespread in clinical practice yet. This review shows a decrease in cardiac function, especially in LV-GLS, starting during pregnancy and lasting up to 13 years after pregnancies complicated by HPD compared to normotensive controls. This could indicate that women with echocardiographic abnormalities measured during their pregnancy may benefit from more strict surveillance and treatment to decrease cardiovascular risks during life.

Since STE might be a tool to detect subclinical cardiac changes, more sensitive than conventional echocardiography,^{12,17,20,22} STE might be promising for prediction and prevention of CVD after HPD.

A recommendation for future research is to measure STE in mother and fetus simultaneously. Growing evidence shows that, like the mothers, the offspring of pregnancies complicated by HPD also have increased risk of developing CVD later in life compared to offspring of uncomplicated pregnancies.⁵⁷⁻⁶² A recent study by Yu et al. already showed signs of diastolic dysfunction in fetuses of mothers with PE compared to

fetuses of healthy mothers.⁶³ This study also shows that STE appears to be more sensitive than conventional echocardiography for the evaluation of fetal cardiac function.⁶³ Early identification of children with altered strain measurement during and after a pregnancy complicated by HPD might help to provide adequate screening and treatment for children at risk for CVD.

One study by Shahul et al. studied the predictive value of abnormal strain during pregnancy in women with CH. Women with CH who had a decreased LV-GLS mid-gestation, had a significantly higher risk of developing superimposed PE.⁶⁴ Future research should continue to focus on the clinical relevance of an abnormal strain during pregnancy, as well as the association between abnormal strain during pregnancy and the development of CVD later in life. Possibly, abnormal strain measurements might help to differentiate within the high-risk group of women with HPD, making appropriate screening and treatment possible based on each women's personal risk profile. Pregnancy may therefore be considered as a window of opportunity for improvement of future health.⁶⁵

CONCLUSION

STE can detect differences between pregnant women with HPD and normotensive pregnant women, mainly in LV-GLS. This parameter is significantly decreased in pregnant women with HPD compared to normotensive pregnant controls. Other deformation values show a significant decrease in women with severe or early-onset PE, with lasting myocardial changes after early-onset PE. Future research should focus on the clinical relevance of an abnormal strain during pregnancy, and its association with the development of CVD later in life.

ACKNOWLEDGMENTS

This research was performed within the framework of Eindhoven MedTech Innovation Center (e/MTIC). We thank drs. H.E.J. de Vries, a professional medical research librarian who helped us with the search.

APPENDIX 1

(("Pre-Eclampsia"[Mesh]) OR (pre eclampsia[tw] OR preeclampsia[tw] OR gestosis[tw] OR toxemia*[tw] OR toxicosis[tw]) OR (("Hypertension, Pregnancy-Induced"[Mesh]) OR ("Pregnancy"[Mesh] AND "Hypertension"[Mesh])) OR ((pregnan*[tiab] OR gestation*[tiab] OR maternal[tiab]) AND hypertens*[tiab])) AND ((speckle*[tw] OR strain*[tw]) AND (echocardiograph*[tw] OR imaging[tw]))

REFERENCES

- 1. Umesawa M, Kobashi G. Epidemiology of hypertensive disorders in pregnancy: Prevalence, risk factors, predictors and prognosis. *Hypertens Res.* 2017;40(3):213-220. doi: 10.1038/hr.2016.126 [doi].
- Hutcheon JA, Lisonkova S, Joseph KS. Epidemiology of pre-eclampsia and the other hypertensive disorders of pregnancy. *Best Pract Res Clin Obstet Gynaecol*. 2011;25(4):391-403. doi: 10.1016/j.bpobgyn.2011.01.006 [doi].
- 3. Wagner SJ, Barac S, Garovic VD. Hypertensive pregnancy disorders: Current concepts. *J Clin Hypertens (Greenwich)*. 2007;9(7):560-566.
- 4. Folk DM. Hypertensive disorders of pregnancy: Overview and current recommendations. *J Midwifery Womens Health*. 2018;63(3):289-300. doi: 10.1111/jmwh.12725 [doi].
- 5. ACOG practice bulletin no. 202: Gestational hypertension and preeclampsia. *Obstet Gynecol.* 2019;133(1):e-e25. doi: 10.1097/AOG.0000000003018 [doi].
- Souza JP, Gulmezoglu AM, Vogel J, et al. Moving beyond essential interventions for reduction of maternal mortality (the WHO multicountry survey on maternal and newborn health): A cross-sectional study. *Lancet.* 2013;381(9879):1747-1755. doi: 10.1016/S0140-6736(13)60686-8 [doi].
- Khan KS, Wojdyla D, Say L, Gulmezoglu AM, Van Look PF. WHO analysis of causes of maternal death: A systematic review. *Lancet*. 2006;367(9516):1066-1074. doi: S0140-6736(06)68397-9 [pii].
- 8. Wu P, Haththotuwa R, Kwok CS, et al. Preeclampsia and future cardiovascular health: A systematic review and meta-analysis. *Circ Cardiovasc Qual Outcomes*. 2017;10(2):10.1161/CIRCOUTCOMES.116.003497. Epub 2017 Feb 22. doi: e003497 [pii].
- Bellamy L, Casas JP, Hingorani AD, Williams DJ. Pre-eclampsia and risk of cardiovascular disease and cancer in later life: Systematic review and meta-analysis. *BMJ*. 2007;335(7627):974. doi: bmj.39335.385301.BE [pii].
- 10. Ghossein-Doha C, Hooijschuur MCE, Spaanderman MEA. Pre-eclampsia: A twilight zone between health and cardiovascular disease? *J Am Coll Cardiol*. 2018;72(1):12-16. doi: S0735-1097(18)34726-0 [pii].
- 11. Kalafat E, Thilaganathan B. Cardiovascular origins of preeclampsia. *Curr Opin Obstet Gynecol.* 2017;29(6):383-389. doi: 10.1097/GCO.00000000000419 [doi].
- 12. Buddeberg BS, Sharma R, O'Driscoll JM, Kaelin Agten A, Khalil A, Thilaganathan B. Cardiac maladaptation in term pregnancies with preeclampsia. *Pregnancy Hypertens*. 2018;13:198-203. doi: S2210-7789(18)30089-8 [pii].
- 13. Melchiorre K, Sutherland GR, Baltabaeva A, Liberati M, Thilaganathan B. Maternal cardiac dysfunction and remodeling in women with preeclampsia at term. *Hypertension*. 2011;57(1):85-93. http://ovidsp.ovid.com/ovidweb.cgi?T=JS&PAGE=reference&D=emed12&NEWS=N&AN=51165113. doi: //dx.doi.org/10.1161/HYPERTENSIONAHA.110.162321.
- Melchiorre K, Sutherland G, Sharma R, Nanni M, Thilaganathan B. Mid-gestational maternal cardiovascular profile in preterm and term pre-eclampsia: A prospective study. *BJOG*. 2013;120(4):496-504. doi: 10.1111/1471-0528.12068 [doi].

- 15. Melchiorre K, Sutherland GR, Watt-Coote I, Liberati M, Thilaganathan B. Severe myocardial impairment and chamber dysfunction in preterm preeclampsia. *Hypertens Pregnancy.* 2012;31(4):454-471. http://ovidsp.ovid.com/ovidweb.cgi?T=JS&PAGE=reference&D=emed13&NEWS=N&AN=365774621. doi: //dx.doi.org/10.3109/10641955.2012.697951.
- 16. Vaught AJ, Kovell LC, Szymanski LM, et al. Acute cardiac effects of severe pre-eclampsia. *J Am Coll Cardiol.* 2018;72(1):1-11. doi: S0735-1097(18)34723-5 [pii].
- 17. Visentin S, Palermo C, Camerin M, et al. Echocardiographic techniques of deformation imaging in the evaluation of maternal cardiovascular system in patients with complicated pregnancies. *BioMed Res Int*. 2017;2017:4139635. http://www.hindawi.com/journals/biomed/ http://ovidsp.ovid.com/ovidweb.cgi?T=JS&PAGE=reference&D=emexa&NEWS=N&AN=618228494. doi: //dx.doi.org/10.1155/2017/4139635.
- 18. Ajmi H, Abid D, Milouchi S, et al. Interest of speckle tracking in the detection of cardiac involvement in pregnant women with hypertensive disorder. *Pregnancy Hypertens*. 2018;11:136-141. doi: S2210-7789(17)30070-3 [pii].
- 19. Bansal M, Kasliwal RR. How do I do it? speckle-tracking echocardiography. *Indian Heart* J. 2013;65(1):117-123. doi: 10.1016/j.ihj.2012.12.004 [doi].
- 20. D'Andrea A, Radmilovic J, Ballo P, et al. Left ventricular hypertrophy or storage disease? the incremental value of speckle tracking strain bull's-eye. *Echocardiography*. 2017;34(5):746-759. doi: 10.1111/echo.13506 [doi].
- 21. Collier P, Phelan D, Klein A. A test in context: Myocardial strain measured by speckletracking echocardiography. *J Am Coll Cardiol*. 2017;69(8):1043-1056. doi: S0735-1097(17)30007-4 [pii].
- 22. Davis EF, Lewandowski AJ, Leeson P. Cardiac dysfunction and preeclampsia can imaging give clues to mechanism? *Circ Cardiovasc Imaging*. 2012;5(6):691-692. http://ovidsp.ovid. com/ovidweb.cgi?T=JS&PAGE=reference&D=emed13&NEWS=N&AN=368097993. doi: //dx.doi.org/10.1161/CIRCIMAGING.112.979831.
- 23. Gorcsan J,3rd, Tanaka H. Echocardiographic assessment of myocardial strain. *J Am Coll Cardiol*. 2011;58(14):1401-1413. doi: 10.1016/j.jacc.2011.06.038 [doi].
- 24. Farsalinos KE, Daraban AM, Unlu S, Thomas JD, Badano LP, Voigt JU. Head-to-head comparison of global longitudinal strain measurements among nine different vendors: The EACVI/ASE inter-vendor comparison study. *JAm Soc Echocardiogr.* 2015;28(10):117-1181, e2. doi: 10.1016/j.echo.2015.06.011 [doi].
- 25. Geyer H, Caracciolo G, Abe H, et al. Assessment of myocardial mechanics using speckle tracking echocardiography: Fundamentals and clinical applications. *J Am Soc Echocardiogr.* 2010;23(4):35-5. doi: 10.1016/j.echo.2010.02.015 [doi].
- 26. Saito M, Khan F, Stoklosa T, Iannaccone A, Negishi K, Marwick TH. Prognostic implications of LV strain risk score in asymptomatic patients with hypertensive heart disease. *JACC Cardiovasc Imaging*. 2016;9(8):911-921. doi: 10.1016/j.jcmg.2015.09.027 [doi].
- 27. Lee WH, Liu YW, Yang LT, Tsai WC. Prognostic value of longitudinal strain of subepicardial myocardium in patients with hypertension. *J Hypertens*. 2016;34(6):1195-1200. doi: 10.1097/HJH.0000000000000903 [doi].
- 28. Moher D, Liberati A, Tetzlaff J, Altman DG, PRISMA Group. Preferred reporting items for systematic reviews and meta-analyses: The PRISMA statement. *PLoS Med.* 2009;6(7):e1000097. doi: 10.1371/journal.pmed.1000097 [doi].

- 29. Cameli M, Mandoli GE, Sciaccaluga C, Mondillo S. More than 10 years of speckle tracking echocardiography: Still a novel technique or a definite tool for clinical practice? *Echocardiography*. 2019. doi: 10.1111/echo.14339 [doi].
- 30. Wells G, Shea B, O'connell D, et al. The newcastle-ottawa scale (NOS) for assessing the quality of nonrandomised studies in meta-analyses http://www.ohri.ca/programs/clinical_epidemiology/oxford.asp. Updated 2000. Accessed February 8, 2019.
- Scholten R, Offringa M, Assendelft W. Inleiding in evidence-based medicine. klinisch handelen gebaseerd op bewijsmateriaal. 4th ed. Houten: Bohn, Stafleu, Van Loghum; 2013.
- 32. Higgins J, Green S (editors). *Cochrane handbook for systematic reviews of interventions*. Version 5.1.0 [updated March 2011]. ed. The Cochrane Collaboration, 2011. Available from www.handbook.cochrane.org.
- Losilla JM, Oliveras I, Marin-Garcia JA, Vives J. Three risk of bias tools lead to opposite conclusions in observational research synthesis. *J Clin Epidemiol.* 2018;101:61-72. doi: S0895-4356(18)30039-8 [pii].
- 34. Liu W, Li Y, Wang W, Li J, Cong J. Layer-specific longitudinal strain analysis by speckle tracking echocardiography in women with early and late onset preeclampsia. *Pregnancy Hypertens*. 2019;17:172-177. doi: S2210-7789(19)30011-X [pii].
- Shahul S, Ramadan H, Nizamuddin J, et al. Activin A and late postpartum cardiac dysfunction among women with hypertensive disorders of pregnancy. *Hypertension*. 2018;72(1):188-193. doi: 10.1161/HYPERTENSIONAHA.118.10888 [doi].
- Shahul S, Medvedofsky D, Wenger JB, et al. Circulating antiangiogenic factors and myocardial dysfunction in hypertensive disorders of pregnancy. *Hypertension*. 2016;67(6):1273-1280. doi: 10.1161/HYPERTENSIONAHA.116.07252 [doi].
- 37. Shahul S, Rhee J, Hacker MR, et al. Subclinical left ventricular dysfunction in preeclamptic women with preserved left ventricular ejection fraction: A 2D speckle-tracking imaging study. *Circ Cardiovasc Imaging.* 2012;5(6):734-739. doi: 10.1161/CIRCIMAGING.112.973818 [doi].
- 38. Cong J, Lee Y, Wang W, Fu X, Wang Z, Li R. Longitudinal strain in layer-specific myocardium in early and late preeclampsia. *J Am Soc Echocardiogr.* 2018;31(6):B14-B141. http://ovidsp. ovid.com/ovidweb.cgi?T=JS&PAGE=reference&D=emexa&NEWS=N&AN=623037277. doi: //dx.doi.org/10.1016/j.echo.2018.04.010.
- 39. Cong J, Fan T, Yang X, Shen J, Cheng G, Zhang Z. Maternal cardiac remodeling and dysfunction in preeclampsia: A three-dimensional speckle-tracking echocardiography study. *Int J Cardiovasc Imaging*. 2015;31(7):1361-1368. doi: 10.1007/s10554-015-0694-y [doi].
- 40. Clemmensen TS, Christensen M, Kronborg CJS, Knudsen UB, Logstrup BB. Long-term follow-up of women with early onset pre-eclampsia shows subclinical impairment of the left ventricular function by two-dimensional speckle tracking echocardiography. *Pregnancy Hypertens*. 2018;14:9-14. doi: S2210-7789(18)30047-3 [pii].
- 41. Cho KI, Kim SM, Shin MS, et al. Impact of gestational hypertension on left ventricular function and geometric pattern. *Circ J*. 2011;75(5):1170-1176. doi: JST.JSTAGE/circj/CJ-10-0763 [pii].

- 42. Xia Y, Li X, Wang S, Xing Z. Three-dimensional speckle tracking imaging technology in evaluation on systolic function of left ventricular in patients with preeclampsia. *Chin J Med Imaging Technol*. 2017;33(3):335-339. http://www.cjmit.com/cjmit/ch/index.aspx http://ovidsp.ovid.com/ovidweb.cgi?T=JS&PAGE=reference&D=emed18&NEWS=N&AN=617367528. doi: //dx.doi.org/10.13929/j.1003-3289.201606171.
- 43. Orabona R, Vizzardi E, Sciatti E, et al. Insights into cardiac alterations after pre-eclampsia: An echocardiographic study. *Ultrasound Obstet Gynecol*. 2017;49(1):124-133. doi: 10.1002/uog.15983 [doi].
- 44. Al-Nashi M, Eriksson MJ, Ostlund E, Bremme K, Kahan T. Cardiac structure and function, and ventricular-arterial interaction 11 years following a pregnancy with preeclampsia. *J Am Soc Hypertens*. 2016;10(4):297-306. doi: 10.1016/j.jash.2016.01.012 [doi].
- 45. Yu L, Zhou Q, Peng Q, Yang Z. Left ventricular function of patients with pregnancyinduced hypertension evaluated using velocity vector imaging echocardiography and N-terminal pro-brain natriuretic peptide. *Echocardiography*. 2018;35(4):459-466. doi: 10.1111/echo.13817 [doi].
- 46. Levine LD, Lewey J, Koelper N, et al. Persistent cardiac dysfunction on echocardiography in african american women with severe preeclampsia. *Pregnancy Hypertens*. 2019;17:127-132. doi: S2210-7789(19)30051-0 [pii].
- 47. Pan G, Chen D, Xu L, et al. Cardiac dysfunction in women with severe preeclampsia detected by tissue doppler and speckle-tracking echocardiography. *Int J Clin Exp Med.* 2019;12(7):9245-9250. http://www.ijcem.com/files/ijcem0088564.pdf http://ovidsp.ovid.com/ovidweb.cgi?T=JS&PAGE=reference&D=emexb&NEWS=N&AN=2002363868.
- 48. Davey DA, MacGillivray I. The classification and definition of the hypertensive disorders of pregnancy. *Am J Obstet Gynecol.* 1988;158(4):892-898. doi: 0002-9378(88)90090-7 [pii].
- 49. Melchiorre K, Sutherland G, Baltabaeva A, Liberati M, Thilaganathan B. Impaired mid-gestational maternal cardiac function and left ventricular remodelling in women who subsequently develop preterm but not term preeclampsia. *Pregnancy Hypertens*. 2011;1(3-4):263-264. http://ovidsp.ovid.com/ovidweb.cgi?T=JS&PAGE=reference&D=emed12&NEWS=N&AN=70551167. doi: //dx.doi.org/10.1016/j.preghy.2011.08.046.
- 50. Vaddamani S, Keepanasseril A, Pillai AA, Kumar B. Maternal cardiovascular dysfunction in women with early onset preeclampsia and late onset preeclampsia: A cross-sectional study. *Pregnancy Hypertens*. 2017;10:247-250. doi: S2210-7789(17)30299-4 [pii].
- 51. Lisonkova S, Sabr Y, Mayer C, Young C, Skoll A, Joseph KS. Maternal morbidity associated with early-onset and late-onset preeclampsia. *Obstet Gynecol.* 2014;124(4):771-781. doi: 10.1097/AOG.00000000000472 [doi].
- 52. Castleman JS, Ganapathy R, Taki F, Lip GY, Steeds RP, Kotecha D. Echocardiographic structure and function in hypertensive disorders of pregnancy: A systematic review. *Circ Cardiovasc Imaging*. 2016;9(9):10.1161/CIRCIMAGING.116.004888.doi: 10.1161/CIRCIMAGING.116.004888.doi: 10.1161/CIRCIMAGING.1161/CIRCIMAGING.1161/CIRCIMAGING.1161/CIRCIMAGING.1161/CIRCIMAGING.1161/CIRCIMAGING.1161/CIRCIMAGING.1161/CIRCIMAGING.1161/CIRCIMAGING.1161/CIRCIMAGING.1161/CIRCIMAGING.1161/CIRCIMAGING.1161/CIRCIM
- 53. Grandi SM, Filion KB, Yoon S, et al. Cardiovascular disease-related morbidity and mortality in women with a history of pregnancy complications. *Circulation*. 2019;139(8):1069-1079. doi: 10.1161/CIRCULATIONAHA.118.036748 [doi].

- 54. Ersboll M, Valeur N, Mogensen UM, et al. Relationship between left ventricular longitudinal deformation and clinical heart failure during admission for acute myocardial infarction: A two-dimensional speckle-tracking study. *J Am Soc Echocardiogr.* 2012;25(12):1280-1289. doi: 10.1016/j.echo.2012.09.006 [doi].
- 55. Sengelov M, Jorgensen PG, Jensen JS, et al. Global longitudinal strain is a superior predictor of all-cause mortality in heart failure with reduced ejection fraction. *JACC Cardiovasc Imaging*. 2015;8(12):1351-1359. doi: S1936-878X(15)00719-6 [pii].
- 56. Shah AM, Claggett B, Sweitzer NK, et al. Prognostic importance of impaired systolic function in heart failure with preserved ejection fraction and the impact of spironolactone. *Circulation*. 2015;132(5):402-414. doi: 10.1161/CIRCULATIONAHA.115.015884 [doi].
- 57. Davis EF, Lazdam M, Lewandowski AJ, et al. Cardiovascular risk factors in children and young adults born to preeclamptic pregnancies: A systematic review. *Pediatrics*. 2012;129(6):e155-e1561.
- 58. Davis EF, Lewandowski AJ, Aye C, et al. Clinical cardiovascular risk during young adulthood in offspring of hypertensive pregnancies: Insights from a 20-year prospective follow-up birth cohort. *BMJ Open.* 2015;5(6):e008136.
- 59. Sacks KN, Friger M, Shoham-Vardi I, et al. Prenatal exposure to preeclampsia as an independent risk factor for long-term cardiovascular morbidity of the offspring. *Pregnancy Hypertension: An International Journal of Women's Cardiovascular Health.* 2018;13:181-186.
- 60. Andraweera PH, Lassi ZS. Cardiovascular risk factors in offspring of preeclamptic pregnancies-systematic review and meta-analysis. *The Journal of pediatrics*. 2019;208:10-113.e6.
- 61. Kajantie E, Eriksson JG, Osmond C, Thornburg K, Barker DJP. Pre-eclampsia is associated with increased risk of stroke in the adult offspring: The helsinki birth cohort study. *Stroke*. 2009;40(4):1176-1180.
- 62. Fugelseth D, Ramstad HB, Kvehaugen AS, Nestaas E, Stoylen A, Staff AC. Myocardial function in offspring 5-8years after pregnancy complicated by preeclampsia. *Early human development*. 2011;87(8):531-535.
- 63. Yu L, Zhou Q, Peng Q, Zeng S, Yang Z. Velocity vector imaging echocardiography and NT-proBNP study of fetal cardiac function in pregnancy-induced maternal hypertension. *J Clin Ultrasound*. 2019;47(5):285-291. doi: 10.1002/jcu.22720 [doi].
- 64. Shahul S, Ramadan H, Mueller A, et al. Abnormal mid-trimester cardiac strain in women with chronic hypertension predates superimposed preeclampsia. *Pregnancy Hypertens*. 2017;10:251-255. doi: S2210-7789(17)30144-7 [pii].
- 65. Mol BWJ, Roberts CT, Thangaratinam S, Magee LA, de Groot, C. J. M., Hofmeyr GJ. Pre-eclampsia. *Lancet*. 2016;387(10022):999-1011. doi: S0140-6736(15)00070-7 [pii].





Heart rate variability in hypertensive pregnancy disorders: a systematic review

S Moors KJJ Staaks MEMH Westerhuis LRC Dekker KMJ Verdurmen SG Oei JOEH van Laar

Pregnancy Hypertension: An International Journal of Women's Cardiovascular Health 20C (2020) pp. 56-68

ABSTRACT

Background

Hypertensive pregnancy disorders (HPD) are associated with dysfunction of the autonomic nervous system. Cardiac autonomic functions can be assessed by heart rate variability (HRV) measurements.

Objective

To study whether HRV detects differences in the function of the autonomic nervous system between pregnant women with HPD compared to normotensive pregnant women and between women with a history of a pregnancy complicated by HPD compared to women with a history of an uncomplicated pregnancy.

Methods

A systematic search was performed in Medline, EMBASE, and CENTRAL to identify studies comparing HRV between pregnant women with HPD or women with a history of HPD to women with (a history of) normotensive pregnancies.

Results

The search identified 523 articles of which 24 were included in this review, including 850 women with (a history of) HPD and 1205 normotensive controls. The included studies showed a large heterogenicity. A decrease in overall HRV was found in preeclampsia (PE), compared to normotensive pregnant controls. A trend is seen towards increased low frequency/high frequency-ratio in women with PE compared to normotensive pregnant controls.

Conclusion

Our systematic review supports the hypothesis a sympathetic overdrive is found in HPD which is associated with a parasympathetic withdrawal. However, the included studies in our review showed a large diversity in the methods applied and their results.

INTRODUCTION

Hypertensive Pregnancy Disorders (HPD), defined as gestational hypertension (GH), chronic hypertension (CH), preeclampsia (PE), and superimposed PE,¹ are pregnancy-specific systematic disorders that globally affect 5-10% of all pregnancies.²⁻⁶

Hypertension is defined as a systolic blood pressure \geq 140 mmHg and/or a diastolic blood pressure \geq 90 mmHg.^{1,7} In GH and PE, hypertension develops after 20 weeks' gestation, with PE also being characterized by proteinuria (>300 mg/day) and/or end organ dysfunction.¹ CH is defined as hypertension discovered preconception or prior to 20 weeks' gestation.⁷ Of all HPD, GH and PE occur the most frequent.⁸

HPD are a major cause of maternal and fetal mortality and morbidity worldwide.^{1,2,9,10} Furthermore, there is growing evidence that HPD are associated with a higher risk of maternal cardiovascular disease (CVD) later in life.¹¹⁻¹³

The etiology of HPD is not exactly known, but several studies state that the primary derangement involves the cardiovascular system.^{14,15} In normal pregnancy, cardiovascular volume load is increased and cardiovascular pressure load is reduced as an adjustment to the early pregnancy drop in peripheral vascular resistance.^{15,16} The autonomic nervous system has a prominent role in these adaptations.^{14,16-18}

HPD might originate from abnormal remodeling of maternal spiral arteries and the absence of adequate cardiovascular adaptations to pregnancy,^{14,15} resulting in a higher blood pressure and an increase in vascular resistance.^{1,19} In addition, women with HPD typically lack the hypervolemia that is associated with normal pregnancy.^{1,20,21} Several studies have stated that HPD develop because of dysfunction of the autonomic nervous system.^{16,20,22,23}

Cardiac autonomic functions can - amongst others - be assessed by heart rate variability (HRV).^{24,25} HRV reflects the impact of central and peripheral circuits of regulation on hemodynamics.^{19,26} Therefore, a possible dysfunction of the autonomic nervous system can be shown by measuring HRV.

The objective of this systematic review is to study whether HRV detects differences in the function of the autonomic nervous system between pregnant women with HPD compared to normotensive pregnant women or between women with a history of a pregnancy complicated by HPD compared to women with a history of an uncomplicated pregnancy.

METHODS

HRV definitions

HRV is the oscillation between subsequent heartbeats, which is generated by heart-brain interactions, cardiovascular hormones, and neuronal pathways that exert effects on the heart, blood vessels and kidneys.^{25,27,28}

Assessment of HRV is most frequently performed with electrocardiography (ECG). The data acquired from ECG can be evaluated and described by using time-domain and frequency-domain features.^{25,29}

Frequency-domain features

Frequency-domain values estimate the distribution of the signal energy within a frequency band. The differences in frequency ranges allows HRV analysis to separate the sympathetic and parasympathetic contributions to the autonomous nervous system.²⁹ The HRV Task Force defined the frequency-domain bandwidths, shown in Table 1.²⁵

The ratio between low frequency (LF) and high frequency (HF) is considered to mirror sympatho/vagal balance or to reflect sympathetic modulations. A low LF/HF-ratio reflects parasympathetic dominance, while a high LF/HF-ratio reflects sympathetic dominance.

The frequency-domain features are usually expressed in absolute values of power (ms^2) or as the natural logarithm (Ln). LF and HF might also be measured in normalized units (n.u.; (LF/(TP-VLF)) and (HF/(TP-VLF))).^{23,25,30}

Time-domain features

Time-domain features quantify the amount of variability in the interbeat interval. The normal-to-normal (NN) intervals are determined, i.e. intervals between adjacent QRS complexes.²⁵

A wide variety of time-domain features can be calculated.^{25,29} Many of these measures correlate closely with each other. Therefore, the HRV Task Force recommended four time-domain features as presented in Table 2.²⁵

HRV in HPD: a systematic review

Table 1. Frequency-domain features according to Heart rate variability Task Force²⁵

Band	Frequency	Period recording	Clinical application
ULF	≤0.003 Hz	24 hours	Slow-acting biological processes like the circadian rhythm
VLF	0.003-0.04 Hz	2-5 minutes	The physiological explanation of the VLF component is not defined very strictly and it is questionable whether a specific physiological process attributes to VLF
LF	0.04-0.15 Hz	2-5 minutes	Some studies suggest that LF is a quantitative marker for sympathetic modulations, other studies describe LF as reflecting both sympathetic and vagal activity.
HF	0.15-0.40 Hz	1 minute	Reflects parasympathetic tone
TP	ULF+VLF+LF+HF	24 hours	Sum of all bandwidths

ULF: ultra low frequency, VLF: very low frequency, LF: low frequency, HF: high frequency, TP: total power, Hz: Hertz.

Table 2. Time-domain features according to Heart Rate Variability Task Force.²⁵

Time-domain feature	Clinical relevance
SDNN	The estimate of overall HRV.
HRV triangular index	The estimate of overall HRV; permits only casual pre-processing of the ECG-signal.
RMSSD	The estimate of short-term components of HRV.
SDANN	The estimate of long-term components of HRV. Intervals in all 5-minutes segments of the entire recording.

SDNN: Standard Deviation of normal-to-normal (NN)-interval; HRV: Heart Rate Variability; RMSSD: Root Mean Square of Successive Differences; SDANN: Standard Deviation of the Averages of the NN-interval. ECG: electrocardiography

61

Study design

Eligibility criteria

Our review included studies that examine HRV in women with HPD compared to healthy normotensive pregnant women or studies that measure HRV in women with a history of HPD compared to normotensive women with a history of a normotensive pregnancy. The analyses in which pregnant women with HPD were compared with normotensive non-pregnant women and the studies in which women with a history of a pregnancy complicated by HPD were compared to women who had never been pregnant were excluded. We included studies that described maternal HRV using linear time- and frequency-domain features. We excluded animal studies, case reports, reviews, meta-analyses, conference abstracts, commentaries, and editorials. Also, studies without a control arm or HPD arm were excluded. No restrictions were applied regarding language or date of publication.

Search strategy

This review is conducted in accordance with the PRISMA guideline for systematic reviews,³¹ and prospectively registered in the PROSPERO International prospective register of systematic reviews (CRD42019124052, available from: http://www.crd. york.ac.uk/PROSPERO/display_record.php?ID=CRD42019124052). With help of a professional librarian, we performed our search in the electronic databases EMBASE, MEDLINE (PubMed) and CENTRAL. Our search was conducted from inception of databases up until October first, 2019. Our search consisted of the following terms and its synonyms: 'heart rate variability', 'gestational hypertension', 'preeclampsia' and a wide variety of its synonyms. The search strings can be found in Appendix 1.

Study selection and data extraction

All titles and abstracts obtained from our search were independently screened by two reviewers (SM and KS). Based on our selection criteria, eligible studies were identified, and subsequently full texts were retrieved. When multiple articles were published by the same authors, we contacted the authors to avoid duplication of participants and consequently excluded duplicates. Furthermore, we hand-searched all references of the selected articles and related articles to identify additional relevant publications. Any disagreement was resolved by consensus and if necessary, a third reviewer (MW) was consulted. Two reviewers (SM and KS) extracted relevant data using a preliminary designed data form. The retrieved data comprised year of publication, maternal features of HRV and study characteristics.

Assessment quality and risk of bias

Quality of the included studies was assessed by using the criteria of the Newcastle-Ottawa Scale (NOS), as recommended by the Dutch Cochrane Center.³¹⁻³³ Nine criteria were assessed, divided over three dimensions: selection, comparability, and exposure. For each criterium, stars can be awarded. As suggested by Losilla et al.,³⁴ studies scoring 0-3, 4-6, and 7-9 stars were deemed to be of low, moderate, and high quality, respectively.

Two authors (SM and KS) completed the data quality and risk of bias assessment independently. Disagreements were resolved by consensus and if needed, a third author (MW) was consulted.

Data analysis

Data from the included studies were aggregated to provide a narrative synthesis of the findings.

RESULTS

PRISMA Flow Figure 1 shows the search process in steps. A total of 24 studies met all our inclusion criteria and were included in our systematic review. All included studies were observational studies; 20 case-control and four longitudinal case-control studies. All studies combined included a total of 2055 women. 21 studies measured HRV during pregnancy, including 787 women with a pregnancy complicated by HPD and 1142 controls with a normotensive pregnancy. One study measured HRV postpartum, including 20 women who have had a pregnancy complicated by HPD and 20 women with a history of a normotensive pregnancy. Two studies analyzed women both during their pregnancy as well as postpartum. Of these, 43 women had HPD and 43 had a normotensive pregnancy. Detailed information regarding the study characteristics of the included studies, the quality assessment and risk of bias are presented in Table 3.



Figure 1. PRISMA Flow diagram demonstrating an overview of the selection process.

Table 3. Characteristics of included studies

Author year	Population§ (n=)	Inclusion criteria (definition)	Exclusion criteria	Mean GA in weeks	Frequency bands in Hz	Medication	Quality NOS + Losilla categories	Study design and measurements
Studies incl	uding women during p	regnancy						
Brown et al. 2008 ₂ಂ	Cases: GH (20) Controls: NTP (20)	HPD definition: ACOG 2002 Age ≥18 years, GA 33-41 weeks, singleton, maternal weight <114kg, no treated comorbidities (DM,thyroid disease, depression)	Cardiac arrhythmias	GH: 36.5±2.4 NTP: 36± 0.5	LF: 0,04-0,15 MF: 0,12-015 HF: 0,15-0,40	Noβ-blockers or anti-arrhythmic drugs	5 (3/0/2) Moderate quality	Case-control ECG 10 min CTG 30 min
Chaswal et al. 2018 ³ಂ	Cases: PE (40) Controls: NTP (40)	HPD definition:, ACOG 2013	Multiple pregnancy, DM, CH, liver/ thyroid/auto-immune/ renal/ inflammatory disease	PE: 29±2.4 NTP: 30±3.6	VLF: not described LF: 0,04-0,15 HF: 0,15-0,4	Not mentioned	7 (3/2/2) High quality	Case-control ECG 5 min
Ekholm et al. 1997 ³⁸	Cases: GH (14) Controls: NTP (16)	HPD definition: BP >140/90 mmHg on two occasions in 3ª trimester, with or without proteinuria ≥0.3 gm/l	Not applicable	GH: 37 (31-40) NTP: 37 (31-40)	LF: 0,00-0,07 MF: 0,07-0,15 HF: 0,15-0,4	No antihypertensive treatment	6 (3/1/2) Moderate quality	Case-control ECG 7 min
Eneroth et al. 1994 ⁴⁸	Cases: PE (13) Controls: NTP (12)	HPD definition: BP ≥140/90 mmHg and proteinuria ≥300mg/24hr or dipstick ≥1+	History of cardiopulmonary disease or DM	PE: 35±3 NTP: 37±4	LF: 0,07-0,15 HF: 0,15-0,4	No medication	5 (3/0/2) Moderate quality	Case-control ECG 5 min

Author year	Population§ (n=)	Inclusion criteria (definition)	Exclusion criteria	Mean GA in weeks	Frequency bands in Hz	Medication	Quality NOS + Losilla categories	Study design and measurements
Euliano et al. 2018 ^{so}	Cases: PE with severe features (37) Controls: NTP (43)	HPD definition: ACOG 2002	Confounding disease state during admission (e.g. vasculitis/renal/ liver disease unrelated to PE), proteinuria before pregnancy, labor beyond 5cm cervical dilation, neuraxial anesthesia	Inclusion:29-41 PE: 32±3.6 NTP: 37±4.3	LF, HF (bandwidth not described)	Antihypertensive drugs and/or MgSO4	4 (2/0/2) Moderate quality	Case-control PPG + ECG 30 min
Faber et al. 2004 ²¹	Cases: - PE (44) - GH (18) - CH (19) Controls: NTP (80)	HPD definition: NHBPEP	Not applicable	PE: 32 (30-36) GH: 36 (31-37) CH: 34 (25-38) NTP: 35 (32-37)	VLF: 0,003- 0,04 LF: 0,04-0,15 HF: 0,15-0,4	Not mentioned	5 (3/0/2) Moderate quality	Case-control ECG 30 min
Flood et al. 2015 ^{s2}	Cases: - Later developed PE (27) - Later developed GH (26) Control: NTP (332)	HPD definition: ACOG	Not applicable	Measurement at 28 week assessment visit	LF: 0,04-0,15 HF: 0,15-0,5	Not mentioned,	5 (2/1/2) Moderate quality	Secondary analysis of prospective cohort ECG 5min
Guo et al. 2018 ⁴○	Cases: - PE (19) - GH (41) Controls: NTP (60)	HPD definition: ACOG 2013 Nulliparous, single living fetus after 32 weeks	CH, DM, any endocrinal disorders, cardiac arrhythmias, infectious diseases, cancer, smokers, or drinkers	PE: 35±2.0 GH: 36±3.2 NTP: 35±2.7	LF: 0,04-0,15 HF: 0,15-0,4	Antihypertensive drugs. No other medication	5 (3/0/2) Moderate quality	Case-control Holter 24 hours

Table 3. Continued.

Author year	Population§ (n=)	Inclusion criteria (definition)	Exclusion criteria	Mean GA in weeks	Frequency bands in Hz	Medication	Quality NOS + Losilla categories	Study design and measurements
Khan et al. 2014 ⁴⁹	Cases: PE (15) Controls: NTP (33)	HPD definition: not specified Risk factors for GH	Controls had no risk factors for GH	>20 weeks of gestation, no mean presented	VLF: 0,003-0,04 LF: 0,04-0,15 HF: 0,15-0,4	Not mentioned	5 (3/0/2) Moderate quality	Case-control ECG 10 min
Khlybova et al. 2008 ⁴⁵	Cases: - Mild PE (12) - Moderate PE (10) - CH: (18) Controls: NTP (15)	HPD definition: Not specified	Not applicable	Mild PE: 34±0.5 Moderate PE:33±1.3 CH: 35±0.9 NTP: 34±0.5	LF, HF (bandwidth not described)	Antihypertensive drugs	4 (3/0/1) Moderate quality	Case-control ECG 500 cycles
2017 ²⁶	Cases: - Mild- moderate PE (44) - Severe PE (32) Controls: NTP (30)	HPD definition: Mild-moderate PE: SBP 140-159 mmHg and DBP 90-109 mmHg Severe PE: SBP ≥160mmHg and DBP 110 mmHg, and/ or thrombocytopenia, serum creatinine >1.1 mg/l, liver transaminases >2x7, pulmonary edema, cerebral or visual disturbances	Multiple pregnancy, eclampsia, pre- existing diseases (e.g. DM, metabolic syndrome, cardiac or renal diseases, thyrotoxycosis and CH)	Inclusion: 34-40 Mild/moderate PE: 37±.3.6 Severe PE: 37 ±2.5 NTP: 37±1.8	VLF, LF, HF (bandwidths not described)	Antihypertensive drugs	6 (4/0/2) Moderate quality	Case-control ECG 10 min Non-invasive fetal ECG
Lewinsky et al. 1998 ⁵³	Cases: PE (15) Controls: NTP (25)	HPD definition: BP >140/90 mmHg on two occasions at least 6h apart and proteinuria >300 mg/24hr	History of hypertension, renal disease, DM, medication with exception of iron subelementation	PE: 33±3 NTP: 35±4	VLF:0-0,05 MF:0,05-0,15 HF:0,2-0,5	Measurements before start of MgSO4 or antihypertensive therapy	4 (2/0/2) Moderate quality	Case-control ECG, not mentioned how many min. (short term)

Continued.
Table 3.

Author year	Population§ (n=)	Inclusion criteria (definition)	Exclusion criteria	Mean GA in weeks	Frequency bands in Hz	Medication	Quality NOS + Losilla categories	Study design and measurements
Metsaars et al. 2006 42	Cases: Early-onset HPD (37) Late-onset PE (8) Controls: NTP (29)	HPD defined as GH with fetal growth restriction, severe PE, HELLP or eclampsia HPD definition: ACOG 1988 Fetal growth restriction: EFW <p10< th=""><th>Not defined</th><th>Inclusion:24-34 EO-HPD: 29 (24-34) LO-PE: 32 (31-33) NTP: 32 (31- 33)</th><th>LF: 0,04-0,15 HF: 0,15-0,4</th><th>MgSO4 and/or antihypertensive drugs</th><th>6 (3/1/2) Moderate quality</th><th>Case-control Continuous HR non- invasive finger arterial pressure waveform, unknown duration</th></p10<>	Not defined	Inclusion:24-34 EO-HPD: 29 (24-34) LO-PE: 32 (31-33) NTP: 32 (31- 33)	LF: 0,04-0,15 HF: 0,15-0,4	MgSO4 and/or antihypertensive drugs	6 (3/1/2) Moderate quality	Case-control Continuous HR non- invasive finger arterial pressure waveform, unknown duration
Musa et al. 2016 ⁴⁴	Cases: PE (60) Controls: NTP (60)	HPD definition: ACOG 2013	Thyroid/renal/liver disease, hypertension, DM, medication for hypertension	PE: 34±4.3 NTP: 33±4.0	LnVLF, LnLF, LnHF (bandwidths not mentioned, ref. Task Force)	All women enrolled before receiving any medication	5 (3/0/2) Moderate quality	Case-control ECG 5 min
Pal et al. 2009 ⁴⁶	Cases: With risk factors, developed GH (27) Controls: NTP (38)	HPD definition: Not defined	Oral contraceptives prior to pregnancy	- 12 - 24 - 31	LF, HF (bandwidths not mentioned)	Not mentioned	6 (3/1/2) Moderate quality	Longitudinal ECG 10 min
Pal et al. 2011 ³⁹	Cases: With risk factors GH, developed GH (31) Controls: NTP (38)	HPD definition: Joydev et al.	Not defined	- 12 - 24 - 31	LF, HF (bandwidths not mentioned)	Not mentioned	6 (3/1/2) Moderate quality	Longitudinal ECG 10 min

Table 3. Continued.

Author year	Population§ (n=)	Inclusion criteria (definition)	Exclusion criteria	Mean GA in weeks	Frequency bands in Hz	Medication	Quality NOS + Losilla categories	Study design and measurements
Swansburg et al. 2005 ⁴³	Cases: PE (9) Controls: NTP (18)	HPD definition: NHBPEP Maternal age >16years, singleton pregnancy Gestational age 32-40 weeks	No exclusion criteria	PE: 37±2.6 NTP: 36±2.0	LF: 0,04-0,15 HF: 0,15-0,4	Three women used medication (gluticasone, terbutaline sulfate, labetalol, methyldopa, ranitidine)	5 (3/0/2) Moderate quality	Case-control ECG 20 min
Tejera et al. 2012 ⁴7	Cases: - PE (27) - GH (55) Controls: NTP (135)	HPD definition: NHBPEP Maternal age >16years, singleton	Not defined	PE: 27 (range 7-40) GH: 25 (range 7-40) NTP: 24 (range 6-40)	VLF: =0,04 LF: 0,04-0,15 HF: 0,15-0,4	Antihypertensive drugs	4 (2/0/2) Moderate quality	Case-control ECG 10 min
Weber et al. 2017 37	Cases: - PE < AD34 (10) - PE > AD34 (14) Controls: - NTP < GA34 (30) - NTP > GA34 (42)	HPD definition: ACOG 2013	PE and pre-existing diseases (e.g. insulin- dependent DM, cardiovascular or renal diseases, etc.)	Inclusion: 28-38 EO-PE:31±2 LO-PE: 36±3 NTP: 31±2 and 36±3	LnLF: 0,04-0,15 LnHF: 0,15-0,4	Not mentioned	7 (3/2/2) High quality	Case-control ECG 10 min
Yang et al. 2000 ⁵⁵	Cases: PE (11) Controls: NTP (17)	HPD definition: ISSHP 2000	Diabetic neuropathy, cardiac arrhythmia or other cardiovascular diseases that affect HRV	PE: 35±1 NTP: 34±1	LF: 0,04-0,15 HF: 0,15-0,4	No hypnotics or autonomic blockers	7 (3/2/2) High quality	Case-control ECG 5 min
ntinued.								

ී ස								
Table								

70

Author year	Population§ (n=)	Inclusion criteria (definition)	Exclusion criteria	Mean GA in weeks	Frequency bands in Hz	Medication	Quality NOS + Losilla categories	Study design and measurements
Yokusoglu et al. 2009 ³⁶	Cases: PE (34) Controls: NTP (29)	HPD definition: ISSHP 2000	Not described	Inclusion: 27-34 PE: 29±3.8 NTP: 30±3.6	Only time domain	No medication	8 (4/2/2) High quality	Case-control Holter 24 hours
Studies inclu	uding women postpar	tum						
Murphy et al. 2015 ⁵¹	Cases: PE (20) Controls: NTP (20)	HPD definition: Canadian Hypertensive Disorders of Pregnancy Working Group, Magee et al.	HELLP, pre- pregnancy history of hypertension, DM, renal or cardiovascular disease or smoking	6-8 months postpartum	LF: 0,04-0,15 HF: 0,15-0,4	No clear mentioning of medication postpartum	4 (3/0/1) Moderate quality	Case-control ECG 10 min
Studies inclu	uding women both du	ring pregnancy and post	partum					
Eneroth et al. 1999	Cases: PE (15) Controls: NTP (15)	HPD definition: BP ≥140/90 mmHg in at least two readings more than 6h apart and proteinuria ≥300mg/24hr or dipstick ≥1+)	History of hypertension, DM or renal disease	PE: 33±1.6 NTP: 33±2.0 and 3-6 months postpartum	VLF:0,0033-0,04 LF:0,04-0,15 HF:0,15-0,4	Nomedication	5 (3/0/2) Moderate quality	Longitudinal Holter 24 hour

Table 3. Continued.

Author year	Population§ (n=)	Inclusion criteria (definition)	Exclusion criteria	Mean GA in weeks	Frequency bands in Hz	Medication	Quality NOS + Losilla categories	Study design and measurements
Heiskanen et al. 2011 ¹⁸	Cases: HPD (GH or PE) (28) Controls: NTP (28)	HPD definition: BP ≥140/90 mmHg after 20 weeks GA with previously normal BP and with or without proteinuria, measured three times	Pregnant women with superimposed PE	HPD: 35±0.7 NTP: 34±0.6 and 3 months postpartum	VLF: 0-0,04 LF: 0,04-0,15 HF: 0,15-0,4	Not mentioned	5 (3/0/2) Moderate quality	ECG 10 min

§ population of which the analyses are included in the review. ACOG: The American College of Obstetricians and Gynecologists; BP: blood pressure; CH: chronic hypertension; DBP: diastolic blood pressure; DM: Diabetes Mellitus; GA: gestational age; GH: gestational hypertension; HELLP: Hemolysis, Elevated Liver Enzymes, Low Platelets syndrome; Hz: Hertz; NTP: normotensive pregnancy; ISSHP: International Society of Hypertension in Pregnancy; NHBPEP: National High Blood Pressure Education Program; MgSO₄: magnesium sulfate NOS: Newcastle-Ottawa Scale; PE: preeclampsia; PPG: Photoplethysmography; PPI: Partus Prematurus Imminens (Threatened preterm labor); SBP: systolic blood pressure.

71

Quality assessment

The risk of bias, as assessed by the NOS, ranges from 5 to 8 stars. Four studies were classified as high quality.^{30,35-37} The remaining 20 articles were of moderate quality. No study was considered to be of low quality (Table 3).

Frequency-domain features

The results of all frequency-domain features during pregnancy are presented in Table 4. The postpartum results of frequency-domain features are shown in Table 5.

Total Power (TP)

During pregnancy

When comparing GH to normotensive pregnant women, TP showed no difference (2 studies).^{20,38} A lower TP was found at 12, 24 and 31 weeks of gestation in women who would develop GH later in pregnancy compared to normotensive pregnant controls (1 study).³⁹ With regard to natural logarithm of TP (LnTP), a decrease was found in GH compared to normotensive pregnant women.⁴⁰

Comparing women with PE to normotensive pregnant women, TP showed no difference (3 studies)⁴¹⁻⁴³ or a decrease (1 study).³⁰ LnTP showed no difference (1 study)⁴⁴ or a decrease (1 study)⁴⁰ in women with PE compared to normotensive pregnant controls.

Studies comparing women with a variety of HPD to normotensive controls found no difference in $\rm TP.^{18,42}$

Postpartum

No differences were found in TP after either GH or PE compared to after a normotensive pregnancy (1 study).¹⁸ After PE, TP was decreased compared to after a normotensive pregnancy (1study).⁴¹

Low Frequency (LF)

During pregnancy

In CH, LF was increased (1 study)⁴⁵ or comparable (1 study)²¹ to normotensive controls. Comparing GH with normotensive controls, no difference in LF was found (4 studies).^{18,20,21,38} LF(n.u.) was increased in women who would develop GH at 12, 24 and 31 weeks of gestation compared to normotensive controls.^{39,46} Studies comparing LnLF between GH and normotensive controls found an increase in case of GH (1 study)⁴⁰ or no significant difference (1 study).⁴⁷

Studies comparing LF between PE and normotensive women found no difference (5 studies),^{21,41,43,45,48} an increased LF in PE (2 studies),^{42,49} or a decreased LF in PE (3 studies).^{26,30,50} With regard to LF(n.u.), an increase in PE (3 studies)^{30,44,49} or no significant difference (1 study)³⁵ was found compared to normotensive controls. Comparing LnLF between PE and normotensive controls, an increase in PE (2 studies)^{40,44} or no difference (3 studies)^{35,37,47} was found.

Studies comparing women with a variety of HPD to normotensive controls found no difference in LF. $^{\rm 18,42}$

Postpartum

No difference was found in LF postpartum after GH or PE (2 studies)^{18,41} compared to normotensive controls, also not when calculated as LF(n.u.; 1 study).⁵¹

High Frequency (HF)

During pregnancy

Comparing CH with normotensive pregnant women, an increased HF (1 study)⁴⁵ or no difference (1 study)²¹ was found.

In GH, an increased HF (1 study)³⁸ or no difference (2 studies)^{20.21} was found compared to normotensive controls. HF(n.u.) was decreased at 12 weeks of gestation in women who would develop GH compared to normotensive controls. At 24 and 31 weeks of gestation, no difference in HF(n.u.) was found between these groups.^{39,46} One study found a decreased HF at 28 weeks of gestation in women who would later develop GH compared to normotensive controls.⁵² No difference in LnHF was found between GH and normotensive controls.^{40,47}

In PE, an increased HF (1 study)⁴³, a decreased HF (5 studies)^{26,30,45,48,49} or no difference (3 studies)^{21,41,53} was found compared to normotensive controls. One study found no difference in HF at 28 weeks of gestation in women who would later develop PE compared to normotensive controls.⁵² All three studies comparing HF(n.u.) between PE and normotensive controls found a decreased HF(n.u.) in PE.^{30,44,49} LnHF was found to be decreased in PE (1 study)³⁵ or comparable to normotensive controls (4 studies).^{37,40,44,47} Studies comparing all HPD with normotensive controls showed an increased HF (1 study)¹⁸ or no difference in HF (1 study)⁴².

Postpartum

HF was not different after either PE or GH compared to normotensive controls.^{18,41}

Low frequency/High frequency-ratio (LF/HF-ratio)

During pregnancy

In CH, an increased LF/HF-ratio (1 study)⁴⁵ or no difference in LF/HF-ratio (1 study)²¹ was found compared to normotensive controls.

Comparing GH and normotensive controls, no difference in LF/HF-ratio was found (3 studies).^{18,21,47} However, an increased LF/HF-ratio was found in all three trimesters in women who would later develop GH compared to normotensive controls.^{39,46} The Ln(LF/HF-ratio) was increased in GH compared to normotensive controls (1 study).⁴⁰

In PE, an increased LF/HF-ratio (3 studies)^{26,30,49} or no difference in LF/HF-ratio (2 studies)^{21,47} was found compared to normotensive controls. LF/HF-ratio was increased in moderate PE compared to normotensive controls, while it was not different between

Chapter 3

mild PE and normotensive controls.⁴⁵ The Ln(LF/HF-ratio) was found to be increased in PE (3 studies)^{35,40,44} or not different from normotensive controls (1 study).³⁷ Studies comparing all HPD with normotensive controls found no difference in LF/HF-ratio(2 studies).^{18,42}

Postpartum

LF/HF-ratio was not different after a pregnancy complicated by either GH or PE compared to normotensive controls.^{18,51}

Table 4. Freq.	uency dor	main feature	s, during p	regnancy wo	men						
Author Year	Brown 2008	Chaswal 2018	Ekholm 1997	Eneroth 1994	Eneroth 1999	Euliano 2018	Faber 2004	Flood 2015	Guo 2018	Heiskanen 2011	Khan 2014
Epoch No. patients	10 min 40	5 min 80	7 min 30	10 min 25	24 hours 30	30 min 80	30 min 161	5min 385	24 hours 120	10 min 120	10 min 48
TP	GH: =	PE: ↓	GH: =		PE: =					HPD: =	
ULF											
VLF					PE: =					HPD: =	PE: ↓
LF	GH: =	PE: \	GH: =	PE: =	PE: =	PE: ↓	PE: = GH: = CH: =			HPD: =	PE: ↑
MF			GH: =								
HF	GH: =	PE: ←	GH:↑	PE:	PE: =		PE: = GH: = CH: =	Before PE: = Before C.H.		HPD: 1	PE:
LF/HF		PE: ↑					PE: = GH: = CH: =			HPD: =	PE: 1
LF/(LF+HF)						PE: ↓					
MF/HF			GH: =								
LF(n.u.)		PE: ↑									PE ↑
HF(n.u.)		PE: ↓									PE: ↓
LnTP									PE: ↓		
									GH: ↓		
LnVLF											
LnLF									ΡΕ: Υ		
									GH:↑		
LnHF									PE: = GH: =		
LnLF/HF									РЕ:		
									GH: ↑		

HRV in HPD: a systematic review

;
e
P
iti
ō
0
4
e
ab
H

76

Author Year	Khlybova 2008	Lakhno 2017	Lewinsky 1998	Metsaars 2006	Musa 2016	Pal 2009	Pal 2011	Swansburg 2005	Tejera 2012	Weber 2017	Yang 2000
Epoch No. women	500 cycles 55	10 min 121	Unknown 40	Unknown 74	5 min 120	10 min 65	10 min 69	20 min 27	10 min 217	10 min 96	5 min 28
TP				LO-PE = EO-HPD =			GH: GA 12 ↓ GA 24: ↓ GA 31: ↓	PE: =			
NLF											
VLF	Mild PE: =	PE: ↓	PE: =								
	Moderate PE: ↓ CH: ↑										
LF	Mild PE: =	PE: ↓	×	LO-PE =				PE: =			
	Moderate PE: = CH: 个			ЕО-НРD ↑							
MF			PE: =								
ΗF	Mild PE: \	PE: ↓	PE: =	LO-PE =				PE: ↑			
	Moderate PE: ↓ CH: ↑			EO-HPD =							
LF/HF	Mild PE: =	PE: ↑		LO-PE =		GH: GA 12 Τ	GH: GA 12 Τ		GH: = PE: =		
	Moderate PE: ↑			EO-HPD =		GA 24: ↑	GA 24: ↑				
	CH:↑					GA 31: ↑	GA 31: Υ				
LF/(LF+HF)											
MF/HF											
LF(n.u.)					PE: →	GH: GA 12 1	GH: GA 12 1				PE =
						GA 24: 1 GA 31: ↑	GA 24: 'I' GA 31: ↑				

Table 4. Continued.

Author Year	Khlybova 2008	Lakhno 2017	Lewinsky 1998	Metsaars 2006	Musa 2016	Pal 2009	Pal 2011	Swansburg 2005	Tejera 2012	Weber 2017	Yang 2000
Epoch No. women	500 cycles 55	10 min 121	Unknown 40	Unknown 74	5 min 120	10 min 65	10 min 69	20 min 27	10 min 217	10 min 96	5 min 28
HF(n.u.)					PE: ←	GH: GA 12: ↓ GA 24: = GA 31: =	GH: GA 12: ↓ GA 24: = GA 31: =				
LnTP					PE: =						
LnVLF					PE: ↑						
LnLF					PE: A				PE: = GH: =	PE: = (eo-PE = lo-PE =)	PE =
LnHF					PE:=				PE: = GH: =	PE: = (eo-PE =, lo-PE Φ)	PE ←
LnLF/HF					PE: ↑					PE: = (eo-PE =, lo-PE =)	$PE \uparrow$
	-				1.00						

↑: significant increase, ↓: significant decrease, =: no significant difference compared to a control group of normotensive pregnant women. Empty cell: parameter not analyzed in this study; GA: gestational age; PE: preeclampsia; GH: gestational hypertension; CH: chronic hypertension; HPD: hypertensive pregnancy disorder; eo-PE: early-onset PE; lo-PE: late-onset PE. No. patients: number of patients included in the analyses.

Table 5. Frequency domain features postpartum

Author	Eneroth	Heiskanen	Murphy	
Year	1999	2011	2015	
Epoch	24 hours	10 min	10 min	
No. women	30	56	40	
TP	PE:↓	HPD: =		
ULF				
VLF	PE: =	HPD: =		
LF	PE: =	HPD: =		
MF				
HF	PE:↓	HPD: =		
LF/HF		HPD: =	PE: =	
LF/(LF+HF)				
MF/HF				
LF(n.u.)			PE: =	
HF(n.u.)				
LnTP				
LnVLF				
LnLF				
LnHF				
I nI F/HF				

 \uparrow : significant increase, \downarrow : significant decrease, =: no significant difference compared to a control group of women with a history of a normotensive pregnancy.

Empty cell: parameter not analyzed in this study; PE: preeclampsia; HPD: hypertensive pregnancy disorders; No. patients: number of patients included in the analyses.

Time-domain features

The results of the time-domain features of HRV are presented in Table 6. Eleven studies reported results during pregnancy,^{21,26,50,30,36,37,39,40,44-46} and one study described results postpartum.⁵¹

Standard deviation of all NN intervals (SDNN)

During pregnancy

In CH, no difference was found in SDNN compared to normotensive controls.²¹ One study found a decrease in SDNN in GH compared to normotensive controls.²¹ The Ln(SDNN) was decreased in GH compared to normotensive controls in one study.⁴⁰ A decreased SDNN was found at 12 weeks of gestation in women who would later develop GH compared to normotensive controls,³⁹ whereas another study showed no difference between these groups.⁴⁶ No significant differences were found at 24 or 31 weeks of gestation.^{39,46} In PE, SDNN was found to be decreased (3 studies)^{26,30,36} or not different (2 studies)^{21,37} from normotensive controls. SDNN was found to be increased in early-onset PE and not different in late-onset PE compared to normotensive controls.³⁷

Ln(SDNN) was decreased in PE (1 study) or comparable to normotensive controls (1 study).44

Postpartum

Postpartum, SDNN was found to be significantly lower after PE compared to normotensive controls. $^{\rm 51}$

HRV triangular index

During pregnancy

In CH, HRV triangular index was significantly higher compared to normotensive controls. $^{\rm 45}$

One study showed HRV to be significantly lower in PE compared to normotensive controls.³⁶ HRV triangular index was found to be lower in moderate PE and comparable between mild PE and normotensive controls.⁴⁵

Standard deviation of the averages of NN intervals (SDANN)

During pregnancy

SDANN was shown to be decreased in PE compared to normotensive controls (1 study).³⁶ Another study showed Ln(SDANN) in either PE or GH to be equal to normotensive controls.⁴⁰

Root mean square of successive differences (RMSSD)

During pregnancy

RMSSD was not different between CH and normotensive controls (1 study).²¹

In GH, RMSS was not different from normotensive controls (1 study).²¹ No difference in RMSSD between women who would develop GH and normotensive controls at 12 or 24 weeks of gestation was found. At 31 weeks, RMSSD was decreased in women who would develop GH compared to normotensive controls.^{39,46}

In PE, RMSSD was increased (1 study),³⁷ decreased (2 studies),^{26,30} or comparable to normotensive women (2 studies).^{21,36} The Ln(RMSSD) was not different between PE and normotensive controls.⁴⁴

Postpartum

 RMSSD was decreased after a pregnancy complicated by PE compared to normotensive controls. 51

78

Table 6. Time	domain f	eatures, dur	ing pregn:	ancy								
Author Year	Brown 2008	Chaswal 2018	Euliano 2018	Faber 2004	Guo 2018	Khlybova 2008	Lakhno 2017	Musa 2016	Pal 2009	Pal 2011	Weber 2017	Yokusoglu 2009
Epoch No. women	10min 40	5 min 80	30 min 80	30 min 161	24 hours 120	500 cycles 55	10 min 121	5 min 120	10 min 65	10 min 69	10 min 96	24 hours 63
Mean NN	GH =			PE: = GH: ↓ CH: =					GH: GA 12: ↓ GA 31: ↓ GA 31: ↓	GH: GA 12: ↓ GA 24: ↓ GA 31: =		
SDNN		PE: ←		PE: = GH: ↓ CH: =			PE: ←		GH: GA 12: = GA 24: = GA 31: =	GH: GA 12: ↓ GA 24: = GA 31: =	PE: = (total =, <34 =, >34 ↑)	PE: ←
Ln (SDNN)					PE:↓ GH:↓			PE: =				
RMSSD		PE: ←		PE: = GH: = CH: =			PE: ←		GH: GA 12: = GA 24: = GA 31: ↓	GH: GA 12: = GA 24: = GA 31: ↓	PE: ↑ (total ↑, <34 =, >34 ↑)	PE:=
Ln(RMSSD)								PE:=				
SDANN Ln(SDANN5)					PE: = GH: =							ÞE: ←
pNN50												PE: 1

ntinued.	
0	
%	
ole	
Tat	

Author Year	Brown 2008	Chaswal 2018	Euliano 2018	Faber 2004	Guo 2018	Khlybova 2008	Lakhno 2017	Musa 2016	Pal 2009	Pal 2011	Weber 2017	Yokusoglu 2009
Epoch No. women	10min 40	5 min 80	30 min 80	30 min 161	24 hours 120	500 cycles 55	10 min 121	5 min 120	10 min 65	10 min 69	10 min 96	24 hours 63
pRR50%			PE: =			Mild PE: =	PE: ↓					
						Moderate PE ↓						
						CH:↑						
HRV						Mild PE: =						PE: ↓
triangular						Moderate PE ↓						
index						CH:↑						
WPSUM13				PE: =								
				GH:=								
				CH: =								
PLVAR20				PE: =								
				GH: =								
				CH: =								

↑: significant increase, ↓: significant decrease, =: no significant difference compared to a control group of normotensive pregnant women. Empty cell: parameter not analyzed in this study; PE: preeclampsia; GH: gestational hypertension; GA: gestational age; No. patients: number of patients included in the analyses.

80

Table 7. Time domain features, postpartum

Author Year	Murphy 2015
Epoch No. women	10 min 40
Mean NN	PE:↓
SDNN	PE:↓
Ln (SDNN)	
RMSSD	PE:↓
Ln(RMSSD)	
SDANN	
pNN50	PE:↓
pRR50%	PE:↓
HRV triangular index	
WPSUM13	PE:↓

 \uparrow : significant increase, \downarrow : significant decrease, =: no significant difference compared to a control group of women with a history of a normotensive pregnancy.

X: parameter not analyzed in this study; PE: preeclampsia; No. patients: number of patients included in the analyses.

HRV changes in GH and PE

A decrease in overall HRV was found in GH and PE, compared to normotensive pregnant controls. A trend is seen towards increased LF/HF-ratio in women with (moderate) PE compared to normotensive pregnant controls. An increased LF/HF-ratio was found in two longitudinal studies in women who would later in their pregnancy develop GH. However, this difference in LF/HF-ratio was not found in other studies in women with GH compared to normotensive pregnant controls.

DISCUSSION

This systematic review provides an overview of the available literature on HRV in HPD. The frequency-domain features results show a wide variety, possibly due to the heterogenicity of the data. Normalized units of LF and HF show more consistency between the studies. In pregnant women with PE, three out of four studies showed an increased LF(n.u.),^{30,44,49} and both longitudinal studies showed in increased LF(n.u.) in GH.^{39,46} HF(n.u.) was decreased in all three studies measuring it during pregnancy with PE.^{30,44,49} The LF/HF-ratio is increased in women with (moderate) PE compared to normotensive pregnant controls in four out of the six studies.^{26,30,45,49} No study demonstrated a decrease in the LF/HF-ratio. Similar results were found for GH and CH.

In time-domain features, a decrease in overall HRV was shown in PE, with both SDNN and HRV triangular index being significantly lower compared to normotensive pregnant controls. Postpartum, SDNN was measured in one study showing a significant decrease after PE compared to women with a history of a normotensive pregnancy.

These results may suggest a sympathetic overdrive of the autonomous nervous system in HPD, possibly associated with parasympathetic withdrawal. This could explain the differences in adaptations of the cardiovascular system to pregnancy between HPD and normotensive pregnancies.

Sympathetic overactivity is also stated to be a precursor of essential hypertension and CVD, years before the onset of symptoms.⁵⁴⁻⁵⁶ The sympathetic overdrive seen in HPD could explain the higher risk of developing essential hypertension and CVD in later life.

However, caution should be applied to these conclusions, given the large diversity in the results of the included studies.

Two previous reviews regarding autonomic function of pregnant women with PE showed the same wide variety in results with a trend towards more autonomic dysfunction in women with HPD compared to normotensive pregnant controls.^{17,27} Our review is the first to present the results of all HPD both during pregnancy and postpartum after a pregnancy complicated by HPD. Thereby, it provides a complete overview given the shared pathophysiological mechanisms and the long-term persistent effects of these complicated pregnancies.

We were unable to perform a meta-analysis due to heterogenicity of data in the included studies regarding, amongst others, the used definitions of HPD, the mode and duration of recording HRV, and the gestational age or period postpartum.

In the majority of studies, non-invasive ECG recordings were used. Standard ECG recording is a practical way to assess the autonomic nervous system in a clinical setting.²⁵ However, there is a wide range in duration of ECG recordings. The "gold standard" is a 24h measurement. Subsequently, the recording should be divided into 5-minute segments.²⁵ The length of the recording affects both the frequency- and time-domain. Longer recordings show an increased overall HRV. Therefore, comparing and pooling HRV features measured in different time periods is inappropriate.^{25,57}

Furthermore, there is a large range in gestational age of the included pregnancies. During early pregnancy, an increase in sympathetic nervous system activity is seen.⁵⁸ When gestation advances, the LF/HF-ratio increases during normal uncomplicated pregnancy.^{59,60} Therefore, the large variety in gestational age of the included women makes pooling of data even more difficult.

Chapter 3

Another contributor to the heterogenicity is the variety in bandwidths used for the frequency-domain features. The bandwidths as recommended by the HRV Task Force were used by all but four studies.^{38,48,52,53} Three studies reported no information regarding the bandwidths used.^{26,44,45}

For this review, we chose to exclude the analyses in which pregnant women with HPD were compared with non-pregnant women, since HRV is altered in normotensive pregnant women compared to normotensive non-pregnant women.^{30,53}

Imbalance autonomic nervous system

A major finding in our systematic review is a trend towards an elevated sympathetic tone in HPD and a reduced parasympathetic tone. Although the pathophysiologic mechanisms that cause HPD remain to be elucidated, the imbalance of the autonomic nervous system could contribute to the origin of HPD. Figure 2 illustrates a schematic overview of how the autonomic imbalance is hypothesized to lead to HPD, based on several studies and reviews.^{15,16,19,61-63} Increased sympathetic activity can cause an increase in peripheral resistance, which subsequently leads to high blood pressure.^{15,16,19,61-63} Furthermore, an increased sympathetic tone may foster placental ischemic/reperfusion events, leading to the release of abnormal soluble placental factors into the maternal circulation.⁶¹ This results in endothelial dysfunction and adrenergic receptor-induced vasoconstriction, which aggravates placental ischemia.^{16,61} In the end, these mechanisms could contribute to the development of HPD.

Effect of medication on HRV

According to standard protocol in several international guidelines, women with HPD are likely to receive routine medication such as antihypertensive drug therapy, and in severe cases also corticosteroids and magnesium sulfate (MgSO J).^{1,64} This introduces an important confounding factor in many studies. Khlybova et al. state that administrating antihypertensive drugs may increase maternal HRV, due to inhibition of beta-adrenergic receptors and/or a decrease in the amount of myocardial beta-adrenergic receptors.⁴⁵ To our knowledge, there are no studies regarding the effect of MgSO, and corticosteroids on maternal HRV. The effect of both drugs on fetal HRV, however, has been well documented.⁶⁵⁻⁶⁹ Given the mechanism of action of both drugs, it is likely that maternal HRV is affected by its admission. However, none of the included studies reported on the administration of corticosteroids, little is reported regarding MgSO, and whether HRV was measured before or after administration. In order to differentiate between the effect of medication on maternal HRV or merely HPD as a condition, further research regarding the effect of medication on maternal HRV in pregnant women is mandatory.







Strengths and limitations

One of the strengths of this systematic review is our rigorous search with the help of a professional librarian. Furthermore, we applied no restrictions regarding language or year of publication. Another strength of this review is the moderate to high quality of the included studies.

The limitation of this review was the wide variation of the included studies regarding the definitions of HPD, frequency-domain bandwidths, the duration and method of HRV measurement, and gestational age or period postpartum. Due to this heterogenicity, a meta-analysis could not be performed.

HRV is a non-invasive and accurate method to assess autonomic function and is also a good predictor of the risk of CVD.^{25,29,70} However, HRV per se has certain limitations. Whereas HF power primarily reflects parasympathetic influence, LF power is not a pure marker of sympathetic drive but reflects both sympathetic and parasympathetic influences.^{25,70} Furthermore, HRV is not an ideal index of autonomic activity and reactivity. Therefore, in clinical practice it could be useful to also perform other autonomic function tests, like cardiovascular reflex tests and baroreflex sensitivity, to assess autonomic dysfunctions in HPD.

Recommendations for future research

This review shows a large diversity in used methods to measure HRV. There is urgent need for standardized methods to measure HRV that would preferably use the bandwidths and duration of the measurement as suggested by the HRV Task Force.²⁵

Mother and fetus can be measured simultaneously during pregnancy. Lakhno showed that in case of PE, both maternal and fetal HRV were significantly altered compared to normotensive pregnancies, with a decreased SDNN, RMSSD, LF, and HF.²⁶ Furthermore, less fetal accelerations and more fetal decelerations were demonstrated during complicated pregnancies.²⁶ There is growing evidence that children born after pregnancies complicated by HPD have a higher risk of developing CVD later in life.⁷¹⁻⁷⁴ Compared to the offspring of uncomplicated pregnancies, the offspring of pregnancies complicated by HPD have a higher blood pressure and more subclinical changes in cardiac structure in childhood or young adulthood.^{73,75,76} Longitudinal studies with long term neonatal follow-up are needed to investigate the clinical relevance of abnormal fetal HRV measured during pregnancy.

In addition, longitudinal cohort studies can investigate whether HRV is able to predict HPD in a pre-clinical stage. The studies by Pal et al. are the only longitudinal studies included in this review.^{39,46} They showed a significantly higher LF/HF-ratio early in pregnancy in women who would later develop GH, compared to normotensive pregnant women of similar gestational age and pregnant women with risk factors for GH who would not develop GH.^{39,46} The study by Subha et al. found similar results, with a higher LF/HF-ratio in all three trimesters in high risk women who would develop GH, compared to high risk women who would remain to have a normotensive pregnancy.⁶⁰ These results are promising and indicate the need for further research. HRV might be helpful to compose a prediction model to identify women at risk for developing HPD.

CONCLUSION

Our systematic review supports the hypothesis that a sympathetic overdrive is found in HPD which is associated with a parasympathetic withdrawal.

However, the included studies in our review showed large diversity in the methods applied and the results.

ACKNOWLEDGMENTS

This research was performed within the framework of Eindhoven MedTech Innovation Center (e/MTIC). We thank Drs. H.E.J. de Vries, a professional medical research librarian who helped us with the search.

APPENDIX 1

Search string

((("Heart Rate"[Mesh] OR heart rate*[tiab]) AND variabilit*[tw]) OR HRV[tiab]) AND ((("Hypertension, Pregnancy-Induced"[Mesh] OR ("Pregnancy"[Mesh] AND "Hypertension"[Mesh])) OR ((pregnan*[tiab] OR gestation*[tiab] OR maternal[tiab]) AND hypertens*[tiab])) OR ("Pre-Eclampsia"[Mesh] OR pre eclampsia[tiab] OR preeclampsia[tiab]))

REFERENCES

- 1. ACOG Practice Bulletin No. 202: Gestational Hypertension and Preeclampsia. *Obstet Gynecol.* 2019;133(1):e-e25. doi:10.1097/AOG.000000000003018 [doi]
- 2. Umesawa M, Kobashi G. Epidemiology of hypertensive disorders in pregnancy: prevalence, risk factors, predictors and prognosis. *Hypertens Res.* 2017;40(3):213-220. doi:10.1038/hr.2016.126 [doi]
- 3. Wagner SJ, Barac S, Garovic VD. Hypertensive pregnancy disorders: current concepts. *J Clin Hypertens (Greenwich)*. 2007;9(7):560-566.
- 4. Hutcheon JA, Lisonkova S, Joseph KS. Epidemiology of pre-eclampsia and the other hypertensive disorders of pregnancy. *Best Pract Res Obstet Gynaecol.* 2011;25(4):391-403. doi:10.1016/j.bpobgyn.2011.01.006 [doi]
- 5. Kalafat E, Thilaganathan B. Cardiovascular origins of preeclampsia. *Curr Opin Obstet Gynecol.* 2017;29(6):383-389. doi:10.1097/GCO.00000000000419 [doi]
- 6. Folk DM. Hypertensive Disorders of Pregnancy: Overview and Current Recommendations. *J Midwifery Womens Health.* 2018;63(3):289-300. doi:10.1111/jmwh.12725
- 7. ACOG Practice Bulletin No. 203: Chronic Hypertension in Pregnancy. *Obstet Gynecol.* 2019;133(1):e26-e50. doi:10.1097/AOG.000000000003020
- 8. Shah S, Gupta A. Hypertensive Disorders of Pregnancy. *Cardiol Clin.* 2019;37(3):345-354. doi:10.1016/j.ccl.2019.04.008
- Saleem S, McClure EM, Goudar SS, et al. A prospective study of maternal, fetal and neonatal deaths in low- and middle-income countries. *Bull World Health Organ*. 2014;92(8):605-612. doi:10.2471/BLT.13.127464
- Khan KS, Wojdyla D, Say L, Gulmezoglu AM, Look PF Van. WHO analysis of causes of maternal death: a systematic review. *Lancet (London, England)*. 2006;367(9516):1066-1074. doi:S0140-6736(06)68397-9 [pii]
- Bellamy L, Casas JP, Hingorani AD, Williams DJ. Pre-eclampsia and risk of cardiovascular disease and cancer in later life: systematic review and meta-analysis. *BMJ*. 2007;335(7627):974. doi:bmj.39335.385301.BE [pii]
- 12. Wu P, Haththotuwa R, Kwok CS, et al. Preeclampsia and Future Cardiovascular Health: A Systematic Review and Meta-Analysis. *Circ Qual outcomes*. 2017;10(2):10.1161/ CIRCOUTCOMES.116.003497. Epub 2017 Feb 22. doi:e003497 [pii]
- 13. Tooher J, Thornton C, Makris A, Ogle R, Korda A, Hennessy A. All Hypertensive Disorders of Pregnancy Increase the Risk of Future Cardiovascular Disease. *Hypertension*. 2017;70(4):798-803. doi:10.1161/HYPERTENSIONAHA.117.09246
- 14. Logue OC, George EM, Bidwell GL 3rd. Preeclampsia and the brain: neural control of cardiovascular changes during pregnancy and neurological outcomes of preeclampsia. *Clin Sci (Lond)*. 2016;130(16):1417-1434. doi:10.1042/CS20160108
- 15. Ghossein-Doha C, Hooijschuur MCE, Spaanderman MEA. Pre-Eclampsia: A Twilight Zone Between Health and Cardiovascular Disease? *J Am Coll Cardiol*. 2018;72(1):12-16. doi:S0735-1097(18)34726-0 [pii]
- 16. Reyes LM, Usselman CW, Davenport MH, Steinback CD. Sympathetic Nervous System Regulation in Human Normotensive and Hypertensive Pregnancies. *Hypertens (Dallas, Tex 1979)*. 2018;71(5):793-803. doi:10.1161/HYPERTENSIONAHA.117.10766

- 17. Yousif D, Bellos I, Penzlin AI, et al. Autonomic Dysfunction in Preeclampsia: A Systematic Review. *Front Neurol.* 2019;10:816. doi:10.3389/fneur.2019.00816
- Heiskanen N, Saarelainen H, Karkkainen H, et al. Cardiovascular autonomic responses to head-up tilt in gestational hypertension and normal pregnancy. *Blood Press*. 2011;20(2):84-91. doi://dx.doi.org/10.3109/08037051.2010.532313
- 19. Schobel HP, Fischer T, Heuszer K, Geiger H, Schmieder RE. Preeclampsia -- a state of sympathetic overactivity. *N Engl J Med.* 1996;335(20):1480-1485. doi:10.1056/ NEJM199611143352002
- 20. Brown CA, Lee CT, Hains SM, Kisilevsky BS. Maternal heart rate variability and fetal behavior in hypertensive and normotensive pregnancies. *Biol Res Nurs*. 2008;10(2):134-144. doi:10.1177/1099800408322942 [doi]
- 21. Faber R, Baumert M, Stepan H, Wessel N, Voss A, Walther T. Baroreflex sensitivity, heart rate, and blood pressure variability in hypertensive pregnancy disorders. *J Hum Hypertens*. 2004;18(10):707-712. doi:10.1038/sj.jhh.1001730 [doi]
- 22. Guyenet PG. The sympathetic control of blood pressure. *Nat Rev Neurosci*. 2006;7(5):335-346.doi:10.1038/nrn1902
- 23. Julius S, Majahalme S. The changing face of sympathetic overactivity in hypertension. Ann Med. 2000;32(5):365-370. doi:10.3109/07853890008995939
- 24. Musa S. Sympathetic activity in preeclampsia, a study of heart rate variability. *J Hypertens*. 2018;36:e5. doi://dx.doi.org/10.1097/01.hjh.0000548004.87380.fa
- 25. Task Force of The European Society of Cardiology and The North American Society of Pacing and Electrophysiology. Heart rate variability. Standards of measurement, physiological interpretation, and clinical use. *Eur Heart J.* 1996;17(3):354-381.
- 26. Lakhno I. Autonomic imbalance captures maternal and fetal circulatory response to pre-eclampsia. *Clin Hypertens*. 2017;23:x. eCollection 2017. doi:10.1186/s40885-016-0061-x [doi]
- 27. Rang S, Wolf H, Montfrans GA, Karemaker JM. Non-invasive assessment of autonomic cardiovascular control in normal human pregnancy and pregnancy- associated hypertensive disorders: a review. *J Hypertens*. 2002;20(11):2111-2119.
- 28. Baier V, Baumert M, Caminal P, Vallverdu M, Faber R, Voss A. Hidden Markov models based on symbolic dynamics for statistical modeling of cardiovascular control in hypertensive pregnancy disorders. *IEEE Trans Biomed Eng.* 2006;53(1):140-143. doi:10.1109/TBME.2005.859812 [doi]
- 29. Acharya UR, Joseph KP, Kannathal N, Lim C, Suri J. Heart rate variability: a review. *Med Biol Eng Comput.* 2006;44(12):1031-1051. doi:10.1007/s11517-006-0119-0
- 30. Chaswal M, Kapoor R, Batra A, Verma S, Yadav BS. Heart Rate Variability and Cardiovascular Reflex Tests for Assessment of Autonomic Functions in Preeclampsia. *Int J Hypertens*. 2018;2018:8163824. doi:10.1155/2018/8163824 [doi]
- 31. Moher D, Liberati A, Tetzlaff J, Altman DG, Group P. Preferred reporting items for systematic reviews and meta-analyses: the PRISMA statement. *PLoS Med.* 2009;6(7):e1000097.doi:10.1371/journal.pmed.1000097 [doi]
- 32. Cochrane Netherlands. Website Dutch Cochrane Center, subsection Assessment forms and other downloads. Accessed November 15, 2018. https://netherlands.cochrane.org/beoordelingsformulieren-en-andere-downloads
- 33. Higgins JPT, (editors) GS. Cochrane Handbook for Systematic Reviews of Interventions. Version 5. The Cochrane Collaboration, 2011. Available from www.handbook.cochrane. org.

- 34. Losilla JM, Oliveras I, Marin-Garcia JA, Vives J. Three risk of bias tools lead to opposite conclusions in observational research synthesis. *J Clin Epidemiol.* 2018;101:61-72. doi:S0895-4356(18)30039-8 [pii]
- 35. Yang CC, Chao TC, Kuo TB, Yin CS, Chen HI. Preeclamptic pregnancy is associated with increased sympathetic and decreased parasympathetic control of HR. *Am J Physiol Circ Physiol.* 2000;278(4):1269. doi:10.1152/ajpheart.2000.278.4.H1269 [doi]
- 36. Yokusoglu M, Dede M, Uzun M, et al. Cardiac autonomic balance is impaired in preeclampsia. *Turkiye Klin J Med Sci.* 2009;29(3):605-610. http://tipbilimleri. turkiyeklinikleri.com/download_pdf.php?id=54798
- 37. Weber TM, Lackner HK, Roessler A, et al. Heart rate variability and baroreceptor reflex sensitivity in early- versus late-onset preeclampsia. *PLoS One*. 2017;12(10):e0186521. doi:10.1371/journal.pone.0186521 [doi]
- Ekholm EM, Tahvanainen KU, Metsala T. Heart rate and blood pressure variabilities are increased in pregnancy-induced hypertension. *Am J Obstet Gynecol*. 1997;177(5):1208-1212. doi:S0002-9378(97)70041-3 [pii]
- 39. Pal GK, Shyma P, Habeebullah S, Pal P, Nanda N, Shyjus P. Vagal withdrawal and sympathetic overactivity contribute to the genesis of early-onset pregnancy-induced hypertension. *Int J Hypertens*. 2011;2011:361417. doi:10.4061/2011/361417 [doi]
- 40. Guo J, Liu G, Guo G. Association of insulin resistance and autonomic tone in patients with pregnancy-induced hypertension. *Clin Exp Hypertens (New York, NY 1993)*. 2018;40(5):476-480. doi:10.1080/10641963.2017.1403619 [doi]
- 41. Eneroth E, Westgren M, Ericsson M, Lindblad LE, Storck N. 24-hour ECG frequencydomain measures in preeclamptic and healthy pregnant women during and after pregnancy. *Hypertens pregnancy*. 1999;18(1):1-9.
- 42. Metsaars WP, Ganzevoort W, Karemaker JM, Rang S, Wolf H. Increased sympathetic activity present in early hypertensive pregnancy is not lowered by plasma volume expansion. *Hypertens pregnancy*. 2006;25(3):143-157. doi:PP88046K55T24783 [pii]
- 43. Swansburg ML, Brown CA, Hains SM, Smith GN, Kisilevsky BS. Maternal cardiac autonomic function and fetal heart rate in preeclamptic compared to normotensive pregnancies. *Can J Cardiovasc Nurs*. 2005;15(3):42-52.
- 44. Musa SM, Adam I, Lutfi MF. Heart Rate Variability and Autonomic Modulations in Preeclampsia. *PLoS One*. 2016;11(4):e0152704. doi:10.1371/journal.pone.0152704 [doi]
- 45. Khlybova S V, Tsirkin VI, Dvorianskii SA, Makarova IA, Trukhin AN. Heart rate variability in normal and complicated pregnancies. *Fiziol Cheloveka*. 2008;34(5):97-105.
- 46. Pal GK, Shyma P, Habeebullah S, Shyjus P, Pal P. Spectral analysis of heart rate variability for early prediction of pregnancy-induced hypertension. *Clin Exp Hypertens (New York, NY 1993)*. 2009;31(4):330-341. doi:10.1080/10641960802621333 [pii]
- 47. Tejera E, Areias MJ, Rodrigues AI, Nieto-Villar JM, Rebelo I. Blood pressure and heart rate variability complexity analysis in pregnant women with hypertension. *Hypertens pregnancy*. 2012;31(1):91-106. doi:10.3109/10641955.2010.544801 [doi]
- 48. Eneroth-Grimfors E, Westgren M, Ericson M, Ihrman-Sandahl C, Lindblad LE. Autonomic cardiovascular control in normal and pre-eclamptic pregnancy. *Acta Obstet Gynecol Scand*. 1994;73(9):680-684.
- 49. Khan G, Ishrat N, Zulquarnain Z. Analysis of heart rate variability in pre-eclamptic pregnancy: a study employing frequency domain analysis. *Int J Reprod Contraception*, *Obstet Gynecol*. 2014;3(4):1037. doi:10.5455/2320-1770.ijrcog20141232

- 50. Euliano TY, Michalopoulos K, Singh S, et al. Photoplethysmography and Heart Rate Variability for the Diagnosis of Preeclampsia. *Anesth Analg.* 2018;126(3):913-919. doi:10.1213/ANE.0000000002532 [doi]
- 51. Murphy MS, Seaborn GE, Redfearn DP, Smith GN. Reduced Heart Rate Variability and Altered Cardiac Conduction after Pre-Eclampsia. *PLoS One*. 2015;10(9):e0138664. doi:10.1371/journal.pone.0138664 [doi]
- 52. Flood P, McKinley P, Monk C, et al. Beat-to-beat heart rate and blood pressure variability and hypertensive disease in pregnancy. *Am J Perinatol.* 2015;32(11):1050-1058. doi:10.1055/s-0035-1548542 [doi]
- 53. Lewinsky RM, Riskin-Mashiah S. Autonomic imbalance in preeclampsia: evidence for increased sympathetic tone in response to the supine-pressor test. *Obstet Gynecol.* 1998;91(6):935-939. doi:S0029-7844(98)00105-7 [pii]
- 54. Wulsin LR, Horn PS, Perry JL, Massaro JM, D'Agostino RB. Autonomic Imbalance as a Predictor of Metabolic Risks, Cardiovascular Disease, Diabetes, and Mortality. *J Clin Endocrinol Metab.* 2015;100(6):2443-2448. doi:10.1210/jc.2015-1748
- 55. Pal GK, Pal P, Nanda N, Amudharaj D, Karthik S. Spectral analysis of heart rate variability (HRV) may predict the future development of essential hypertension. *Med Hypotheses*. 2009;72(2):183-185. doi:10.1016/j.mehy.2008.07.060
- Kubota Y, Chen LY, Whitsel EA, Folsom AR. Heart rate variability and lifetime risk of cardiovascular disease: the Atherosclerosis Risk in Communities Study. Ann Epidemiol. 2017;27(10):619-625.e2. doi:10.1016/j.annepidem.2017.08.024
- 57. Saul JP, Albrecht P, Berger RD, Cohen RJ. Analysis of long term heart rate variability: methods, 1/f scaling and implications. *Comput Cardiol*. 1988;14:419. https://www.ncbi. nlm.nih.gov/pubmed/11542156
- 58. Fu Q. Hemodynamic and Electrocardiographic Aspects of Uncomplicated Singleton Pregnancy. *Adv Exp Med Biol.* 2018;1065:413-431. doi:10.1007/978-3-319-77932-4_26
- 59. M Balajewicz-Nowak, A Furgala, K Pitynski, P Thor, H Huras, K Rytlewski. The dynamics of autonomic nervous system activity and hemodynamic changes in pregnant women. *Neuroendocrinol Lett.* 2016;37(1):70-77.
- 60. Subha M, Pal P, Pal GK, Habeebullah S, Adithan C, Sridhar MG. Decreased baroreflex sensitivity is linked to sympathovagal imbalance, low-grade inflammation, and oxidative stress in pregnancy-induced hypertension. *Clin Exp Hypertens (New York, NY 1993)*. 2016;38(8):666-672. doi:10.1080/10641963.2016.1200596 [doi]
- 61. Spradley FT. Sympathetic nervous system control of vascular function and blood pressure during pregnancy and preeclampsia. *J Hypertens*. 2019;37(3):476-487. doi:10.1097/HJH.00000000001901
- 62. Greenwood JP, Scott EM, Walker JJ, Stoker JB, Mary DASG. The magnitude of sympathetic hyperactivity in pregnancy-induced hypertension and preeclampsia. *Am J Hypertens*. 2003;16(3):194-199. doi:10.1016/S0895-7061(02)03256-9
- 63. Greenwood JP, Scott EM, Stoker JB, Walker JJ, Mary DASG. Sympathetic Neural Mechanisms in Normal and Hypertensive Pregnancy in Humans. *Circulation*. 2001;104(18):2200-2204. doi:10.1161/hc4301.098253
- 64. NICE project Team. NICE Guideline Hypertension in Pregnancy: Diagnosis and Management.; 2019.
- 65. Noben L, Verdurmen KMJ, Warmerdam GJJ, Vullings R, Oei SG, van Laar JOEH. The fetal electrocardiogram to detect the effects of betamethasone on fetal heart rate variability. *Early Hum Dev.* 2019;130:57-64. doi:10.1016/j.earlhumdev.2019.01.011

Chapter 3

- 67. Verdurmen KMJ, Hulsenboom ADJ, van Laar JOEH, Oei SG. Effect of tocolytic drugs on fetal heart rate variability: a systematic review. *J Matern Fetal Neonatal Med.* 2017;30(20):2387-2394. doi:10.1080/14767058.2016.1249844 [doi]
- 68. Nensi A, Silva DA De, von Dadelszen P, et al. Effect of magnesium sulphate on fetal heart rate parameters: a systematic review. *J Obstet Gynaecol Can.* 2014;36(12):1055-1064. doi:S1701-2163(15)30382-0 [pii]
- 69. Verdurmen KMJ, Warmerdam GJJ, Lempersz C, et al. The influence of betamethasone on fetal heart rate variability, obtained by non-invasive fetal electrocardiogram recordings. *Early Hum Dev.* 2018;119:8-14. doi:S0378-3782(17)30150-0 [pii]
- 70. Thayer JF, Yamamoto SS, Brosschot JF. The relationship of autonomic imbalance, heart rate variability and cardiovascular disease risk factors. *Int J Cardiol.* 2010;141(2):122-131. doi:10.1016/j.ijcard.2009.09.543
- Davis EF, Lazdam M, Lewandowski AJ, et al. Cardiovascular Risk Factors in Children and Young Adults Born to Preeclamptic Pregnancies: A Systematic Review. *Pediatrics*. 2012;129(6):e155-e1561. doi:10.1542/peds.2011-3093
- 72. Tenhola S, Rahiala E, Martikainen A, Halonen P, Voutilainen R. Blood Pressure, Serum Lipids, Fasting Insulin, and Adrenal Hormones in 12-Year-Old Children Born with Maternal Preeclampsia. *J Clin Endocrinol Metab.* 2003;88(3):1217-1222. doi:10.1210/jc.2002-020903
- 73. Davis EF, Lewandowski AJ, Aye C, et al. Clinical cardiovascular risk during young adulthood in offspring of hypertensive pregnancies: insights from a 20-year prospective follow-up birth cohort. *BMJ Open.* 2015;5(6):e008136. doi:10.1136/ bmjopen-2015-008136
- 74. Sacks KN, Friger M, Shoham-Vardi I, et al. Prenatal exposure to preeclampsia as an independent risk factor for long-term cardiovascular morbidity of the offspring. *Pregnancy Hypertens An Int J Women's Cardiovasc Heal*. 2018;13:181-186. doi:10.1016/j. preghy.2018.06.013
- 75. Andraweera PH, Lassi ZS. Cardiovascular Risk Factors in Offspring of Preeclamptic Pregnancies-Systematic Review and Meta-Analysis. *J Pediatr.* 2019;208:104-113.e6. doi:10.1016/j.jpeds.2018.12.008
- 76. Timpka S, Macdonald-Wallis C, Hughes AD, et al. Hypertensive Disorders of Pregnancy and Offspring Cardiac Structure and Function in Adolescence. *J Am Heart Assoc.* 2016;5(11). doi:10.1161/JAHA.116.003906





Changes in maternal heart rate variability in response to the administration of routine obstetric medications in hospitalized patients; study protocol for a cohort study (MAMA-heart study)

> M Bester* S Moors* R Joshi MB van der Hout-van der Jagt SG Oei M Mischi R Vullings JOEH van Laar

> > * equal contribution

Submitted

ABSTRACT

Introduction

Pregnancy is a period of continuous change in the maternal cardiovascular system, partly mediated by the autonomic nervous system (ANS). However, the ANS does not always adapt sufficiently to support the increasing demands of pregnancy – a scenario associated with various pregnancy complications, e.g. hypertensive pregnancy disorders (HDP) and preterm birth (PTB). Early detection of these complications (both leading causes of perinatal morbidity and mortality) could enable actionable interventions to mitigate the burden of disease.

Owing to the association of HPD and PTB with autonomic dysfunction, monitoring maternal heart rate variability (mHRV) – a proxy measure for ANS activity – could aid in the early detection of these complications. However, the pathway to mHRV alteration in relation to HPD or PTB remains unclear. Patients diagnosed with these complications are routinely administered obstetric medications, including antenatal corticosteroids, which affect fetal HRV. Little is known on how these medications (particularly corticosteroids) affect mHRV. Subsequently, administering these may confound mHRV measurements in this cohort. Therefore, our study will investigate how these medications affect mHRV.

Methods

A longitudinal, observational cohort study will be conducted in a tertiary obstetric care unit. We will include 61 women admitted between 23 5/7 and 33 6/7 weeks' gestation with an indication to receive corticosteroids antenatally. Continuous photoplethysmography (PPG) measurements will be acquired with a wrist-worn device throughout subjects' hospitalization to determine mHRV, facilitating within-patient comparisons of the effect of corticosteroids and other routinely administered obstetric medications on mHRV. Additionally, 24 hours of PPG measurements will be acquired at home at six weeks postpartum to enable a comparison between antenatal and postpartum mHRV.

Trial registration

Dutch Trial Register, NL8204; registered on December 6, 2019.

INTRODUCTION

Pregnancy is a period of continuous anatomical and physiological changes in both mother and fetus.¹ During this period, most maternal physiological systems undergo considerable adaptation to support the growing fetus. Some of the most prominent changes needed to sustain the increasing metabolic demands of the maternal-fetal dyad occur in the maternal cardiovascular system.¹⁻³

These maternal cardiovascular adaptations involve, amongst others, changes in blood pressure and heart rate (HR).¹ The main mechanisms mediating these changes are related to the endocrine and the autonomic nervous systems (ANS).^{2,4} However, in some cases, the ANS does not sufficiently adapt to support the increasing demands of pregnancy – a scenario which is associated with various pregnancy complications.⁵ Two prominent examples are hypertensive pregnancy disorders (HPD) and preterm birth (PTB), both of which are leading causes of worldwide perinatal and maternal morbidity and mortality.⁶⁻⁹ Although the exact etiology remains uncertain, both conditions are believed to be associated with dysfunctional autonomic regulation.^{5,10-13} Since early detection of HPD and PTB (i.e. birth before 37 weeks of gestation) is challenging, alleviating the burden of these complications remains an important challenge in perinatology.

Early detection of HPD and imminent PTB is important and actionable, since effective risk-mitigating interventions exist. While significant effort has been devoted to developing prognostic models for the early detection of pregnancy complications, clinically implementable progress has been modest.¹⁴ Challenges include, on the one hand, that most of these models have not been clinically validated, while on the other, these models often require clinical markers that are either impractical to acquire or simply unavailable.^{14,15}

An alternative practical opportunity for the early detection of deterioration in maternal health is through continuously tracking changes in maternal heart rate variability (mHRV), as a proxy measure for the regulation offered by the ANS.¹⁶⁻¹⁸ mHRV can be readily monitored using unobtrusive wearable technologies at home, e.g. wristwatches employing photoplethysmography (PPG).¹⁹ Hence, it would be worthwhile to investigate whether continuous antenatal monitoring of mHRV may aid in the early detection of pregnancy complications.

Detecting pregnancy complications via mHRV requires a clear understanding of how these complications might affect mHRV. Considering previously published literature on this topic, many studies have explored how mHRV changes in pregnancies complicated by HPD,^{16,20-22} though often with conflicting results.²³ Contrarily, only one study describes the relationship between PTB and mHRV.⁵ Clearly, this remains an open research area.

97

An important limitation in studying HRV in this patient cohort is the difficulty in obtaining mHRV measurements that are not confounded by the use of medication. Typically, the onset of HPD and PTB is sudden, resulting in swift hospitalization to obstetric care units (OCUs) where patients frequently receive routine obstetric medications such as tocolytic drugs, magnesium sulfate (MgSO₄), corticosteroids, or antihypertensive treatment.²⁴⁻²⁶ These medications possibly affect mHRV, thus acting as potential confounders in attempts to assess maternal health from mHRV measurements. However, the changes in mHRV resulting from this confounding effect remains largely unknown.²³

Several studies have, in fact, urged investigation into the possible confounding effects of these medications on mHRV as this might help in better identifying changes in mHRV related to pregnancy complications.^{21,23} Conducting such studies, however, is challenging. One of the primary reasons is because patients admitted to OCUs do not routinely undergo continuous HR monitoring (e.g. via ECG). Other factors include logistical challenges such as the urgent admission of patients to OCU along with, typically, speedy administration of one or more medications. Possibly, because of these reasons, many studies investigating mHRV do not address the use of medication,^{20,22} while others only conduct short measurements before the administration of medication.^{5,16}

To our knowledge, only two studies – one in hospitalized women, the other in rats – have investigated the changes in mHRV in response to the administration of routinely used obstetric medication. Weissman et al. found that a tocolytic drug (atisoban) – which is used to arrest preterm contractions – had no effect on mHRV.²⁷ Contrarily, Zhou et al. observed changes in mHRV when administering MgSO₄ to rats with medically induced preeclampsia (a type of HPD). While the rats' condition resulted in lowered HRV, this was partly normalized after administering MgSO₄.²⁸

In addition to tocolytics, another widely used type of medication in obstetrics is corticosteroids. These are antenatally administered to women with threatened PTB between 24 and 34 weeks of gestation to induce fetal lung maturation, thereby reducing the risk of neonatal morbidity and mortality.²⁹ Fetal HRV (fHRV) is known to be affected by corticosteroids.³⁰⁻³³ Subsequently, it is likely that these will also affect mHRV. Yet, the effect of corticosteroids on HRV in adults has only been studied in a small group (n = 12) of healthy male subjects.³⁴ In this study, while the HR of participants increased, HRV remained unchanged. However, we cannot assume a similar response in pregnant women, as HRV not only differs between sexes,³⁵ but is also impacted by the duration of gestation.^{20,36,37}

Consequently, in this paper, we describe a study design to investigate changes in mHRV in response to routinely administered obstetric medications in a cohort of patients with pregnancy complications. Quantifying these changes is important as it not only fills a knowledge gap but could also aid future studies in distinguishing changes in mHRV

owing to pregnancy complications from those resulting from the use of medication. A clearer understanding of how pregnancy complications affect mHRV could in turn enable predictive monitoring of the same, allowing timely implementation of risk-mitigating interventions.

METHODS AND ANALYSIS

Aim of the study

This study aims to investigate the effect of routinely administered obstetric medications on mHRV in patients hospitalized due to pregnancy complications.

Clinical setting

This longitudinal, observational cohort study will be conducted at the OCU of Máxima Medical Center (Máxima MC), Veldhoven, The Netherlands, starting July 2020. The study cohort will comprise patients admitted to the OCU between 23 5/7 and 33 6/7 weeks of gestation with an indication to receive corticosteroids antenatally. Since Máxima MC is a tertiary obstetric referral center, high-risk patients from neighboring secondary care hospitals are often transferred here. Subsequently, the study cohort comprises of patients directly admitted to Máxima MC, as well as those transferred from elsewhere.

Clinical data acquisition

Longitudinal PPG measurements will be continuously acquired with the Philips Data Logger (PDL, Philips Research, Eindhoven, the Netherlands). The PDL – shown in Figure 1 – is a non-invasive wrist-worn device (CE-marked) that acquires PPG data (32 Hz) through optical sensing that measure changes in blood volume. Previous studies have utilized a predecessor of this device to collect PPG measurements in free-living conditions.^{38,39}

PPG measurements capture the time intervals between pulses resulting from subsequent heart beats, serving as a measure of HR, from which HRV can be calculated.⁴⁰ Furthermore, the PDL also records movement data using a tri-axial accelerometer (range: ± 8 G, sampled at 32 Hz), which can aid in filtering out motion artifacts. The PDL offloads acquired data to a mobile phone via Bluetooth. Data are not displayed on either the PDL or the mobile phone, ensuring that acquired data cannot influence clinical decision making.

In addition to PPG measurements, the study utilizes patient data routinely collected in electronic patient files. These data – detailed in the *Study Parameter* section – include maternal-fetal health parameters and routine measurements, e.g. cardiotocography (CTG).





Figure 1: The Philips Data Logger. This device will be employed in this study to acquire PPG and accelerometer data. The device does not display this PPG and accelerometer data, it only displays the time.

Routinely administered medications in obstetric care settings

Owing to their clinical state, the patient cohort participating in this study will be administered one or more obstetric medications as part of their standard clinical care. All medications administered during this study are part of standard care and not influenced by study participation.

When pregnancy complications are diagnosed before 34 weeks of gestation, patients receive corticosteroids (betamethasone). Owing to its frequent use and effects on fHRV, our study design focuses on this medication. A course of betamethasone (Celestone Chrondose®, Schering AG, Berlin, Germany) consists of two 11.4 mg injections administered intramuscularly, each consisting of 50% betamethasone phosphate for quick uptake (\approx 1 hour) and 50% betamethasone acetate for slow release to facilitate sustained exposure.⁴¹⁻⁴³ Although the pharmacokinetics of betamethasone in the maternal system are not fully known, its maximum effect and terminal half-life (i.e. time until the drug concentration in plasma reduces by 50%) are believed to lie within 0.5-3 hours and within 6-12 hours after administration, respectively.⁴³⁻⁴⁶ Betamethasone's biological half-life – which relates to its effect on the hypothalamus-pituitary-adrenal axis – is 36-59 hours.^{47,48} This medication is cleared from the maternal system within 48 hours.⁴⁴

Patients will typically receive other obstetric medications in addition to corticosteroids. In cases of threatened PTB, patients are likely to receive tocolytic drugs such as nifedipine (taken orally) or atosiban (administered intravenously) to attenuate uterine contractions.²⁵ Furthermore, patients in the study population can also receive MgSO₄, which is given

intravenously and prescribed for either fetal neuroprotection (in case of PTB <30 weeks' gestation) or maternal neuroprotection (in case of severe HPD).^{24–26} Patients with HPD might also be administered anti-hypertensive medications such as labetalol, methyldopa, nifedipine, or nicardipine, which are administered either orally or intravenously.

Study design

The study comprises two periods of PPG measurements in the same study population. The *primary phase* will assess the effect of obstetric medications on mHRV, based on PPG data gathered throughout subjects' hospitalization in the OCU. The *secondary phase* - added to compare cardiovascular features between the antenatal and postpartum periods – consists of 24-hour PPG measurements at six weeks postpartum, acquired in free-living conditions at home.

<u>Primary phase</u>: Enrollment in the *primary phase* occurs at patient admission to the OCU at Máxima MC. Due to the typically speedy administration of betamethasone after admission, PPG measurements with the PDL will start as soon as possible. Subjects will in so far as is reasonable wear the PDL throughout their hospitalization. If possible, subjects will wear the PDL on their non-dominant hand to reduce motion artifacts in the PPG measurements.

We define the principal PPG measurement epochs around the administration of the second betamethasone injection (administered 24 hours after the first), as the study population will likely contain patients transferred from other hospitals who have already received their first injection elsewhere. The principal baseline measurement epoch – shown in Figure 2 in light blue – is the two hours leading up to the second injection. Since the terminal half-life of betamethasone is 6–12 hours,^{43–46} we plan this baseline measurement 22 hours after the first injection. The principal post-medication measurement epoch is the four hours after the injection (shown as dark red in Figure 2), as betamethasone's maximum effect occurs 30 minutes to 3 hours after administration.^{43–46} We will exclude and replace subjects for whom less than 50% of reliable PPG data is available in each of these principal measurements epochs.

Since PPG measurements will be longitudinal, there is a possibility to incorporate multiple measurement epochs to improve the accuracy of the analysis. Subsequently, we define a set of additional baseline and post-medication PPG measurements centered on the first betamethasone injection where possible. A third baseline PPG measurement is defined 48 hours after the second injection, although this might be unavailable if the subject is transferred to another hospital or goes into labor.

Participation in the *primary phase* ends upon discharge from Máxima MC. Where possible, subjects who go into labor during the primary phase will also wear the PDL during delivery.



Figure 2: Baseline and post-medication measurements acquired in the primary phase of the study. The principal measurement epochs occur around the second betamethasone injection, while additional measurement epochs are defined around the first betamethasone injection and 48 hours

<u>Secondary phase</u>: Participants from the *primary phase* have the option of participating in the *secondary phase*. This consists of wearing the PDL at six weeks postpartum for a

24-hour monitoring period in free-living conditions at home.

Primary and secondary analyses

after the second injection.

Primary phase: The primary analysis will determine the effect of administering betamethasone on mHRV. Secondary analyses will explore the effect of other medications on mHRV, compare cardiovascular parameters between subgroups (e.g. stratified by diagnosis), assess cardiovascular parameters during delivery and evaluate similarities between trends in PPG and CTG measurements.

Secondary phase: PPG measurements acquired in the *secondary phase* will further facilitate secondary analyses, including a within-patient comparison of cardiovascular parameters between the antenatal and postpartum periods. If the sample size allows, we will also compare postpartum parameters between subgroups (stratified by diagnosis).

Study parameters

We will assess cardiovascular parameters derived from the PPG measurements to perform our analyses. These include HR, HRV features (e.g. SDNN, RMSSD, HF, LF and pNN50) and features based on the morphology of the PPG waveform (e.g. pulse area and large artery stiffness index⁴⁹). To describe the study cohort, we will also collect the following data from patient records:

- 1. Maternal condition:
 - patient characteristics, including age, BMI and ethnicity
 - pregnancy characteristics, including gestational age, results of prenatal screening, and complications in pregnancy
 - obstetric history, including gravidity, parity, and previous pregnancy or labor complications
 - family history, including genetic or congenital diseases or a history of hypertension or preeclampsia
 - medical condition, including preexisting diseases (i.e. cardiovascular disease, pre-existing hypertension, autoimmune disorders, neurologic disorders)
 - routine measurements, including blood pressure, laboratory test results, physical examination results, ultrasound results.
- 2. Fetal/neonatal condition, including fetal growth, congenital diseases, birth weight, APGAR score, CTG measurements and umbilical cord blood gases.
- 3. Labor and delivery, including mode of delivery and clinical notes.
- 4. Administration of medications, including timing, dosage, and reasons for administration.

The electronic medical records only contain information relevant to a patient's hospitalization or appointments at Máxima MC. Subsequently, we will contact subjects who did not deliver at Máxima MC for basic details of their delivery (i.e. birth weight and gestational age).

For subjects who participate in the *secondary phase* and have their postpartum appointments at Máxima MC, information on their postpartum condition (e.g. postpartum complications and standard checkup measurements) will also be collected from their electronic medical records.

Subject inclusion and exclusion criteria

Patients admitted to the OCU at Máxima MC who are going to receive one or both dosages of betamethasone injection(s) are eligible for inclusion. Table 1 outlines the entire inclusion and exclusion criteria.

Chapter 4

Table 1: Inclusion and exclusion c	criteria for MAMA-hart study
------------------------------------	------------------------------

-		_			
_Ir	iclusion criteria	E	Exclusion criteria		
-	Age 18 years and above Gestational age 23 5/7 to 33 6/7 weeks	-	History of severe arrhythmia and/or maternal congenital heart disease		
-	Yet to receive the second	-	Cushing's disease		
	betamethasone injection	-	Known allergies to hard plastic (e.g. used in		
-	Proficient in Dutch or English		sport watches) or elastic band material		
		-	Wounds, injuries or infectious diseases on wrist where the PDL will be worn		
		-	Tattoo location on wrist interfering with the positioning of the PDL		
		-	Both wrists unavailable for wearing the PDL (e.g. owing to intravenous lines)		
		-	Dexamethasone (another brand of corticosteroid) administered instead of betamethasone		

Retrospectively, if subjects are identified to be incorrectly enrolled (i.e. not meeting the full eligibility criteria), they will be excluded from the study analysis and replaced with a new subject.

Sample size

In accordance with the primary analysis (i.e. the effect of administering betamethasone on mHRV during pregnancy), we designed the study to detect the differences in mean maternal HR (mHR) between the baseline and post-medication epochs. Subsequently, we calculated the sample size using a two-sided T-test for a confidence interval of 95% and a power assumption of 80%. The expected variation in HR measurements was estimated from studies assessing HR in pregnant populations.^{11,50,51} No prior research was available to guide the decision concerning effect size, but the heterogeneity of the cohort could necessitate being able to detect a small effect. Subsequently, we selected an effect size of 10%.

Based on these parameters, we calculated a sample size of 43 using R (version 3.5.3, RStudio Inc., Boston, MA, USA). Adding in a safety margin of 20% increased the sample size to 61. We will perform an interim analysis after including 43 subjects to assess whether further inclusions are necessary. The study is powered for the primary analysis , and not specifically powered for secondary analyses.

Statistical analysis

A within-subject comparison of cardiovascular features (derived from PPG measurements) in the baseline and post-medication epochs will be carried out for the primary analysis, i.e. determining the changes in mHRV in response to administering betamethasone. For secondary analyses in both *phases*, we will perform within-subject as

well as between-group comparisons. We will test normality assumptions with a Shapiro-Wilk test and subsequently compare continuous variables using a paired T-test or a Wilcoxon matched-pairs test, and categorical variables using a χ^2 -test or Fisher's exact test, chosen as appropriate. P<0.05 is considered significant for a two-tailed test.

Data handling and storage

We will adhere to the European General Data Protection Regulation (GDPR) and the Dutch Personal Data Protection Act ("Uitvoeringswet AVG") for data processing and analyses. Subsequently, subject data will be de-identified.

We will use Research Manager (version 5.51.0, Research Manager, Deventer, The Netherlands) for the case report form and data handling. Personal data will be stored in accordance with Good Clinical Practice guidelines. Analyses are carried out under the Eindhoven MedTech Innovation Center framework, a collaboration between Máxima MC, Philips Research and the Eindhoven University of Technology.

Ethics and dissemination

The Medical Ethics Committee of Máxima MC, Veldhoven, The Netherlands, confirmed that the study neither imposes any changes in general practice, nor does it burden participants. Therefore, in line with the Declaration of Helsinki, a waiver for ethical approval was granted (N19.112).

All investigators agree to publish the study results in an international peer-reviewed journal, regardless of whether the outcomes align with the stated hypotheses. The full study protocol (version 2.0, dated January 20, 2020) is available upon request.

Patient and public involvement

Patients and the public were not involved in the design or planning of the study.

DISCUSSION

This study is one of only a few to explore the effect of administering routine obstetric medications on mHRV,^{27,28} and the first to investigate changes in mHRV resulting from antenatal administration of corticosteroids (betamethasone). Apart from a small number of human and animal studies addressing the effect of this medication on HR and HRV in adult males and in baboons,^{34,52,53} research has focused on assessing changes in fHRV – demonstrating that administering betamethasone significantly decreases fHRV parameters.^{30,33,54}

A possible reason for this disparity in research interest between maternal and fetal HRV is that fHRV is an established parameter in assessing fetal well-being, while mHRV is not yet

widely used in clinical practice.⁵⁵ Even so, fetal HR (fHR) is not continuously monitored in this cohort. Consequently, in a previous study that assessed the effect of betamethasone on fHRV, Verdurmen et al. had to deliberately incorporate fHR measurements into clinical workflow. This, as well as the typical urgency of administrating corticosteroids, imposed logistical challenges on their study.⁵⁴ Since our clinical setting is comparable, we will likely encounter similar obstacles. Subsequently, we designed the study to mitigate two main logistical challenges: firstly, acquiring sufficiently long and unconfounded baseline measurements; secondly, ensuring that the study protocol seamlessly fits into standard clinical workflow.

Verdurmen et al.'s baseline fHRV-measurements were planned five days after the second betamethasone injection, since betamethasone affects fHRV for four days.⁵⁴ However, many subjects were discharged or gave birth before this time point.⁵⁴ It is also impractical to acquire the principal baseline measurement before betamethasone administration since transferred subjects will likely have already received the first injection before inclusion in the study. Even for subjects directly admitted to Máxima MC, the epoch before the first injection could be too short to provide a sufficiently long baseline measurement. Furthermore, since admission is typically urgent and unexpected, patients are likely physiologically stressed during this epoch, which can affect HRV parameter.⁵⁶

Therefore, we define our principal PPG baseline measurement as the epoch preceding the second betamethasone injection. Based on the pharmacokinetics of the drug,⁴³⁻⁴⁶ levels of betamethasone in the maternal system are expected to decrease in the 24 hours between the first and second injection. Consequently, we hypothesize that if betamethasone affects mHRV, comparing epochs before and after the second injection will indicate this.

Still, it is possible that the first betamethasone injection could confound our principal baseline measurement. Even though studies show that the terminal half-life of betamethasone is 6 to 12 hours,⁴³⁻⁴⁶ the medication clears the maternal system only after 48 hours. Considering this, we introduce additional measurements to the analysis where possible to improve accuracy, both before administration of betamethasone as well as 48 hours thereafter.

It is uncertain how long mHRV will remain affected by the medication. It should be noted that the biological half-life of betamethasone is 36 to 59 hours;^{47,48} therefore, this could also confound measurements. However, the biological half-life was not determined in a pregnant population, and pregnancy is known to affect the pharmacokinetics of drugs.^{57,58} Literature has shown that fHRV remains affected for four days,⁵⁴ but Ballard et al. have also demonstrated that betamethasone's terminal half-life in the mother is

half of that in the fetus.⁴⁴ Subsequently, it is plausible that there is a less sustained effect on mHRV than on fHRV.

The second logistical challenge concerns ensuring that the study fits seamlessly into standard clinical workflow. Ideally, acquiring measurements should minimally disrupt clinical workflow and not burden patients or clinical staff. To overcome this logistical challenge, a wristwatch-like device (the PDL) is employed for collecting mHR since its ease of use requires minimal clinical intervention.

An ECG Holter monitor is an alternative device that might offer higher accuracy in determining mHRV. However, this approach is, both, more obtrusive and cumbersome for the patient and staff. Furthermore, in addition to high compliance in wrist-worn monitoring in pregnant populations,¹⁹ HRV determined from PPG measurements sampled above 25 Hz (PDL: 32 Hz) can be as reliable as that calculated from ECG.⁵⁹

Our study design not only addresses logistical challenges identified from previous studies, but also offers opportunities for additional exploratory analyses. Since PPG measurements are non-invasive, a dataset representing the periods of hospitalization of participants can be collected. Subsequently, we could analyze trends in mHR in conjunction with trends in routinely collected fHR, potentially offering insights into the utility of mHR monitoring as an indicator for fetal well-being. Ramadam et al. have already demonstrated that mHR mimics fHR characteristics,⁶⁰ while Gonçalves et al. have suggested that analyzing both maternal and fetal HR – compared to only fHR – could improve the detection of fetal acidemia.⁶¹

Furthermore, incorporating measurements from the *secondary phase* (i.e. at six weeks postpartum) allows for analysis of mHRV throughout the perinatal period (i.e. antepartum, intrapartum and postpartum), which – to our knowledge – has not been assessed. Insights into the postpartum period could be particularly useful, since literature on this is limited.⁶²⁻⁶⁴

For the results of the study to be applicable in clinical practice, participants will represent a cohort of women who typically receive corticosteroids, i.e. patients with varying characteristics and diagnoses (e.g. HDP, threatened PTB), and who subsequently receive multiple medications. The heterogeneity in characteristics and diagnoses could serve as limitations, as they will likely also influence mHRV. We account for this heterogeneity by focusing on within-patient comparisons when assessing the effect of betamethasone on mHRV, emphasizing the relative change between the epoch before and after the administration of medication, and averaging results across subjects. However, the heterogeneity could also be advantageous, as it allows us to explore analyses of subgroups stratified by diagnosis, e.g. differences in cardiovascular parameters of subjects who deliver prematurely and those who deliver at term. The administration of multiple obstetric medications is an unavoidable limitation in this study design. Although medications administered in conjunction with corticosteroids could confound measurements, there is also the opportunity to explore their effects on mHRV. Limited literature exists in this regard, aside from one study which determined that a tocolytic drug had no significant effect on mHRV.²⁷

Still, the most prominent knowledge gap concerns betamethasone. This is the first study to investigate the effect of administering this medication on mHRV. Results from this study could reduce the confounding effect of betamethasone in studies employing mHRV to investigate the autonomic dysregulation associated with pregnancy complications such as HPD or threatened PTB. An improved understanding of how mHRV is altered could facilitate earlier diagnosis through tracking deteriorations in mHRV. In turn, early detection could enable prevention or better management of these complications. alleviating some of the burden they place on women, families and society.

ACKNOWLEDGMENTS

This research was performed within the framework of Eindhoven MedTech Innovation Center, Eindhoven, The Netherlands – a collaboration between Máxima MC. Philips Research and the Eindhoven University of Technology.

REFERENCES

- 1. Tan EK, Tan EL. Alterations in physiology and anatomy during pregnancy. Best Pract Res Clin Obstet Gynaecol. 2013:27(6):791-802.
- 2. Fu Q. Hemodynamic and Electrocardiographic Aspects of Uncomplicated Singleton Pregnancy. Adv Exp Med Biol. 2018;1065:413-431.
- 3. Soma-Pillay P. Nelson-Piercy C. Tolppanen H. Mebazaa A. Physiological changes in pregnancy. Cardiovasc J Afr. 2016:27(2):89-94.
- 4. May L. Cardiac Physiology of Pregnancy. In: Teriung R. editor. Comprehensive Physiology. Hoboken, NJ, USA: John Wiley & Sons, Inc.; 2015. p. 1325-44.
- Khlybova SV, Tsirkin VI, Dvoryanskii SA, Makarova IA, Trukhin AN. Heart rate variability 5 in normal and complicated pregnancies. Hum Physiol. 2008 Sep 1:34(5):625-32.
- 6. Frey HA, Klebanoff MA. The epidemiology, etiology, and costs of preterm birth. Semin Fetal Neonatal Med. 2016 Apr;21(2):68-73.
- 7. Umesawa M, Kobashi G, Epidemiology of hypertensive disorders in pregnancy: prevalence, risk factors, predictors and prognosis. Hypertens Res Off J Jpn Soc Hypertens. 2017;40(3):213-20.
- 8. Saleem S. McClure EM. Goudar SS. Patel A. Esamai F. Garces A. et al. A prospective study of maternal, fetal and neonatal deaths in low- and middle-income countries. Bull World Health Organ. 2014 Aug 1;92(8):605–12.
- 9. Khan KS, Wojdyla D, Say L, Gulmezoglu AM, Look PFV. WHO analysis of causes of maternal death: a systematic review. Lancet Lond Engl. 2006 Apr;367(9516):1066-74.
- 10. Reves LM, Usselman CW, Davenport MH, Steinback CD. Sympathetic Nervous System Regulation in Human Normotensive and Hypertensive Pregnancies. Hypertens Dallas Tex 1979. 2018;71(5):793-803.
- 11. Brown CA, Lee CT, Hains SMJ, Kisilevsky BS. Maternal heart rate variability and fetal behavior in hypertensive and normotensive pregnancies. Biol Res Nurs. 2008 Oct:10(2):134-44.
- 12. Guyenet PG. The sympathetic control of blood pressure. Nat Rev Neurosci. 2006 May;7(5):335-46.
- 13. Julius S, Majahalme S. The changing face of sympathetic overactivity in hypertension. Ann Med. 2000 Jan 1;32(5):365-70.
- 14. Kleinrouweler CE, Cheong-See FM, Collins GS, Kwee A, Thangaratinam S, Khan KS, et al. Prognostic models in obstetrics: available, but far from applicable. Am J Obstet Gynecol. 2016 Jan;214(1):79-90.e36.
- 15. De Kat AC, Hirst J, Woodward M, Kennedy S, Peters SA. Prediction models for preeclampsia: A systematic review. Pregnancy Hypertens. 2019 Apr;16:48-66.
- 16. Musa S. Sympathetic activity in preeclampsia, a study of heart rate variability. J Hypertens. 2018;36:e5-e5.
- 17. Task Force of The European Society of Cardiology and The North American Society of Pacing and Electrophysiology. Heart rate variability. Standards of measurement, physiological interpretation, and clinical use. Eur Heart J. 1996;17(3):354-81.
- 18. Yousif D, Bellos I, Penzlin AI, Hijazi MM, Illigens BM-W, Pinter A, et al. Autonomic Dysfunction in Preeclampsia: A Systematic Review. Front Neurol. 2019;10.

- 19. Grym K, Niela-Vilén H, Ekholm E, Hamari L, Azimi I, Rahmani A, et al. Feasibility of smart wristbands for continuous monitoring during pregnancy and one month after birth. BMC Pregnancy Childbirth. 2019 Dec;19(1):34.
- 20. Chaswal M, Kapoor R, Batra A, Verma S, Yadav BS. Heart Rate Variability and Cardiovascular Reflex Tests for Assessment of Autonomic Functions in Preeclampsia. Int J Hypertens. 2018 Sep;2018:8163824–8163824.
- 21. Weber TM, Lackner HK, Roessler A, Papousek I, Kolovetsiou-Kreiner V, Lucovnik M, et al. Heart rate variability and baroreceptor reflex sensitivity in early- versus late-onset preeclampsia. Frasch MG, editor. PLOS ONE. 2017 Oct 20;12(10):e0186521.
- 22. Heiskanen N, Saarelainen H, Karkkainen H, Valtonen P, Lyyra-Laitinen T, Laitinen T, et al. Cardiovascular autonomic responses to head-up tilt in gestational hypertension and normal pregnancy. Blood Press. 2011;20(2):84–91.
- 23. Moors S, Staaks KJJ, Westerhuis MEMH, Dekker LRC, Verdurmen KMJ, Oei SG, et al. Heart rate variability in hypertensive pregnancy disorders: A systematic review. Pregnancy Hypertens. 2020 Mar;20:56–68.
- 24. ACOG Practice Bulletin No. 202: Gestational Hypertension and Preeclampsia. Obstet Gynecol. 2019 Jan;133(1):e1.
- 25. ACOG practice bulletin no. 127: Management of preterm labor. Obstet Gynecol. 2012 Jun;119(6):1308–1317.
- ACOG Practice Bulletin No. 188: Prelabor Rupture of Membranes. Obstet Gynecol. 2018 Jan;131(1):e1–e14.
- 27. Weissman A, Tobia RS, Burke YZ, Maxymovski O, Drugan A. The effects of oxytocin and atosiban on the modulation of heart rate in pregnant women. J Matern Fetal Neonatal Med. 2017 Feb;30(3):329–33.
- 28. Zhou Q, Shen J, Zhou G, Shen L, Zhou S, Li X. Effects of magnesium sulfate on heart rate, blood pressure variability and baroreflex sensitivity in preeclamptic rats treated with L-NAME. Hypertens Pregnancy. 2013 Nov;32(4):422–31.
- 29. Roberts D, Brown J, Medley N, Dalziel SR. Antenatal corticosteroids for accelerating fetal lung maturation for women at risk of preterm birth. Cochrane Pregnancy and Childbirth Group, editor. Cochrane Database Syst Rev. 2017 Mar 21;
- 30. Noben L, Verdurmen KMJ, Warmerdam GJJ, Vullings R, Oei SG, van Laar JOEH. The fetal electrocardiogram to detect the effects of betamethasone on fetal heart rate variability. Early Hum Dev. 2019 Mar;130:57–64.
- 31. Nensi A, Silva DAD, von Dadelszen P, Sawchuck D, Synnes AR, Crane J, et al. Effect of magnesium sulphate on fetal heart rate parameters: a systematic review. J Obstet Gynaecol Can JOGC J Obstet Gynecol Can JOGC. 2014 Dec;36(12):1055–64.
- 32. Verdurmen KMJ, Hulsenboom ADJ, van Laar JOEH, Oei SG. Effect of tocolytic drugs on fetal heart rate variability: a systematic review. J Matern Fetal Neonatal Med. 2017 Oct 18;30(20):2387–94.
- Verdurmen K, Van LJ, Oei S. Corticosteroids and fetal heart rate variability. J Matern Fetal Neonatal Med. 2014;27:361–361.
- 34. Cottin F, Malcurat V, Zorgati H, Prieur F, Labsy Z, Do MC, et al. Effect of oral glucocorticoid intake on autonomic cardiovascular control. SpringerPlus. 2015 Dec;4(1):622.
- 35. Koenig J, Thayer JF. Sex differences in healthy human heart rate variability: A metaanalysis. Neurosci Biobehav Rev. 2016 May;64:288–310.

- 36. Eneroth E, Westgren M, Ericsson M, Lindblad LE, Storck N. 24-hour ECG frequencydomain measures in preeclamptic and healthy pregnant women during and after pregnancy. Hypertens Pregnancy. 1999;18(1):1–9.
- Lewinsky RM, Riskin-Mashiah S. Autonomic imbalance in preeclampsia: evidence for increased sympathetic tone in response to the supine-pressor test. Obstet Gynecol. 1998 Jun;91(6):935–9.
- 38. Eerikäinen LM, Bonomi AG, Schipper F, Dekker LRC, Vullings R, de Morree HM, et al. Comparison between electrocardiogram- and photoplethysmogram-derived features for atrial fibrillation detection in free-living conditions. Physiol Meas. 2018 08;39(8):084001.
- 39. Radha M, de Groot K, Rajani N, Wong CCP, Kobold N, Vos V, et al. Estimating blood pressure trends and the nocturnal dip from photoplethysmography. Physiol Meas. 2019 Feb 26;40(2):025006.
- 40. Allen J. Photoplethysmography and its application in clinical physiological measurement. Physiol Meas. 2007 Mar 1;28(3):R1–39.
- 41. Kemp MW, Newnham JP, Challis JG, Jobe AH, Stock SJ. The clinical use of corticosteroids in pregnancy. Hum Reprod Update. 2015 Nov 20;dmv047.
- 42. Shanks AL, Grasch JL, Quinney SK, Haas DM. Controversies in antenatal corticosteroids. Semin Fetal Neonatal Med. 2019 Jun;24(3):182–8.
- 43. Salem II, Najib NM. Pharmacokinetics of Betamethasone After Single-Dose Intramuscular Administration of Betamethasone Phosphate and Betamethasone Acetate to Healthy Subjects. Clin Ther. 2012 Jan;34(1):214–20.
- 44. Ballard L, Ballard A. Scientific basis and therapeutic regimens for use of antenatal glucocorticoids. 1995;173(1):9.
- 45. Moss TJM, Doherty DA, Nitsos I, Harding R, Newnham JP. Pharmacokinetics of betamethasone after maternal or fetal intramuscular administration. Am J Obstet Gynecol. 2003 Dec;189(6):1751–7.
- 46. He C, Fan H, Tan J, Zou J, Zhu Y, Yang K, et al. Pharmacokinetics of betamethasone and betamethasone 17-monopropionate in Chinese healthy volunteers after intramuscular injection of betamethasone phosphate/betamethasone dipropionate. Arzneimittelforschung. 2011;61(7):417–420.
- 47. Jobe AH, Milad MA, Peppard T, Jusko WJ. Pharmacokinetics and Pharmacodynamics of Intramuscular and Oral Betamethasone and Dexamethasone in Reproductive Age Women in India. Clin Transl Sci. 2019 Dec 13;cts.12724.
- 48. MSD. Celestone Chronodose, suspensie voor injectie 5,7 mg/ml. 2018 Apr. Report No.: RVG 05399.
- 49. Elgendi M. On the Analysis of Fingertip Photoplethysmogram Signals. Curr Cardiol Rev. 2012 Jun 1;8(1):14–25.
- 50. Voss A, Malberg H, Schumann A, Wessel N, Walther T, Stepan H, et al. Baroreflex sensitivity, heart rate, and blood pressure variability in normal pregnancy. Am J Hypertens. 2000 Nov 1;13(11):1218–25.
- 51. Alam T, Choudhary A, D S. Maternal heart rate variability during different trimesters of pregnancy. Natl J Physiol Pharm Pharmacol. 2018;8(9):1475.
- 52. Koenen SV, Mecenas CA, Smith GS, Jenkins S, Nathanielsz PW. Effects of maternal betamethasone administration on fetal and maternal blood pressure and heart rate in the baboon at 0.7 of gestation. Am J Obstet Gynecol. 2002 Apr;186(4):812–7.

Chapter 4

- 53. Brotman DJ, Girod JP, Garcia MJ, Patel JV, Gupta M, Posch A, et al. Effects of Short-Term Glucocorticoids on Cardiovascular Biomarkers. J Clin Endocrinol Metab. 2005 Jun;90(6):3202–8.
- 54. Verdurmen KMJ, Warmerdam GJJ, Lempersz C, Hulsenboom ADJ, Renckens J, Dieleman JP, et al. The influence of betamethasone on fetal heart rate variability, obtained by non-invasive fetal electrocardiogram recordings. Early Hum Dev. 2018 Apr;119:8–14.
- 55. Ayres-de-Campos D, Spong CY, Chandraharan E, FIGO Intrapartum Fetal Monitoring Expert Consensus Panel. FIGO consensus guidelines on intrapartum fetal monitoring: Cardiotocography. Int J Gynecol Obstet. 2015 Oct;131(1):13–24.
- 56. Kim H-G, Cheon E-J, Bai D-S, Lee YH, Koo B-H. Stress and Heart Rate Variability: A Meta-Analysis and Review of the Literature. Psychiatry Investig. 2018 Mar 25;15(3):235–45.
- 57. Ansari J, Carvalho B, Shafer SL, Flood P. Pharmacokinetics and Pharmacodynamics of Drugs Commonly Used in Pregnancy and Parturition: Anesth Analg. 2016 Mar;122(3):786–804.
- 58. Feghali M, Venkataramanan R, Caritis S. Pharmacokinetics of drugs in pregnancy. Semin Perinatol. 2015 Nov;39(7):512–9.
- 59. Choi A, Shin H. Photoplethysmography sampling frequency: pilot assessment of how low can we go to analyze pulse rate variability with reliability? Physiol Meas. 2017 Mar 1;38(3):586–600.
- 60. Ramadan MK, Fasih R, Itani S, Salem Wehbe GR, Badr DA. Characteristics of fetal and maternal heart rate tracings during labor: A prospective observational study. J Neonatal-Perinat Med. 2020 Jan 4;12(4):405–10.
- 61. Gonçalves H, Pinto P, Silva M, Ayres-de-Campos D, Bernardes J. Toward the improvement in fetal monitoring during labor with the inclusion of maternal heart rate analysis. Med Biol Eng Comput. 2016 Apr;54(4):691–9.
- 62. Samways JW, Vause S, Kontopantelis E, Eddleston J, Ingleby S, Roberts A, et al. Maternal heart rate during the first 48 h postpartum: a retrospective cross sectional study. Eur J Obstet Gynecol Reprod Biol. 2016 Nov;206:41–7.
- 63. Mahendru AA, Everett TR, Wilkinson IB, Lees CC, McEniery CM. A longitudinal study of maternal cardiovascular function from preconception to the postpartum period: J Hypertens. 2014 Apr;32(4):849–56.
- 64. Groer MW, Jevitt CM, Sahebzamani F, Beckstead JW, Keefe DL. Breastfeeding Status and Maternal Cardiovascular Variables Across the Postpartum. J Womens Health. 2013 May;22(5):453–9.





Management of intrapartum fetal distress in The Netherlands: a clinical practice survey

LM Bullens S Moors PJ van Runnard Heimel MB van der Hout-van der Jagt SG Oei

European Journal of Obstetrics and Gynecology and Reproductive Biology 2016;205:48-53

ABSTRACT

Objective

Solid evidence on the effect of intrauterine resuscitation on neonatal outcome is limited, and superiority of one intervention over the others is not clear. We therefore surveyed the clinical practice variation in fetal monitoring and the management of fetal distress during labor, in Dutch labor wards. In addition, we have compared recommendations from international guidelines.

Study design

We conducted a survey among all 86 Dutch hospitals, using a questionnaire on fetal monitoring and management of fetal distress. In addition, we requested international guidelines of 28 western countries to study international recommendations regarding labor management.

Results

The response rate of the national survey was 100%. Labor wards of all hospitals use ctg for fetal monitoring, 98% use additional fetal scalp blood sampling, and 23% use stanalysis. When fetal distress is suspected, oxytocin is discontinued and tocolytic drugs are applied in all hospitals. Nearly all hospitals (98%) use maternal reposition for fetal resuscitation, 33% use amnioinfusion, and 58% provide maternal hyperoxygenation. Management is mainly based on the Dutch national guideline (58%) or on local guidelines (26%). Eight international guidelines on fetal monitoring were obtained for analysis. Fetal scalp blood sampling facilities are recommended in all the obtained guidelines. Use of st-analysis is recommended in three guidelines and advised against in three guidelines. Five guidelines also advised on intrauterine resuscitation: discontinuation of oxytocin and use of tocolytic drugs was advised in all guidelines, whereas maternal hyperoxygenation was recommended in two guidelines and advised against in one guideline.

Conclusion

Nationwide clinical practice, and recommendations from international guidelines agree on the use of fetal scalp blood sampling in addition to cardiotocography during labor. The opinion on the use of st-analysis differs per clinic and per guideline. Discontinuation of oxytocin, administration of tocolytic drugs and maternal repositioning are rather uniform, on national and international level. However, there is a large variation in the use of amnioinfusion and maternal hyperoxygenation, which may be explained by the contradictory recommendations of the different guidelines.

INTRODUCTION

Non-reassuring fetal heart rate (FHR) patterns frequently occur during labor. They may be indicative for impaired fetal oxygenation, which eventually may lead to fetal asphyxia.^{1,2} As fetal asphyxia is associated with hypoxic-ischemic encephalopathy and even fetal death, timely intervention is important to optimize neonatal outcome.

Intrauterine resuscitation is defined as interventions with the intention to improve fetal oxygenation during labor, in the presence of suspected fetal distress. Depending on the presumable cause of the abnormal FHR pattern, these interventions aim to restore oxygenation of the fetus. Possible actions consist of alleviation of cord compression and/ or improvement of uteroplacental blood flow.^{3,4}

Improvement of the intrauterine condition of the fetus can avoid termination of the delivery, thereby preventing a cesarean section or vaginally assisted delivery. In the case of an emergency cesarean section, intrauterine resuscitation may restore fetal oxygenation during the decision to incision time period.

Several techniques to improve fetal oxygenation are used in clinical practice. The most commonly used interventions are discontinuation of oxytocin infusion, maternal repositioning, amnioinfusion, maternal hyperoxygenation, and the use of tocolytic agents. Unfortunately, solid evidence on the effect of each of these interventions on neonatal outcome is limited, and superiority of one intervention over the others is not clear.⁵ This lack of evidence leads to differences in recommendations in two important guidelines on fetal monitoring during labor and intrapartum management.^{4,6} For example, the Practice Bulletin of the American Congress of Obstetricians and Gynecologists (ACOG) recommends maternal hyperoxygenation in the presence of fetal distress, whereas the Royal College of Obstetricians and Gynaecologist (RCOG) in the United Kingdom advises against this intervention.^{4,6} The Dutch Society of Obstetricians and Gynecologists (NVOG) is currently working on a recommendation regarding the use of maternal hyperoxygenation for fetal distress.⁷

Apparently, different guidelines have different recommendations regarding the management of fetal distress during labor. This may lead to clinical practice variation regarding the management of fetal distress during labor.

We aim to investigate the clinical practice variation in Dutch delivery wards, with specific interest in the local methods used for fetal monitoring, and the actions undertaken in suspected fetal distress. Local guidelines, as well as intervention techniques to improve fetal oxygenation are inventoried. Hence, we conducted a survey among all Dutch hospitals. In addition, we requested the national guidelines from 28 Western countries regarding fetal monitoring and fetal distress.

MATERIALS AND METHODS

A clinical practice survey was conducted from August to October 2015 in all 86 Dutch hospitals. Also, we aimed to obtain international guidelines of 25 European countries, the USA, Canada, and Australia & New Zealand.

Survey among Dutch obstetricians

Per hospital, one obstetrician was asked to complete a survey comprising twelve multiplechoice questions on fetal monitoring and common interventions regarding suspected fetal distress (Appendix A, original version in Dutch).

Topics included: methods used for fetal monitoring; classification method of the cardiotocogram (CTG); how to diagnose fetal distress; the indication and use of intrauterine resuscitation techniques and the use of national and/or international guidelines. Respondents were able to give more than one answer to each question and were free to add more options if their answer was not listed. In case of unclear responses, we contacted the respondent to clarify the answers.

In The Netherlands, low risk deliveries are managed by primary care midwives. These primary midwife practices are excluded from this survey, since most resuscitation techniques are not available during home births. Hence, in the presence of fetal distress, the parturient will be referred to a hospital.

Statistical analysis national survey

After all questionnaires were returned, we analyzed the results using SPSS (IBM SPSS Statistics, version 23). Categorical variables were expressed as frequencies and percentages.

Survey of national guidelines of European countries

We searched the Internet for international guidelines on fetal monitoring and resuscitation of 25 European countries, the USA, Canada, and Australia & New Zealand. If guidelines were not freely available, we approached the corresponding national societies of obstetricians and gynecologists and requested their national guideline on fetal monitoring and/or intrauterine resuscitation during labor.

If a guideline was not available in English, it was translated by a native speaker or by the use of an online translation program. We compared the recommendations as stated in the different guidelines. The following details were abstracted from the guidelines and listed: methods of fetal monitoring systems during labor (intermittent auscultation, fetal heart rate pattern, fetal scalp blood sampling and ST-analysis), the classification system to judge the fetal heart rate pattern, and recommendations on the use of intrauterine resuscitation techniques.

Ethical considerations

As confirmed by the Medical Ethics Committee of Máxima Medical Center, Veldhoven, The Netherlands, our study does not involve any patient data and imposes no changes in general practice. Therefore, according to the Declaration of Helsinki, no ethical approval was required.

RESULTS

Survey among Dutch obstetricians

A total of 86 obstetricians, representing all 86 Dutch hospitals, completed the questionnaire. Hospitals include eight university hospitals, 39 general teaching hospitals, and 39 non-teaching hospitals. The response rate was 100%.

Besides the national guideline on fetal monitoring provided by the NVOG, various local protocols exist on the diagnosis and management of fetal distress during labor. In The Netherlands, the guideline of the NVOG is frequently used (58%), sometimes in combination with a local guideline (26%). In 36% of the hospitals, only local protocols are used. The American guideline, provided by the American College of Obstetricians and Gynecologists (ACOG) is used in one hospital (1%), while the British guideline, provided by the Royal College of Obstetricians and Gynaecologists (RCOG) is used in six hospitals (7%). Results are shown in Fig. 1.



Figure 1. Percentage of Dutch hospitals using the represented guidelines on fetal monitoring and fetal resuscitation.

Practice variation in management of fetal distress

Chapter 5

All hospitals used fetal CTG to estimate fetal well-being during labor. Besides, in 23% (N = 20) of the hospitals ST-analysis is used to monitor fetal condition, while in 98% (N=84) fetal scalp blood sampling is used in addition to CTG.

In most of the hospitals (95%), the (modified) FIGO classification is used to classify the CTG. In two hospitals the Fischer classification is used.⁸ In two non-teaching hospitals no structural classification system is implemented.

Preferences regarding the use of resuscitation techniques are different among the hospitals. Fig. 2 shows the percentage of hospitals that use each of the mentioned resuscitation techniques. The most used tocolytic agents are oxytocin antagonists, 6.75 mg atosiban administered intravenously. Furthermore, beta-mimetics, 5-10 mg ritodrine administered intravenously, or nitroglycerin, administered oromucosally with a dose of 0.4 mg are in use.

When fetal distress is suspected, immediate delivery may be expedited, depending on the clinical situation. In almost all hospitals (98%), the attending staff will try to improve fetal oxygenation in expectation of an emergency cesarean section or vaginally assisted birth. Improvement of the intrauterine condition of the fetus may also avoid termination of the delivery, thereby preventing a cesarean section or vaginally assisted delivery. In 86% of the Dutch delivery wards, intrauterine resuscitation techniques are applied as an attempt to prevent immediate delivery.



Figure 2. Percentage of Dutch hospitals using the represented intrauterine resuscitation techniques in the presence of suspected intrapartum fetal distress.

Survey of national guidelines of European countries

We were able to obtain national guidelines on fetal monitoring of the following countries: United Kingdom (2015),⁶ United States of America (USA, 2009),⁴ Canada (2007),⁹ Australia & New Zealand (2014),¹⁰ Germany (2013),¹¹ Ireland (2014),¹² Portugal (2012),¹³ and Denmark (2008).¹⁴ The years refer to the date of publication of the most recent version of each guideline. Five of them also contained recommendations on the use of intrauterine resuscitation techniques.^{4,6,9,10,12} We were not able to obtain the guidelines of the remaining 20 countries. The national societies of Luxembourg and Finland reported they had no national guideline and used guidelines of surrounding countries.

All the above-mentioned guidelines were freely online available, apart from the Portuguese guideline that was kindly provided by the Portuguese Federation of Obstetrics and Gynecology. Germany and Portugal did have guidelines on fetal monitoring, but not on fetal resuscitations. Denmark did have a guideline on amnioinfusion, but no recommendations regarding the other resuscitation techniques were reported. Recommendations from the obtained guidelines are listed in Tables 1 and 2.

Table 1. Recommendations from national and international guidelines on fetal monitoring during labor.

Country	Intermittent auscultation	CTG	Classification	FBS	STAN	SpO ₂
			system			- 2
Netherlands	Low risk	High risk	FIGO	YES	YES	-
USA	-	YES	FIGO	-	-	-
UK	Low risk	High risk	FIGO	YES	-	-
Ireland	Low risk	High risk	FIGO	YES	-	-
Canada	Low risk	High risk	FIGO	YES	NO	NO
Australia	Low risk	High risk	-	YES	NO	NO
& New Zealand						
Germany	Low risk*	High risk	FIGO	YES	NO	-
Portugal	NO	YES	FIGO	YES	YES	-
Denmark	Low risk*	High risk	FIGO	YES	YES	-

Low risk = recommended in low risk population. High risk = recommended in high risk population. CTG = cardiotocogram. FIGO = (modified) FIGO classification

FBS = fetal scalp blood sampling. STAN = ST-analysis. SpO₂ = fetal pulse oximetry

- = not mentioned. * = intermittent electronic fetal heart rate monitoring allowed in a low risk population, under certain circumstances.

Table 2. Recommendation from national and international guidelines regarding intrauterine resuscitation during labor.

Country	Maternal repositioning	O ₂	Stop oxytocin	Tocolytic agent	Amnioinfusion	IV fluid bolus
Netherlands	YES	-	YES	YES	#	-
USA	YES	YES	YES	YES	YES	YES
UK	YES	NO	YES	YES	NO	YES
Ireland	YES	-	YES	YES	-	NO
Canada	YES	YES	YES	-	YES	YES
Australia	YES	-	YES	YES	NO	YES
& New Zealand						

O₂ = maternal hyperoxygenation. - = not mentioned. # = not recommended nor discouraged

COMMENT

The Netherlands

The Dutch national guideline on fetal monitoring during labor promotes the use of fetal scalp blood sampling, in combination with CTG.⁷ The Dutch national guideline on fetal monitoring provided by the NVOG does not advise on how the individual parameters of the CTG should be measured. Fetal heart rate can be monitored externally, or internally with a fetal scalp electrode. Uterine contractions can be monitored externally with an ultrasound transducer or with an electrode patch.¹⁵ Internal tocodynamometry can be performed using an intrauterine pressure catheter. A Cochrane review by Bakker et al. shows no superiority of internal over external tocodynamometry in induced or augmented labor.¹⁶

The use of ST-analysis is not promoted, since its use will not decrease the incidence of intrapartum acidosis. However, its use is not discouraged either, since it may be cost-effective in comparison to the use of only CTG and fetal scalp blood sampling.¹⁷⁻¹⁹ As a result, our nationwide survey on the diagnosis and management of intrapartum fetal distress in Dutch labor wards shows a large practice variation on the use of ST-analysis. In contrast, the use of fetal scalp blood sampling is rather uniform. In all but one hospital (98%), fetal scalp blood sampling facilities are available. However, ongoing discussion exists on the use of pH or lactate as a marker for fetal well-being during labor. The Cochrane review by East et al. concludes that fetal scalp blood lactate estimation is more likely to succeed than pH.²⁰ Nevertheless, due to the lack of long term neonatal follow up, no choice of preference has been made so far. Besides, there is no clear answer yet on the question which level of fetal scalp blood lactate indicates the need for intervention during labor.²¹⁻²⁴ As a consequence, in most Dutch hospitals fetal scalp blood pH is still used as a measure for fetal well-being during labor.

Regarding the use of intrauterine resuscitation techniques, a large practice variation was shown in the use of amnioinfusion and maternal hyperoxygenation. According to the Dutch guideline, the use of amnioinfusion is not helpful to improve neonatal outcome.⁷ However, it may decrease the presence of variable decelerations in the fetal heart rate pattern. Therefore the use of this intervention is not promoted, nor discouraged. In 33% of the Dutch hospitals amnioinfusion is used in the presence of fetal distress. So far, no recommendations are made on the use of maternal hyperoxygenation in the Dutch guideline. In 58% of the hospitals, this intervention is commonly used to promote fetal oxygenation.

In almost all hospitals, discontinuation of oxytocin, maternal repositioning and administration of tocolytic drugs is common practice. The administration of tocolytic drugs, preferably atosiban, is actually recommended in the Dutch guideline.⁷ Atosiban has a similar effect on uterine pressure as ritodrine, but has significantly less side effects on maternal condition.²⁵ However, the effect of atosiban on neonatal outcome is not investigated, in contrast to beta-mimetic drugs and ritodrine. A Cochrane review concludes that betamimetic therapy appears to improve abnormal fetal heart rate tracings.²⁶ They state that there is not enough evidence based on clinically important outcomes to evaluate the use of betamimetics for suspected fetal distress. Another systematic review suggests a positive effect of ritodrine, terbutaline, MgSO₄, orciprenaline and nitroglycerine on fetal well-being.⁵ Tocolytic drugs may decrease the need for emergency delivery without increasing maternal and fetal adverse side-effects.²⁷

The available literature to base practical recommendations on, is scarce. Therefore, local guidelines are typically based on results of small, non-randomized studies and expert opinions. Hence, the difference in delivery ward management regarding intrauterine resuscitation may be caused by the lack of strong evidence to promote or refuse certain techniques. Dutch labor wards use different national and international guidelines for their local protocol.

We believe the results are illustrative for the delivery ward management in our country. The response rate was 100%. Although it is very likely that most respondents are aware of the common delivery ward management in their hospitals, we cannot exclude that other staff members in certain cases practice other policies.

International guidelines

We aimed to compare the recommendations from the Dutch guideline on fetal monitoring and intrauterine resuscitation, with international guidelines. We managed to obtain eight international guidelines on intrapartum monitoring.^{4,6,9-14} Five of them also advised on the use of intrauterine resuscitation techniques.^{4,6,9,10,12} Canada, the United Kingdom and Australia & New Zealand have elaborate, evidence-based guidelines on antenatal and intrapartum fetal surveillance.^{6,9,10} These guidelines describe various fetal monitoring techniques, extensive CTG interpretation guidelines and management recommendations in case of non-reassuring fetal heart rate patterns. The Irish and American guidelines provide recommendations regarding intrauterine resuscitation, without an overview of the supporting literature.^{4,12} Other guidelines we obtained were exclusively on fetal monitoring during labor, or on a specific intrauterine resuscitation technique, e.g. the Danish guideline on amnioinfusion.²⁸ We assume that more European countries do have national guidelines, but unfortunately we were not able to obtain more than eight guidelines for analysis.

By comparing the eight different guidelines, we identified various contradictory recommendations. For example, the Practice Bulletin of the ACOG recommends maternal hyperoxygenation in the presence of fetal distress, whereas RCOG in the United Kingdom advises against this intervention.^{4,6} Also, amnioinfusion is recommended in Denmark, Canada and the United States, but advised against in the United Kingdom, Australia and New Zealand.^{4,6,9,10,28} The Netherlands did not state an explicit recommendation on the use of amnioinfusion.⁷

Evidence regarding the effect of the various intrauterine resuscitation techniques is limited, and sometimes contradictory. As a consequence, guidelines are mainly based on low-level evidence and consensus. Also, it is not clear how long the effect of intrauterine resuscitation should be awaited, before an emergency delivery is indicated.

To come to funded recommendations, the effect of the various resuscitation techniques should be investigated in randomized controlled trials. The technique studied could be compared to expectant management, or to another resuscitation technique. Since most of the interventions have become 'common practice', and therefore cannot be withholded, it will be difficult to conduct a randomized controlled trial. In our hospital (Máxima Medical Center), we have started a randomized controlled trial to investigate the effect of maternal hyperoxygenation on fetal distress during labor (EudraCT number 2015-001654-15, Dutch Trial Register number 5461, Central Committee on Research Involving Human Subjects number NL53018.000.15). We hope more studies to investigate the benefit of other resuscitation techniques will follow and lead to clear and uniform recommendations.

ACKNOWLEDGEMENTS

This research was performed within the framework of the IMPULS perinatology. We thank all gynecologists who agreed to participate to this survey.

APPENDIX I.

Questionnaire regarding diagnosis and management of fetal distress during labor (original questions in Dutch)

- Does your hospital have a delivery ward? A Yes
 B No (end of the questionnaire)
- What techniques for fetal monitoring are used? A Cardiotocogram B Cardiotocogram and/or fetal scalp blood sampling C Cardiotocogram and/or fetal scalp blood sampling and/or ST-analysis D Other...
- Is the cardiotocogram classified using a classification system? A Yes, the FIGO classification system is used B Yes, another classification system is used C No, no classification system is used
- 4. How fetal distress is diagnosed?
 A Suspicion due to an abnormal CTG
 B Confirmed using fetal scalp blood sampling
 C Confirmed using ST-analysis
 D Other...
- 5. Which action are undertaken in case of suspected fetal distress?
 A Confirmation of impaired fetal condition using fetal scalp blood sampling
 B Immediate delivery (vaginally assisted birth or cesarean section)
 C Application of intra-uterine resuscitation techniques
- 6. Which intrauterine resuscitation techniques are used in your hospital? A Discontinuation of oxytocin infusion B Use of tocolytic agents C Maternal repositioning D Amnioinfusion E Maternal hyperoxygenation F Other...

Practice variation in management of fetal distress

- 7. In case a tocolytic drug is administered, which drug and dose are used? A Ritodrine
 B Atosiban
 C Fenoterol
 D Other...
 - Dose:
- Are intra-uterine resuscitation techniques applied while waiting for an emergency cesarean section or vaginally assisted delivery?
 A Yes
 B No
- 9. If yes, which techniques are applied?
 A Discontinuation of oxytocin infusion
 B Use of tocolytic agents
 C Maternal repositioning
 D Amnioinfusion
 E Maternal hyperoxygenation
 F Other...
- On what bases one decides which intervention is applied? A Based on a guideline B Based on my own experience/what I have learnt
- 11. In case intervention is based on a guideline, which guideline is used?
 A NVOG
 B RCOG
 C ACOG
 D Local guideline
 E Other...
- 12. Which interventions do you think are effective when applied for fetal distress? A Discontinuation of oxytocin infusion
 B Use of tocolytic agents
 C Maternal repositioning
 D Amnioinfusion
 E Maternal hyperoxygenation
 F Intravenous fluid administration (not for correction of hypotension)
 G Other...

REFERENCES

- 1. Graham EM, Ruis KA, Hartman AL, Northington FJ, Fox HE. A systematic review of the role of intrapartum hypoxia-ischemia in the causation of neonatal encephalopathy. AJOG. 2008;199:587-95.
- 2. Parer JT. Effects of fetal asphyxia on brain cell structure and function: limits of tolerance. Comp Biochem Physiol A: Mol Integr Physiol. 1998;119:711-6.
- 3. Simpson KR. Intrauterine resuscitation during labor: should maternal oxygen administration be a first-line measure? Semin Fetal Neonatal Med. 2008;13:362-7.
- 4. American College of Obstetricians and Gynecologists. Practice bulletin no. 116: management of intrapartum fetal heart rate tracings. Obstet Gynecol 2010;116:1232-40.
- 5. Bullens LM, Van Runnard Heimel PJ, Van der Hout-van der Jagt MB, Oei SG. Interventions for intrauterine resuscitation in suspected fetal distress during term labor: a systematic review. Obstet Gynecol Surv. 2015;70:524-39.
- 6. National Collaborating Centre for Women's and Children's Health (UK). Intrapartum care: care of healthy women and their babies during childbirth. London: National Institute for Health and Care Excellence (UK); 2014.
- 7. Nederlandse Vereniging voor Obstetrie en Gynaecologie. Intrapartum fetal monitoring at term [Intrapartum foetale bewaking a terme] [internet]. Utrecht, The Netherlands: NVOG; May 2014 [updated May 2015]. Available from: http://nvog-documenten.nl/ uploaded/docs/NVOG%20richtlijn%20foetale%20bewaking%2019-05-2014%20 update%2028-5-2015.pdf. [Dutch]
- 8. Berg D, Brandt H, Ekert WD, Fischer M, Gennser G, Halberstadt E, et al. Kardiotokographie. Diagnostische Methoden in der Perinatologie. Stuttgart - New York: Georg Thieme Verlag; 1973. [German]
- Liston R, Sawchuck D, Young D: the Society of Obstetricians and Gynaecologists of Canada. Fetal health surveillance: antepartum and intrapartum consensus guideline. J Obstet Gynecol Can. 2007;29(9 Suppl. 4):S3-56.
- 10. The Royal Australian and New Zealand College of Obstetricians and Gynaecologists. Intrapartum Fetal Surveillance. Clinical Guideline, third ed. Available from: www.ranzcog. edu.au/intrapartum-fetal-surveillance-clinical-guidelines.html; May 2016.
- 11. Deutsche Gesellschaft für Gynäkologie und Guburtshilfe, Arbeitsgemeischaft Maternofetale Medizin, Deutsche Gezellschaft für Pränatal und Geburtzmedizin, Deutsche Gesellschaft für Perinatale Medizin. S1 Leitlinie: Anwendung des CTG während Schwangerschaft und Geburt. Available from: www.awmf.org/uploads/tx_szleitlinien/015-036l_S1_CTG_Schwangerschaft_Ge- burt_2014-06.pdf; May 2016. [German]
- 12. Institute of Obstetricians and Gynecologistst, Royal College of Physicians of Ireland and Directorate of Stategy and Clinical Programmes Health Service Executive. Clinical practice guideline: Intrapartum fetal heart rate monitoring. Version 1.2. Available from: www.hse.ie/eng/about/Who/clinical/natclinprog/obsandgynaeprogramme/guide6.pdf; May 2016.
- 13. Graça LM. Monitorização fetal intra-parto; 2012. [Portuguese]
- 14. Palmgren Colov N, Hedegaard M, Hvidman L, Stener Jørgensen J, Lenstrup C. Fosterovervågning under fødslen ved hjælp af STAN. Available from: http://clin.au.dk/ fileadmin/www.ki.au.dk/forskning/forskningsenheder/ gyn__kologisk-obstetrisk_ afd__y/logistics/sandbjerg_m_der/sandb-jerg_2008/stan.pdf; May 2016. [Danish]

- 15. Vlemminx MW, de Lau H, Vullings R, Peters CH, Oei SG. Electrohysterography. A promising alternative for monitoring contractions. Ned Tijdschr Geneeskd 2015;159:A8535. [Dutch]
- 16. Bakker JJ, Janssen PF, van Halem K, van der Goes BY, Papatsonis DN, van der Post JA, et al. Internal versus external tocodynamometry during induced or augmented labour. Cochrane Database Syst Rev. 2013;8:CD006947.
- 17. Vijgen SM, Westerhuis ME, Opmeer BC, Visser GH, Moons KG, Porath MM, et al. Cost-effectiveness of cardiotocography plus ST-analysis of the fetal electrocardiogram compared with cardiotocography only. Acta Obstet Gynecol Scand. 2011;90:772-8.
- 18. Heintz E, Brodtkorb TH, Nelson N, Levin LA. The long-term cost-effectiveness of fetal monitoring during labour: a comparison of cardiotocography complemented with ST analysis versus cardiotocography alone. BJOG 2008;115:1676-87.
- 19. Van 't Hooft J, Vink M, Opmeer BC, Ensing S, Kwee A, Mol BW. ST-analysis in electronic foetal monitoring is cost-effective from both the maternal and neonatal perspective. J Matern Fetal Neonatal Med. 2016;29:3260-5.
- 20. East CE, Leader LR, Sheehan P, Henshall NE, Colditz PB, Lau R. Intrapartum fetal scalp lactate sampling for fetal assessment in the presence of a non-reassuring fetal heart rate trace. Cochrane Database Syst Rev. 2015;1:CD006174.
- Kruger K, Hallberg B, Blennow M, Kublickas M, Westgren M. Predictive value of fetal scalp blood lactate concentration and pH as markers of neurologic disability. AJOG. 1999;181(5 Pt. 1):1072-8.
- 22. Allen RM, Bowling FG, Oats JJ. Determining the fetal scalp lactate level that indicates the need for intervention in labour. Aust N Z J Obstet Gynaecol. 2004;44:549-52.
- 23. Ramanah R, Martin A, Clement MC, Maillet R, Riethmuller D. Fetal scalp lactate microsampling for non-reassuring fetal status during labor: a prospective observational study. Fetal Diagn Ther. 2010;27:14-9.
- 24. Heinis AM, Spaanderman ME, Gunnewiek JM, Lotgering FK. Scalp blood lactate for intra-partum assessment of fetal metabolic acidosis. Acta Obstet Gynecol Scand. 2011;90:1107-14.
- 25. De Heus R, Mulder EJ, Derks JB, Kurver PH, van Wolfswinkel L, Visser GH. A prospective randomized trial of acute tocolysis in term labour with atosiban or ritodrine. Eur J Obstet Gynecol Reprod Biol. 2008;139:139-45.
- 26. Kulier R, Hofmeyr GJ. Tocolytics for suspected intrapartum fetal distress. Cochrane Database Syst Rev. 2000;2:CD000035.
- 27. Briozzo L, Martinez A, Nozar M, Fiol V, Pons J, Alonso J. Tocolysis and delayed delivery versus emergency delivery in cases of non-reassuring fetal status during labor. J Obstet Gynaecol Res. 2007;33:266-73.
- 28. Brix Westergaard H, Krebs L, Weber T, Bek Helmig R, Stener Jørgensen Jan, et al. Amnioinfusion under fødslen. Available from: http://clin.au.dk/fileadmin/www.ki.au. dk/forskning/for- skningsenheder/gyn_kologisk-obstetrisk_afd__y/logistics/sandbjerg_m_der/sandbjerg_2008/amnioinfusion.pdf; May 2016. [Danish]





Intrauterine resuscitation during term labor by maternal hyperoxygenation: a randomized controlled trial (INTEREST O2)

> LM Bullens ODJ Hulsenboom S Moors R Joshi PJ van Runnard Heimel MB van der Hout-van der Jagt ER van den Heuvel SG Oei

Trials. 2018 Mar 23;19(1):195

ABSTRACT

Background

Perinatal asphyxia is, even in developed countries, one the major causes of neonatal morbidity and mortality. Therefore, if fetal distress during labor is suspected, one should try to restore fetal oxygen levels, or aim for immediate delivery. However, studies on the effect of intrauterine resuscitation during labor are scarce. We designed a randomized controlled trial to investigate the effect of maternal hyperoxygenation on the fetal condition. In this study, maternal hyperoxygenation is induced for the treatment of fetal distress during the second stage of term labor.

Methods

This study is a single-center randomized controlled trial, performed in a tertiary hospital in The Netherlands. In case of a suboptimal or abnormal fetal heart rate pattern during the second stage of term labor, a total of 116 patients will be randomized to the control group, where normal care is provided, or to the intervention group, where before normal care 100% oxygen is supplied to the mother by a non-rebreathing mask until delivery. The primary outcome is change in fetal heart rate pattern. Secondary outcomes are Apgar score, mode of delivery, admission to the neonatal intensive care unit and maternal side effects. In addition, blood gas values and malondialdehyde are determined in umbilical cord blood.

Discussion

This study will be the first randomized controlled trial to investigate the effect of maternal hyperoxygenation for fetal distress during labor. This intervention should only be recommended as a treatment for intrapartum fetal distress, when improvement of the fetal condition is likely and outweighs maternal and neonatal side effects.

BACKGROUND

Labor contractions cause alterations in intrauterine pressure, and can thereby affect uterine and umbilical blood flow.¹⁻⁵ These fluctuations in blood flow towards the fetus can negatively influence oxygen flow and blood pressure.¹⁻⁵ Through chemo- and baroreceptor responses, these changes in fetal oxygenation and blood pressure affect fetal heart rate (FHR).^{1.2.6.7} Hence, non-reassuring FHR patterns, for example, FHR decelerations, may be a sign of fetal hypoxia.⁸⁻¹⁰ Prolonged fetal hypoxia may lead to an increased risk of fetal morbidity, including renal insufficiency, pulmonary hypertension, necrotizing enterocolitis and hypoxic-ischemic encephalopathy and fetal death.¹¹⁻¹² A prospective cohort study of term neonates in 2010 showed that 48% of admissions to Neonatal Intensive Care Units (NICUs) of these neonates were related to perinatal asphyxia (defined by the authors as a 5 minute Apgar score <7). The neonatal mortality rate was 8% in this study, the largest proportion of which (71%, N = 12/17) was related to asphyxia.¹³

Methods to directly measure fetal oxygenation during labor are unavailable, while methods for the continuous intrapartum monitoring of pH, saturation (SpO_2) , partial carbon dioxide pressure (pCO_2) , and partial oxygen pressure (pO_2) are not yet suitable for clinical practice.¹⁴⁻¹⁶ Therefore, the cardiotocogram (CTG), with occasional fetal scalp blood sampling (FSBS), is still the method of first choice to estimate fetal wellbeing during labor. The CTG has very good specificity but poor sensitivity for fetal wellbeing.¹⁷ In other words, if the FHR pattern is reassuring the fetus is very likely to be well-oxygenated. However, when FHR patterns are non-reassuring, the fetal condition is unclear and fetal distress cannot be ruled out.

Instead of aiming for immediate delivery in the presence of suspected fetal distress, one may try to improve fetal oxygenation to avoid an invasive intervention. Several intrauterine resuscitation techniques are used in clinical practice and described in the literature.^{18,19} However, robust evidence regarding their effect on neonatal outcome is limited.¹⁸ One of the interventions that still raises discussion is the administration of additional oxygen to the mother to treat fetal distress during labor.^{18,20-23}

Summary of findings from clinical studies

In the past decades, several studies have investigated the effect of maternal hyperoxygenation on maternal and fetal oxygenation. Indeed they found increasing maternal pO_2 and $fetal SpO_2$ and pO_2 levels,²⁴ but unfortunately these studies are mainly performed in the non-compromised fetus.²⁵⁻²⁷ Furthermore, only a few non-randomized studies of poor quality have been performed in the distressed fetus.²⁸⁻³² These studies suggest an improvement in FHR patterns and fetal scalp pH when 100% oxygen is applied to the mother. Based on these publications, a Cochrane review from 2012 concludes that "there is not enough evidence to support the use of prophylactic oxygen therapy

for women in labor, nor to evaluate its effectiveness for fetal distress", due to the lack of randomized controlled trials (RCTs).³³

An important concern in the use of maternal hyperoxygenation for fetal distress is the potential negative effect on umbilical cord pH. In a study by Thorp et al, 86 term parturients were randomized to receive additional oxygen or normal care, during the second stage of labor.³⁴ The main outcome measures were cord blood gas and co-oximetry values. The mean cord blood gas values did not significantly differ between the intervention and control group. However, they found significantly more arterial pH values <7.20 in the group receiving extra oxygen. The lowest arterial pH (pHa) value that was found was 7.09. They also found that the duration of oxygen therapy was inversely related to arterial cord pH, while Apgar scores and hospital admission rates did not differ between the groups. The authors concluded that prolonged oxygen treatment during the second stage of labor leads to a deterioration of cord blood gas values at birth. An important remark is the fact that in this study only patients with reassuring FHR patterns were included. Therefore, (ominous) fetal hypoxia at the start of oxygen delivery was very unlikely. Thus, this study did not address the effect of maternal hyperoxygenation in case of suspected fetal distress.

Another frequently stated argument to discourage maternal hyperoxygenation as standard care, is the potential increase in free oxygen radicals in both mother and fetus.^{35,36} An increase in the markers for free oxygen radical production has been seen for the use of high fractions of inspired oxygen and in the presence of non-reassuring FHR patterns.³⁵⁻³⁸ Also, lipid peroxide concentrations in arterial cord blood are higher after uncomplicated vaginal delivery compared with those after elective cesarean section.³⁹ To a certain degree, free oxygen radicals are physiological and known to be higher in the presence of several maternal and fetal conditions, such as preeclampsia, diabetes, smoking, intrauterine growth restriction and fetal distress.^{37,39-42} The effect of maternal hyperoxygenation on free oxygen radical release, in response to non-reassuring fetal status, has not yet been investigated. What we do know is that neonatal resuscitation with 100% oxygen may lead to an increase in neonatal mortality and morbidity, including bronchopulmonary disease and retinopathy, mainly in premature infants.⁴³⁻⁴⁶ However, the increase in fetal pO_{2} due to maternal hyperoxygenation will never reach the levels obtained by the direct application of 100% oxygen directly to the fetus.²³ To our knowledge, the clinical implication of increased free radical production due to maternal hyperoxygenation has not been investigated. Studies that use maternal hyperoxygenation as a treatment for the growth restricted fetus did not report any harmful effects.^{47,48}

With regard to the mother, some potential side effects have to be taken into account. The use of high fractions of inspired oxygen in the absence of tissue hypoxia may cause toxic effects as a result of oxidative stress.^{49,50} This may for example lead to mucosal inflammation, hypoperfusion and pneumonitis.⁵¹ A reversible vasoconstriction of

approximately 10% in the maternal brain has been described.⁵² However, this is not expected to cause any harm.^{53,54} Administration of 100% oxygen during labor is not investigated. However, it is well investigated for the treatment of cluster headaches, and no severe side effects (e.g. hypoventilation and fainting) were reported.⁵⁴

Inhaling high fractions of inspired oxygen will increase the concentration of free oxygen radicals in maternal blood.³⁵ Despite the adverse effects of free oxygen radicals that have been described,⁵⁵ it is unlikely these will cause clinically relevant tissue damage due to the mature anti-oxidant system in the adult.^{35,36} Also, the Dutch pharmacovigilance center Lareb has not been informed of any side effects due to oxygen therapy.⁵⁶

Current recommendations on the use of maternal hyperoxygenation

Based on current knowledge, it is difficult to determine whether the beneficial effects outweigh the potential side effects. As a consequence, recommendations in international guidelines and use in clinical practice are non-uniform.²⁰ Maternal hyperoxygenation during labor is often used in the United States of America to increase oxygen transport towards the fetus.²¹ The American College of Obstetricians and Gynecologists' (ACOG) guideline on fetal resuscitation recommends the administration of oxygen to the mother in case of fetal distress.⁵⁷ In contrast, the Royal College of Obstetricians and Gynaecologists explicitly states in their Green Top Guideline to not apply maternal oxygenation for reasons other than maternal hypoxia, until the beneficial effect is proven.⁵⁸ A recent discussion on benefit and harm of maternal hyperoxygenation in the American Journal of Obstetrics and Gynaecology (AJOG) emphasises the current lack of evidence.^{21,23} In fact, several reviews underline an urgent need for an RCT, investigating the effect of maternal hyperoxygenation on the fetal condition.^{21-23,33}

METHODS

Aim

The aim of this study is to investigate the effect of maternal hyperoxygenation with 100% oxygen on the fetal condition during the second stage of labor, in the presence of suspected fetal distress during term labor. Also, we investigate the potential side effects, to formulate recommendations for international clinical practice and future research.

Study design

This study will be a single-center RCT, performed in a tertiary hospital in The Netherlands, comparing maternal hyperoxygenation for the treatment of fetal distress during the second stage of labor with conventional care. All procedures and timeframes are displayed in Figure 1 (according to the Standard Protocol Items: Recommendations for Interventional Trials (SPIRIT)).⁵⁹ Additional file 2 contains the complete SPIRIT checklist.

Activity/	CRF ⁵ (Y/N)	Staff member	Time to	Pre-study	Pre-study	Study	Study	Post-study
assessment			complete	(screening/	(randomization)	(during	(post	
			(minutes)	consent)		labor)	partum)	
Assessment of eligibility	No	Doctor/midwife/	1	\times				
		(co)investigator						
Supply oral and written	No	Doctor/midwife/	5	×				
study information		(co)investigator						
Obtain and store informed	No	Doctor/midwife/	S	×				
consent		(co)investigator						
Classification of CTG* and	Yes	Doctor/midwife/	1		×			
randomization		(co)investigator						
Intervention group: apply O_2	Yes	Obstetric nurse	Ţ			×		
by non-rebreathing mask								
Fill in and store CRF ^{\$}	Yes	Doctor/midwife/	5					×
		(co)investigator						
Draw umbilical cord blood	No	Doctor/midwife	3			×		
and sent to lab								
Store blood for future	No	Laboratory staff	5				\times	
examination								
Register data on	Yes	(Co)investigator	10					×
demographics, delivery								
and neonatal outcome in								
Research Manager								
Analyse FHR [#] registrations	No	(Co)investigator,	15					×
		experts						
Analyse cord blood for $MDA^{\&}$	No	Laboratory staff	د.					×
Interim analysis	No	(Co)investigator	60					After including 58
								participants
Report SAE/SUSARs	Yes	Principal	~·					As needed
		investigator						throughout protocol
Figure 1. The schedule of forms	and procedu	res, according to the	e Standard P	rotocol Items:	Recommendations	s for Interv	/entional T	rials (SPIRIT).

case report form, $CTG = cardiotocogram, #FHR = fetal heart rate, <math>^{\&}MDA = malondial dehyde$ ^{\$}CRF :

Participants

The study population will be drawn from parturients, admitted to the labor ward of a tertiary hospital (Máxima Medical Center, Veldhoven, The Netherlands), where approximately 2200 deliveries occur annually, of which approximately 1900 term births. CTG and, if necessary, FSBS are generally used for fetal monitoring during labor. Maternal repositioning, discontinuation of administration of oxytocin, use of tocolytic drugs and intermittent pushing are common interventions to achieve intrauterine resuscitation. while amnioinfusion and maternal hyperoxygenation are never applied as standard care in our center.

Inclusion criteria

Pregnant women ≥ 18 years, in term labor, and with an intended vaginal delivery of a singleton in cephalic presentation can participate in this study.

Exclusion criteria

Exclusion criteria are determined with focus on the risk of excessive production of free oxygen radicals, and reducing the influence of other factors affecting FHR pattern. These are a recent use of any of the following medication: corticosteroids, antihypertensives, magnesium sulphate, amiodaron, opioids, adriamycin, bleomycin, actinomycin, menadione, promazine, thioridazine or chloroquine, or the use of tobacco, recreational drugs or alcohol during pregnancy. Parturients suffering from pre-existing cardiac disease, pulmonary disease with the use of medication, diabetes, hyperthyroidism or anemia (hemoglobin < 6.5 mmol/l or 10.5 gr/dL) will also be excluded. Fetal factors leading to exclusion are: suspected infection during labor (need for antibiotics), congenital malformations and normal or preterminal FHR pattern, or prolonged bradycardia (according to the modified FIGO classification, Figure 2).^{60,61}

	Baseline heart frequency	Variability Reactivity	Decelerations
Normal CTG	• 110-150 bpm	Accelerations 5-25 bpm	 Early uniform decelerations Uncomplicated variable decelerations (loss of <60 beats)
Intermediary CTG	 100-110 bpm 150-170 bpm Short bradycardia episode < 100 bpm for >3 min < 80 bpm for >2 min 	 >25 bpm (sattatory pattern) <5 bpm >40 min 	 Uncomplicated variable decelerations (loss of >60 beats)
	 A combination of 2 or sev 	eral intermediary observation	ns will result in an abnormal CTG
Abnormal CTG	 >170 bpm Persistent bradycardia <100 bpm for > 10 min <80 bpm for > 3 min (without an increasing tendency) 	 <5 bpm for >60 min Sinusoidal pattern 	Complicated variable decelerations with a duration of >6 0sec Repeated late uniform decelerations
Preterminal CTG	 Total lack of variability (< bradycardia 	2 bpm) and reactivity with or	without decelerations or

Figure 2. The modified FIGO classification.

Patient recruitment and randomization

All patients eligible to be included in this study will antepartum be asked to participate when they visit the outpatient's clinic, or when they are admitted to the delivery ward. All patients will receive oral and written information about the study from the attending midwife, doctor or a co-investigator. After informed consent, and only in case of suboptimal or abnormal FHR patterns during the second stage of labor, randomization is performed using sealed, opaque envelops. The allocation sequence is computergenerated using random blocks of four or six patients.

Intervention and control group

Patients will randomly be assigned to one of the two arms of the study:

- Control group: normal care (according to the local standard) is provided, and preferably started at least 10 minutes after the onset a suboptimal or abnormal FHR pattern, according to the modified FIGO criteria (Figure 2).^{60,61}
- Intervention group: in case of a suboptimal or abnormal FHR pattern according to the modified FIGO criteria, 100% oxygen is applied to the mother at 10l/min via a non-rebreathing mask, and continued until delivery. Co-interventions (normal care) may be initiated after 10 minutes of oxygen administration without a satisfactory effect on FHR, to investigate the effect of only maternal hyperoxygenation on FHR, without risking prolonged fetal hypoxia. In case a patient needs to undergo a cesarean section, oxygen administration will be continued until the fetus is born.

Obviously, in case the delivery room staff believes additional interventions should be applied for safety reasons, the study protocol can be overruled any time.

Study outcomes and data analysis

The primary outcome is the percentage reduction in the depth and duration of FHR deceleration in the intervention group in comparison with the control group. Secondary outcomes include fetal, neonatal and maternal outcomes.

Fetal outcome

FHR changes

Changes in specific features of the CTG including:

- Decelerations with loss of internal variability (beat to beat variability of <5 beats per minute (BPM))
- Decelerations in combination with tachycardia of bradycardia (> 160 or < 110 BPM)
- Unassignable baseline
- Phase-rectified signal averaging (PRSA); a relatively new technique to determine fetal heart rate variability, by estimating the accelerative (AC_{prsa}) and decelerative capacity (DC_{prsa}) of the fetal heart. This technique is explained in the articles by Bauer and Huhn.^{62,63}
- Change in modified FIGO classification (Figure 2).^{60,61}

In the next paragraph methodology regarding the comparison of FHR tracings and timeframes are described more detailed.

Neonatal outcome

This includes Apgar score, NICU admission, venous and arterial umbilical cord blood gas analysis (pH, lactate, base excess, pO₂ and pCO₂) and malondialdehyde, (MDA, a marker for free oxygen radical production) in arterial and venous umbilical cord blood. Information on neonatal admission is a standard part of the maternal hospital chart. Determination of 1- and 5-minute Apgar score and venous and arterial umbilical cord blood gas analysis (pH, lactate, base excess, pO_2 and pCO_2) are common practice. A cord blood gas analysis will be performed immediately after birth, by the ABL 90 flex blood gas analyzer (Radiometer Benelux BV, Zoetermeer, The Netherlands), in both venous and arterial cord blood. Two additional blood samples (one venous and one arterial sample) are drawn from the umbilical cord in heparin tubes, and immediately centrifuged and stored at the laboratory of Máxima Medical Center at -20°C. Once all samples are collected, they will be transported to the laboratory of Genetic and Metabolic Diseases of the Academic Medical Center Amsterdam. The Netherlands, where total (free and bound) MDA will be determined as the 2.4-dinitrophenylhydrazine (DNPH) derivative. A stable isotopically labelled analogue (²H₂-MDA) will be added as internal standard, where after alkaline hydrolysation, deproteinisation and derivatisation with DNPH, and MDA-hydrazone will be analyzed by HPLC-MS/MS and positive electrospray. Samples will be injected on an LC-18-DB analytical column (250 × 4.6 mm, 5 µm particles, Supelco) hyphenated to a Quattro Premier XE mass spectrometer (Waters Corporation, Milford, MA), using an Acquity UPLC system (Waters Corporation, Milford, MA). Analytes and internal standard will be eluted with acetonitrile/water/acetic acid (50/50/0.2) and detected in multiple reaction monitoring mode for the transitions m/z 235 \rightarrow m/z 159; $m/z 237 \rightarrow m/z 161$.

Maternal outcome

Maternal outcome measures include the mode of delivery, side effects and reasons for discontinuation of oxygen administration. Side effects include a headache, dizziness, discomfort of the non-rebreathing mask and any other complaint mentioned by the participant. The delivery room staff will register on the case report form (CRF) if the parturient experiences any side effects and/or if there are reasons for eventual discontinuation of oxygen administration. Also, a short questionnaire will be used to investigate experiences of all the participants with this study, to gain insight in how laboring women experience receiving additional oxygen by a non-rebreathing mask, compared to receiving normal care.

Analysis of outcome measures regarding FHR pattern

Changes in FHR pattern

The digital CTG tracings will be extracted from Chipsoft EZIS (Amsterdam, The Netherlands) and analyzed using Matlab 2015a (MathWorks Inc USA). For the computerized CTG analysis we will use a custom-made algorithm, based on the OxSys system,⁶⁴ that will first be validated by an expert panel. This expert panel will also manually classify the CTG to one of the FIGO categories.^{60,61}Regarding the analysis of specific CTG features, we searched the literature for CTG features that are likely related to neonatal outcome. A large variety of CTG features have been investigated in relation to neonatal outcome, with varying results. However, three features are consistently mentioned as related to neonatal outcome:

- decelerations with loss of internal variability
- decelerations in combination with tachycardia or bradycardia
- periods with unassignable baseline 3,60,64-71

Besides, AC_{PRSA} and DC_{PRSA} turned out to predict acidemia better than short-term variation.^{62,72,73} We therefore include this parameter in as an outcome measure.

What is the timeframe of interest?

All patients serve as their own control with changes in FHR being compared before and after the start of the study protocol, irrespective of whether the patients belonged to the control or the intervention group. Additionally, results of the intervention group and control group will also be compared.

For the analysis where patients serve as their own control, the timeframes of interest for outcomes related to changes in FHR are as follows:

- Control group: 10 minutes before and after the start of the study protocol. In total 20 minutes of data will be analyzed (Figure 3).
- Intervention group: 10 minutes before the start of the study protocol up to 15 minutes after start of the study protocol. The timeframe of interest after the start of the study protocol is determined as the period between 5 and 15 minutes after maternal hyperoxygenation is initiated, motivated by the expectation that it will take 5 minutes for maternal pO₂ to increase to a maximum of approximately 475 mmHg.²⁴ After that, the effect of the intervention will be observed for 10 minutes. In total 20 minutes of data will be analyzed (Figure 4).



Figure 3. The timeframe of interest for analysis of outcome measures where patients serve as their own control: the control group.



Figure 4. The timeframe of interest for analysis of outcome measures where patients serve as their own control: the intervention group.

These periods are established because during this period, maternal hyperoxygenation can be compared to no treatment. Furthermore, we will also compare the periods from the start of the study until birth, although these results may be influenced by other interventions that may have been applied.
Maternal hyperoxygenation: study protocol of an RCT

Other study endpoints and parameters

Duration of the second stage of labor, duration of time for which supplemental oxygen was received, baseline characteristics (infant sex, gestational age and birth weight, maternal age and parity) are recorded.

Hypothesis

We hypothesize that maternal hyperoxygenation will improve FHR, without any severe maternal side effects. We do not expect a difference in rates of vacuum-assisted delivery or secondary cesarean sections, nor Apgar scores or umbilical cord pH values, due to the relatively small sample size. Furthermore, we expect larger concentrations of MDA in the intervention group than in the control group.

Handling and storage of data and documents

Data will be handled anonymously and we will adhere to the Dutch Personal Data Protection Act (in Dutch: De Wet Bescherming Persoonsgegevens, WBP). A secured subject identification code list will be used to link a study number to a patients name and date of birth. This file is password protected available only to the main investigator (LB). All other information will contain only the study number and no data directly referring to the patient. Fetal blood gas values will be stored in the neonates' hospital chart since this is part of conventional care. Laboratory results regarding markers for free oxygen radicals will be coded and will therefore be anonymous. All data will be stored for 15 years, in accordance with the Good Clinical Practice guidelines.

Statistical analysis

Sample size calculation

The study consists of two study groups: one group with suboptimal FHR patterns, and one group with abnormal FHR patterns. We aim for 90% power and a level of significance of 0.05 in both groups. In one small, non-randomized study, a reduction in FHR decelerations (type II dips) of 50 to 100% was noted.²⁸ This is the only study that reports on FHR changes as a result of maternal hyperoxygenation. Based on the available literature, we expect at least 50% improvement in the oxygen-group and 0% in the control-group in both suboptimal and abnormal FHR patterns.²⁸ We estimated a mean improvement of 50% with a standard deviation of 50% in each group. A power analysis performed in G*Power 3.0.10 (Kiel University, Germany) for a two-tailed Mann Whitney test (assuming that data will not be equally distributed) resulted in a sample size of 58 patients in each study group, given an anticipated 20% of missing data. Since we have two separate study groups (suboptimal and abnormal FHR group) we need 116 patients to participate.

Data analysis

SPSS (version 24, IBM, Armonk, NY) will be used to perform statistical analysis of the study results. Assuming non-normal distribution, the primary clinical outcome will

be analyzed with a Mann Whitney U test for differences between the intervention and control group, and a Wilcoxon Matched-Pairs test for changes within the same participant. When outcome data is found to be normally distributed, independent samples t-tests (two-tailed) will be used to analyze differences between the intervention and control group, and paired t-tests for changes within the same participant. Outcome measures will be calculated for the combined group and the subgroups of suboptimal and abnormal FHR tracings, and for small for gestational age (SGA, growth percentile <p10) and appropriate for gestational age (AGA) neonates. In the intervention group, oxygen may not be applied due to practical concerns such as very quick progression of labor. Therefore, we will perform both per-protocol and intention-to-treat analysis. In the perprotocol analysis parturients that actually received oxygen will be compared to those who did not receive oxygen. Besides, unjust inclusions will be excluded from this analysis.

Interim analysis

On account of safety concerns, an interim analysis will be performed when 50% of the patients are included in the study. In this analysis, we compare the number of neonates with a 5- minute Apgar score < 7 and/or pHa< 7.05, the number of admission to NICU and perinatal death in both groups (all neonates that received oxygen in both suboptimal and abnormal CTG group versus 'conventional care' group). In case the interim analysis shows a significant difference, we will terminate the study. This interim analysis is performed exclusively for safety reasons: since the primary outcome measure (fetal heart rate) will not be analyzed during the interim analysis, and power analysis is based on the primary outcome, adjustment of the significance level is not required.

Public disclosure and publication policy

All investigators agree to publish the study results in an international peer-reviewed journal, even if the results do not correspond to the hypothesis as stated in the methods section of the protocol. The results will be offered for publication after all the investigators agree on the content of the article. The full protocol (version 8, date 1st March 2017) is available upon request.

DISCUSSION

This study is the first RCT to investigate the effect of maternal hyperoxygenation for fetal distress during labor.^{18,33} So far, the effects of supplemental oxygenation in the presence of FHR abnormalities have only been investigated in small, non-randomized studies. Due to the lack of concrete results from clinical trials, it is hard to compare the beneficial effects of maternal hyperoxygenation to the potential side effects. As a result, recommendations on the use this intervention for fetal distress in international guidelines are non-uniform.²⁰ Thus, the results of this study will help in filling an internationally recognized 'research gap'.

We believe patient safety is carefully addressed in this study, and ethical concerns are limited. One of the major concerns of administering high fractions of oxygen, is the increase in free oxygen radicals. Whether this has a clinical effect remains unclear. We excluded all patients with a higher a priori risk of exposure to increased free oxygen radical levels from this study.

Both practical and safety issues led to limitations of this study. An important limitation is the primary outcome measure. We recognize that changes in FHR as a primary outcome measure is not optimal since FHR does not accurately reflect fetal oxygenation and acidbase balance.^{60,75,76} However, we believe this is the 'best available' method to record changes in the fetal condition during labor. Furthermore, we assume that if no beneficial effect on FHR can be shown, an improvement in neonatal outcome is unlikely. Ideally, neonatal outcome measures such as Apgar score and umbilical cord pH are the outcome measures of first choice. However, a study with appropriate power to address these outcome measures would need a very large sample size. Since the potentially harmful effects have not been properly investigated yet, we chose to not expose a large group of women and their fetuses to this intervention. If a positive effect on FHR pattern without severe side effects can be confirmed by this study, we will perform a larger multicenter RCT to investigate the effect on Apgar score and cord blood gas values.

In this study, we focus on the fetal condition during the second stage of labor and shortterm neonatal outcome. This implies that abnormalities in FHR patterns during the first stage of labor are not taken into account. We believe that the randomization process will limit its influence. With regard to the neonatal period, we did not arrange long-term follow-up, as we do not expect any clinically relevant side effects that can be attributed to maternal hyperoxygenation. Besides, the sample size is too small to draw firm conclusions on long-term neonatal effects in this study.

Power analysis of the current study is based on the expected effect on the primary outcome measure and is not powered to find any significant differences in Apgar score and umbilical cord blood gas values. In the power analysis, we used an expected improvement in deceleration depth and duration of 50%. This value is based on small, non-randomized studies, and may be overestimated. On the other hand, this is the only available data. Also, we believe it is unlikely that a limited improvement in deceleration depth and duration has clinical relevance. The sample size is calculated for each of subgroups of suboptimal and abnormal FHR tracings. We believe it is important to assess the effect of the intervention in these subgroups as fetuses having lower initial pO₂ levels may profit more from maternal hyperoxygenation.²⁹

Regarding the subgroups of AGA and SGA infants, we did not increase our sample size to reach an adequate number of participants in the SGA group. Nevertheless, we find

it interesting to see whether there is a different effect of maternal hyperoxygenation in SGA compared to AGA infants.

Due to organizational challenges, it is not possible to conduct a double-blinded trial. Hence patients and delivery room staff are not blinded to the patients' allocation to a study group and may lead to observer bias. However, analysis of FHR tracings will be done using a computerized algorithm and the investigators judging the CTGs and secondary outcome measures are blinded to the study arm, to minimize bias.

To investigate the effect of maternal hyperoxygenation in the presence of fetal distress on the release of free oxygen radicals, MDA is estimated in umbilical cord blood. MDA is the peroxidation product of membrane polyunsaturated fatty acids. We chose to measure this marker for oxidative stress because it is used in former studies performed during labor and it is related to vaginal birth. non-reassuring FHR tracings. maternal hyperoxygenation and acidemia in arterial cord blood.^{36,37,39,41} We realize that differences in values in umbilical cord blood may be confounded by mode and duration of delivery; therefore, we will correct the results for the mode of delivery. A practical ground to choose this marker is that this is the only marker for oxidative stress that can be accurately estimated in Dutch laboratories. In the intervention group, oxygen administration will be continued until delivery to enable analysis of its effect on cord blood gas values and MDA.

Despite some important limitations of this study, we believe this is the best possible way to perform a study while restricting safety issues. If the results do not show any improvement in FHR, we believe maternal hyperoxygenation should not be used as a treatment for fetal distress. However, if a beneficial effect is demonstrated, we will design a multicenter RCT to investigate the effect on neonatal outcome.

ACKNOWLEDGEMENTS

This research was performed within the framework of the IMPULS perinatology, in collaboration with Philips Healthcare, Eindhoven, The Netherlands.

REFERENCES

- 1. Caldeyro-Barcia R, Mendez-Bauer C, Poseiro J, Escarena L, Pose S, Bieniarz A. Control of the human fetal heart rate during labor. In: Cassels DE, editor. The heart and circulation in the newborn and infant. New York: Grune & Stratton; 1966. p. 7-36.
- 2. Murray ML. Antepartal and intrapartal fetal monitoring. 3rd ed. New York: Springer Publishing Company; 2007.
- 3. Westgate JA, Wibbens B, Bennet L, Wassink G, Parer JT, Gunn AJ. The intrapartum deceleration in center stage: a physiologic approach to the interpretation of fetal heart rate changes in labor. Am J Obstet Gynecol. 2007;197:236 e1-11.
- 4. Ball RH, Parer JT. The physiologic mechanisms of variable decelerations. Am J Obstet Gynecol. 1992;166:1683-8.
- 5. Bennet L, Gunn AJ. The fetal heart rate response to hypoxia: insights from animal models. Clin Perinatol. 2009;36:655-72.
- 6. Freeman RK, Garite TJ, Nageotte MP. Fetal heart rate monitoring. 3rd ed. Philadelphia: Lippincott Williams & Wilkins; 2000.
- 7. Hanson MA. Do we now understand the control of the fetal circulation? Eur J Obstet Gynecol Reprod Biol. 1997;75:55-61.
- 8. Mendez-Bauer C, Arnt IC, Gulin L, Escarcena L, Caldeyro-Barcia R. Relationship between blood pH and heart rate in the human fetus during labor. Am J Obstet Gynecol. 1967;97:530-45.
- 9. Elliott C, Warrick PA, Graham E, Hamilton EF. Graded classification of fetal heart rate tracings: association with neonatal metabolic acidosis and neurologic morbidity. Am J Obstet Gynecol. 2010;202:258 e1-8.
- 10. Kubli FW, Hon EH, Khazin AF, Takemura H. Observations on heart rat and pH in the human fetus during labor. Am J Obstet Gynecol. 1969;104:1190-206.
- 11. Graham A, Ruis KA, Hartman A, Northington F, Fox H. A systematic review of the role of intrapartum hypoxia-ischemia in the causation of neonatal encephalopathy. Am J Obstet Gynecol. 2008;199:587-95.
- 12. Martin-Ancel A, Garcia-Alix A, Gaya F, Cabanas F, Burgueros M, Quero J. Multiple organ involvement in perinatal asphyxia. J Pediatr. 1995;127:786-793.
- 13. Evers AC, Brouwers HA, Hukkelhoven CW, Nikkels PG, Boon J, van Egmond-Linden A, et al. Perinatal mortality and severe morbidity in low and high risk term pregnancies in the Netherlands: prospective cohort study. BMJ. 2010;341:c5639.
- 14. McNamara HM, Dildy GA. Continuous intrapartum pH, pO2, pCO2, and SpO2 monitoring. Obstet Gynecol Clin North Am. 1999;4:671-93.
- 15. East CE, Begg L, Colditz PB, Lau R. Fetal pulse oximetry for fetal assessment in labour. Cochrane Database Syst Rev. 2014;10:CD0004075.
- 16. Dildy GA, van den Berg PP, Katz M, Clark SL, Jongsma HW, Nijhuis JG, et al. Intrapartum fetal pulse oximetry: fetal oxygen saturation trends during labor and relation to delivery outcome. Am J Obstet Gynecol. 1994;171:679-84.
- 17. Schiermeier S, Pidner von Steinburg S, Thieme A, Reinhard J, Daumer M, Scholz M, Hatzmann W, Schneider KT. Sensitivity and specificity of intrapartum computerised FIGO criteria for cardiotocography and fetal scalp pH during labour: multicentre, observational study. BJOG. 2008;115:1557-63.

- 18. Bullens LM, van Runnard Heimel PJ, van der Hout-van der Jagt MB, Oei SG. Interventions for Intrauterine Resuscitation in Suspected Fetal Distress During Term Labor: A Systematic Review. Obstet Gynecol Surv. 2015;70:524-39.
- 19. Simpson KR. Intrauterine resuscitation during labor: review of current methods and supportive evidence. J Midwifery Womens Health. 2007;52:229-37.
- 20. Bullens LM, Moors S, van Runnard Heimel PJ, van der Hout-van der Jagt MB, Oei SG. Practice variation in the management of intrapartum fetal distress in The Netherlands and the Western world. Eur J Obstet Gynecol Reprod Biol. 2016;205:48-53.
- 21. Hamel MS, Anderson BL, Rouse DJ. Oxygen for intrauterine resuscitation: of unproved benefit and potentially harmful. Am J Obstet Gynecol. 2014;211:124-7.
- 22. Hamel MS, Hughes BL, Rouse DJ. Whither oxygen for intrauterine resuscitation? Am J Obstet Gynecol. 2015;212:461-2.
- 23. Garite TJ, Nageotte MP, Parer JT. Should we really avoid giving oxygen to mothers with concering fetal heart rate patterns? Am J Obstet Gynecol. 2015;212:459-60.
- 24. Vasicka A, Quilligan EJ, Aznar R, Lipsitz PJ, Bloor BM. Oxygen tension in maternal and fetal blood, amniotic fluid, and cerebrospinal fluid of the mother and the baby. Am J Obstet Gynecol. 1960;79:1041-7.
- 25. Aldrich CJ, Wyatt JS, Spencer JA, Reynolds EO, Delpy DT. The effect of maternal oxygen administration on human fetal cerebral oxygenation measured during labour by near infrared spectroscopy. Br J Obstet Gynaecol. 1994;101:509-13.
- 26. McNamara H, Johnson N, Lilford R. The effect on fetal arteriolar oxygen saturation resulting from giving oxygen to the mother measured by pulse oximetry. Br J Obstet Gynaecol. 1993;100:446-9.
- 27. Khazin AF, Hon EH, Hehre FW. Effects of maternal hyperoxia on the fetus. I. Oxygen tension. Am J Obstet Gynecol. 1971;109:628-37.
- 28. Althabe O, Schwarcz R, Pose S, Escarcena L, Caldeyro-Barcia R. Effects on fetal heart rate and fetal pO2 of oxygen administration to the mother. Am J Obstet Gynecol. 1967;98:858-70.
- 29. Haydon ML, Gorenberg DM, Nageotte MP, Ghamsary M, Rumney PJ, Patillo C, et al. The effect of maternal oxygen administration on fetal pulse oximetry during labor in fetuses with nonreassuring fetal heart rate patterns. Am J Obstet Gynecol. 2006;195:735-8.
- 30. Gare DJ, Shime J, Paul WM, Hoskins M. Oxygen administration during labor. Am J Obstet Gynecol. 1969;105:954-61.
- 31. Hidaka A, Komatani M, Ikeda H, Kitanaka T, Okada K, Sugawa T. A comparative study of intrauterine fetal resuscitation by beta-stimulant and O2 inhalation. Asia Oceania J Obstet Gynaecol. 1987;13:195-200.
- 32. Dildy GA, Clark SL, Loucks CA. Intrapartum fetal pulse oximetry: the effects of maternal hyperoxia on fetal arterial oxygen saturation. Am J Obstet Gynecol. 1994;171:1120-4.
- 33. Fawole B, Hofmeyr GJ. Maternal oxygen administration for fetal distress. Cochrane Database Syst Rev. 2012;12:CD000136.
- 34. Thorp JA, Trobough T, Evans R, Hedrick J, Yeast JD. The effect of maternal oxygen administration during the second stage of labor on umbilical cord blood gas values: a randomized controlled prospective trial. Am J Obstet Gynecol. 1995;172:465-74.
- 35. Nesterenko TH, Acun C, Mohamed MA, Mohamed AN, Karcher D, Larsen J Jr, et al. Is it a safe practice to administer oxygen during uncomplicated delivery: a randomized controlled trial? Early Hum Dev. 2012;88:677-81.

- 36. Khaw KS, Wang CC, Ngan Kee WD, Pang CP, Rogers MS. Effects of high inspired oxygen fraction during elective caesarean section under spinal anaesthesia on maternal and fetal oxygenation and lipid peroxidation. Br J Anaesth. 2002;88:18-23.
- 37. Dede FS, Guney Y, Dede H, Koca C, Dilbaz B, Bilgihan A. Lipid peroxidation and antioxidant activity in patients in labor with nonreassuring fetal status. Eur J Obstet Gynecol Reprod Biol. 2006;124:27-31.
- 38. Yalcin S, Aydogan H, Kucuk A, Yuce HH, Altay N, Karahan MA, et al. Supplemental oxygen in elective cesarean section under spinal anesthesia: Handle the sword with care. Braz J Anesthesiol. 2013;63:393-7.
- 39. Rogers MS, Mongelli JM, Tsang KH, Wang CC, Law KP. Lipid peroxidation in cord blood after birth: the effect of labor. Br J Obstet Gynaecol. 1998;105:739-44.
- 40. Blackburn S. Free radicals in perinatal and neonatal care, part 2: oxidative stress during the perinatal and neonatal period. J Perinal Neonatal Nurs. 2006;20:125-7.
- 41. Wang W, Pang CC, Rogers MS, Chang AM. Lipid peroxidation in cord blood at birth. Am J Obstet Gynecol. 1996;174:62-5.
- 42. Nordström L, Arulkumaran S. Intrapartum fetal hypoxia and biochemical markers: a review. Obstet Gynecol Surv. 1998;53:645-57.
- 43. Saugstad OD, Ramji S, Soll RF, Vento M. Resuscitation of newborn infants with 21% or 100% oxygen: an updated systematic review and meta-analysis. Neonatology. 2008;94:176-82.
- 44. Carlo WA, Finer NN, Walsh MC, Rich W, Gantz MG, Laptook AR, et al: SUPPORT Study Group of the Eunice Kennedy Shriver NICHD Neonatal Research Network. Target ranges of oxygen saturation in extremely preterm infants. N Engl J Med. 2010;362:1959-69.
- 45. Davis PG, Tan A, O'Donnell CP, Schulze A. Resuscitation of newborn infants with 100% oxygen or air: a systematic review and meta-analysis. Lancet. 2004;364:1329-33.
- 46. Rabi Y, Rabi D, Yee W. Room air resuscitation of the depressed newborn: a systematic review and meta-analysis. Resuscitation. 2007;72:353-63.
- 47. Brantberg A, Sonesson SE. Central arterial hemodynamics in small-for-gestational-age fetuses before and during maternal hyperoxygenation: a Doppler velocimetric study with particular attention to the aortic isthmus. Ultrasound Obstet Gynecol.1999;14:237-43.
- 48. Bartnicki J, Saling E. Influence of maternal oxygen administration on the computeranalysed fetal heart rate patterns in small-for-gestational-age fetuses. Gynecol Obstet Invest. 1994;37:172-5.
- 49. Duling BR. Microvascular responses to alterations in oxygen tension. Circ Res. 1972;31:481-9.
- 50. Cornet AD, Kooter AJ, Peters MJ, Smulders YM. The potential harm of oxygen therapy in medical emergenciesThe potential harm of oxygen therapy in medical emergencies. Crit Care. 2013;17:313.
- 51. Sjöberg F, Singer M. The medical use of oxygen: a time for critical reappraisal. J Intern Med. 2013;274:505-28.
- 52. Fitch W. Cerebral blood flow: physiological principles and methods of measurement. In: Sebel PS, Fitch W, editors. Monitoring the Central Nervous System. Oxford: Blackwell Science; 1994. p. 78-117.
- 53. Watson NA, Beards SC, Altaf N, Kassner A, Jackson A. The effect of hyperoxia on cerebral blood flow: a study in healthy volunteers using magnetic resonance phase-contrast angiography. Eur J Anaesthesiol. 2000;17:152-9.

- 54. Bennett MH, French C, Schnabel A, Wasiak J, Kranke P. Normobaric and hyperbaric oxygen therapy for migraine and cluster headache. Cochrane Database Syst Rev. 2008;3:CD005219.
- 55. Kehrer JP, Klotz LO. Free radicals and related reactive species as mediators of tissue injury and disease: implications for health. Crit Rev Toxicol. 2015;45:765-98.
- 56. Lareb. 's Hertogenbosch, The Netherlands. 2017. https://www.lareb.nl/nl/ databank/Result?formGroup=&atc=V03AN01&drug=ZUURSTOF+MEDICINAAL+ %28ZUURSTOF%29. Accessed March 21 2017. [Dutch]
- 57. American Congress of Obstetricians and Gynecologistst. Practice bulletin no. 116: management of intrapartum fetal heart rate tracings. Obstet Gynecol. 2010;116:1232-40.
- 58. Royal College of Obstetricians and Gynaecologistst. Intrapartum care, NICE guideline 190. December 2014, updated February 2017. https://www.nice.org.uk/guidance/ cg190/chapter/Recommendations. Accessed May 2017.
- 59. Chan AW, Tetzlaff JM, Gøtzsche PC, Altman DG, Mann H, Berlin JA, et al. SPIRIT 2013 explanation and elaboration: guidance for protocols of clinical trials._BMJ. 2013;346:e7586.
- 60. Ayres-de-Campos D, Spong CY, Chandraharan E; for the FIGO Intrapartum Fetal Monitoring Expert Consensus Panel. FIGO consensus guidelines on intrapartum fetal monitoring: Cardiotocography. Int J Gynecol Obstet. 2015;131:13-24.
- Neoventa. Mölndal, Sweden. http://www.neoventa.com/2015/11/bigger-is-not-alwaysbetter/. Accessed November 6th 2017.
- 62. Bauer A, Kantelhardt JW, Barthel P, Schneider R, Mäkikallio T, Ulm K, Hnatkova K, et al. Deceleration capacity of heart rate as a predictor of mortality after myocardial infarction: cohort study. Lancet. 2006;367:1674-81.
- 63. Huhn EA, Lobmaier S, Fischer T, Schneider R, Bauer A, Schneider KT, Schmidt G. New computerized fetal heart rate analysis for suveillance of intrauterine growth restriction. Prenat Diagn. 2011;31:509-14.
- 64. Georgieva A, Payne SJ, Moulden M, Redman CWG. Computerized fetal heart rate analysis in labor: detection of intervals with un-assignable baseline. Physiol Meas. 2011;32:1549-60.
- 65. Ozden S, Demirci F. Significance for fetal outcome of poor prognostic features in fetal heart rate traces with variable decelerations. Arch Gynecol Obstet. 1999;262:141-9.
- 66. Gaziano EP. A study of variable decelerations in association with other heart rate patterns during monitored labor. Am J Obstet Gynecol. 1979;135:360-3.
- 67. Krebs HB, Petres RE, Dunn LJ. Intrapartum fetal heart rate monitoring. VIII. Atypical variable decelerations. Am J Obstet Gynecol. 1983;145:297-305.
- 68. Hamilton E, Warrick P, O'Keeffe D. Variable decelerations: do size and shape matter? J Matern Fetal Neonatal Med. 2012;25:648-53.
- 69. Kazandi M, Sendag F, Akercan F, Terek MC, Gundem G. Different types of variable decelerations and their effects to neonatal outcome. Singapore Med J. 2003;44:243-7.
- 70. Holzmann M, Wretler S, Cnattingius S, Nordström L. Cardiotocography patterns and risk of intrapartum fetal acidemia. J Perinal Med. 2015;43:473-9.
- 71. Georgieva A, Payne SJ, Moulden M, Redman CW. Relation of fetal heart rate signals with unassignable baseline to poor neonatal state at birth. Med Biol Eng Comput. 2012;50:717-25.

Chapter 6

- 72. Georgieva A, Papageroghiou AT, Payne SJ, Moulden M, Redman CW. Phase-rectified signal averaging for intrapartum electronic fetal heart rate monitoring is related to acidaemia at birth. BJOG. 2014;121:889-94.
- 73. Lobmaier SM, Mensing van Charante N, Ferrazzi E, Giussani DA, Shaw CJ, Müller A, et al.; TRUFFLE investigators. Phase-rectified signal averaging method to predict perinatal outcome in infants with very preterm fetal growth restriction- a secondary analysis of TRUFFLE-trial. Am J Obstet Gynecol. 2016;215:630.e1-7.
- 74. Pocock SJ. Group sequential methods in the design and analysis of clinical trials. Biometrika. 1977;64:191-9.
- 75. James LS, Morishima HO, Daniel SS, Bowe ET, Cohen H, Niemann WH. Mechanism of late deceleration of the fetal heart rate. Am J Obstet Gynecol. 1972;113:578-82.
- 76. Morishima HO, Daniel SS, Richards RT, James LS. The effect of increased maternal PaO2 upon the fetus during labor. Am J Obstet Gynecol. 1975;123:257-64.





The effect of intrauterine resuscitation by maternal hyperoxygenation on perinatal and maternal outcome: a randomized controlled trial

> S Moors LM Bullens PJ van Runnard Heimel JP Dieleman W Kulik DL Bakkeren ER van den Heuvel MB van der Hout-van der Jagt SG Oei

American Journal of Obstetrics & Gynecology MFM; 2020;2:100102

ABSTRACT

Background

Maternal hyperoxygenation is widely used during labor as an intrauterine resuscitation technique. However, robust evidence regarding its beneficial effect and potential side effects is scarce and previous studies show conflicting results.

Objective

To assess the effect of maternal hyperoxygenation upon suspected fetal distress during the second stage of term labor on fetal heart rate (FHR), neonatal outcome, maternal side effects, and mode of delivery.

Study design

In a single-center randomized controlled trial in a tertiary hospital in The Netherlands, participants were randomized in case of an intermediary or abnormal FHR pattern during the second stage of term labor, to receive either conventional care or 100% oxygen at 10 L/min until delivery. The primary outcome was the change in FHR pattern. Prespecified secondary outcomes were Apgar score, umbilical cord blood gas analysis, neonatal intensive care unit admission, perinatal death, free oxygen radical activity, maternal side effects, and mode of delivery. We performed subgroup analyses for intermediary and abnormal FHR, and for small for gestational age fetuses.

Results

From March 2016 through April 2018, 117 women were included. FHR patterns could be analyzed in 71 women. Changes in FHR (defined as improvement, equal, or deterioration) in favor of maternal hyperoxygenation were significant (OR 5.7, 95% CI 1.7-19.1) using ordinal logistic regression. Apgar score, umbilical cord blood gas analysis, free oxygen radicals, and mode of delivery showed no significant differences between the intervention and control group. Among women with an abnormal FHR, there were fewer episiotomies on fetal indication in the intervention group (25%) than in the control group (65%, P<0.01).

Conclusion

Maternal hyperoxygenation has a positive effect on the FHR in the presence of suspected fetal distress during the second stage of labor. There was no significant difference in the mode of delivery or neonatal outcome, however, significantly fewer episiotomies on fetal indication were performed following maternal hyperoxygenation in the subgroup with abnormal FHR pattern.

INTRODUCTION

Maternal hyperoxygenation is widely used to improve the fetal condition in case of suspected fetal distress during labor.¹⁻³ However, robust evidence regarding its effect is scarce and conflicting, and recommendations in international guidelines are inconsistent.⁴⁻⁶

Several small, non-randomized studies in distressed fetuses show an improvement in fetal heart rate (FHR) pattern or fetal scalp pH as a result of maternal hyperoxygenation.⁷⁻¹² Fetuses with the lowest initial oxygen saturation appear to benefit the most.⁹

In contrast, other studies report potentially harmful effects of prophylactic use of maternal hyperoxygenation,¹³⁻¹⁶ such as significantly more arterial cord blood pH levels <7.20 after maternal hyperoxygenation.¹⁵ A recent RCT among distressed fetuses during labor with >6cm cervical dilation (i.e. combined first and second stage of labor) showed no difference in umbilical artery lactate after maternal hyperoxygenation.¹⁷

Another argument against maternal hyperoxygenation for intrauterine resuscitation is the potential increase in free oxygen radicals, potentially causing cell-damage.^{16,18,19} Previous research shows that the production of free oxygen radicals is increased in case of fetal distress, and after inhalation of high fractions of oxygen in the presence of a normal fetal condition.²⁰ The effect of maternal hyperoxygenation for non-reassuring fetal status on free oxygen radical activity has not been investigated yet.

To our knowledge, there are no RCTs studying the effect of maternal hyperoxygenation in the presence of suspected fetal distress only including women during the second stage of labor. Several reviews underline the need for such a study.¹⁻³ However, before the safety of maternal hyperoxygenation has been further investigated, it is not desirable to perform an RCT powered for neonatal morbidity, requiring a very large study population.² Change in FHR could serve as a surrogate outcome, because a non-reassuring pattern of FHR is considered to be indicative of potential fetal hypoxia.²¹⁻²⁴

Prior to setting up a clinical trial, we first investigated the effect of maternal hyperoxygenation in the distressed fetus with a mathematical model. Simulations with this model indicate that maternal hyperoxygenation leads to an increase in pO₂ in all placental compartments leading to an increased fetal oxygenation and amelioration of the FHR pattern.²⁵

Therefore we performed an RCT in which our objective was to assess the effect of maternal hyperoxygenation in case of suspected fetal distress during the second stage of term labor on FHR, neonatal outcome, maternal side effects, and mode of delivery. We hypothesize that maternal hyperoxygenation will improve FHR, without any severe side effects.

MATERIALS AND METHODS

Study design

This RCT was conducted in a tertiary teaching hospital in The Netherlands. The study protocol has been published previously.²⁶ We included women aged \geq 18 years, in term labor, with an intended vaginal delivery of a singleton fetus in cephalic presentation.

The exclusion criteria were determined with the focus on the risk of excessive free oxygen radical production as well as on reducing the influence of other factors that could affect FHR.

Exclusion criteria were: use of tobacco, recreational drugs, or alcohol during pregnancy, pre-existing cardiac disease, pulmonary disease requiring medication, diabetes, hyperthyroidism, anemia (hemoglobin <10.5 gr/dL) or recent use of corticosteroids, anti-hypertensives, magnesium sulphate, amiodarone, opioids, adriamycin, bleomycin, actinomycin, menadione, promazine, thioridazine, or chloroquine. Fetal factors leading to exclusion were congenital malformations, signs of infection during labor requiring antibiotics, normal or pre-terminal FHR pattern, or prolonged bradycardia according to the modified International Federation of Gynecology and Obstetrics (FIGO) classification (Table 1).^{27,28}

Table 1. Fetal heart rate classification according to modified FIGO criteria, adapted from Yli et al.²⁸

	Baseline heart	FHR variability and	FHR decelerations
	irequency	reactivity	
Normal F HR	• 110–150 bpm	Accelerations 5–25 bpm	 Early uniform decelerations Uncomplicated variable decelerations (loss of < 60 beats)
Intermediary FHR	 100–110 bpm 150–170 bpm Short bradycardia episode 100 bpm for > 3 minutes 80 bpm for > 2 minutes 	 Saltatory pattern (> 25 bpm) < 5 bpm for > 40 minutes 	Uncomplicated variable decelerations (loss of > 60 beats)
A combination of classification	wo or several intermediary	observations will result	in an abnormal FHR
Abnormal FHR	 > 170 bpm Persistent bradycardia < 100 bpm for > 10 minutes < 80 bpm for > 3 minutes (without an increasing tendency) 	 < 5 bpm for > 60 minutes Sinusoidal pattern 	Complicated variable decelerations with a duration of > 60 seconds Repeated late uniform decelerations
<i>Pre-terminal</i> FHR	 Total lack of variability (decelerations or bradycar 	< 2 bpm) and reactivity dia	with or without

Bpm = beats per minute. FHR = fetal heart rate.

Intervention and randomization

All women eligible were asked antepartum to participate during outpatient visits or when they are admitted to the delivery ward before the delivery started. In the presence of suspected fetal distress during the second stage of labor, as defined by intermediary or abnormal FHR pattern according to the modified FIGO classification,²⁷ women were randomized to receive either maternal hyperoxygenation with 100% oxygen at 10 L/min delivered by a non-rebreathing mask (fraction of inspired oxygen approximately 0.80)²⁹ until delivery (intervention group) or conventional care according to local standards without maternal hyperoxygenation (control group). Eligible women were assigned a sealed opaque envelope with treatment allocation, which was opened during the second stage of labor in case of intermediary or abnormal FHR patterns, as classified by the midwife, resident, or obstetrician on duty. Unopened envelopes were reused.

In case co-interventions were required, they were preferably initiated 10 minutes after randomization. A computer-generated allocation sequence is used with random blocks of four or six patients. Stratification was applied for intermediary or abnormal FHR.

Primary outcome

The primary outcome was the change of FHR pattern after randomization into the study. As part of the primary outcome, the depth and duration of decelerations will be analyzed using a computerized algorithm,²⁶ the results of which will be published separately. For this report, we analyzed the change in FHR as evaluated by clinicians. An expert team of three blinded gynecologists classified the FHR according to the FIGO classification.²⁷ The time-frame of interest was 10 minutes before and the period 5 to 15 minutes after start of the study protocol (Figure 1). In case of discrepancy, all three reached consensus by discussion. The change in FHR FIGO classification was categorized as deteriorated, equal or improved after the start of the study protocol (for example, when FHR was classified as *abnormal* before start of the study and subsequently classified as *intermediary* or *normal* after start of the study, the change was categorized as improved FIGO classification).



Figure 1. Time-frames of interest. The time-frames of interest for analysis of outcome measures where patients serve as their own controls.

Secondary outcomes

Neonatal outcomes included 1- and 5-minute Apgar score, venous and arterial umbilical cord blood gas (UCBG) values (pH, base excess, pCO_2), neonatal intensive care unit (NICU) admission and perinatal death. Analysis of UCBG was performed using the ABL90FlexPlus blood gas analyzer (Radiometer Benelux BV, Zoetermeer, The Netherlands). To assure validation of accurate paired UCBG samples, the Modified Westgate-Criteria were used.^{30,31} In case only one valid sample was available, this was categorized as venous as suggested by White et al..³⁰

Due to the brief lifespan of free oxygen radicals, it is extremely difficult to detect these directly.³² Therefore, we measured Malondialdehyde (MDA) as a surrogate marker of free oxygen radical activity. MDA, a by-product of lipid peroxidation, is considered a non-invasive biomarker for free radical damage.²⁰ To assess MDA, two additional blood samples (one venous and one arterial) were drawn from the umbilical cord in heparinized tubes and immediately centrifuged and stored at -20°C. Samples were analyzed in duplo by stable isotope dilution with HPLC-tandem mass-spectrometry (Quattro Premier XE (Waters, Milford, MA, USA), and means were determined.

Maternal outcomes included mode of delivery, patient self-reported side effects of oxygen admission, and reasons for discontinuation

Statistical analyses

The sample size calculation was described in detail in the published protocol.^{26,33} To detect the expected effect with an α of 0.05 and 90% power, a sample size of 96 was required. To accommodate for 20% missing data, we planned to enroll 116 women. IBM SPSS Statistics software (version 25; IBM, Armonk, NY, USA) was used for the statistical analyses. Change in FHR was analyzed using ordinal logistic regression. A kappa for multiple raters was calculated.³⁴ For comparison of continuous variables, an independent T-test or Mann-Whitney U test was used as appropriate. For categorical variables, the χ^2 -test or Fisher's exact test was used as appropriate. *P*<0.05 for a 2-tailed test was considered significant.

The analyses were performed for the overall study population, as well as for the subgroups of intermediary FHR pattern, abnormal FHR pattern, and small for gestational age (SGA) fetuses (<10th percentile). The primary analyses were intention-to-treat. We anticipated that in some cases it might not be possible to apply oxygen according to the study protocol, due to practical concerns. Therefore, we also performed per-protocol analyses, excluding women receiving oxygen for <5 minutes, since the optimal effect of maternal hyperoxygenation cannot be reached within that period.³⁵ Furthermore, we also performed a sensitivity analysis to address possible confounding issues due to missingness.

Interim analysis

A pre-planned interim analysis was performed after 50% of the women were included to investigate potential safety issues regarding oxygen administration. There were no significant differences in the number of neonates with 5-minute Apgar score <7, arterial umbilical cord blood pH <7.05, NICU admissions, or perinatal deaths between the intervention and control group.

RESULTS

Between March 2016 and April 2018, a total of 376 women gave informed consent for the study. Of those, 117 women developed abnormal or intermediary FHR during the second stage of labor and were therefore randomized into the study. A total of 57 women were assigned to receive oxygen and 60 women were assigned to receive conventional care (CONSORT flow diagram, Figure 2).

The treatment groups were comparable regarding baseline characteristics (Table 2).

Within the first 10 minutes after randomization, 10 women (8.5%) received additional intrauterine resuscitation techniques: eight (13.3%) in the control group and two (3.5%) in the intervention group (P=0.10). These interventions included maternal repositioning, discontinuation of pushing, and adjustment of oxytocin dosage.

FHR pattern

For all women, FHR was available until birth. Since 46 women (39%) delivered within 15 minutes after enrollment, the FHR data sets were not complete for the post-study time-frame of interest (Figure 1), hence no pre-post analysis could be performed for this group. From 71 of 117 women (61%), the FHR pattern was available for the complete pre and post randomization time-frames of interest and, therefore, change in FHR pattern could be determined. The blinded gynecologists had discrepancy in FHR analyses in 25 cases (35%). The overall kappa was 0.54 [95% CI 0.41-0.68]. To this end, consensus on the final FIGO classification was reached by discussion between all three physicians.

Improvement of FHR was seen over four times more often in the intervention group compared to the control group (2.9 % vs. 13.9%, Table 3). Furthermore, deterioration of FHR was seen four times less often in the intervention group than in the control group (8.3% vs 34.3%). These changes in FHR were significant with an odds ratio (OR) of 5.7 (95% CI 1.7-19.1). None of the subgroup analyses for changes in FHR showed significance.



* Four women had signs of infection and were treated with antibiotics, two women had diabetes, two women smoked during pregnancy, one woman delivered prematurely, one woman delivered at 42 weeks and had anemia, and there were three fetuses with congenital abnormalities (cheilognathopalatoschisis, hypospadias, and pyelectasis).

** Three women had signs of infection and were treated with antibiotics, two women smoked during pregnancy, one woman delivered prematurely, and one woman delivered at 42 weeks.

Figure 2. Trial flow diagram. A total of 117 women were allocated to the intervention or control group. Both an intention-to-treat and per-protocol analysis were performed. For the FHR analysis, 71 women were eligible based on availability of complete FHR time-frames.

Table 2. Patient characteristics of women randomized to hyperoxygenation treatment or conventional treatment.

	Maternal hyperoxygenation n=57	Conventional care n=60
Maternal age (years)	31.8 ± 4.2	30.7 ±3.4
Gestational age (days)	279 ±9.0	280 ±8.8
Parity ≥1	22 (38.6%)	27 (45%)
BMI (kg/m2)	25 ± 4.7	24 ± 5.0
Fetal sex male	30 (52.6%)	26 (43.3%)
Birthweight (grams)	3510 ± 471.7	3541.4 ±560.2
SGA	7 (12.3%)	5 (8.3%)
Intermediary FHR pattern	25 (43.9%)	34 (56.7%)
Abnormal FHR pattern	32 (56.1%)	26 (43.3%)

Data are mean±SD or n (%). BMI = body-mass index, SGA = small for gestational age, FHR= fetal heart rate

Table 3. Changes in FHR according to FIGO classification following maternal hyperoxygenation versus conventional care in the total study population.

	Deterioration FHR	Equal FHR	Improvement FHR	Odds ratio (95% CI)
Total study population (n=71)				
Maternal hyperoxygenation (n=36)	3 (8.3%)	28 (77.8%)	5 (13.9%)	57/17101)
Conventional care (n=35)	12 (34.3%)	22 (62.9%)	1 (2.9%)	5.7 (1.7-19.1)
Abnormal FHR (n=34)				
Maternal hyperoxygenation (n=20)	1 (5.0%)	15 (75%)	4 (20%)	
Conventional care (n=14)	4 (28.6%)	9 (64.3%)	1 (7.1%)	5.2 (0.9-30.2)
Intermediary FHR (n=37)				
Maternal hyperoxygenation (n=16)	2 (12.5%)	13 (81.3%)	1 (6.3%)	
Conventional care (n=21)	8 (38.1%)	13 (61.9%)	0	5.0 (0.9-27.3)
SGA (n=9)				
Maternal hyperoxygenation (n=6)	1 (16.7%)	3 (50%)	2 (33.3%)	
Conventional care (n=3)	1 (33.3%)	2 (66.7%)	0	4.3 (0.2-84.6)

Data are n (%).CI= confidence interval. FHR= fetal heart rate. SGA = small for gestational age. Odds ratio of ordinal regression analysis

Neonatal outcome and mode of delivery

Four neonates had 5-minute Apgar score <7, three in the control group (5.0%), and one (1.8%) in the intervention group (P=0.62, Table 4). In the intervention group 13 neonates (28.9%) had arterial cord blood pH <7.20, compared to 22 (41.5%) in the control group (P=0.19). No neonatal deaths occurred in this study.

Table 4. Neonatal outcome and mode of delivery following maternal hyperoxygenation orconventional care in the intention-to-treat analysis.

Outcome parameter	Maternal	Conventional	Р
	hyperoxygenation	care	
Neonatal outcome			
Median 1-minute Apgar score ¥	9 (8.5-9)	9 (8.25-9)	0.77
[n= 57 O2/ 60 control]			
Neonates with 1-minute Apgar score <5 ¶ [n= 57 O2/ 60 control]	4 (7%)	4 (6.7%)	1.00
Median 5-minute Apgar Score ¥ [n= 57 O2/60 control]	10 (10-10)	10 (10-10)	0.13
Neonates with 5-minute Apgar score<7 ‡ [n= 57 O2/ 60 control]	1 (1.8%)	3 (5.0%)	0.62
Median arterial pH ¥ [n=45 O2/ 53 control]	7.22 (7.19-7.26)	7.20 (7.16-7.27)	0.35
pH arterial <7.05 ‡ [n= 45 O2/ 53 control]	1 (2.2%)	0	0.46
Median venous pH ¥ [n= 57 O2/ 60 control]	7.30 (7.26-7.34)	7.30 (7.26-7.35)	0.94
Arterial base excess ¥ [n=45 O2/ 51 control]	-6 (-8/-3)	-6 (-8/-4)	0.69
Arterial pCO ₂ ¥ [n=45 O2/ 52 control]	56 (51.5-59.5)	57 (52-62)	0.54
Arterial Malondialdehyde ¥ [n=41 O2/48control]	4.45 (3.68-5.35)	4.15 (3.40-4.75)	0.15
Venous Malondialdehyde § [n=45O2/54control]	4.67±1.27	4.38±1.15	0.21
Neonatal Intensive Care Unit admissions ‡ [n= 57	1 (1.8%)	2 (3.3%)	1.00
O2/60 control]			
Mode of delivery outcome			
Episiotomy all indications ¶ [n=57O2/60control]	30 (52.6%)	33 (55%)	0.80
Episiotomy fetal indication ¶ [n=57 O2/60control]	17 (29.8%)	27 (45.0%)	0.09
Assisted delivery all indications ¶ [n=57 O2/ 60	7 (12%)	9 (15%)	0.67
control] Of which:			
Cesarean section	2 (3.5%)	0	
Vacuum-assisted delivery	5 (8.8%)	7 (12%)	
Cesarean section after failed vacuum- assisted delivery	0	2 (3.4%)	
Assisted delivery fetal indication ‡ [n= 57 O2/60 control]	4 (7.0%)	6 (10.0%)	0.74
Active second stage of labor in minutes ¥ [n= 57 O2/ 60 control]	35 (20-64)	25 (14-58)	0.32

Data are mean \pm SD, median (IQR), or n (%). Data are analyzed by ¥; Mann-Whitney U test, ¶; χ^2 test, ‡ Fisher's exact test, or §; Independent t-test. pCO₂= Partial carbon dioxide pressure.

In the subgroup analyses, we did not find significant differences between treatment groups in Apgar scores or NICU admissions (Table 5). However, in the subgroup with abnormal FHR patterns, fewer episiotomies for fetal indication were performed in the intervention group than in the control group (n=8, 25% vs n=17, 65%, P<0.01).

Table 5. Subgroup analysis of neonatal outcomes and mode of delivery on intention-to-treat basis.

Outcome parameter Subgroup	Maternal hyperoxygenation	Conventional care	Р
Neonates with 1-minute Apgar score <5 ‡			
Abnormal FHR [n=32 O2/ 26 control]	3 (9%)	3 (12%)	1.00
Intermediary FHR [n=25 O2 /34 control]	1 (4%)	1 (3%)	1.00
SGA [n=7 O2 / 5 control]	1 (14%)	0	1.00
Neonates with 5-minute Apgar score <7 ‡			
Abnormal FHR [n=32 O2/ 26 control]	0	2 (8%)	0.20
Intermediary FHR [n=25 O2 /34 control]	0	1 (3%)	1.00
SGA [n=7 O2 / 5 control]	0	0	-
Arterial pH <7.05 ±			
Abnormal FHR [n=22 O2/ 24 control]	1 (5%)	0	0.48
Intermediary FHR [n=23 O2 / 29control]	0	0	-
SGA [n=6 O2 / 5 control]	0	0	-
Venous pH <7.10 ±			
Abnormal FHR [n=32 O2/26 control]	1 (3%)	0	1.00
Intermediary FHR [n=25 O2 /34 control]	0	0	
SGA [n=7 O2 / 5 control]	0	0	-
Arterial pCO, ¥			
Abnormal FHR [n=22 O2/23 control]	57 (52-61)	54 (49-62)	0.52
Intermediary FHR [n=23 O2 / 29control]	55 (51-59)	57 (53-62)	0.14
SGA [n=6 O2 / 5 control]	55 (49-57)	54 (42-64)	1.00
Venous pCO ₂ ¥			
Abnormal FHR [n= 32 O2/ 25 control]	45 (40-49)	44 (41-47)	0.75
Intermediary FHR[n=25 O2/34control]	38 (36-43)	42 (39-45)	0.04
SGA [n=7 O2 / 5 control]	39 (37-45)	41 (36-47)	0.54
Arterial Malondialdehyde ¥			
Abnormal FHR [n=24 O2/21 control]	4.65 (3.86-5.45)	4.15 (3.40-4.45)	0.07
Intermediary FHR [n=17 O2 / 27control]	4.20 (3.30-5.35)	4.20 (3.35-5.20)	0.96
SGA [n=6 O2 / 4 control]	5.15 (3.03-6.98)	4.55 (4.31-4.75)	0.73
Venous Malondialdehyde ¥			
Abnormal FHR [n=26 O2/23 control]	4.60 (3.99-5.43)	4.40 (3.95-5.00)	0.50
Intermediary FHR [n=19 O2 / 31control]	4.60 (3.95-5.40)	4.45 (3.75-5.20)	0.37
SGA [n=7 O2 / 5 control]	4.60 (3.68-6.86)	4.20 (3.43-5.03)	0.66
Enisiotomy fetal indication			
Abnormal FHR ¶ [n=32O2/ 26control]	8 (25%)	17 (65%)	<0.01
Intermediary FHR ¶[n=2502/34control]	9 (36%)	10 (29%)	0.59
SGA‡ [n=7 O2 / 5 control]	4 (57%)	4 (100%)	0.24
Assisted delivery, all indications ⁺			
Abnormal EHR [$n=32$ $O2/26$ control]	5 (16%)	5 (19%)	0.74
Intermediary FHR [n=25 O2 /34 control]	2 (8%)	4 (12%)	1.00
SGA[n=7 O2 / 5 control]	2 (29%)	1 (25%)	1.00

Table 5. Continued.

Outcome parameter	Maternal	Conventional	Р
Subgroup	hyperoxygenation	care	
Assisted delivery, fetal indications‡			
Abnormal FHR [n=32 O2/ 26 control]	3 (9%)	4 (15%)	0.69
Intermediary FHR [n=25 O2/34 control]	1 (4%)	2 (6%)	1.00
SGA [n=7 O2 / 5 control]	0	0	-
Active second stage of labor in minutes $¥$			
Abnormal FHR [n=32 O2/ 26 control]	26 (17-56)	21 (12-55)	0.14
Intermediary FHR [n=25 O2/34 control]	42 (27-74)	36 (16-61)	0.54
SGA [n=7 O2 / 5 control]	59 (38-79)	24 (8-47)	0.052

Data are median (IQR), or n (%). Data are analyzed by ¥; Mann-Whitney U test, ¶; χ^2 test, ‡; Fisher's exact test, or §; Independent t-test. O2=maternal hyperoxygenation group. pCO₂ = Partial carbon dioxide pressure

Maternal outcome

The median duration of oxygen admission was 12 minutes (range 0-75 minutes). In 19 women (33%) oxygen administration was stopped before the infant was born, mostly due to the discomfort of the facemask (n=17, 89%). In 36 women (63%) no side effects were reported.

Per-protocol analysis

We performed a per-protocol analysis which excluded 13 women from the intervention group who had oxygen administration for <5 minutes. No women in the control group received additional oxygen. In addition, we excluded 20 women who had been included despite the presence of exclusion criteria. In the per-protocol analysis, maternal hyperoxygenation had a significant positive effect on FHR classification in the abnormal FHR subgroup (P=0.045). All other results of the per-protocol analyses were similar to the intention-to-treat analyses (Supplementary Tables 1-2).

Post-hoc analysis

In a post-hoc analysis, we calculated the Pearsons' and Spearman's correlation coefficient to explore the relation between the duration of oxygen administration and arterial cord blood parameters. No significant correlation was found between duration of oxygen administration and arterial pH, base excess, pCO₂, or MDA.

Sensitivity analysis

For our primary outcome, we had 39% missing data due to women who delivered within the time-frame of interest. First, we ran an analysis to compare the women with and without missing FHR time-frames (i.e. missing data for the primary outcome). We found multiparity (P<0.001), and assisted delivery (P=0.02) to be significantly higher in the missing group All other parameters (including FHR at baseline) were equal between these groups. Secondly, we performed a sensitivity analysis using ordinal regression analysis correcting for multiparity and assisted delivery as possible confounders for missingness. In line with the results from the primary analyses, the sensitivity analysis showed a significant positive effect of maternal hyperoxygenation on FHR classification with an OR of 6.4 (95% CI 1.9-22.2). Furthermore, in this sensitivity analysis, a significant positive effect of maternal hyperoxygenation on FHR was shown in the subgroup of patients with abnormal FHR (OR 6.2, 95% CI 1.05-36.5). The subgroups of intermediary FHR and SGA showed the same non-significant trend of a positive effect of maternal hyperoxygenation on FHR as seen in the complete-case analyses (OR 5.6 95% CI 0.8-38.6 and OR 5.2 95% CI 0.3-102.3, respectively).

COMMENT

Principal findings

This study shows that maternal hyperoxygenation has a positive effect on FHR pattern in case of fetal distress during the second stage of labor, as compared to conventional care.

We found no adverse effects regarding neonatal outcome, mode of delivery, or formation of MDA. We did find fewer episiotomies on fetal indication following maternal hyperoxygenation in the subgroup with abnormal FHR pattern.

Results and clinical implications

The positive effect of maternal hyperoxygenation on FHR cannot be explained by the use of cointerventions, which was comparable in both groups. There is a relatively high frequency of deterioration of FHR pattern in both groups. However, this is not uncommon in women at the onset of the second stage of labor. The analyses of FHR pattern in the subgroups showed no significant differences in the complete-case analyses. This may be explained by small sample sizes due to the number of missing values (39% for FHR pattern analysis), which was higher than the accommodated 20%. To correct for possible confounders of missingness, we ran a sensitivity analysis which showed a similar significant positive effect as the complete-case analysis.

An important concern regarding the use of maternal oxygen administration during labor is a potential decrease in umbilical cord pH.^{2,15} Although our study was not powered for this outcome, there are no indications in our results to confirm this effect. Moreover, in the intermediary FHR subgroup, pCO_2 was significantly lower in hyperoxygenated women than in conventionally treated women.

Thorp et al. found that the duration of prophylactic oxygen therapy was inversely correlated with arterial cord pH.¹⁵ This relation was not found in our study with distressed fetuses.

Effect of maternal hyperoxygenation on perinatal and maternal outcome (INTEREST O2 study)

A Cochrane review from 2012 on maternal oxygen administration for fetal distress identified two RCT on the prophylactic use of oxygen.^{1,15,36} Since then, one RCT has been performed that included patients with suspected fetal distress. In this study by Raghuraman et al.,¹⁷ women were included during labor with >6cm cervical dilation and type II FHR tracings. Patients were randomized to receive either maternal hyperoxygenation until birth or to the control group, breathing room air. The primary outcome was umbilical artery lactate. In line with our results, they found all umbilical artery blood gas components to be similar in both groups. Their results did not report results on FHR, Apgar score, and free oxygen radical activity.

We found 5-minute Apgar score <7 almost three times more frequent in the control group, albeit not significant. There were no previous studies investigating Apgar score after maternal hyperoxygenation. In the abnormal FHR subgroup, significantly fewer episiotomies on fetal indication were performed, and a similar trend was seen in the total group. These findings correspond with results from Haydon et al, which show that fetuses with the lowest initial oxygen saturations benefit the most from maternal hyperoxygenation.⁹

In this study, oxygen administration was stopped in 33% before the infant was born. We hypothesize that this high rate of discontinuation is partly due to the lack of evidence regarding the effect of maternal hyperoxygenation as an intrauterine resuscitation technique.

With the positive findings from this study, especially the positive effect on FHR and the decrease in episiotomy rate for fetal distress, we expect women to be more willing to continue oxygen supplementation.

Research implications

Based on our study results, we cannot subscribe to the theoretical increased risk of increased oxygen radical production.² Mean MDA values were not significantly different between the intervention and control group. However, the clinical implications of increased free oxygen radical activity in the neonate remain unclear. Furthermore, possible long-term effects of maternal hyperoxygenation were not taken into account in this study.

Strengths and limitations

This is the first RCT investigating the effect of maternal hyperoxygenation in the presence of fetal distress, only including women during the second stage of labor.¹ In addition, this study takes into account both the beneficial and harmful effects of maternal hyperoxygenation. However, practical and safety issues led to some limitations. Ideally, the primary outcome should have been neonatal morbidity. Achieving sufficient power to address this outcome measure would require a sample size of over 10,000 women.² Because some studies raised concerns about the potentially harmful effects of maternal hyperoxygenation, we chose not to expose such a large group of women and their fetuses

to this intervention before its safety has been further investigated. Hence we took FHR as the primary outcome of our study, which represents a surrogate endpoint, of fetal wellbeing. Currently, FHR is deemed the best available method to record changes in fetal condition during labor, despite its poor specificity.^{21-24.27.37} We focused on analyzing features in the FHR pattern that are previously related to fetal acidosis. It is therefore unlikely to expect improvement in neonatal outcome if no beneficial effect on FHR can be shown.

Our intended time-frame after the start of the study protocol was 10 minutes after the start of the study in the control group.²⁶ In the intervention group, our time-frame of interest was the period between 5 and 15 minutes after maternal hyperoxygenation was initiated, since it will take 5 minutes for maternal pO2 to increase to a maximum.^{26,35} To allow blinding and limit potential bias for the gynecologists who classified the FHR, we chose to equalize the time-frames for the intervention and the control group. Further investigations are needed to detect whether the amelioration in FHR pattern translates into improved fetal metabolic status and neonatal outcome.

Despite the widespread use of oxygen for intrauterine resuscitation, there is no guideline regarding the optimal dose or duration. The dose used in this study (100% oxygen at 10L/min), applied via a non-rebreathing facemask, is consistent with other studies.^{11,15,17,38} Furthermore, the expected effect of 100% oxygen is greater than that of 40% oxygen.⁹ Due to practical impossibilities and in line with other studies, we have chosen not to use a sham procedure in the control group.^{11,15,17,38} This prohibited blinding of health care providers during labor, which could potentially cause bias for the outcomes episiotomy and assisted delivery. We aimed to minimize the influence of bias by blinding the FHR expert panel for treatment.

Due to the complex setting of the study, it was difficult to adhere to the protocol, despite intensive training of all study personnel. A total of 21 women were included despite having exclusion criteria. In addition, we included one woman more than we originally aimed for. Factors that may have contributed to this violation of the study protocol are the acute obstetric situation, strict exclusion criteria, and the use of envelopes for randomization. To investigate the effect of these patients on the study results, the perprotocol analysis was performed with and without exclusion of the incorrectly included women. This did not alter the results, nor did the results indicate safety hazards.

CONCLUSION

Maternal hyperoxygenation has a positive effect on the FHR pattern in the presence of fetal distress during the second stage of labor. There was no significant difference in mode of delivery or neonatal outcome, however, significantly fewer episiotomies on fetal indication were performed following maternal hyperoxygenation in the subgroup with abnormal FHR pattern. A larger study powered for improvement in neonatal outcome is needed to propose strong recommendations for clinical practice.

ACKNOWLEDGMENTS

This research was performed within the framework of Eindhoven MedTech Innovation Center (e/MTIC). We thank all the obstetric staff members of Máxima Medical Center for their help with this study, as well as all the laboratory staff. We also thank Bernice Wieland and Julia Smith, both medical students at Máxima Medical Center, for their help with the data collection. The study was conceived and designed by LMB, SM, MBvdH, PJvRH, and SGO. SGO was the initiator of this project. SM and LMB performed data collection/ entry. SM, LMB, MBvdH, PJvRH, EvdH, and SGO were responsible for the analysis and interpretation of data. EvdH and JD advised on the statistical analysis (power analysis and description of the statistical tests that were used to analyze the study results and the interim analysis). DB and WK performed the malondialdehyde analyses. SM prepared the draft manuscript. All other authors revised this manuscript, and they all read and approved the final manuscript. The study sponsor (Máxima Medical Center Board of Management) was not involved in the design, implementation, data analysis or any reporting of this study.

SUPPLEMENTARY TABLES

Supplementary Table 1. Changes in FHR according to FIGO classification following maternal hyperoxygenation versus conventional care in the per protocol analysis

	Deterioration	Equal	Improvement	Р
	FHR	FHR	FHR	
Total study population (n=50)				
Maternal hyperoxygenation (n=20)	2 (9.5%)	16 (76.2%)	3 (14.3%)	0.02
Conventional care (n=29)	11 (37.9%)	18 (62.1%)	0	0.02
Abnormal FHR (n=23)				
Maternal hyperoxygenation (n=13)	0	12 (85.7%)	2 (14.3%)	0.045
Conventional care (n=9)	3 (33.3%)	6 (66.7%)	0	0.045
Intermediary FHR (n=27)				
Maternal hyperoxygenation (n=7)	2 (28.6%)	4 (57.1%)	1 (14.3%)	0.22
Conventional care (n=20)	8 (40%)	12 (60%)	0	0.22
SGA (n=5)				
Maternal hyperoxygenation (n=3)	0	2 (66.7%)	1 (33.3%)	0.22
Conventional care (n=2)	1 (50%)	1 (50%)	0	0.33

Data are n (%). FHR= fetal heart rate. SGA = small for gestational age. The differences are analyzed with a χ^2 test since ordinal regression analysis was not possible because of too many cells with zero frequencies.

Supplementary Table 2. Neonatal outcome and mode of delivery following maternal hyperoxygenation or conventional care in the per protocol analysis

Outcome parameter	Maternal	Conventional	Р
	hyperoxygenation	care	
Neonatal outcome			
1- minute Apgar score ¥ [n= 31 O2 / 53 control]	9 (9-9)	9 (9-9)	0.52
Neonates with 1-minute Apgar Score <5 ‡ [n=31 O2 / 53 control]	1 (3.2%)	4 (7.5%)	0.65
5-minute Apgar Score ¥ [n= 31 O2 / 53 control]	10 (10-10)	10 (9.5-10)	0.15
Neonates with 5-minute Apgar Score <7 ‡ [n=31 O2 / 53 control]	0	3 (5.7%)	0.29
pH arterial § [n=22 O2 / 47 control]	7.23±0.05	7.21±0.06	0.20
pH arterial <7.05 ‡ [n=22 O2 / 47 control]	0	0	-
pH venous ¥ [n= 31 O2 / 53 control]	7.30 (7.27-7.34)	7.31 (7.27-7.35)	0.50
Base Excess arterial § [n=22 O2 / 4 5 control]	-4.73±2.55	-5.87±2.68	0.10
pCO ₂ arterial § [n=22 O2 / 4 6 control]	55.4±5.4	56.4±7.2	0.56
Malondialdehyde arterial ¥ [n=2 5 O2 / 4 2 control]	4.45 (3.68-5.75)	4.05 (3.31-4.84)	0.13
Malondialdehyde venous § [n=26 O2 / 47 control]	4.82±1.40	4.34±1.22	0.13
NICU admissions ‡ [n=31 O2 / 53 control]	0	2 (3.8%)	0.53
Mode of delivery outcome			
Episiotomy, all indications ¶ [n=31 O2 / 53 control]	18 (58.1%)	28 (52.8%)	0.64
Episiotomy, fetal indication ¶ [n=31 O2 / 53 control]	10 (32.3%)	22 (41.5%)	0.40
Assisted delivery, all indications ¶ [n=31 O2 / 53 control]	4 (12.9%)	10 (18.9%)	0.48
Assisted delivery, fetal indication ¶ [n=31 O2 / 53 control]	1 (3.2%)	5 (9.4%)	0.41
Active second stage of labor in minutes § [n=31 O2 / 53 control]	46.9±28.7	37.6±30.3	0.17

Data are mean±SD, median (IQR), or n (%). Data are analyzed by ¥; Mann-Whitney U test, ¶; χ^2 test, ‡ Fisher's exact test, or §; Independent t-test. pCO₂ = Partial carbon dioxide pressure

REFERENCES

- 1. Fawole B, Hofmeyr GJ. Maternal oxygen administration for fetal distress. *Cochrane Database Syst Rev.* 2012;12:CD000136. doi: 10.1002/14651858.CD000136.pub2 [doi].
- 2. Hamel MS, Anderson BL, Rouse DJ. Oxygen for intrauterine resuscitation: Of unproved benefit and potentially harmful. *Am J Obstet Gynecol.* 2014;211(2):124-127. doi: 10.1016/j.ajog.2014.01.004 [doi].
- 3. Bullens LM, van Runnard Heimel, P J, van der Hout-van der Jagt, M B, Oei SG. Interventions for intrauterine resuscitation in suspected fetal distress during term labor: A systematic review. *Obstet Gynecol Surv.* 2015;70(8):524-539. doi: 10.1097/ OGX.00000000000215 [doi].
- 4. Bullens LM, Moors S, van Runnard Heimel, P J, van der Hout-van der Jagt, M B, Oei SG. Practice variation in the management of intrapartum fetal distress in the netherlands and the western world. *Eur J Obstet Gynecol Reprod Biol*. 2016;205:48-53. doi: 10.1016/j. ejogrb.2016.08.012 [doi].
- 5. American College of Obstetricians and Gynecologists. Practice bulletin no. 116: Management of intrapartum fetal heart rate tracings. *Obstet Gynecol*. 2010;116(5):1232-1240. doi: 10.1097/AOG.0b013e3182004fa9 [doi].
- 6. Delgado Nunes V, Gholitabar M, Sims JM, Bewley S, Guideline Development Group. Intrapartum care of healthy women and their babies: Summary of updated NICE guidance. *BMJ*. 2014;349:g6886. doi: 10.1136/bmj.g6886 [doi].
- 7. Althabe O, Jr, Schwarcz RL, Pose SV, Escarcena L, Caldeyro-Barcia R. Effects on fetal heart rate and fetal pO2 of oxygen administration to the mother. *Am J Obstet Gynecol.* 1967;98(6):858-870. doi: 0002-9378(67)90205-0 [pii].
- 8. Gare DJ, Shime J, Paul WM, Hoskins M. Oxygen administration during labor. *Am J Obstet* Gynecol. 1969;105(6):954-961. doi: 0002-9378(69)90104-5 [pii].
- Haydon ML, Gorenberg DM, Nageotte MP, et al. The effect of maternal oxygen administration on fetal pulse oximetry during labor in fetuses with nonreassuring fetal heart rate patterns. *Am J Obstet Gynecol.* 2006;195(3):735-738. doi: S0002-9378(06)00867-2 [pii].
- 10. Hidaka A, Komatani M, Ikeda H, Kitanaka T, Okada K, Sugawa T. A comparative study of intrauterine fetal resuscitation by beta-stimulant and O2 inhalation. *Asia Oceania J Obstet Gynaecol.* 1987;13(2):195-200.
- 11. Simpson KR, James DC. Efficacy of intrauterine resuscitation techniques in improving fetal oxygen status during labor. *Obstet Gynecol.* 2005;105(6):1362-1368. doi: 105/6/1362 [pii].
- 12. Dildy GA, Clark SL, Loucks CA. Intrapartum fetal pulse oximetry: The effects of maternal hyperoxia on fetal arterial oxygen saturation. *Am J Obstet Gynecol*. 1994;171(4):1120-1124. doi: 0002-9378(94)90048-5 [pii].
- 13. Perreault C, Blaise GA, Meloche R. Maternal inspired oxygen concentration and fetal oxygenation during caesarean section. *Can J Anaesth*. 1992;39(2):155-157.doi: 10.1007/BF03008647 [doi].
- 14. Saling E. Effect of oxygen inhalation by the mother on the blood gases and acid-base equilibrium of the fetus. *Geburtshilfe Frauenheilkd*. 1963;23:528-538.

Chapter 7

- 15. Thorp JA, Trobough T, Evans R, Hedrick J, Yeast JD. The effect of maternal oxygen administration during the second stage of labor on umbilical cord blood gas values: A randomized controlled prospective trial. *Am J Obstet Gynecol*. 1995;172(2 Pt 1):465-474. doi: 0002-9378(95)90558-8 [pii].
- 16. Nesterenko TH, Acun C, Mohamed MA, et al. Is it a safe practice to administer oxygen during uncomplicated delivery: A randomized controlled trial? *Early Hum Dev.* 2012;88(8):677-681. doi: 10.1016/j.earlhumdev.2012.02.007 [doi].
- 17. Raghuraman N, Wan L, Temming LA, et al. Effect of oxygen vs room air on intrauterine fetal resuscitation: A randomized noninferiority clinical trial. *JAMA Pediatr.* 2018. doi: 10.1001/jamapediatrics.2018.1208 [doi].
- 18. Khaw KS, Ngan Kee WD. Fetal effects of maternal supplementary oxygen during caesarean section. *Curr Opin Anaesthesiol.* 2004;17(4):309-313. doi: 10.1097/01. aco.0000137089.37484.5e [doi].
- 19. Torres-Cuevas I, Parra-Llorca A, Sanchez-Illana A, et al. Oxygen and oxidative stress in the perinatal period. *Redox Biol.* 2017;12:674-681. doi: S2213-2317(17)30057-5 [pii].
- 20. Dede FS, Guney Y, Dede H, Koca C, Dilbaz B, Bilgihan A. Lipid peroxidation and antioxidant activity in patients in labor with nonreassuring fetal status. *Eur J Obstet Gynecol Reprod Biol.* 2006;124(1):27-31. doi: S0301-2115(05)00210-1 [pii].
- 21. Devane D, Lalor JG, Daly S, McGuire W, Cuthbert A, Smith V. Cardiotocography versus intermittent auscultation of fetal heart on admission to labour ward for assessment of fetal wellbeing. *The Cochrane database of systematic reviews*. 2017;1:CD005122.
- 22. Alfirevic Z, Devane D, Gyte GM, Cuthbert A. Continuous cardiotocography (CTG) as a form of electronic fetal monitoring (EFM) for fetal assessment during labour. *The Cochrane database of systematic reviews*. 2017;2:CD006066.
- 23. Elliott C, Warrick PA, Graham E, Hamilton EF. Graded classification of fetal heart rate tracings: Association with neonatal metabolic acidosis and neurologic morbidity. *Obstet Gynecol.* 2010;202(3):258.e1-8. doi: 10.1016/j.ajog.2009.06.026.
- 24. American College of Obstetricians, and Gynecologists. ACOG practice bulletin no. 106: Intrapartum fetal heart rate monitoring: Nomenclature, interpretation, and general management principles. *Obstet Gynecol.* 2009;114(1):192-202. doi: 10.1097/ AOG.0b013e3181aef106.
- 25. Bullens LM, van der Hout-van der Jagt, MB, Van Runnard Heimel, PJ, Oei G. A simulation model to study maternal hyperoxygenation during labor. *Acta Obstet Gynecol Scand*. 2014;93(12):1268-1275. doi: 10.1111/aogs.12486 [doi].
- 26. Bullens LM, Hulsenboom ADJ, Moors S, et al. Intrauterine resuscitation during the second stage of term labour by maternal hyperoxygenation versus conventional care: Study protocol for a randomised controlled trial (INTEREST O2). *Trials*. 2018;19(1):195-x. doi: 10.1186/s13063-018-2567-x [doi].
- 27. Ayres-de-Campos D, Spong CY, Chandraharan E, FIGO Intrapartum Fetal Monitoring Expert Consensus Panel. FIGO consensus guidelines on intrapartum fetal monitoring: Cardiotocography. *Int J Gynaecol Obstet*. 2015;131(1):13-24. doi: 10.1016/j. ijgo.2015.06.020 [doi].
- Yli B, Hahn T, Kessler J, Lie HK, Martinussen M. Bigger is not always better... the validity of the US randomized trial of STAN for norway. Mölndal, sweden: Neoventa medical. http://www.neoventa.com/2015/11/bigger-isnot- always-better/. Accessed 6 Nov, 2017.

- 29. Parillo JE. Critical care medicine: Principles of diagnosis and management in the adult. Fourth edi ed. Philadelphia, PA: Elsevier Inc; 2014.
- 30. White CR, Doherty DA, Kohan R, Newnham JP, Pennell CE. Evaluation of selection criteria for validating paired umbilical cord blood gas samples: An observational study. *BJOG.* 2012;119(7):857-865. doi: 10.1111/j.1471-0528.2012.03308.x [doi].
- 31. Westgate J, Garibaldi JM, Greene KR. Umbilical cord blood gas analysis at delivery: A time for quality data. *Br J Obstet Gynaecol*. 1994;101(12):1054-1063.
- 32. Longini M, Belvisi E, Proietti F, Bazzini F, Buonocore G, Perrone S. Oxidative stress biomarkers: Establishment of reference values for isoprostanes, AOPP, and NPBI in cord blood. *Mediators Inflamm*. 2017;2017:1758432. doi: 10.1155/2017/1758432 [doi].
- 33. Bullens LM, Hulsenboom ADJ, Moors S, et al. Correction to: Intrauterine resuscitation during the second stage of term labour by maternal hyperoxygenation versus conventional care: Study protocol for a randomised controlled trial (INTEREST O2) (trials (2018) 19 (195) DOI: 10.1186/s13063-018-2567-x). *Trials*. 2018;19(1).
- 34. Nicholls D. Kappa for multiple raters. http://imaging.mrc-cbu.cam.ac.uk/statswiki/FAQ/kappa/multiple. Accessed 20 February, 2020.
- 35. Vasicka A, Quilligan EJ, Aznar R, Lipsitz PJ, Bloor BM. Oxygen tension in maternal and fetal blood, amniotic fluid, and cerebrospinal fluid of the mother and the baby. *Am J Obstet Gynecol.* 1960;79:1041-1047. doi: 0002-9378(60)90508-1 [pii].
- 36. Sirimai K, Atisook R, Boriboonhirunsarn D. The correlation of intrapartum maternal oxygen administration and umbilical cord blood gas values. *Acta Obstetricia et Gynecologica Scandinavica Supplement*. 1997;76(167:2):90.
- 37. Schiermeier S, Pildner von Steinburg S, Thieme A, et al. Sensitivity and specificity of intrapartum computerised FIGO criteria for cardiotocography and fetal scalp pH during labour: Multicentre, observational study. *BJOG : an international journal of obstetrics and gynaecology*. 2008;115(12):1557-1563. doi: 10.1111/j.1471-0528.2008.01857.x.
- 38. Simon VB, Fong A, Nageotte MP. Supplemental oxygen study: A randomized controlled study on the effect of maternal oxygen supplementation during planned cesarean delivery on umbilical cord gases. *Am J Perinatol.* 2018;35(1):84-89. doi: 10.1055/s-0037-1606184 [doi].





A randomized controlled trial studying the effect of maternal hyperoxygenation on fetal heart rate in case of fetal distress

> S Moors R Joshi LM Bullens NHM van Oostrum M Regis ER van den Heuvel JOEH van Laar SG Oei MB van der Hout-van der Jagt

Physiol Meas. 2020 Oct 13. doi: 10.1088/1361-6579/abc0b6 Online ahead of print

ABSTRACT

Objective

To investigate the effect of maternal hyperoxygenation on fetal heart rate (FHR) when applied for suspected fetal distress during the second stage of term labor.

Approach

A single-center randomized controlled trial was conducted in a tertiary care hospital in The Netherlands. Participants were included during the second stage of labor in case of an intermediary or abnormal FHR pattern. Patients were randomized to receive either 100% oxygen at 10L/min until delivery, or conventional care without additional oxygen. The primary outcome was the change in FHR pattern before and after the onset of the study, measured as the change in depth and duration of FHR decelerations. Secondary outcome measures were features based on phase-rectified signal averaging (PRSA), baseline assignability, and deceleration characteristics of the FHR pattern.

Main results

Between March 2016 and April 2018, 117 women were included. The FHR pattern could be analyzed for 71 participants, the other 46 women delivered before the end of the post time-frame. A 2.3% reduction in depth and duration of FHR decelerations was found after maternal hyperoxygenation, compared to a 10% increase in the control group (p=0.24). Maternal hyperoxygenation had a significantly positive effect on PRSA metrics, with a decrease in PRSA-acceleration capacity (p=0.03) and PRSA-deceleration capacity (p=0.02) in the intervention group compared to the control group.

Significance

The difference in depth and duration of decelerations after the start of the study was not significantly different between both study groups. A statistically significant positive effect on PRSA-deceleration capacity and PRSA-acceleration capacity was found after maternal hyperoxygenation, which might be associated with a positive effect on neonatal outcome.

Trial registration

The study was registered in the EudraCT database (2015-001654-15) on April 3rd, 2015 and in the Dutch Trial Register (NTR5461) on October 20th, 2015. URL: https://www.clinicaltrialsregister.eu/ctr-search/search?query=2015-001654-15 and http://www.trialregister.nl/trialreg/admin/rctview.asp?TC=5461. Date of initial participant enrollment: March 2nd, 2016.

INTRODUCTION

During labor, contractions cause alterations in intrauterine pressure and may thereby lead to intermittent interruption of uterine and umbilical blood flow.^{1,2} This may decrease oxygen delivery to the fetus, leading to reduced fetal oxygenation, and eventually to fetal hypoxia and acidosis.^{1,2} Timely clinical interventions, such as intrauterine resuscitation, may avoid these detrimental effects.^{3,4}

In some countries, maternal hyperoxygenation is commonly used to improve fetal oxygenation status through increased oxygen delivery to the placenta.³⁻⁸Indeed, Vasicka et al. showed increased partial oxygen pressure (pO_2) levels in all placental and fetal compartments already after five minutes.⁹ However, concerns are raised about potential side effects of maternal hyperoxygenation, such as an increase in free oxygen radicals and a decrease in arterial cord pH.¹⁰⁻¹³ Unfortunately, the number of clinical studies on the effects of maternal hyperoxygenation in case of fetal distress is limited and their results are inconclusive.¹⁴⁻²⁰ Hence, several reviews have urged for a randomized controlled trial (RCT) to study the effects of maternal hyperoxygenation during the second stage of labor in the presence of fetal distress.^{3.5.7}

Therefore, we conducted a RCT, the INTEREST O2 study, in which maternal hyperoxygenation was compared to conventional care in case of suspected fetal distress during the second stage of labor.²¹ The maternal and neonatal outcomes from this study have been published previously.²² No harmful side-effects of maternal hyperoxygenation were observed, especially no reduction in Apgar Score or arterial umbilical cord blood pH and no increase in free oxygen radical activity. Moreover, a significantly positive effect on change in International Federation of Gynecology and Obstetrics (FIGO) classification was found following maternal hyperoxygenation.²² Although such visual FHR analysis is one of the most used methods to evaluate FHR, it is known to have a large intra- and inter-observer variability.^{23,24} Recent advances in computerized analysis of FHR could provide additional insights in the physiological changes caused by maternal hyperoxygenation.

The detailed analyses of FHR features presented in this paper are based on the link between FHR and the autonomic nervous system (ANS) in response to reduced fetal oxygenation.^{1,25} FHR decelerations are a well-known fetal adaptation to hypoxia to reduce myocardial work and oxygen requirements.^{2,26} We will evaluate the depth and duration of FHR decelerations as these are related broadly to the severity of the fetal hypoxia.^{2,27}

Furthermore, changes in fetal well-being can also be measured using phase-rectified signal averaging (PRSA). This innovative method of FHR analysis can quantify the acceleration and deceleration capacity of FHR, thereby providing insights into the compensatory activation of the ANS in case of fetal distress.^{25,28}

Finally, we will also evaluate other important FHR features mentioned in literature as being related to unfavorable neonatal outcomes, namely periods with unstable baseline, decelerations in combination with tachycardia or bradycardia, and decelerations with loss of internal variability.^{2,27,36,37,28-35}

METHODS

Study design

The INTEREST O2 study, a single-center RCT, was conducted at a tertiary care teaching hospital in the Netherlands. The study protocol has been published previously.²¹ Women, aged \geq 18 years, with a term non-anomalous singleton in cephalic presentation and intended vaginal delivery, were considered eligible for this study. Exclusion criteria were mainly based on the risk of increased free oxygen radical activity: maternal fever during labor with the need of antibiotic therapy, the use of tobacco, recreational drugs, or alcohol during pregnancy, pulmonary disease requiring medication, pre-existing cardiac disease, diabetes, hyperthyroidism, anemia (hemoglobin <10.5 g/dL) or recent use of any of the following medications: anti-hypertensives, magnesium sulfate, corticosteroids, amiodarone, adriamycin, bleomycin, actinomycin, menadione, opioids, promazine, thioridazine, or chloroquine. Eligible subjects were only randomized in the study if they presented with suspected fetal distress during the second stage of labor. Suspected fetal distress was defined as an intermediary or abnormal FHR pattern according to the modified FIGO classification, see Figure 1.^{27.38}

	Baseline heart frequency	FHR variability and reactivity	FHR decelerations
Normal FHR	• 110–150 bpm	Accelerations 5–25 bpm	Early uniform decelerations Uncomplicated variable decelerations (loss of < 60 beats)
Intermediary FHR	100–110 bpm 150–170 bpm Short bradycardia episode, defined as: < 100 bpm for > 3 minutes < 80 bpm for > 2 minutes	Saltatory pattern (> 25 bpm) < 5 bpm for > 40 minutes	Uncomplicated variable decelerations (loss of > 60 beats)
A combination of two	or several intermediary obser	vations will result in an abnorm	hal FHR classification
Abnormal FHR	 > 170 bpm Persistent bradycardia < 100 bpm for > 10 minutes < 80 bpm for > 3 minutes (without an increasing tendency) 	 < 5 bpm for > 60 minutes Sinusoidal pattern 	Complicated variable decelerations with a duration of > 60 seconds Repeated late uniform decelerations
Pre-terminal FHR	Total lack of variability (< 2	bpm) and reactivity with or wi	thout decelerations or bradycardia

Figure 1. Classification of FHR according to modified FIGO criteria Bpm = beats per minute. FHR = fetal heart rate.

Intervention and randomization

All women eligible to be included in this study received antepartum oral and written information about the study when they visited the outpatient clinic or when they were admitted to the delivery ward. After informed consent was obtained, two sealed opaque envelopes with treatment allocation were assigned per patient during the first stage of labor, one in case of future intermediary FHR and one in case of future abnormal FHR. The allocation sequence was computer-generated with the use of random blocks of four or six patients. Stratification was applied for intermediary or abnormal FHR.

Only in case of suspected fetal distress during the second stage of labor, women were randomized in the study by opening the appropriate envelope (i.e. intermediary or abnormal FHR). Unopened envelopes were reused.

Patients were randomized to receive either conventional care without additional oxygen (control group), or maternal hyperoxygenation with 100% oxygen at 10 L/min via a non-rebreathing mask until delivery (intervention group). Preferably, other intrauterine resuscitation techniques, such as maternal positioning, admission of tocolytic drugs, or adjustment of oxytocin dosage, were not initiated within the first 10 minutes after randomization. However, the obstetric staff could overrule the study protocol at any time if deemed necessary.

Primary outcome

The primary outcome was the change in FHR pattern after the onset of the study protocol, measured as the percentage change in the depth and duration of FHR deceleration in the intervention group compared to the control group. To be able to analyze decelerations of the FHR, visually or through computerized analysis, the baseline of the FHR needs to be identified. Subsequently, the depth and duration of FHR decelerations can be calculated.

The change in FHR was first analyzed per patient, each with a "pre-study" and "poststudy" 10-min time-frame with respect to the start of the protocol as shown in Figure 2. The FHR time-frames of interest correspond to the 10 minutes before start of the study protocol (pre-study time-frame) and the period 5 to 15 minutes after the start of the study protocol (post-study time-frame). The difference between the pre and post scores was provided as a single score per parameter per patient. Results were subsequently compared between the intervention and the control group.

By definition, women who delivered within 15 minutes after the start of the study did not have complete time-sets of FHR in the post time-frame, therefore changes in FHR pattern could not be analyzed in these cases.



Figure 2. The 10-minute time-frames of interest, pre and post-initiation of the study protocol. Patients in the intervention group receive standard care with the start of oxygen administration immediately after time=0, while patients from the control group will continue to receive standard care without oxygen.

FHR-baseline assignment

The baseline of the FHR is defined by the FIGO as "the mean level of the most horizontal and least oscillatory FHR segment".²⁷ A FHR baseline can sometimes be unassignable, due to poor signal quality or when FHR is unstable.

To determine the baseline, two methods were used, namely visual evaluation and computerized analysis. Visual evaluation of the baseline was done by a team consisting of a consultant-obstetrician (NvO) and an obstetric resident (LB), both blinded for treatment allocation. Both clinicians independently visually analyzed whether a FHR baseline was assignable for each time-frame. For assigning the FHR baseline, they could use both the FHR pattern and the uterine contraction pattern. Any disagreement was resolved by consensus. In the event no baseline could be assigned in the 10-minute time-frame of interest, the time-frame of interest was extended with the 10 minutes preceding the original time-frame.

The visual analysis was then used to validate a preexisting computerized algorithm for baseline evaluation, henceforth referred to as the OxSys algorithm.^{29,30,39} The uterine contraction pattern is not taken into account in the FHR baseline evaluation by this algorithm.

The results of the visual analysis by clinicians were used as the gold standard for baseline assignment when comparing the two methods. An interobserver agreement of $>0.60^{40}$ was set as a cut-off to determine whether the Oxsys algorithm will be used for assignment of baseline values for further calculations. In case of a lower interobserver agreement, baseline values determined from visual inspections will be used for further calculations.

Primary outcome of FHR decelerations: deceleration surface

An integral measure of the depth and duration of FHR decelerations was assessed by calculating the area under the curve with respect to the baseline (Figure 3). This so-called deceleration surface was determined for both the pre-study and post-study 10-minute window (Figure 2), also if the corresponding baseline was determined from a 20-minute window. FHR deceleration surface was not analyzed for FHR baseline values ≤110bpm, as this is by definition a bradycardia.

Per subject, the normalized percentage change in deceleration surface between the preand post-study window was calculated. Subsequently, these changes were compared between the intervention and control group.



Figure 3. An example to showcase the deceleration surface (in pink), a single integrated measure to capture the depth and duration of FHR decelerations.

Secondary outcomes

Unstable baseline

The instability of FHR may indicate compromised fetal condition.³⁰ Therefore, all cases in which a baseline could not be assigned in the original 10-minute time-frame due to unstable FHR were compared between the intervention and the control group.

FHR variability: Phase-rectified signal averaging

PRSA is a non-traditional computerized measure to characterize heart rate variability (HRV).⁴¹ More details about this signal processing technique can be found in the article by Bauer et al.⁴¹ PRSA is based on two parameters, *T* determining the periods of average increase or decrease in the FHR (around the anchor points) that count as accelerations and decelerations respectively, and *L* that represents the period over which the data around the anchor points are averaged.⁴¹ For these analyses, we used *T*= 20 anchor points (corresponds to 5 seconds) and *L*= 100 anchor points (corresponds to 25 seconds). The acceleration- and deceleration-related modulations identified through PRSA can be characterized by specific features to quantitatively study changes in autonomic regulation.^{41,42} Specifically, the average acceleration capacity (PRSA-AAC) can be used to characterize the acceleration capacity of the fetal heart.⁴¹ Similarly, the average deceleration capacity (PRSA-ADC) can be used to characterize the deceleration capacity of the fetal heart.⁴³

Characteristics of FHR decelerations

The following two aspects of FHR decelerations were analyzed visually by the same two clinicians (NvO and LB);

- Presence of decelerations with loss of internal variability (beat-to-beat variability < 5 bpm)
- Presence of decelerations in combination with tachycardia (defined as >160 bpm) or bradycardia (defined as < 110 bpm)

Disagreement was resolved by consensus.

Traditional heart rate variability metrics

In addition to the primary and secondary outcomes described in the study protocol,²¹ we also analyzed the following traditional HRV metrics; short term variability (STV), standard deviation of the NN interval (SDNN), and the root mean squared successive differences of NN intervals (RMSSD). To calculate STV, the algorithm described by Dawes/Redman was used.⁴⁴

Signal processing

During labor, the FHR was continuously recorded with either Doppler ultrasound or fetal scalp electrode using Philips Avalon FM30 (Philips Healthcare, Best, The Netherlands). To perform the computerized FHR analysis, the digital FHR tracings were extracted from the electronic patient file (HIX, ChipSoft, Amsterdam, The Netherlands), anonymized and subsequently analyzed offline using MATLAB 2018 software (MathWorks Inc., Natick, MA, USA). For all computerized analysis, the FHR data, acquired at a sampling rate of 4Hz, were preprocessed. First, to remove potential measurement artifacts, all FHR values above 250 and below 20 beats per minute (bpm), as well as those FHR values that differed by more than 25 bpm from the preceding value were filtered out and considered

to be a missing value. Next, only those time-frames that had less than 50% of missing data were retained for further analysis. Finally, all missing data were interpolated by employing shape-preserving, cubic spline interpolation.

Intention-to-treat and per-protocol analyses

The primary analyses were intention-to-treat. In some cases, oxygen may not have been applied according to the study protocol, due to practical concerns such as fast progression of labor. To anticipate for these cases, additional per-protocol analyses were performed. For the per-protocol analyses, all women receiving <5 minutes of oxygen were excluded, since it takes 5 minutes of maternal hyperoxygenation for maternal pO2 to increase to a maximum.⁹ Furthermore, unjust inclusions were also excluded from the per-protocol analyses.

Sample size calculation

A sample size calculation was performed as described previously.^{21,45} To provide a power of 90% at a two-sided significance level of 0.05, 96 women (48 in each group) were required. To accommodate for 20% missing data, 116 participants were planned to be included in the study.

Statistical analyses of outcome parameters

IBM SPSS Statistics software (version 25; IBM, Armonk, NY, USA) was used for all statistical analyses. In case of non-normally distributed outcomes, continuous variables were analyzed with a Mann-Whitney U test for differences between the intervention and control group and a Wilcoxon matched-pairs test for changes within the same participant. In case of normal distribution, continuous data were analyzed using independent samples t-tests (two-tailed) for differences between the intervention and control group, and paired t-tests were used for changes within the same participant. For categorical variables, the χ^2 test or Fisher's exact test was used depending on the expected number of observations per category. P<0.05 for a two-tailed test was considered significant. For calculation of the interobserver agreement of baseline assignability (i.e. whether a baseline can be assigned) between the visual analysis of the clinicians and the computerized analysis, Cohen's kappa was used.

All analyses were performed for the total study population (i.e. the combined group of intermediary and abnormal FHR), as well as for the subgroups of *intermediary FHR*, *abnormal FHR*, and *small for gestational age neonates* (birthweight <10th percentile).

When missing data exceeds the 20% accommodation window of the study design, additional or sensitivity statistical analyses will be performed to evaluate risk factor imbalances between treatment groups (violation of the randomization due to missingness) and evaluate the sensitivity of these potential imbalances.

To analyze whether possible risk factors of missingness were likely to have affected the primary outcome, the Mann-Whitney U statistic will be applied to the following parameters to compare women with and without missing FHR time-frames (i.e. missing data for FHR analyses): FIGO classification before start of the study, gestational age, birth weight, Apgar Score at one and five minutes, arterial umbilical cord gas parameters (pH, Base Excess, pCO_2 and Malondialdehyde) episiotomy on fetal indication, assisted delivery, maternal age, maternal body mass index, and multiparity. In the event of imbalances, a propensity score matching approach will be used to correct for the observed imbalances using a SAS algorithm for matching (PSMatch_Multi).

Ethical approval

We conducted this study in accordance with the Declaration of Helsinki. The Central Committee on Research Involving Human Subjects approved this study (protocol number NL53018.000.15). The trial was registered at the EudraCT database (2015-001654-15) and at the Dutch Trial Register (NTR5461).

RESULTS

Between March 2016 and April 2018, a total of 376 women gave informed consent for the study during outpatient visits or during the first stage of labor (CONSORT flow diagram, Figure 4). During the second stage of labor, 117 women were randomized due to an *intermediary* or *abnormal* FHR pattern. Of these, 57 women were assigned to the intervention group and 60 women were assigned to the control group. In total, 46 women delivered within 15 minutes after the start of the study. Hence, the complete FHR timeframes suitable for FHR analyses were available for the remaining 71 women (61%). Of these, 36 women (51%) originated from the intervention group and 35 women (49%) from the control group.

The baseline characteristics of the included women are presented in Table 1. Groups were comparable regarding baseline characteristics, including gravidity and parity. Within the first 10 minutes after randomization, 7 women (9.9%) received additional intrauterine resuscitation, two (5.6%) in the intervention group and five (14.3%) in the control group (p=0.26). These intrauterine resuscitation techniques included discontinuation of pushing, adjustment of the dosage of oxytocin infusion, and maternal repositioning.



§ A total of 21 women from the intervention group and 25 women from the control group gave birth within 15 minutes after start of the study and by definition these women did not have complete time-sets of FHR in the post timeframe, therefore change in FHR pattern could not be analyzed in these cases.

* Four women had signs of infection and were treated with antibiotics, two women smoked during pregnancy, one woman had diabetes, one woman delivered prematurely, one fetus had congenital abnormalities (cheilognathopalatoschisis), and there was one case of fetal bradycardia prior to the start of the study (FHR baseline of 105bpm).

** Three women had signs of infection and were treated with antibiotics, two women smoked during pregnancy, and one woman delivered at 42 weeks of gestation.

¶ In the intervention group, the six women with oxygen admission <5min and the 10 women with exclusion criteria were excluded in the per-protocol analysis.

 ${\sf Y}$ In the control group, the six women with exclusion criteria were excluded from the per-protocol analysis

Figure 4. Trial flow diagram.

A total of 117 patients were allocated to either the intervention or control group. For the fetal heart rate (FHR) analysis, 71 women were eligible based on the availability of complete FHR time-frames. Both an intention-to-treat and per-protocol analysis were performed. FHR = fetal heart rate

	Intervention group	Control group	Р
	n=36	n=35	
Maternal age (years)	31±3	32±4	0.42
Gestational age (days)	280±9	278±9	0.27
Gravidity >1	22 (39%)	32 (53%)	0.11
Parity ≥1	8 (22%)	11 (31%)	0.38
BMI (kg/m2)	23 (20-27)	24 (21-27)	0.10
Fetal sex male	21 (58%)	15 (43%)	0.19
Birth weight (grams)	3610±623	3431±489	0.18
SGA	6 (17%)	3 (9%)	0.48
Abnormal FHR pattern	20 (56%)	14 (40%)	0.19

Data are mean±SD, median (IQR) or n (%). BMI = body-mass index, SGA = small for gestational age, FHR= fetal heart rate

Analyses of the FHR

Baseline assignability

Comparing the baseline assignability of the *computerized analysis* with the gold standard of *visual analysis* (i.e. whether a baseline could be assigned or not), a poor interobserver agreement was found, with kappa values ranging between 0.13 and 0.60 (Supplementary Table 1). Since this interobserver variability of baseline assignability was <0.60, all further analyses were based on the baseline value as assigned by the clinicians. For 27 women, a 20-minute time-frame was needed to be able to determine a baseline (Supplementary Table 1).

Depth and duration of decelerations

In one subject, the FHR baseline was <110 bpm and was therefore excluded from deceleration surface analysis. The deceleration surface in the pre-study and post-study time-frames are presented in Table 2 for both the intervention and control group. Figure 5 shows an example of the changes in FHR traces provoked by maternal hyperoxygenation.

The change in depth and duration of decelerations between the pre-study and poststudy time-frame are presented in Table 3. The difference was not significant (p=0.24) between the maternal hyperoxygenation group (-2.3%) and the control group (+10%).

Unstable baseline

The change in the number of unstable baselines was not significantly different (p=0.48) between the intervention (+9%) and control group (-6%, Supplementary Table 1).



Figure 5. An example of the changes in the depth and duration of FHR decelerations after start of the study provoked by maternal hyperoxygenation in the intervention group (39% decrease in deceleration surface following maternal hyperoxygenation).

PRSA-based parameters

The PRSA values of the pre and post-study time-frames are presented in Table 2, whereas the differences after start of the study are presented in Table 3. The averaged PRSA measures are visualized in Figure 6.

Prior to the start of the study, the PRSA-ADC was similar between the intervention and the control group (p=0.23). After start of the study, the intervention group showed a 3% decrease of PRSA-ADC, compared to before the onset of the study. The control group showed an increase of 20% in PRSA-ADC after the onset of the study. This difference was statistically significant (p=0.02).

In the pre-study time-frame, PRSA-AAC was significantly higher in the intervention group than in the control group (7.01 vs 4.52, p<0.01). After the start of the study protocol, the intervention group showed a 4% decrease in PRSA-AAC between pre and post. The control group showed an increase in PRSA-AAC of 37% between pre and post. This difference was statistically significant (p=0.03).



Figure 6. Averaged PRSA measures for the intervention and the control group.

	Pre-study time-frame			Post-study time-frame		
	Intervention	Control	p-value	Intervention	Control	p-value
	N=36	N=35		N=36	N=35	
Deceleration surface	17.42 [10.65 to 20.65]	11.23 [6.70 to 25.05]	0.28	13.97 [8.94 to 21.86]	17.84 [10.67 to 23.27]	0.16
PRSA – ADC (FHR/s)	-7.60 [-9.17 to -6.00]	-6.55 [-7.75 to -4.57]	0.23	-6.92 [-8.94 to -5.53]	-7.60 [-9.67 to 5.19]	0.57
PRSA – AAC (FHR/s)	7.01 [4.91 to 8.23]	4.52 [3.88 to 7.25]	0.005	5.99 [4.09 to 8.60]	6.30 [4.60 to 8.75]	0.71
STV (ms)	2.41 ± 1.06	2.07 ± 0.93	0.16	2.44 ± 1.33	2.38 ± 1.07	0.84
SDNN (ms)	88.83 [59.47 to 108.68]	78.49 [41.60 to 106.64]	0.22	91.16±51.31	96.91 ± 46.33	0.62
RMSSD (ms)	9.85 [6.42 to 12.36]	8.20 [5.99 to 11.49]	0.32	9.86 ± 4.96	9.57 ± 3.89	0.78
All calculations are based AAC; average accelerati root mean squared succ	I on the results of the adjusted on capacity. ADC, average de essive differences of NN inte	d algorithm. Data are mean± 4 sceleration capacity. STV; shc srvals.	standard de ort term va	eviation or median [IQR].P riability. SDNN; standard	SSA; Phase-rectified signal deviation of the NN interv	averaging. II. RMSSD;
Table 3. Changes in com	puterized FHR before and af	fter randomization				

	Intervention (n=36)	Control (n=35)	٩
Δ Deceleration surface in %	-2.31 [-39.46 to 68.30]	10.00[-14.97 to 127.85]	0.24
△ PRSA – ADC in %	-2.75 [-27.94 to 35.69]	19.75 [-9.24 to 67.11]	0.02
Δ PRSA – AAC in %	-3.93 [-39.56 to 30.75]	37.14 [-20.63 to 93.73]	0.03
∆ STV in %	-2.18 [-25.55 to 26.55]	13.11 [-24.56 to 53.25]	0.27
Δ SDNN in %	0.78 [-31.33 to 43.59]	11.25 [-9.15 to 61.29]	0.16
A RMSSD in %	06 [-27.75 to 32.87]	3.41 [-12.75 to 40.67]	0.33

median [IQR] PRSA; SDNN; standard iability. are time-frame). Data vari term < short t STV; pre t deceleration capacity. to with respect Phase-rectified signal averaging. AAC; average acceleration capacity. ADC; average deceleration deviation of the NN interval. RMSSD; root mean squared successive differences of NN intervals. pre time-frame minus -frame r percentage difference (post timepresented as are Changes

Traditional heart rate variability metrics

The results of the analyses of STV, SDNN, and RMSSD in the pre- and post-study timeframes are presented in Tables 2, whereas the differences in these parameters after the start of the study are presented in Table 3. No significant differences were found in STV, SDNN, or RMSSD between the intervention and the control group.

Characterization of decelerations

The results of the analyses of the characteristics of FHR decelerations are presented in Supplementary Tables 2 and 3. No statistically significant differences were seen between the intervention and the control group in decelerations with loss of internal variability or in decelerations in combination with tachycardia (p=0.67 and p=0.11, respectively).

Subgroup analysis

The results of the analysis on the subgroups *intermediary FHR, abnormal FHR*, and *small for gestational age* neonates are presented in Table 4. The change in deceleration surface after start of the study protocol was not significant in either of the subgroups.

Table 4. Changes in computerized FHR before and after randomization, subgroup analyses

	Intervention	Control	р
Δ Deceleration surface in % Abnormal FHR [n= 33] Intermediary FHR [n= 37] SGA [n= 9]	-7.09 [-47.90 to 25.23] 67.66 [-18.58 to 103.03] -19.10 [-83.13 to 55.68]	-0.38 [-30.89 to 90.71] 21.46 [-12.19 to 130.67] -21.85 [-24.63 to 447.85]	0.34 0.94 0.55
∆ PRSA – ADC in % Abnormal FHR [n= 34] Intermediary FHR [n= 37] SGA [n= 9]	-3.75 [-31.21 to 26.44] 4.68 [-23.13 to 42.84] -20.96 [-70.43 to 14.38]	29.92 [3.63 to 63.68] 17.08 [-11.32 to 70.70] 26.61 [7.92 to 66.41]	0.06 0.21 0.17
∆ PRSA – AAC in % Abnormal FHR [n= 34] Intermediary FHR [n= 37] SGA [n= 9]	-5.06 [-34.26 to 23.46] -3.93 [-26.31 to 50.42] -38.85 [-69.71 to 81.07]	37.58 [4.34 to 108.98] 30.33 [-28.93 to 92.46] 37.14 [-20.63 to 91.20]	0.02 0.44 0.38

Changes are presented as percentage difference (post time-frame minus pre time-frame with respect to pre time-frame). Data are median [IQR] PRSA; Phase-rectified signal averaging. AAC; average acceleration capacity. ADC; average deceleration capacity.

Per-protocol analysis

A per-protocol analysis was performed in which 6 women from the intervention group were excluded who received oxygen administration for <5 minutes. No women from the control group received supplementary oxygen. In addition, 16 women were excluded in the per-protocol analyses because they were included despite the presence of exclusion criteria. The per-protocol analysis therefore included a total of 49 patients of whom 20

Table 2. Measurements of FHR for the intervention and control group

women were allocated to receive oxygen and 29 women were allocated to the control group.(Figure 4).

All results of the per-protocol analyses were in line with the results of the intention-to-treat analyses (Table 5).

Table 5. Per-protocol analyses, changes in computerized FHR before and after randomization

	Intervention (n=21)	Control (n=29)	р
Δ Deceleration surface in %	-4.48 [-44.22 to 62.64]	9.19 [-14.02 to 128.91]	0.18
Δ PRSA – ADC in %	-0.46 [-31.21 to 26.44]	19.75 [-2.08 to 76.54]	0.02
Δ PRSA – AAC in %	-3.25 [-33.72 to 39.97]	41.20 [-17.81 to 99.10]	0.048

Changes are presented as percentage difference (post time-frame minus pre time-frame with respect to pre time-frame). Data are median [IQR] PRSA; Phase-rectified signal averaging. AAC; average acceleration capacity. ADC; average deceleration capacity.

Additional or sensitivity analyses

As the fraction of missing data exceeded the accommodated 20%, sensitivity analyses were performed. As shown in Table 2 and Supplementary Table 1, only parameter PRSA-AAC showed an imbalance between the intervention and the control group in the prestudy time-frame. Hence, a propensity score matched-pair analysis was performed for this parameter. Using a caliper of 1.3 (approximately 50% of the standard deviation and 10% of the range in PRSA-AAC pre-study values) 28 pairs were obtained.

Using a mixed effects model on the logarithmically transformed pre- and post-study differences, using the pair as random effect and the intervention as fixed effect, we obtain a p-value for the intervention equal to p=0.007, thus confirming the significant difference in the change of PRSA-AAC between the intervention and the control group after correcting for imbalances.

In the group with missing data for FHR analyses, multiparity (p<0.001) and assisted delivery (p=0.02) were significantly higher compared to the group without missing FHR time-frames, all other parameters did not differ between the two groups.

Both assisted delivery and multiparity were not associated with the difference in depth and duration of decelerations after start of the study (p=0.38 and 0.89, respectively).

DISCUSSION

Principal findings

In this study, the effect of maternal hyperoxygenation during the second stage of term labor was analyzed using specific FHR features that are associated with poor neonatal outcome. Differences in depth and duration of decelerations were seen between maternal hyperoxygenation (-2.3% deceleration surface) compared to the control group (+10% deceleration surface) in favor of maternal hyperoxygenation. These differences were not statistically significant (p=0.24). This may be (partly) due to the higher than accommodated fraction of missing data.

Heart rate variability metrics

The differences in STV, SDNN, and RMSSD were not statistically significant between the intervention and the control group. Caution should be applied when interpreting these parameters, due to the autocorrelation function and zero-order interpolation of the Philips Avalon FM30 monitoring system which calculates an *average* heart rate in beats per minute.⁴⁶ The device outputs the equally sampled FHR at 4Hz. When calculated at this low sampling rate, traditional HRV metrics such as STV, SDNN and RMSSD might remain unchanged in the event of fetal acidemia.⁴⁷⁻⁴⁹

In contrast to other measures of FHR variability, PRSA-based parameters can be used at a sampling rate of 4Hz to calculate both the variation in FHR as well as the speed of those changes.⁴¹ Van Scheepen et al. showed that PRSA-metrics measured at 4Hz have a high correlation with signals measured with the use of fetal ECG (1000Hz), especially when T is larger than 15, which is the case in our study.⁴⁶ Moreover, by eliminating artifacts and signal perturbations in FHR, the PRSA-based approach can be used to separately quantify the acceleration and deceleration-related components in the underlying signal.^{28,37,41} Fetal compensatory mechanisms to repetitive transient hypoxic stress during labor can therefore be measured using PRSA-based parameters, as the compensatory activation of the autonomic nervous system is reflected by an increase in PRSA-AAC and PRSA-ADC.^{25,50}

The difference in PRSA-based parameters (i.e. the delta values) in the two study arms was found to be statistically significant. The PRSA-ADC showed a 3% reduction in the intervention group after start of maternal hyperoxygenation, whereas a 20% increase was found in the control group after start of the study (p=0.02). For PRSA-AAC, a 4% reduction was seen in the intervention group compared to a 37% increase in the control group (p=0.03). Since the primary outcome was the relative change, and the study was powered for this outcome, our study may lack power with the current number of participants to demonstrate a difference between treatment groups for the post study data.

Previously, several studies have reported on PRSA test characteristics to predict fetal acidemia.^{25,28,37,50} Rivolta et al. presented an in-vivo sheep model, where fetuses were exposed to repetitive umbilical cord occlusions. Both PRSA-AAC and PRSA-ADC progressively increased with phases of acute hypoxic-acidemia.⁵⁰ In fact, Lobmaier et al. showed that antepartum measured PRSA-AAC is a better predictor of short-term perinatal outcome than short-term variability of FHR in severely growth-restricted fetuses.³⁷ Weyrich et al. and Georgieva et al. studied the prognostic value of PRSA-ADC in intrapartum FHR measurements.^{25,28} Using ROC-analyses, they both showed that increasing values of PRSA-DC were related to low pH at birth, and that PRSA-ADC appears to predict acidemia more accurately than short-term variability.^{25,28} Therefore, the significant difference in PRSA-based parameters in favor of hyperoxygenation found in our study may positively reflect on neonatal outcome.

Baseline assignment

Various computerized algorithms are developed to analyze FHR signals to overcome the subjectivity of visual FHR analysis.⁵¹ Several of these computerized algorithms have a high agreement and reliability.⁵¹

As the time-frames of interest in our study are relatively short for baseline determination (10 minutes), we chose to implement the OxSys algorithm, which had previously been validated for short 15-minute segments of FHR during labor.²⁹ However, since the interobserver agreement found in our study between the OxSys algorithm and *visual analysis* was poor (kappa < 0.60), the baseline assignment by this computerized baseline algorithm could not be used.

This prompted us to evaluate whether the Omniview-SisPorto system (Speculum, Lisbon, Portugal),⁵² which is based on the FIGO guidelines, was able to determine the FHR baseline in the 10-minute segments. Unfortunately, we experienced similar difficulties with this algorithm for detecting the baseline. The inability to assign the FHR baseline is a common problem of computerized FHR analysis in the intrapartum period, especially during the second stage of labor, due to greater signal instability and the frequent occurrence of signal loss and artifacts.^{27,29,30,53} It is likely that the limited duration of the time-frames in our study has further reduced the applicability of the computerized algorithms. However, for the purpose of this study, it was considered unethical to withhold other intrauterine resuscitation techniques longer than 10-minutes in the event of suspected fetal distress. Therefore, we chose not the change the time-frames and based all further analyses on the baseline value as assigned by the clinicians. Our results underline the need for a more robust method to analyze shorter segments of FHR tracings.

Clinical and research implications

Up to now, mostly small, non-randomized studies have investigated the effect of maternal hyperoxygenation in the presence of fetal distress during the second stage of labor. Those studies showed a positive effect of maternal hyperoxygenation on FHR pattern and fetal scalp pH.^{14,15,18,19,54,55} Only one other RCT studying maternal hyperoxygenation has been performed in patients with suspected fetal distress. In this study by Raghuraman et al. with a noninferiority design,²⁰ women with >6cm cervical dilatation (stage 1 and 2 of labor) and suspected fetal distress were randomized to breathing room air or to maternal hyperoxygenation until birth. They showed no significant difference in umbilical artery lactate between both groups, which was their primary outcome.

The primary outcome of the current study was the change of FHR pattern. This outcome was explicitly chosen as the primary outcome, rather than e.g. neonatal morbidity, as this would require a sample size of over 10,000 women.⁵ Therefore, in the current study, FHR pattern was used as a surrogate marker with fewer patients needed because a non-reassuring pattern of FHR is considered to be indicative of potential fetal hypoxia.⁵⁶⁻⁵⁹ We recognize that FHR monitoring has a high sensitivity but a relatively low specificity for fetal acidosis.^{27,60} Therefore we also evaluated the change in FHR by using specific FHR features that are related to unfavorable neonatal outcome.^{2,27,36,37,28-35}

The previously reported significant positive effects of maternal hyperoxygenation on the FIGO classification during the second stage of labor are concordant with our current results.²² Furthermore, in our previous study, no negative side effects were found, especially no reduction in arterial umbilical cord blood pH or increase in free oxygen radical activity.²² Moreover, in the group with abnormal FHR, a significant reduction was seen in the rate of episiotomies on fetal indication in the intervention group, likely due to the amelioration of the FHR pattern.

Although the difference in deceleration surface is not statistically significant in the current study, possibly due to the higher than accommodated fraction of missing data, the positive effect of maternal hyperoxygenation might be significant in a larger cohort. As maternal hyperoxygenation is shown to have no harmful side-effects and has a significantly positive effect on FIGO classification and PRSA-based parameters. Therefore, we recommend to study the effect of maternal hyperoxygenation in a large study powered for neonatal outcome.

Our results suggest that the use of maternal hyperoxygenation in case of suspected fetal distress does not have to be discouraged, which is in line with the guideline of the American College of Obstetrics and Gynecology.⁴

Strengths and limitations

This study is the first RCT investigating the effect of maternal hyperoxygenation on FHR in the presence of suspected fetal distress during the second stage of labor. By limiting co-interventions in the first 10 minutes after initiation of the study protocol, the pure effect of maternal hyperoxygenation on FHR could be studied more accurately. It was considered unethical to withhold additional interventions for a longer time, due to the risk of prolonged fetal hypoxia. Moreover, the obstetric staff could overrule the study protocol at any time if deemed necessary, which was done in seven women (9.9%).

By using several computerized methods to evaluate the effect of maternal hyperoxygenation on FHR, our analysis provides additional knowledge on the physiological changes caused by maternal hyperoxygenation. In contrast to visual analyses of FHR, these computerized analyses were not influenced by intra- and interobserver variability.

However, practical issues led to some limitations. In the sample size calculation, we accommodated for 20% missing data. Since 46 women (39%) gave birth before the end of the study time-frame, FHR analyses could only be performed in 71 women (61%). Therefore, we performed sensitivity analyses to address the potential bias caused by missing data. As the matched-pair analysis showed a similar result as the primary analysis, we believe the difference in PRSA-AAC prior to the start of the study does not influence the study results on which our conclusions are based.

Unfortunately, due to organizational challenges, it was not possible to blind patients and the medical care team to the treatment allocation. Yet, the risk of bias was minimized due to, on the one hand, a blinded visual analysis of FHR tracings by the investigators, and on the other hand, a computerized calculation of deceleration surface and PRSA-based parameters.

CONCLUSION

In patients with maternal hyperoxygenation during the second stage of labor in the presence of fetal distress, the difference in depth and duration of decelerations after the start of the study was not significantly different compared to the control group. However, a significantly positive effect on PRSA-ADC and PRSA-ACC was found after maternal hyperoxygenation. Since these effects may be associated with an improvement of neonatal outcome, we now deem it justified to initiate a study powered for neonatal outcome.

	Pre-study time-fr	ame		Post-study tin	ne-frame	
	Intervention N=36	Control N=35	p-value	Intervention N=36	Control N=35	p-value
	Original OxSys algo	orithm		2	8	
Computerized assignability of baseline	30 (83%)	26 (74%)	0.35	23 (64%)	26 (74%)	0.34
Visual assignability of baseline	26 (72%)	30 (86%)	0.16	29(81%)	28 (80%)	0.95
Agreement computerized and visual assignability baseline	Kappa= 0.21	Kappa= 0.13	~	Kappa= 0.60	Kappa = 0.19	
	Original OxSys algo	orithm with addit	ion baseline	<pre></pre>	assignable	
	Intervention	Control	p-value	Intervention	Control	p-value
	N=36	N=35		N=36	N=35	
Computerized assignability of baseline	27 (75%)	26 (74%)	0.95	23 (64%)	23 (66%)	0.87
Visual assignability of baseline	26 (72%)	30 (86%)	0.16	29(81%)	28 (80%)	0.95
Agreement computerized and visual assignability baseline	Kappa= 0.21	Kappa= 0.13	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	Kappa= 0.60	Kappa = 0.37	
	If needed, 20 minu	te time-frame ¶				
	Intervention	Control	p-value	Intervention	Control	p-value
	N=36	N=35		N=36	N=35	
Computerized assignability of baseline	32 (89%)	30 (86%)	0.74	28 (78%)	26 (74%)	0.73
Visual assignability of baseline	36 (100%)	35 (100%)	1.00	36 (100%)	35 (100%)	
Agreement computerized and visual assignability baseline	Kappa =	Kappa =		Kappa=	Kappa=	
	incalculable§	incalculable	.00	incalculable§	incalculable§	
Data are or n (%).§ kappa cannot be calculated in the event c If In the event no baseline could be assigned in the original	of one parameter bein. 10-minute time-fram	g 100%. e of interest, the	time-frame	of interest was	extended to 20	minutes, bv

Supplementary Table 1. Baseline assignability of FHR for the intervention and control group

SUPPLEMENTARY TABLES

was 5 Ē 10-minute in the original eceding the original time-frame. assign 9 0 ne could ¶ In the eventructure including the 10 minutes pr

Supplementary Table 2. Decelerations with loss of internal variability

	Intervention	Control	P-value
Decrease after start of study	5 (14%)	3 (9%)	
No change after start of study	30 (83%)	29 (85%)	0.67
Increase after start of study	1 (3%)	2 (6%)	

Data are n (%).

Supplementary Table 3. Decelerations in combination with tachycardia or bradycardia

	Intervention	Control	P-value
Decrease after start of study	6 (17%)	3 (9%)	
No change after start of study	23 (64%)	29 (85%)	0.11
Increase after start of study	7 (19%)	2 (6%)	

Data are n (%).

REFERENCES

- 1. Yli BM, Kjellmer I. Pathophysiology of foetal oxygenation and cell damage during labour. Best Pract Res Clin Obstet Gynaecol. 2016;30:9-21. doi:10.1016/j.bpobgyn.2015.05.004
- 2. Westgate JA, Wibbens B, Bennet L, Wassink G, Parer JT, Gunn AJ. The intrapartum deceleration in center stage: a physiologic approach to the interpretation of fetal heart rate changes in labor. *Am J Obstet Gynecol.* 2007;197(3):236.e-236.11. doi:S0002-9378(07)00429-2 [pii]
- 3. Bullens LM, van Runnard Heimel PJ, van der Hout-van der Jagt MB, Oei SG. Interventions for Intrauterine Resuscitation in Suspected Fetal Distress During Term Labor: A Systematic Review. *Obstet Gynecol Surv.* 2015;70(8):524-539. doi:10.1097/ OGX.00000000000215 [doi]
- 4. The American College of Obstetricians and Gynecologists. Practice bulletin no. 116: Management of intrapartum fetal heart rate tracings. *Obstet Gynecol*. 2010;116(5):1232-1240. doi:10.1097/AOG.0b013e3182004fa9
- 5. Hamel MS, Anderson BL, Rouse DJ. Oxygen for intrauterine resuscitation: of unproved benefit and potentially harmful. *Am J Obstet Gynecol.* 2014;211(2):124-127. doi:10.1016/j.ajog.2014.01.004 [doi]
- 6. Garite TJ, Nageotte MP, Parer JT. Should we really avoid giving oxygen to mothers with concerning fetal heart rate patterns? *Am J Obstet Gynecol.* 2015;212(4):45-60, 459.e1. doi:10.1016/j.ajog.2015.01.058 [doi]
- 7. Fawole B, Hofmeyr GJ. Maternal oxygen administration for fetal distress. *Cochrane database Syst Rev.* 2012;12:CD000136. doi:10.1002/14651858.CD000136.pub2 [doi]
- 8. Bullens LM, Moors S, van Runnard Heimel PJ, van der Hout-van der Jagt MB, Oei SG. Practice variation in the management of intrapartum fetal distress in The Netherlands and the Western world. *Eur J Obstet Gynecol Reprod Biol.* 2016;205. doi:10.1016/j. ejogrb.2016.08.012
- Vasicka A, Quilligan EJ, Aznar R, Lipsitz PJ, Bloor BM. Oxygen tension in maternal and fetal blood, amniotic fluid, and cerebrospinal fluid of the mother and the baby. *Am J Obstet Gynecol.* 1960;79:1041-1047. doi:0002-9378(60)90508-1 [pii]
- 10. Thorp JA, Trobough T, Evans R, Hedrick J, Yeast JD. The effect of maternal oxygen administration during the second stage of labor on umbilical cord blood gas values: a randomized controlled prospective trial. *Am J Obstet Gynecol.* 1995;172(2 Pt 1):465-474. doi:0002-9378(95)90558-8 [pii]
- 11. Lofaso F, Dauger S, Matrot B, Vardon G, Gaultier C, Gallego J. Inhibitory effects of repeated hyperoxia on breathing in newborn mice. *Eur Respir J.* 2007;29(1):18-24. doi:10.1183/09031936.00111705
- 12. Nesterenko TH, Acun C, Mohamed MA, et al. Is it a safe practice to administer oxygen during uncomplicated delivery: a randomized controlled trial? *Early Hum Dev.* 2012;88(8):677-681. doi:10.1016/j.earlhumdev.2012.02.007 [doi]
- 13. Saling E. Effect of oxygen inhalation by the mother on the blood gases and acid-base equilibrium of the fetus. *Geburtshilfe Frauenheilkd*. 1963;23:528-538.
- 14. Althabe Jr O, Schwarcz RL, Pose S V, Escarcena L, Caldeyro-Barcia R. Effects on fetal heart rate and fetal pO2 of oxygen administration to the mother. *Am J Obstet Gynecol.* 1967;98(6):858-870. doi:0002-9378(67)90205-0 [pii]

- 15. Simpson KR, James DC. Efficacy of intrauterine resuscitation techniques in improving fetal oxygen status during labor. *Obstet Gynecol.* 2005;105(6):1362-1368. doi:105/6/1362 [pii]
- 16. Dildy GA, Clark SL, Loucks CA. Intrapartum fetal pulse oximetry: the effects of maternal hyperoxia on fetal arterial oxygen saturation. *Am J Obstet Gynecol*. 1994;171(4):1120-1124. doi:0002-9378(94)90048-5 [pii]
- 17. Gare DJ, Shime J, Paul WM, Hoskins M. Oxygen administration during labor. *Am J Obstet* Gynecol. 1969;105(6):954-961. doi:0002-9378(69)90104-5 [pii]
- 18. Haydon ML, Gorenberg DM, Nageotte MP, et al. The effect of maternal oxygen administration on fetal pulse oximetry during labor in fetuses with nonreassuring fetal heart rate patterns. *Am J Obstet Gynecol*. 2006;195(3):735-738. doi:S0002-9378(06)00867-2 [pii]
- 19. Hidaka A, Komatani M, Ikeda H, Kitanaka T, Okada K, Sugawa T. A comparative study of intrauterine fetal resuscitation by beta-stimulant and O2 inhalation. *Asia-Oceania J Obstet Gynaecol.* 1987;13(2):195-200.
- Raghuraman N, Wan L, Temming LA, et al. Effect of Oxygen vs Room Air on Intrauterine Fetal Resuscitation: A Randomized Noninferiority Clinical Trial. JAMA Pediatr. 2018;179(9):818-823. doi:10.1001/jamapediatrics.2018.1208 [doi]
- 21. Bullens LM, Hulsenboom ADJ, Moors S, et al. Intrauterine resuscitation during the second stage of term labour by maternal hyperoxygenation versus conventional care: Study protocol for a randomised controlled trial (INTEREST O2). *Trials.* 2018;19(1). doi:10.1186/s13063-018-2567-x
- 22. Moors S, Bullens LM, van Runnard Heimel PJ, et al. The effect of intrauterine resuscitation by maternal hyperoxygenation on perinatal and maternal outcome; a randomized controlled trial. *Am J Obs Gynecol MFM*. 2020;2:100102.
- 23. Bernardes J, Ayres-de-Campos D. Poor reliability of visual analysis of fetal heart rate tracings: what should be done about it? *Am J Obstet Gynecol*. 2012;206(6):e6. doi:10.1016/j.ajog.2012.02.027
- 24. Ayres-de-Campos D, Bernardes J, Costa-Pereira A, Pereira-Leite L. Inconsistencies in classification by experts of cardiotocograms and subsequent clinical decision. *Br J Obstet Gynaecol*. 1999;106(12):1307-1310. doi:10.1111/j.1471-0528.1999.tb08187.x
- 25. Weyrich J, Ortiz JU, Muller A, et al. Intrapartum PRSA: a new method to predict fetal acidosis?-a case-control study. *Arch Gynecol Obstet*. 2020;301(1):137-142. doi:10.1007/s00404-019-05419-y
- 26. Fletcher AJW, Gardner DS, Edwards CMB, Fowden AL, Giussani DA. Development of the ovine fetal cardiovascular defense to hypoxemia towards full term. *Am J Physiol Heart Circ Physiol*. 2006;291(6):H3023-34. doi:10.1152/ajpheart.00504.2006
- 27. Ayres-de-Campos D, Spong CY, Chandraharan E, Panel FIFMEC. FIGO consensus guidelines on intrapartum fetal monitoring: Cardiotocography. *Int J Gynaecol Obstet*. 2015;131(1):13-24. doi:10.1016/j.ijgo.2015.06.020 [doi]
- 28. Georgieva A, Papageorghiou AT, Payne SJ, Moulden M, Redman CW. Phase-rectified signal averaging for intrapartum electronic fetal heart rate monitoring is related to acidaemia at birth. *BJOG*. 2014;121(7):889-894. doi:10.1111/1471-0528.12568 [doi]
- Georgieva A, Payne SJ, Moulden M, Redman CW. Computerized fetal heart rate analysis in labor: detection of intervals with un-assignable baseline. *Physiol Meas*. 2011;32(10):1549-1560. doi:10.1088/0967-3334/32/10/004 [doi]

- 30. Georgieva A, Payne SJ, Moulden M, Redman CW. Relation of fetal heart rate signals with unassignable baseline to poor neonatal state at birth. *Med Biol Eng Comput*. 2012;50(7):717-725. doi:10.1007/s11517-012-0923-7 [doi]
- 31. Ozden S, Demirci F. Significance for fetal outcome of poor prognostic features in fetal heart rate traces with variable decelerations. *Arch Gynecol Obstet*. 1999;262(3-4):141-149. doi:10.1007/s004040050242
- 32. Gaziano EP. A study of variable decelerations in association with other heart rate patterns during monitored labor. *Am J Obstet Gynecol.* 1979;135(3):360-363. doi:10.1016/0002-9378(79)90705-1
- 33. Kazandi M, Sendag F, Akercan F, Terek MC, Gundem G. Different types of variable decelerations and their effects to neonatal outcome. *Singapore Med J.* 2003;44(5):243-247.
- 34. Holzmann M, Wretler S, Cnattingius S, Nordstrom L. Cardiotocography patterns and risk of intrapartum fetal acidemia. *J Perinat Med.* 2015;43(4):473-479. doi:10.1515/jpm-2014-0105
- 35. Hamilton E, Warrick P, O'Keeffe D. Variable decelerations: do size and shape matter? *J Matern Fetal Neonatal Med.* 2012;25(6):648-653. doi:10.3109/14767058.2011.594118
- 36. Krebs HB, Petres RE, Dunn LJ. Intrapartum fetal heart rate monitoring. VIII. Atypical variable decelerations. *Am J Obstet Gynecol.* 1983;145(3):297-305.
- 37. Lobmaier SM, Mensing van Charante N, Ferrazzi E, et al. Phase-rectified signal averaging method to predict perinatal outcome in infants with very preterm fetal growth restriction- a secondary analysis of TRUFFLE-trial. *Am J Obstet Gynecol.* 2016;215(5):630.e1-630.e7.doi:10.1016/j.ajog.2016.06.024
- 38. Yli B, Hahn T, Kessler J, Lie H, Martinussen M. Bigger is not always better... the validity of the US randomized trial of STAN for norway. Mölndal, sweden: Neoventa medical. Accessed March 2, 2020. http://www.neoventa.com/2015/11/bigger-is-not-always-better/
- Georgieva A, Payne SJ, Redman CW. Computerised electronic foetal heart rate monitoring in labour: automated contraction identification. *Med Biol Eng Comput.* 2009;47(12):1315-1320. doi:10.1007/s11517-009-0538-9 [doi]
- 40. Landis JR, Koch GG. The measurement of observer agreement for categorical data. *Biometrics*. 1977;33(1):159-174.
- 41. Bauer A, Kantelhardt JW, Bunde A, et al. Phase-rectified signal averaging detects quasiperiodicities in non-stationary data. *Phys A Stat Mech its Appl.* 2006;364:423-434. doi:10.1016/j.physa.2005.08.080
- 42. Bauer A, Kantelhardt JW, Barthel P, et al. Deceleration capacity of heart rate as a predictor of mortality after myocardial infarction: cohort study. *Lancet*. 2006;367(9523):1674-1681. doi:10.1016/S0140-6736(06)68735-7
- 43. Joshi R, Kommers D, Long X, et al. Cardiorespiratory coupling in preterm infants. *J Appl Physiol*. 2019;126(1):202-213. doi:10.1152/japplphysiol.00722.2018
- 44. Dawes GS, Visser GH, Goodman JD, Redman CW. Numerical analysis of the human fetal heart rate: the quality of ultrasound records. *Am J Obstet Gynecol*. 1981;141(1):43-52. doi:10.1016/0002-9378(81)90673-6
- 45. Bullens LM, Hulsenboom ADJ, Moors S, et al. Correction to: Intrauterine resuscitation during the second stage of term labour by maternal hyperoxygenation versus conventional care: Study protocol for a randomised controlled trial (INTEREST O2) (Trials (2018) 19 (195) DOI: 10.1186/s13063-018-2567-x). *Trials*. 2018;19(1). doi:10.1186/s13063-018-2963-2

- 46. van Scheepen JAM, Koster MPH, Vasak B, Redman C, Franx A, Georgieva A. Effect of signal acquisition method on the fetal heart rate analysis with phase rectified signal averaging. *Physiol Meas*. 2016;37(12):2245-2259. doi:10.1088/1361-6579/37/12/2245
- 47. Mahdiani S, Jeyhani V, Peltokangas M, Vehkaoja A. Is 50 Hz high enough ECG sampling frequency for accurate HRV analysis? *Conf Proc . Annu Int Conf IEEE Eng Med Biol Soc IEEE Eng Med Biol Soc Annu Conf.* 2015;2015:5948-5951. doi:10.1109/EMBC.2015.7319746
- 48. Kwon O, Jeong J, Kim H Bin, et al. Electrocardiogram Sampling Frequency Range Acceptable for Heart Rate Variability Analysis. *Healthc Inform Res.* 2018;24(3):198-206. doi:10.4258/hir.2018.24.3.198
- 49. Durosier LD, Green G, Batkin I, et al. Sampling rate of heart rate variability impacts the ability to detect acidemia in ovine fetuses near-term. *Front Pediatr.* 2014;2:38. doi:10.3389/fped.2014.00038
- 50. Rivolta MW, Stampalija T, Casati D, et al. Acceleration and deceleration capacity of fetal heart rate in an in-vivo sheep model. *PLoS One.* 2014;9(8):e104193. doi:10.1371/journal. pone.0104193
- 51. Nunes I, Ayres-de-Campos D. Computer analysis of foetal monitoring signals. *Best Pract Res Clin Obstet Gynaecol.* 2016;30:68-78. doi:10.1016/j.bpobgyn.2015.02.009
- 52. Ayres-de-Campos D, Rei M, Nunes I, Sousa P, Bernardes J. SisPorto 4.0–computer analysis following the 2015 FIGO Guidelines for intrapartum fetal monitoring. *J Matern Neonatal Med*. 2017;30(1):62-67. doi:10.3109/14767058.2016.1161750
- 53. Annunziata ML, Tagliaferri S, Esposito FG, et al. Computerized analysis of fetal heart rate variability signal during the stages of labor. *J Obstet Gynaecol Res.* 2016;42(3):258-265. doi:10.1111/jog.12908
- 54. Khazin AF, Hon EH, Hehre FW. Effects of maternal hyperoxia on the fetus. I. Oxygen tension. *Am J Obstet Gynecol*. 1971;109(4):628-637. doi:0002-9378(71)90639-9 [pii]
- 55. McNamara H, Johnson N, Lilford R. The effect on fetal arteriolar oxygen saturation resulting from giving oxygen to the mother measured by pulse oximetry. *Br J Obstet Gynaecol*. 1993;100(5):446-449.
- 56. Devane D, Lalor JG, Daly S, McGuire W, Cuthbert A, Smith V. Cardiotocography versus intermittent auscultation of fetal heart on admission to labour ward for assessment of fetal wellbeing. *Cochrane database Syst Rev.* 2017;1:CD005122. doi:10.1002/14651858. CD005122.pub5 [doi]
- 57. Alfirevic Z, Devane D, Gyte GM. Continuous cardiotocography (CTG) as a form of electronic fetal monitoring (EFM) for fetal assessment during labour. *Cochrane database Syst Rev.* 2013;(5):CD0060(5):CD006066. doi:10.1002/14651858.CD006066.pub2 [doi]
- 58. American College of Obstetricians and Gynecologists. ACOG Practice Bulletin No. 106: Intrapartum fetal heart rate monitoring: nomenclature, interpretation, and general management principles. *Obstet Gynecol.* 2009;114(1):192-202. doi:10.1097/ AOG.0b013e3181aef106
- 59. Elliott C, Warrick PA, Graham E, Hamilton EF. Graded classification of fetal heart rate tracings: association with neonatal metabolic acidosis and neurologic morbidity. *Am J Obstet Gynecol.* 2010;202(3):258.e1-8. doi:10.1016/j.ajog.2009.06.026
- 60. Schiermeier S, Pildner von Steinburg S, Thieme A, et al. Sensitivity and specificity of intrapartum computerised FIGO criteria for cardiotocography and fetal scalp pH during labour: multicentre, observational study. *BJOG*. 2008;115(12):1557-1563. doi:10.1111/j.1471-0528.2008.01857.x





General discussion and future perspectives

GENERAL DISCUSSION AND FUTURE PERSPECTIVES

Obstetric caregivers want pregnancy and delivery to be completed as safe as possible, since complications can have serious and lifelong consequences. In this thesis, several methods are addressed to help caregivers monitor and manage maternal and fetal complications during pregnancy and delivery. This thesis is subdivided into two parts, on the monitoring and management of maternal and fetal risks, respectively.

In this chapter, the main outcomes of the studies and its clinical implications are discussed and suggestions for future research are made.

MAIN FINDINGS

PART I – MONITORING AND MANAGEMENT OF THE MOTHER AT RISK OF HYPERTENSIVE PREGNANCY DISORDERS

Hypertensive pregnancy disorders (HPD) are associated with maternal, placental, fetal, and neonatal morbidity and mortality.¹⁻⁴ Therefore, adequate prediction of HPD is very important to enable early diagnosis and management. As current prediction models for HPD are limited in their predictive abilities, there is a need for new, more accurate methods. In part I of this thesis, we investigate whether two promising prediction methods, namely speckle tracking echocardiography (STE) and heart rate variability (HRV), are suitable to detect differences between women with (a history of) HPD and normotensive controls.

Is STE a suitable method to detect differences in cardiac function in pregnant women with HPD or women with a history of HPD compared to normotensive women?

Although the exact etiology of HPD is not fully known, several studies state that the primary derangement in HPD involves the cardiovascular system and abnormal placentation.⁵⁻⁷ The inability of the maternal cardiovascular system to adequately adapt to pregnancy causes the maternal myocardium to subtly change in shape, size, and function.^{5.8} The systematic review presented in **chapter 2** revealed that the STE parameter left ventricular (LV) global longitudinal strain (LV-GLS) is significantly decreased in pregnant women with HPD compared to normotensive pregnant women. A decrease in LV-GLS is associated with subclinical stages of multiple cardiac diseases.⁹ Hence, a decreased LV-GLS in women with HPD may be an indicator of subclinical deterioration of the myocardium.

The other two directions of strain, LV global radial strain (LV-GRS) and LV global circumferential strain (LV-GRS), were found to be decreased in women with early-onset and severe preeclampsia, whereas no differences were found in women with late-onset

and mild preeclampsia. These two parameters might therefore be associated with a more severe course of HPD. However, little is known on the predictive value of abnormal strain during pregnancy. Possibly, STE results might show deterioration before the clinical onset of HPD and might therefore aid in the prediction and early diagnosis of HPD. Up till now, only one study investigated the prognostic value of STE in HPD, which showed that women with chronic hypertension and decreased LV-GLS mid-trimester had a significantly higher risk of developing superimposed preeclampsia.¹⁰ STE therefore has tremendous clinical potential. Future studies are needed to further investigate the value of this diagnostic method during pregnancy.

Risk of cardiovascular disease after pregnancies complicated by HPD

Several large studies and meta-analyses have consistently demonstrated an increased risk of cardiovascular disease (CVD) in women with a history of HPD, especially after preeclampsia, compared to women with normotensive pregnancies.^{11,12} Furthermore, the manifestation of CVD occurs earlier in women with a history of HPD compared to uncomplicated controls.¹² Hence, adequate follow-up of women with a history of HPD may allow early identification of CVD and provide opportunities for prevention and management of CVD.

Ideally, the CVD screening would focus on subclinical changes with a high predictive value of the actual development of the disease. Unfortunately, conventional echocardiographic features are unsuitable, as these parameters typically remain normal in the subclinical phase of cardiac disease.⁹ In contrast, advanced techniques like STE can allow for earlier detection of these subtle myocardial changes.¹³ The review provided in chapter 2 showed a decrease in cardiac function measured with STE, especially in LV-GLS, lasting up to 13 years after pregnancies complicated by HPD in women without clinical symptoms of heart disease. Possibly, these abnormal strain measurements after pregnancy may provide the desired differentiation in the risk of CVD within the high-risk group of women with a history of HPD. This could enable appropriate screening and management based on each women's personal risk profile.

Does HRV detect differences in the function of the autonomic nervous system in pregnant women with HPD or women with a history of HPD compared to normotensive women?

The autonomic nervous system (ANS) has a prominent role in the cardiovascular adaptations during pregnancy.¹⁴⁻¹⁶ As HPD might originate from the inadequacy of such adaptations, differences in the functioning of the ANS – assessed by HRV – may be found between HPD and normotensive controls. Hence, in **chapter 3**, the differences in HRV were reviewed between pregnant women with HPD and normotensive pregnant women and between women with a history of HPD and women with a history of a normotensive pregnancy.

The results from the included studies showed a wide variety, possibly due to the heterogenicity of the data. Moreover, long-acting medication might influence HRV results, introducing an important confounding factor in many studies. Therefore, it is hard to draw firm conclusions based on available literature. However, a trend is seen towards an elevated sympathetic and a reduced parasympathetic tone in case of HPD. This imbalance of the ANS could contribute to the origin of HPD,^{16,17} as illustrated in Figure 1. The increased sympathetic tone causes an increase in peripheral vascular resistance, leading to a higher heart rate and blood pressure. Moreover, the increase in sympathetic activity may expedite placental ischemic/reperfusion events, which subsequently leads to the release of abnormal soluble placental factors. These placental factors may cause endothelial dysfunction and vasoconstriction, which aggravates placental ischemia and leads to a higher blood pressure. The combination of these events could contribute to the development of HPD.^{16,17}



Figure 1. Schematic overview of how the autonomic imbalance is hypothesized to lead to HPD. Adapted from Moors et al.¹⁸

Changes in the sympathovagal balance, reflected in changes in HRV, might occur before the clinical onset of HPD. Thereby, these early changes might aid in the detection of HPD in preclinical stages. Three longitudinal studies demonstrated that in the first trimester, before the clinical onset of hypertension, the LF/HF ratio was significantly higher in high-risk women who developed gestational hypertension later in their pregnancy compared to the high-risk group that did not.¹⁹⁻²¹ These results are promising and indicate the need for further longitudinal research.

However, before such future HRV studies are initiated, the confounding effect of certain medications should be clarified. As discussed above, the use of medication might influence HRV results, making it harder to interpret the results. Women with HPD are likely to receive routine medication such as antihypertensive drug therapy and, in severe cases, antenatally administered corticosteroids and magnesium sulfate. Unfortunately, little is known about how these medications affect maternal HRV. In contrast, several studies have demonstrated that corticosteroids and magnesium sulfate affect fetal HRV.²²⁻²⁵ Given the mechanism of action of both drugs, it is likely that maternal HRV is also affected by its admission. Moreover, it is hypothesized that antihypertensive drugs may also affect maternal HRV.^{17,26} Hence, it is important to quantify the effect of these medications on maternal HRV to improve the interpretation of HRV study results.

What is the effect of routine obstetric medication on maternal HRV?

To quantify the effect of routine obstetric medication on maternal HRV, we designed and initiated a longitudinal cohort study, known as the MAMA-hart study. A detailed description of the study protocol can be found in **chapter 4** of this thesis. Antenatally administered corticosteroids are chosen as the main focus of this study due to its frequent use and known effect on fetal HRV. HRV features will be compared between the epoch prior to medication admission and the epoch after medication admission, thereby quantifying the effect of medication on maternal HRV.

Participants are likely to also receive other obstetric medication as part of standard care, which is an unavoidable limitation in this study design. Although the administration of multiple drugs could confound the measurements, it also offers the opportunity to explore the effect of those medications on maternal HRV. This may contribute to also limit their confounding effect in future HRV studies.

All participants will wear a wrist-worn device that acquires continuous photoplethysmography (PPG) measurements, from which HRV features will be derived. Even though electrocardiography (ECG) is the more traditional method to measure HRV, the use of PPG measurement has several advantages over ECG. Firstly, the wrist-worn device that measures PPG is non-invasive, non-obstructive, and simple in its use. Secondly, continuous measurements with a wrist-worn device are less costly than continuous ECG measurements.²⁷ Although ECG measurements offer a higher sampling rate of the heart rate than PPG, literature shows that if PPG is acquired with a sampling rate of \geq 25 Hz, HRV measurements using PPG can be as reliable as those derived from ECG.²⁸ Hence, if HRV might turn out to be an effective method in prediction and early detection of HPD, the use of PPG measurement with a wrist-worn device might be better suited for clinical practice than ECG.

It is therefore expected that the results from the MAMA-hart study will reduce the confounding effect of routine obstetric medication in studies employing HRV. Thereby, future studies can more accurately distinguish changes in HRV associated with pregnancy complications, from those resulting from the use of medication. An improved understanding of possible HRV changes associated with pregnancy complications could in turn facilitate earlier diagnosis of those complications, allowing timely implementation of risk-mitigating interventions.

PART II – MONITORING AND MANAGEMENT OF THE FETUS AT RISK FOR FETAL DISTRESS

Monitoring and management of fetal distress during labor is complex. Obstetricians are challenged to, on the one hand, perform timely interventions to prevent fetal hypoxia and asphyxia, but, on the other hand, prevent unnecessary interventions due to their potential harm to both the mother and her child. Moreover, in case of suspected fetal distress, the healthcare team has to decide whether the risk of fetal hypoxia is so severe that immediate delivery of the baby is needed, or if fetal oxygenation can be restored with the use of intrauterine resuscitation techniques. To help with this decision, several methods can be used to monitor the fetal condition during labor.

In part II of this thesis, we first discuss which monitoring and management strategies are recommended in guidelines and used in clinical practice. Subsequently, we focus on the effect of one frequently debated intrauterine resuscitation technique: maternal hyperoxygenation.

Which methods are recommended in international guidelines regarding the monitoring and management of fetal distress during labor?

In **chapter 5** we compared recommendations regarding fetal monitoring and management of suspected fetal distress from the national guidelines of several Western countries. All of the obtained guidelines recommended the use of cardiotocography in high-risk population during delivery. Although it is deemed the best available method to monitor fetal condition during labor in high-risk population, it has some shortcomings, and additional information regarding the fetal condition might be desired. Two main methods can acquire such additional information, namely fetal scalp blood sampling (FSBS) and analysis of the ST segment of the fetal ECG (short ST analysis). With respect to monitoring, all guidelines recommended the use of FSBS if additional testing is required. A Cochrane review reported an increase in instrumental deliveries and a decrease in neonatal acidosis following FSBS.²⁹

In contrast to FSBS, the recommendations on the use of ST-analysis were conflicting, with three guidelines promoting this monitoring technique and three others advising against its use. This might be because, even though the use of ST-analysis significantly decreases the need for FSBS as well as the number of vaginal assisted deliveries,³⁰ the use of this technique does not cause significant decrease in other outcomes like cesarean sections, neonatal intensive care admissions, neonatal acidosis and perinatal deaths.³⁰ Previous studies have shown that ST-analysis is influenced by electrode placement and head orientation of the fetus.^{31,32} This confounding effect may be partly corrected by using relative ST-analysis, in which the rise in T/QRS fragment is analyses as a percentage from the baseline.³³ To our knowledge, there are no trials comparing cardiotocography with FSBS to cardiotocography with (relative) ST-analysis. The variations in international guidelines regarding these techniques emphasize the need for such studies.

Five guidelines also contained recommendations on intrauterine resuscitation as a method for management of fetal distress. Their recommendations were consistent concerning the use of tocolytic drugs and discontinuation of oxytocin, which were advised in all guidelines. In contrast, the recommendations regarding maternal hyperoxygenation and amnioinfusion were contradictory.

How can these differences in international guidelines be explained? The main reason seems to be the lack of solid evidence. Hence, recommendations need to be made on limited available evidence, which may cause variation in the interpretation of the study results. The differences in international guidelines create ambiguity for health care professionals, which may lead to variations in clinical practice. Therefore, future research should focus on providing more clarity, especially with regard to maternal hyperoxygenation, amnioinfusion and ST-analysis.

Which fetal monitoring methods are used in Dutch clinical practice, and which intrauterine resuscitation techniques are utilized in case of suspected fetal distress?

To investigate how fetal distress is diagnosed and managed in the Netherlands, in 2015 a nationwide survey was conducted in all Dutch hospitals, the results of which are also presented in chapter 5. Cardiotocography was used in all hospitals to monitor the fetal condition. In addition, FSBS was available and used in 98% of Dutch hospitals. This was an increase compared to the 87% availability found in 2009.³⁴ In contrast, the use of ST-analysis decreased from 30% in 2012³⁵ to 23% in 2015.

The following intrauterine resuscitation techniques are commonly used in all Dutch hospitals; maternal repositioning, discontinuation of oxytocin, and the use of tocolytic drugs. The use of maternal hyperoxygenation and amnioinfusion is inconsistent in Dutch hospitals, thus reflecting conflicting international guidelines. This inconsistency also exposes the lack of recommendations regarding these two interventions in the Dutch guideline at the moment of this survey.

The results from this nationwide survey show that, even in a country as small as the Netherlands, a large practice variation is present in monitoring and management of fetal distress during labor. This is likely due to a lack of solid scientific evidence. Moreover, the lack of recommendations on certain aspects of monitoring and management of fetal distress in the Dutch national guideline, as well as the different interpretation of study results among delivery room staff, may extend this variation even more. This may be extremely confusing for a patient who gives birth to her first child in Dutch hospital A, moves to another city, and gives birth to her next child in Dutch hospital B.

What is the effect of intrauterine resuscitation by maternal hyperoxygenation during the second stage of term labor on neonatal and maternal outcome?

Maternal hyperoxygenation is one of the most debated intrauterine resuscitation techniques. Although it is frequently used, robust evidence regarding its effect is scarce and conflicting.^{36,37} Consequently, the need for more research has been underlined in multiple reviews.^{36,37}

Therefore, we designed and performed the INTEREST O2 study, which is the first randomized controlled trial (RCT) evaluating the effect of maternal hyperoxygenation in case of suspected fetal distress during the second stage of labor. A detailed description of the study protocol of this single center RCT can be found in **chapter 6**.

Patients were included in case of suspected fetal distress during the second stage of term labor, defined as either an intermediary or an abnormal FHR pattern according to the modified FIGO classification.³⁸ All participants were randomly allocated to receive either maternal hyperoxygenation with 100% oxygen or conventional care without additional oxygen.

The primary outcome of this study was the change in FHR, which may be topic of debate. One could argue that the primary outcome should have been neonatal morbidity. However, achieving sufficient power for the latter would require a very large sample size of over 10,000 women.³⁹ As the effect and potentially harmful side-effects of maternal hyperoxygenation were not properly investigated yet, we chose not to expose such a large group of women and their fetuses to this intervention. Hence, a change in FHR was used as a surrogate marker for neonatal outcome, with fewer patients needed. Currently, FHR is deemed the best available method to monitor the fetal condition during labor, and a non-reassuring FHR pattern is considered to be indicative of potential fetal hypoxia.^{29,40} One of the limitations of FHR monitoring is its low specificity for fetal acidosis.^{38,41} Therefore, we also evaluated the change in FHR using specific FHR features that are related to unfavorable neonatal outcome.

Besides the change in FHR, several secondary outcomes were chosen to analyze the safety and potentially harmful side-effects of maternal hyperoxygenation.

In addition to the complete case analysis, additional subgroup analyses were performed on participants with *intermediary* FHR pattern and with *abnormal* FHR pattern, respectively. Moreover, a third subgroup was analyzed consisting of the patient group with *small for gestational age fetuses* (<10th percentile). With these subgroup analyses, we wanted to analyze more specifically whether one of the subgroups would potentially benefit more than others from maternal hyperoxygenation, as previous studies demonstrated that fetuses with the lowest initial oxygen saturation benefit the most from this intervention.^{42,43}

The perinatal and maternal outcomes of the INTEREST O2 study are described in **chapter 7**. We found no statistically significant differences in Apgar Score, NICU admission, or umbilical cord blood gas. Furthermore, we measured malondialdehyde in umbilical cord blood as a marker to evaluate the potential harm of free oxygen radicals. In contrast to previous studies on the prophylactic use of maternal hyperoxygenation, the levels of malondialdehyde were not significantly different between both groups, implying that we could not observe harmful side effects on the fetus. This rejects the most important argument to discourage the use of this intrauterine resuscitation technique.

Besides the absence of harmful side-effects of maternal hyperoxygenation, a positive effect was seen on the number of episiotomies on fetal indication, as this intervention was carried out less often in mothers receiving extra oxygen, with a statistically significant decrease in the subgroup with abnormal FHR pattern. The reduction in this potentially harmful intervention might indicate a better fetal condition after maternal hyperoxygenation, as fewer fetuses were in such distress that an episiotomy was needed to immediately terminate the delivery.

Does maternal hyperoxygenation, applied in case of suspected fetal distress during the second stage of term labor, affect FHR?

In routine obstetric care, the FHR pattern is typically analyzed with the use of the FIGO classification.³⁸ Hence, we analyzed the change in FIGO classification following hyperoxygenation, as evaluated by clinicians.

The change in FHR could be analyzed for 71 participants (61%). The other 46 participants (39%) gave birth before the end of the time-frame of interest and, therefore, insufficient FHR data existed in the "post-study" time-frame to properly analyze the change in FHR. A significantly positive effect of maternal hyperoxygenation on change in FIGO classification was found. A similar beneficial effect was seen in all subgroup analyses, albeit not statistically significant. Even though this study was not powered for neonatal outcome, the positive effect seen on FIGO classification may positively reflect on neonatal outcome. Larger studies powered for neonatal outcome are needed to confirm this finding.

Although visual analysis of FHR, as used for the FIGO classification, is one of the most used methods to evaluate the FHR, it is known to have a large intra- and inter-observer variability.^{44,45} Computerized analysis of FHR could improve intra- and interindividual reproducibility and could also provide additional insights in physiological changes by assessing FHR features that are not detected visually. Therefore, the effect of maternal hyperoxygenation has also been evaluated using detailed computerized analyses. These results are presented in **chapter 8.** The change in depth and duration of FHR decelerations was not statistically different between maternal hyperoxygenation (decrease of 2.3%) and conventional care (increase of 10%). This may be (partly) due to insufficient power due to the higher than accommodated fraction of missing data.

Besides FHR decelerations, fetal HRV is one of the most important markers to assess fetal wellbeing.⁴⁶ Phase-rectified signal averaging (PRSA) is an innovative non-traditional method to characterize HRV that, in contrast to traditional HRV metrics, can be used at the sampling rate of 4Hz, which is typically used at delivery wards.⁴⁷ A significantly positive effect of maternal hyperoxygenation was seen on PRSA-based parameters, with a decrease of both PRSA-acceleration capacity and PRSA-deceleration capacity in the intervention group and an increase of both these parameters in the control group. These significant differences in PRSA-based parameters indicate less compensatory activation of the ANS after maternal hyperoxygenation, which may positively reflect on neonatal outcome.

In conclusion, the INTEREST O2 study is the first RCT to study the effect of maternal hyperoxygenation on FHR in the second stage of labor. The difference in deceleration depth and duration was not significant, possibly due to the higher than accommodated

fraction of missing data. The change in FIGO classification and PRSA-based parameters did show a significantly positive effect of maternal hyperoxygenation, which may positively reflect on neonatal outcome. As this study shows a potentially beneficial effect of maternal hyperoxygenation, without any harmful side-effects, there is no need to ban this intervention from delivery rooms that are currently using this technique. Larger studies powered for neonatal outcome are needed to provide strong recommendations for the use of maternal hyperoxygenation in clinical practice. Given the results presented in this thesis, there are no longer impediments to withhold the conduction of such large trials.

CLINICAL IMPLICATIONS AND FUTURE PERSPECTIVES

The research presented in this thesis yields multiple opportunities to improve monitoring and management of both maternal and fetal risks during pregnancy and delivery.

Implications for maternal care

This thesis shows that STE is a suitable method to detect differences between pregnant women with HPD and normotensive controls. Its clinical implications may become more evident once the clinical and prognostic value of abnormal strain during pregnancy has been further evaluated. Likely, STE may become a key aspect in the future prediction and early detection of HPD. The significant differences observed in women with HPD may already exist before the clinical onset of HPD. If so, those parameters might be used for HPD risk stratification in addition to the currently used maternal medical history. Women with an increased risk of HPD based on their medical history (e.g. women with obesity, higher age, primigravida) could undergo STE, enabling a more precise and individual risk prediction. This might in turn facilitate a more personal management strategy and contribute to improved medical care for women at risk of HPD.

Besides the abnormalities that can be detected with STE during pregnancies complicated by HPD, this technique can also detect lasting myocardial changes in women with a history of early-onset preeclampsia before the clinical onset of CVD. STE has gained growing importance in several clinical settings over the last decade, confirming its added value in cardiac function analysis.⁹ However, the clinical implications of these STE abnormalities after HPD are not yet known. Do women with a history of HPD have an even higher risk of CVD when they have decreased strain? Or do they have a higher risk of a specific cardiac conditions, like arterial hypertension and ischemic injury? Is there a drug therapy that should be advised for women who had early-onset preeclampsia and have abnormal STE parameters? Additional research is needed to answer these questions and establish which role STE might play in the cardiovascular follow-up after HPD. Possibly, abnormal strain measurements might help to differentiate the risk of
CVD within this high-risk group of women with a history of HPD, enabling appropriate screening and treatment based on each women's personal risk profile.

Similar to STE parameters, differences in HRV between women with (a history of) HPD and normotensive controls are also evaluated in this thesis. Even though some differences in the function of the ANS can be detected with the use of HRV, the large heterogenicity found in current literature combined with the confounding effect of long-acting medication, makes it hard to draw firm conclusions. This impedes the direct clinical applicability of this method. Does this mean that HRV is, thereby, unsuited for the prediction and early detection of HPD, making further research on this topic irrelevant? We believe not. Despite the limitation of confounding factors, there are still indications that HRV can detect a sympathovagal disbalance in HPD, possibly even before the clinical onset of HPD. Hence, when the 'noise' in the HRV signal caused by confounding factors is quantified and eliminated, HRV may still be a promising method for the prediction and early detection of the confounding effect of several medications should be clarified, enabling better interpretation of the results of HRV analyses and, subsequently, enable the use of this method in clinical practice.

To address this knowledge gap, we therefore designed the MAMA-hart study to quantify the effect of routine obstetric medication. The results of this study will enable future studies to eliminate the confounding effect of such medication. Thereby, future research can provide solid conclusions whether HRV can detect differences between HPD and normotensive controls, and, if so, whether longitudinal HRV measurements during pregnancy could aid in the prediction of HPD.

Implications for fetal care

This thesis also contains a comparison of the recommendation in international guidelines for monitoring and management of fetal distress, showing a large variation. Hence, even though the Netherlands and the United Kingdom have great similarities regarding medical care, a patient admitted at a British delivery ward might get a different treatment when fetal distress is suspected compared to a patient from a Dutch delivery ward. Due to the lack of solid evidence, it is difficult to say which of those two patients gets the 'best' medical care. If we don't know this answer as professionals, how can we promise our patients to provide the best possible care to make sure their baby is born as safely as possible? The observed variation underlines the need for high-quality research on this topic. Results from those high-quality studies can enable uniform, evidence-based recommendations, reducing the differences currently found at delivery wards in various Western countries as well as within the Netherlands itself.

Hence, we encourage to further evaluate the potentially beneficial effect of maternal hyperoxygenation as shown from the INTEREST O2 study results in a trial powered

for neonatal outcome. This will provide evidence whether this treatment should be implemented in delivery wards all around the world. Oxygen is a therapy that is available in almost all hospitals, is easy in use, and proven to have minimal side-effects. In the end, maternal hyperoxygenation might contribute to better management of fetal distress during labor, leading to a decrease in perinatal asphyxia with potentially life-long health consequences.

Before a new treatment like maternal hyperoxygenation can be implemented in clinical practice, its effect needs to be proven by of clinical research. Based on the available evidence, doctors try to provide the best care for their patients. Even though traditional RCTs are considered the gold standard to investigate clinical issues, their design may not always be feasible for all therapeutic problems. Other types of RCT design, like a steppedwedge design, may in some cases be better suited to provide adequate evidence on the impact of interventions used in acute settings like the management of fetal distress. The design of a stepped-wedge study involves random and sequential crossover of clusters from control to the intervention until all clusters are exposed, thereby eliminating the need for individual randomization in the acute clinical setting.^{48,49} As the Dutch national guideline currently advises to only use maternal hyperoxygenation in a study setting, an excellent opportunity is provided to set-up a large randomized trial with a steppedwedge design in the Netherlands to evaluate the effect of this intervention on neonatal outcome. Moreover, the stepped-wedge design could also offer the possibility to study the clinical implementation of maternal hyperoxygenation at the delivery wards across the country. In the end, the high level evidence from such a trial can provide insights to improve perinatal care, thereby contributing to a reduction in neonatal morbidity and mortality.

REFERENCES

- 1. Umesawa M, Kobashi G. Epidemiology of hypertensive disorders in pregnancy: prevalence, risk factors, predictors and prognosis. *Hypertens Res.* 2017;40(3):213-220. doi:10.1038/hr.2016.126 [doi]
- Khan KS, Wojdyla D, Say L, Gulmezoglu AM, Look PF Van. WHO analysis of causes of maternal death: a systematic review. *Lancet (London, England)*. 2006;367(9516):1066-1074. doi:S0140-6736(06)68397-9 [pii]
- 3. Saleem S, McClure EM, Goudar SS, et al. A prospective study of maternal, fetal and neonatal deaths in low- and middle-income countries. *Bull World Health Organ.* 2014;92(8):605-612. doi:10.2471/BLT.13.127464
- 4. ACOG Practice Bulletin No. 202: Gestational Hypertension and Preeclampsia. *Obstet Gynecol.* 2019;133(1):e-e25. doi:10.1097/AOG.000000000003018 [doi]
- 5. Kalafat E, Thilaganathan B. Cardiovascular origins of preeclampsia. *Curr Opin Obstet Gynecol.* 2017;29(6):383-389. doi:10.1097/GCO.000000000000419 [doi]
- 6. Buddeberg BS, Sharma R, O'Driscoll JM, Agten AK, Khalil A, Thilaganathan B. Cardiac maladaptation in term pregnancies with preeclampsia. *Pregnancy Hypertens*. 2018;13:198-203. doi:S2210-7789(18)30089-8 [pii]
- 7. Braunthal S, Brateanu A. Hypertension in pregnancy: Pathophysiology and treatment. SAGE open Med. 2019;7:2050312119843700. doi:10.1177/2050312119843700
- 8. Melchiorre K, Sutherland GR, Baltabaeva A, Liberati M, Thilaganathan B. Maternal cardiac dysfunction and remodeling in women with preeclampsia at term. *Hypertension*. 2011;57(1):85-93. doi://dx.doi.org/10.1161/HYPERTENSIONAHA.110.162321
- 9. Cameli M, Mandoli GE, Sciaccaluga C, Mondillo S. More than 10 years of speckle tracking echocardiography: Still a novel technique or a definite tool for clinical practice? *Echocardiography*. Published online April 2019. doi:10.1111/echo.14339 [doi]
- 10. Shahul S, Ramadan H, Mueller A, et al. Abnormal mid-trimester cardiac strain in women with chronic hypertension predates superimposed preeclampsia. *Pregnancy Hypertens*. 2017;10:251-255. doi:S2210-7789(17)30144-7 [pii]
- 11. Wu P, Haththotuwa R, Kwok CS, et al. Preeclampsia and Future Cardiovascular Health: A Systematic Review and Meta-Analysis. *Circ Qual outcomes*. 2017;10(2):10.1161/ CIRCOUTCOMES.116.003497. Epub 2017 Feb 22. doi:e003497 [pii]
- 12. Bellamy L, Casas JP, Hingorani AD, Williams DJ. Pre-eclampsia and risk of cardiovascular disease and cancer in later life: systematic review and meta-analysis. *BMJ*. 2007;335(7627):974. doi:bmj.39335.385301.BE [pii]
- 13. O'Kelly AC, Sharma G, Vaught AJ, Zakaria S. The Use of Echocardiography and Advanced Cardiac Ultrasonography During Pregnancy. *Curr Treat Options Cardiovasc Med*. 2019;21(11):71. doi:10.1007/s11936-019-0785-5
- 14. Logue OC, George EM, Bidwell GL 3rd. Preeclampsia and the brain: neural control of cardiovascular changes during pregnancy and neurological outcomes of preeclampsia. *Clin Sci (Lond)*. 2016;130(16):1417-1434. doi:10.1042/CS20160108
- 15. Yousif D, Bellos I, Penzlin AI, et al. Autonomic Dysfunction in Preeclampsia: A Systematic Review. *Front Neurol*. 2019;10:816. doi:10.3389/fneur.2019.00816
- 16. Reyes LM, Usselman CW, Davenport MH, Steinback CD. Sympathetic Nervous System Regulation in Human Normotensive and Hypertensive Pregnancies. *Hypertens (Dallas, Tex* 1979). 2018;71(5):793-803. doi:10.1161/HYPERTENSIONAHA.117.10766
- 17. Spradley FT. Sympathetic nervous system control of vascular function and blood pressure during pregnancy and preeclampsia. *J Hypertens*. 2019;37(3):476-487. doi:10.1097/HJH.00000000001901

- 18. Moors S, Staaks KJJ, Westerhuis MEMH, et al. Heart rate variability in hypertensive pregnancy disorders: A systematic review. *Pregnancy Hypertens*. 2020;20:56-68. doi:10.1016/j.preghy.2020.03.003
- 19. Subha M, Pal P, Pal GK, Habeebullah S, Adithan C, Sridhar MG. Decreased baroreflex sensitivity is linked to sympathovagal imbalance, low-grade inflammation, and oxidative stress in pregnancy-induced hypertension. *Clin Exp Hypertens (New York, NY 1993)*. 2016;38(8):666-672. doi:10.1080/10641963.2016.1200596 [doi]
- 20. Pal GK, Shyma P, Habeebullah S, Pal P, Nanda N, Shyjus P. Vagal withdrawal and sympathetic overactivity contribute to the genesis of early-onset pregnancy-induced hypertension. *Int J Hypertens*. 2011;2011:361417. doi:10.4061/2011/361417 [doi]
- 21. Pal GK, Shyma P, Habeebullah S, Shyjus P, Pal P. Spectral analysis of heart rate variability for early prediction of pregnancy-induced hypertension. *Clin Exp Hypertens* (*New York*, *NY 1993*). 2009;31(4):330-341. doi:10.1080/10641960802621333 [pii]
- 22. Noben L, Verdurmen KMJ, Warmerdam GJJ, Vullings R, Oei SG, van Laar JOEH. The fetal electrocardiogram to detect the effects of betamethasone on fetal heart rate variability. *Early Hum Dev.* 2019;130:57-64. doi:10.1016/j.earlhumdev.2019.01.011
- Verdurmen KM, Renckens J, van Laar JO, Oei SG. The influence of corticosteroids on fetal heart rate variability: a systematic review of the literature. *Obstet Gynecol Surv.* 2013;68(12):811-824. doi:10.1097/OGX.0000000000000009 [doi]
- 24. Verdurmen KMJ, Hulsenboom ADJ, van Laar JOEH, Oei SG. Effect of tocolytic drugs on fetal heart rate variability: a systematic review. *J Matern Fetal Neonatal Med.* 2017;30(20):2387-2394. doi:10.1080/14767058.2016.1249844 [doi]
- 25. Nensi A, Silva DA De, von Dadelszen P, et al. Effect of magnesium sulphate on fetal heart rate parameters: a systematic review. *J Obstet Gynaecol Can.* 2014;36(12):1055-1064. doi:S1701-2163(15)30382-0 [pii]
- 26. Khlybova SV, Tsirkin VI, Dvorianskii SA, Makarova IA, Trukhin AN. Heart rate variability in normal and complicated pregnancies. *Fiziol Cheloveka*. 2008;34(5):97-105.
- 27. Georgiou K, Larentzakis A V, Khamis NN, Alsuhaibani GI, Alaska YA, Giallafos EJ. Can Wearable Devices Accurately Measure Heart Rate Variability? A Systematic Review. *Folia Med (Plovdiv)*. 2018;60(1):7-20. doi:10.2478/folmed-2018-0012
- 28. Choi A, Shin H. Photoplethysmography sampling frequency: pilot assessment of how low can we go to analyze pulse rate variability with reliability? *Physiol Meas*. 2017;38(3):586-600. doi:10.1088/1361-6579/aa5efa
- 29. Alfirevic Z, Devane D, Gyte GM, Cuthbert A. Continuous cardiotocography (CTG) as a form of electronic fetal monitoring (EFM) for fetal assessment during labour. *Cochrane database Syst Rev.* 2017;2:CD006066. doi:10.1002/14651858.CD006066.pub3 [doi]
- 30. Nederlandse Vereniging voor Obstetrie en Gynaecologie. NVOG-Richtlijn Intrapartum Foetale Bewaking à Terme.; 2019.
- 31. Vullings R, Verdurmen KMJ, Hulsenboom ADJ, et al. The electrical heart axis and ST events in fetal monitoring: A post-hoc analysis following a multicentre randomised controlled trial. *PLoS One*. 2017;12(4):e0175823. doi:10.1371/journal.pone.0175823
- 32. Hulsenboom ADJ, Warmerdam GJJ, Weijers J, et al. Head orientation and electrode placement potentially influence fetal scalp ECG waveform. *PLoS One*. 2019;14(10):e0223282. doi:10.1371/journal.pone.0223282
- 33. Hulsenboom ADJ, Verdurmen KMJ, Vullings R, et al. Relative versus absolute rises in T/QRS ratio by ST analysis of fetal electrocardiograms in labour: A case-control pilot study. *PLoS One*. 2019;14(3):e0214357. doi:10.1371/journal.pone.0214357
- 34. Westerhuis MEMH, Strasser SM, Moons KGM, Mol BWJ, Visser GHA, Kwee A. [Intrapartum foetal monitoring: from stethoscope to ST analysis of the ECG]. *Ned Tijdschr Geneeskd*. 2009;153:B259.

- 35. Nederlandse Vereniging voor Obstetrie en Gynaecologie. *Intrapartum Foetale Bewaking a Terme*.; 2014.
- Bullens LM, van Runnard Heimel PJ, van der Hout-van der Jagt MB, Oei SG. Interventions for Intrauterine Resuscitation in Suspected Fetal Distress During Term Labor: A Systematic Review. *Obstet Gynecol Surv.* 2015;70(8):524-539. doi:10.1097/ OGX.00000000000215 [doi]
- 37. Fawole B, Hofmeyr GJ. Maternal oxygen administration for fetal distress. *Cochrane database Syst Rev.* 2012;12:CD000136.doi:10.1002/14651858.CD000136.pub2[doi]
- Ayres-de-Campos D, Spong CY, Chandraharan E, Panel FIFMEC. FIGO consensus guidelines on intrapartum fetal monitoring: Cardiotocography. *Int J Gynaecol Obstet*. 2015;131(1):13-24. doi:10.1016/j.ijgo.2015.06.020 [doi]
- Hamel MS, Anderson BL, Rouse DJ. Oxygen for intrauterine resuscitation: of unproved benefit and potentially harmful. *Am J Obstet Gynecol*. 2014;211(2):124-127. doi:10.1016/j.ajog.2014.01.004 [doi]
- 40. Devane D, Lalor JG, Daly S, McGuire W, Cuthbert A, Smith V. Cardiotocography versus intermittent auscultation of fetal heart on admission to labour ward for assessment of fetal wellbeing. *Cochrane database Syst Rev.* 2017;1:CD005122. doi:10.1002/14651858. CD005122.pub5 [doi]
- 41. Schiermeier S, Pildner von Steinburg S, Thieme A, et al. Sensitivity and specificity of intrapartum computerised FIGO criteria for cardiotocography and fetal scalp pH during labour: multicentre, observational study. *BJOG*. 2008;115(12):1557-1563. doi:10.1111/j.1471-0528.2008.01857.x
- 42. Haydon ML, Gorenberg DM, Nageotte MP, et al. The effect of maternal oxygen administration on fetal pulse oximetry during labor in fetuses with nonreassuring fetal heart rate patterns. *Am J Obstet Gynecol*. 2006;195(3):735-738. doi:S0002-9378(06)00867-2 [pii]
- 43. Bullens LM, van der Hout-van der Jagt MB, Van Runnard Heimel PJ, Oei G. A simulation model to study maternal hyperoxygenation during labor. *Acta Obstet Gynecol Scand*. 2014;93(12):1268-1275. doi:10.1111/aogs.12486 [doi]
- 44. Bernardes J, Ayres-de-Campos D. Poor reliability of visual analysis of fetal heart rate tracings: what should be done about it? *Am J Obstet Gynecol*. 2012;206(6):e6. doi:10.1016/j.ajog.2012.02.027
- 45. Ayres-de-Campos D, Bernardes J, Costa-Pereira A, Pereira-Leite L. Inconsistencies in classification by experts of cardiotocograms and subsequent clinical decision. *Br J Obstet Gynaecol*. 1999;106(12):1307-1310. doi:10.1111/j.1471-0528.1999.tb08187.x
- 46. Paul RH, Suidan AK, Yeh S, Schifrin BS, Hon EH. Clinical fetal monitoring. VII. The evaluation and significance of intrapartum baseline FHR variability. *Am J Obstet Gynecol.* 1975;123(2):206-210.
- 47. Durosier LD, Green G, Batkin I, et al. Sampling rate of heart rate variability impacts the ability to detect acidemia in ovine fetuses near-term. *Front Pediatr.* 2014;2:38. doi:10.3389/fped.2014.00038
- 48. Hemming K, Haines TP, Chilton PJ, Girling AJ, Lilford RJ. The stepped wedge cluster randomised trial: rationale, design, analysis, and reporting. *BMJ*. 2015;350:h391. doi:10.1136/bmj.h391
- 49. Dekkers OM. [The stepped wedge design]. Ned Tijdschr Geneeskd. 2012;156(9):A4069.





English summary

SUMMARY

Monitoring and management of mother and fetus at risk

Although most pregnancies and deliveries are uneventful, complications during pregnancy and childbirth can have serious and long-term consequences for both the mother and her baby. Monitoring can aid in early detection of those complications and enable appropriate management, thereby preventing adverse outcomes. Therefore, it is very important to adequately monitor and manage mothers and their fetuses.

This thesis describes the monitoring and management of maternal and fetal complications during pregnancy in two parts, respectively.

In **chapter 1**, a general introduction is provided on the physiological adaptations that occur in uncomplicated pregnancies, and on the monitoring and management of fetal and maternal pregnancy complications. Furthermore, the study objectives of this thesis are introduced.

Part I of this thesis concentrates on the monitoring and management of mothers at risk of hypertensive pregnancy disorders (HPD). As HPD are a major cause of maternal and fetal morbidity and mortality worldwide, accurate prediction of HPD followed by appropriate management can have major benefits for maternal, fetal, and neonatal health. Currently, risk prediction is mainly based on maternal history, which has limited predictive ability.^{1,2} To improve the quality of prediction, numerous studies have evaluated the predictive abilities of various tests.³ However, so far, none of those tests showed adequate performance.³

Therefore, there is a need for new methods for accurate prediction and early detection of HPD. In part I of this thesis, two promising prediction methods, namely speckle tracking echocardiography (STE) and heart rate variability (HRV), are evaluated on their ability to distinguish HPD from normotensive pregnancies.

STE is a relatively new diagnostic method that analyzes the motion of tissues in the heart by using acoustic reflections called speckles to measure strain. As HPD causes subtle changes in myocardial tissue, it is hypothesized that STE could aid in the prediction and early detection of HPD. Hence, in **chapter 2**, we performed a systematic review to study whether STE is a suitable method to detect differences in cardiac function in pregnant women with HPD compared to normotensive pregnant women. This review also studies possible STE differences between women with a history of a pregnancy complicated by HPD and women with a history of uncomplicated pregnancy. The databases of Medline (Pubmed), EMBASE, and Central were systematically searched and 200 articles were identified. These articles were screened, leaving 16 articles that met our inclusion criteria. These 16 studies included 870 women with (a history of) HPD and 693 normotensive controls.

Of the three directions of strain, left ventricular (LV) global longitudinal strain was decreased the most in pregnant women with HPD, with a significant decrease in both preeclampsia and gestational hypertension. Also, LV global radial strain and LV global circumferential strain were found to be decreased in women with early-onset and severe preeclampsia compared to normotensive pregnant controls, whereas no differences were found in women with late-onset and mild preeclampsia.

Lasting myocardial changes were seen in women with a history of early-onset preeclampsia, with a significant decrease in all three directions of LV strain. In contrast, no significant alterations in strain were found between women with a history of late-onset preeclampsia compared to a history of uncomplicated pregnancy.

In addition to STE, heart rate variability (HRV) is another promising method to study the differences between normotensive and hypertensive pregnancies. HRV can be used as a proxy measure for autonomic nervous system (ANS) activity, and several studies state that HPD might be associated with a dysfunction of the ANS.^{4,5}

In order to assess the value of HRV in detecting HPD, **Chapter 3** provides the results of a systematic review of previous studies comparing HRV in women with HPD or a history of a pregnancy complicated by HPD with normotensive controls. A systematic search was performed in the databases of Medline (Pubmed), EMBASE, and Central in which initially 523 articles were obtained. A total of 24 studies were found eligible and were included in this review, comprising 850 women with (a history of) HPD and 1205 normotensive controls.

The results of the included studies showed a wide variety, possibly due to the heterogenicity of the data. In frequency-domain features, the low frequency/high frequency-ratio (LF/HF-ratio) was increased in women with (moderate) preeclampsia compared to normotensive pregnant controls in four out of six studies. None of the included studies demonstrated a decrease in LF/HF ratio in patients with preeclampsia. Similar results were found for gestational hypertension and chronic hypertension.

In time-domain features, standard deviation of normal-to-normal interval (SDNN) and HRV triangular index, both indices reflecting overall HRV, were decreased in preeclampsia compared to normotensive pregnant controls. These results may be suggestive of a sympathetic overactivity of the autonomic nervous system, possibly associated with a parasympathetic withdrawal. However, due to the large diversity in the results of the included studies, caution should be applied when drawing these conclusions.

Unfortunately, the included studies reported very little on the medication used in their study population, causing a possibly important confounding factor in many studies. Women with pre-term HPD are likely to receive routine medication such as antenatally administered corticosteroids, however, little is known on how these medications affect maternal HRV.

Therefore, we designed a longitudinal cohort study, known as the MAMA-hart study, to study the effects of several routine obstetrical medications on maternal HRV, as presented in **chapter 4**. With the use of a wrist-worn device, continuous, long-term photoplethysmography (PPG) measurements will be obtained to derive HRV features. The study comprises two phases. In the *primary phase*, patients between 23+5 and 33+6 weeks of gestation with an indication to receive corticosteroids antenatally will be included, and PPG measurements will be obtained throughout the subjects' hospitalization. The study mainly focuses on the effect of antenatally administered corticosteroids, which are frequently used and are known to affect fetal HRV.^{6,7} Yet, there is an evident knowledge gap regarding its effect on maternal HRV. In addition, we will also study Magnesium Sulphate, Nifedipine, Labetalol, and Methyldopa, as they are used routinely in obstetric care.

HRV features will be compared between the epoch prior to the medication admission and the epoch after medication is administered. By quantifying the effect of routinely used obstetric medication on maternal HRV, their confounding effect can be accounted for in studies employing maternal HRV. Hence, future studies will be enabled to distinguish changes in HRV associated with pregnancy complications like HPD and preterm labor from those resulting from the use of medication. This may aid in a better understanding of how dysfunction of the ANS might lead to pregnancy complications, possibly enabling earlier diagnosis and management of these complications.

During the *secondary phase* of the study, continuous PPG measurement will be acquired in the same patients from the primary phase during a 24-hour monitoring period at six weeks postpartum. The same wrist-worn device as used in the primary phase will be worn in free-living conditions at home during the *secondary phase*. This additional measurement offers a unique opportunity for comparison of cardiovascular parameters between the antepartum and postpartum periods, on which little literature exists.

Part I thus presents the results of our research on monitoring and management of mothers at risk during pregnancy. While showing that two promising methods can provide additional information on pregnancies complicated by HDP, we also identified a need for investigating the effects of routine obstetrical medications on maternal HRV.

Part II of this thesis focusses on the monitoring and management of fetal distress during labor, as the risk of compromised fetal oxygenation is increased during labor due to

reduced oxygen delivery caused by uterine contractions.⁸ To monitor the fetal condition and oxygenation during labor, various methods can be used including a combined registration of FHR in relation to uterine contractions (i.e. cardiotocography), the analysis of the ST-segment of the fetal electrocardiogram (ST-analysis), fetal scalp blood sampling (FSBS), and fetal pulse oximetry.

When fetal distress is suspected, different intrauterine resuscitation techniques can be used to optimize the fetal condition, including maternal repositioning, discontinuation of oxytocin administration, administration of tocolytic drugs, amnioinfusion, iv fluid bolus, and maternal hyperoxygenation. However, evidence on the effect of these intrauterine resuscitation techniques is limited and sometimes conflicting.⁹ This lack of robust evidence is likely to result in variation in clinical guidelines. Therefore, in **chapter 5**, the national guidelines of several Western countries were compared regarding their recommendations on fetal monitoring and management of fetal distress. Eight guidelines were obtained on monitoring of fetal distress during labor. All guidelines advised to use FSBS in addition to cardiotocography. The use of ST-analysis as part of fetal monitoring was recommended in three guidelines, while it was discouraged in three other guidelines. Two guidelines advised not to use fetal pulse oximetry, whereas the others did not mention this technique in their guidelines.

Five guidelines contained recommendations on the use of intrauterine resuscitation techniques. All guidelines recommended maternal repositioning, the discontinuation of oxytocin, and the use of tocolytic drugs. Regarding maternal hyperoxygenation, the guidelines were contradictory. In the United States of America and Canada, maternal hyperoxygenation is recommended in case of fetal distress, whereas the national guideline from the United Kingdom advises against this treatment due to the possible harmful side-effects and lack of solid evidence. The recommendations on the use of amnioinfusion and IV fluid bolus also varied between the different guidelines.

It seemed likely that the differences in recommendations would lead to variation in clinical practice. To test this hypothesis, a survey was conducted in all 86 Dutch hospitals on their diagnostic and therapeutic methods in case of suspected fetal distress. All hospitals use cardiotocography to monitor the fetal condition, and additional FSBS is available in 98% of Dutch hospitals. In 23% of the hospitals, ST-analysis is used to monitor the fetal condition. Whilst discontinuation of oxytocin infusion, administration of tocolytic drugs, and maternal repositioning are implemented in almost 100% of Dutch hospitals in case of suspected fetal distress, a large practice variation is shown in the use of maternal hyperoxygenation and amnioinfusion, in 58% and 33% of Dutch hospitals, respectively.

These differences in clinical practice and international recommendations may be attributed to the lack of solid evidence from clinical studies. With regard to maternal

hyperoxygenation, its effect in the presence of suspected fetal distress in women during the second stage of labor has not yet been evaluated in a randomized controlled trial (RCT). However, several small, non-randomized studies as well as a mathematical computer model showed an improvement in fetal heart rate (FHR) pattern and/or fetal scalp pH after applying this technique in case of fetal distress.^{10–14} Supported by the results from these studies, we designed and conducted an RCT to study the effect of this intrauterine resuscitation technique during the second stage of labor, of which the study protocol is described in **chapter 6**. In this study, known as the INTEREST O2 study, patients with suspected fetal distress during the second stage of term labor will be randomized to receive either maternal hyperoxygenation with 100% oxygen at 10L/ min applied by a non-rebreathing mask, or conventional care without additional oxygen. Suspected fetal distress is defined as an intermediary or abnormal FHR tracing according to the modified FIGO classification.¹⁵

The primary outcome of this study is the change in FHR pattern after the start of the study, measured as the change in depth and duration of FHR decelerations. This outcome represents a surrogate endpoint of fetal well-being, as a non-reassuring FHR pattern is considered to be indicative of potential fetal hypoxia.^{16,17} Moreover, a study powered for neonatal morbidity would require a sample size of over 10,000 women.¹⁸ Because some studies raised concerns about the potentially harmful effects of maternal hyperoxygenation, we chose not to expose such a large group of women and their fetuses to this intervention before its safety had been further investigated.

The secondary outcomes of the INTEREST O2 study include the change in FIGO classification, neonatal Apgar score, umbilical cord blood gas analyses, free oxygen radical activity, NICU admission, neonatal death, mode of delivery, and maternal side effects. Furthermore, detailed FHR analyses will be performed based on calculation of unstable baseline, phase-rectified signal averaging (PRSA)-based parameters, decelerations with loss of internal variability, and decelerations in combination with fetal bradycardia or tachycardia. All of these parameters have been previously related to poor neonatal outcome.¹⁹⁻²²

The change in FHR will first be analyzed per patient and subsequently compared between the intervention and the control group. For each patient, the 10-minute epoch before start of the study will be compared to the epoch 5 to 15 minutes after the start of the study.

In addition to the intention-to-treat analyses, per-protocol analyses and subgroup analyses will be performed. Two subgroups consist of cases with either intermediate or abnormal FHR patterns (according to FIGO classification), while the third subgroup analysis comprises the group with small for gestational age neonates (birth weight <p10). Perinatal and maternal outcomes of the INTEREST O2-study are described in **chapter 7**. A total of 117 women were included in the study. To evaluate potential harm from free oxygen radical activity, Malondialdehyde in umbilical cord blood was measured, but was not significantly different between the intervention and the control group. Most secondary clinical outcomes did not significantly differ between groups, e.g. Apgar score, umbilical cord blood gas analyses, and mode of delivery. However, in the subgroup with abnormal FHR tracings, there were fewer episiotomies on fetal indication in the intervention group (25%) compared to the control group (65%, p<0.01). This decrease in the intervention group is considered a favorable outcome, as episiotomies are considered a potentially harmful intervention.²³ Moreover, this reduction in episiotomies on fetal indication might indicate a better fetal condition in the intervention group, as fewer fetuses were in such distress that an episiotomy was indicated to immediately terminate the delivery.

Change in FHR could be analyzed for 71 participants (61%), as the other 46 women (39%) gave birth before the end of the time-frame of interest and therefore insufficient FHR tracing existed in the "post-study" time-frame to properly analyze the change in FHR. In this particular work, we studied the effect on FHR as the change in FIGO classification as evaluated by clinicians. Improvement of FIGO classification (e.g. abnormal to intermediary FIGO category) was seen over four times as often in the intervention group compared to the control group (13.9% vs. 2.9%). Furthermore, deterioration of FIGO classification was lower in the intervention group compared to the control group (8.3% vs. 34.3%). The change in FIGO classification in favor of maternal hyperoxygenation was statistically significant (OR 5.7, 95% CI 1.7-19.1). A similar trend was seen in the subgroup analyses, albeit not significant.

This study was powered for FHR, not for neonatal outcome. Yet, the positive effect of maternal hyperoxygenation on FIGO classification may positively reflect on neonatal outcome, however, larger studies powered for neonatal outcome are needed to confirm this finding.

The primary outcome of the INTEREST O2 study is the change in FHR pattern. Although visual analysis of the FHR, as used for the FIGO classification in chapter 7, is one of the most used methods to evaluate FHR, it is known to have a large intra- and inter-observer variability.^{24,25} Therefore, we also performed a more detailed, computerized analysis of the change in FHR of the INTEREST O2 study, the results of which are presented in **chapter 8**. This detailed analysis provides additional knowledge of the physiological changes caused by maternal hyperoxygenation.

Differences were seen in the depth and duration of FHR decelerations between maternal hyperoxygenation (-2.3%) compared to the control group (+10%). These differences were not statistically significant (p=0.24), which may be (partly) due to the higher than

Chapter 10

accommodated fraction of missing FHR data in the group of women that already gave birth before the end of the study time-frame.

The change in PRSA-average acceleration capacity (PRSA-AAC) and PRSA-deceleration capacity (PRSA-ADC) after the start of the study was statistically significant in favor of maternal hyperoxygenation (p=0.03 and p=0.02, respectively). These significant differences in PRSA-based parameters in favor of maternal hyperoxygenation indicate less compensatory activation of the autonomic nervous system after maternal hyperoxygenation, which may positively reflect on neonatal outcome.

No significant differences were found between the intervention and control group for the following FHR features; periods with unstable FHR baseline, decelerations with loss of internal variability, and decelerations in combination with tachycardia or bradycardia.

The results of the INTEREST O2 study show a significantly positive effect of maternal hyperoxygenation on FHR as evaluated with the use of FIGO classification and PRSA based parameters. The difference in deceleration depth and duration, however, was not significant. No harmful effects of maternal hyperoxygenation were demonstrated.

In **chapter 9**, the main findings of this thesis are discussed and subsequently, recommendations for future research are proposed.

The main conclusions of this thesis are:

- 1. Speckle tracking echocardiography is a suitable method to detect preclinical differences in cardiac function in pregnant women with HPD compared to normotensive pregnant women, especially with the use of LV global longitudinal strain. In women with a history of HPD, only the subgroup of early-onset preeclampsia showed lasting myocardial changes compared to women with a history of a normotensive pregnancy.
- 2. Some differences in autonomic nervous system functioning can be detected with the use of HRV in women with HPD, as a decreased overall HRV was found in preeclampsia, compared to normotensive pregnant controls. A trend is seen towards increased LF/HF-ratio in women with PE compared to normotensive pregnant controls.
- 3. The large heterogenicity in studies reporting on HRV, combined with the lack of knowledge on the possible confounding effect of routine obstetric medication, provide difficulty in the interpretation of current results from literature. Therefore, there is a need for a prospective study on the effect of routine obstetric medication on maternal HRV.
- 4. Major differences can be found in the recommendations of international guidelines regarding the monitoring and management of fetal distress during labor.

- 5. When fetal distress is suspected, the use of discontinuation of oxytocin and administration of tocolytics drugs are implemented in all Dutch hospitals. Maternal repositioning is used in 98% of Dutch hospitals, whereas maternal hyperoxygenation is applied 58% and amnioinfusion is used in 33% of hospitals.
- 6. Maternal hyperoxygenation in case of fetal distress during the second stage of term labor has a positive effect on FHR, with a significant positive effect on FIGO classification and PRSA-based FHR parameters. The depth and duration of decelerations also decreased after maternal hyperoxygenation, albeit not significant.
- 7. No harmful side effects of maternal hyperoxygenation on neonatal or maternal outcome, especially no increase in free oxygen radical activity or decrease in umbilical cord pH.

REFERENCES

- 1. Poon LCY, Kametas NA, Chelemen T, Leal A, Nicolaides KH. Maternal risk factors for hypertensive disorders in pregnancy: a multivariate approach. *J Hum Hypertens*. 2010;24(2):104-110. doi:10.1038/jhh.2009.45
- 2. Verghese L, Alam S, Beski S, Thuraisingham R, Barnes I, MacCallum P. Antenatal screening for pre-eclampsia: evaluation of the NICE and pre-eclampsia community guidelines. *J Obstet Gynaecol*. 2012;32(2):128-131. doi:10.3109/01443615.2011.63 5224
- 3. Townsend R, Khalil A, Premakumar Y, et al. Prediction of pre-eclampsia: review of reviews. *Ultrasound Obstet Gynecol*. 2019;54(1):16-27. doi:10.1002/uog.20117 [doi]
- 4. Reyes LM, Usselman CW, Davenport MH, Steinback CD. Sympathetic Nervous System Regulation in Human Normotensive and Hypertensive Pregnancies. *Hypertens (Dallas, Tex* 1979). 2018;71(5):793-803. doi:10.1161/HYPERTENSIONAHA.117.10766
- 5. Julius S, Majahalme S. The changing face of sympathetic overactivity in hypertension. Ann Med. 2000;32(5):365-370. doi:10.3109/07853890008995939
- Verdurmen KM, Renckens J, van Laar JO, Oei SG. The influence of corticosteroids on fetal heart rate variability: a systematic review of the literature. *Obstet Gynecol Surv*. 2013;68(12):811-824. doi:10.1097/OGX.0000000000000009 [doi]
- 7. Noben L, Verdurmen KMJ, Warmerdam GJJ, Vullings R, Oei SG, van Laar JOEH. The fetal electrocardiogram to detect the effects of betamethasone on fetal heart rate variability. *Early Hum Dev.* 2019;130:57-64. doi:10.1016/j.earlhumdev.2019.01.011
- 8. Yli BM, Kjellmer I. Pathophysiology of foetal oxygenation and cell damage during labour. Best Pract Res Clin Obstet Gynaecol. 2016;30:9-21. doi:10.1016/j.bpobgyn.2015.05.004
- Bullens LM, van Runnard Heimel PJ, van der Hout-van der Jagt MB, Oei SG. Interventions for Intrauterine Resuscitation in Suspected Fetal Distress During Term Labor: A Systematic Review. *Obstet Gynecol Surv.* 2015;70(8):524-539. doi:10.1097/ OGX.00000000000215 [doi]
- 10. Bullens LM, van der Hout-van der Jagt MB, Van Runnard Heimel PJ, Oei G. A simulation model to study maternal hyperoxygenation during labor. *Acta Obstet Gynecol Scand.* 2014;93(12):1268-1275. doi:10.1111/aogs.12486 [doi]
- 11. Althabe Jr O, Schwarcz RL, Pose S V, Escarcena L, Caldeyro-Barcia R. Effects on fetal heart rate and fetal pO2 of oxygen administration to the mother. *Am J Obstet Gynecol.* 1967;98(6):858-870. doi:0002-9378(67)90205-0 [pii]
- 12. Gare DJ, Shime J, Paul WM, Hoskins M. Oxygen administration during labor. *Am J Obstet Gynecol.* 1969;105(6):954-961. doi:0002-9378(69)90104-5 [pii]
- 13. Haydon ML, Gorenberg DM, Nageotte MP, et al. The effect of maternal oxygen administration on fetal pulse oximetry during labor in fetuses with nonreassuring fetal heart rate patterns. *Am J Obstet Gynecol*. 2006;195(3):735-738. doi:S0002-9378(06)00867-2 [pii]
- 14. Hidaka A, Komatani M, Ikeda H, Kitanaka T, Okada K, Sugawa T. A comparative study of intrauterine fetal resuscitation by beta-stimulant and O2 inhalation. *Asia-Oceania J Obstet Gynaecol.* 1987;13(2):195-200.
- 15. Ayres-de-Campos D, Spong CY, Chandraharan E, Panel FIFMEC. FIGO consensus guidelines on intrapartum fetal monitoring: Cardiotocography. *Int J Gynaecol Obstet*. 2015;131(1):13-24. doi:10.1016/j.ijgo.2015.06.020 [doi]

- 16. Alfirevic Z, Devane D, Gyte GM. Continuous cardiotocography (CTG) as a form of electronic fetal monitoring (EFM) for fetal assessment during labour. *Cochrane database Syst Rev.* 2013;(5):CD0060(5):CD006066. doi:10.1002/14651858.CD006066.pub2 [doi]
- 17. Elliott C, Warrick PA, Graham E, Hamilton EF. Graded classification of fetal heart rate tracings: association with neonatal metabolic acidosis and neurologic morbidity. *Am J Obstet Gynecol.* 2010;202(3):258.e1-8. doi:10.1016/j.ajog.2009.06.026
- 18. Hamel MS, Anderson BL, Rouse DJ. Oxygen for intrauterine resuscitation: of unproved benefit and potentially harmful. *Am J Obstet Gynecol*. 2014;211(2):124-127. doi:10.1016/j.ajog.2014.01.004 [doi]
- 19. Georgieva A, Payne SJ, Moulden M, Redman CW. Relation of fetal heart rate signals with unassignable baseline to poor neonatal state at birth. *Med Biol Eng Comput.* 2012;50(7):717-725. doi:10.1007/s11517-012-0923-7 [doi]
- 20. Georgieva A, Papageorghiou AT, Payne SJ, Moulden M, Redman CW. Phase-rectified signal averaging for intrapartum electronic fetal heart rate monitoring is related to acidaemia at birth. *BJOG*. 2014;121(7):889-894. doi:10.1111/1471-0528.12568 [doi]
- 21. Kazandi M, Sendag F, Akercan F, Terek MC, Gundem G. Different types of variable decelerations and their effects to neonatal outcome. *Singapore Med J.* 2003;44(5):243-247.
- 22. Holzmann M, Wretler S, Cnattingius S, Nordstrom L. Cardiotocography patterns and risk of intrapartum fetal acidemia. *J Perinat Med.* 2015;43(4):473-479. doi:10.1515/jpm-2014-0105
- 23. Jiang H, Qian X, Carroli G, Garner P. Selective versus routine use of episiotomy for vaginal birth. *Cochrane database Syst Rev.* 2017;2(2):CD000081. doi:10.1002/14651858. CD000081.pub3
- 24. Bernardes J, Ayres-de-Campos D. Poor reliability of visual analysis of fetal heart rate tracings: what should be done about it? *Am J Obstet Gynecol.* 2012;206(6):e6. doi:10.1016/j.ajog.2012.02.027
- 25. Ayres-de-Campos D, Bernardes J, Costa-Pereira A, Pereira-Leite L. Inconsistencies in classification by experts of cardiotocograms and subsequent clinical decision. *Br J Obstet Gynaecol*. 1999;106(12):1307-1310. doi:10.1111/j.1471-0528.1999.tb08187.x





Nederlandse samenvatting

NEDERLANDSE SAMENVATTING

Alhoewel de meeste zwangerschappen en bevallingen zonder problemen verlopen, kunnen complicaties tijdens de zwangerschap en bevalling ernstige en langdurige gevolgen hebben voor zowel moeders als hun baby's. Monitoring kan bijdragen aan het vroegtijdig opsporen van zulke complicaties, waardoor passende behandeling kan plaatsvinden. Hiermee kunnen nadelige gevolgen voorkomen worden. Het is daarom erg belangrijk om moeders en hun ongeboren kinderen adequaat te monitoren en behandelen.

Dit proefschrift beschrijft de monitoring en behandeling van respectievelijk maternale en foetale complicaties tijdens de zwangerschap in twee gedeeltes. In **hoofdstuk 1** wordt een algemene introductie gegeven over de fysiologische veranderingen die optreden tijdens een ongecompliceerde zwangerschap en over het monitoren en behandelen van foetale en maternale zwangerschapscomplicaties. Tevens worden de onderzoeksdoelstellingen van dit proefschrift geïntroduceerd.

Deel I van dit proefschrift gaat over de monitoring en behandeling van moeders met een risico op hypertensieve aandoeningen in de zwangerschap (HAZ). Aangezien HAZ belangrijke oorzaken zijn van wereldwijde maternale en foetale morbiditeit en mortaliteit kan het nauwkeurige voorspellen van HAZ, gevolgd door gepaste behandeling, grote voordelen hebben voor de maternale, foetale en neonatale gezondheid. Momenteel is de risicovoorspelling voornamelijk gebaseerd op de medische voorgeschiedenis van de moeder, wat een beperkte voorspellende waarde heeft.^{1,2} Tal van studies hebben de voorspellende waarde van verschillende testen onderzocht om zo de kwaliteit van de risicovoorspelling te kunnen verbeteren.³ Echter van geen van de tot nu toe onderzochte testmethodes zijn de testeigenschappen voldoende voor gebruik in de dagelijkse praktijk.³

Daarom is er vraag naar nieuwe methoden voor nauwkeurige voorspelling en vroege opsporing van HAZ. In deel I van dit proefschrift worden twee veelbelovende methodes voor risicovoorspelling, speckle-tracking echocardiografie (STE) en hartslagvariabiliteit, onderzocht ten aanzien van hun vermogen om HAZ te onderscheiden van normotensieve zwangerschappen.

STE is een relatief nieuwe vorm van diagnostiek die de beweging van weefsels in het hart analyseert door gebruik te maken van akoestische reflecties genaamd speckles om daarmee de parameter "strain" te kunnen meten. Aangezien HAZ subtiele veranderingen in het myocard weefsel veroorzaken, wordt verondersteld dat STE zou kunnen bijdragen aan het voorspellen en vroegtijdig vaststellen van HAZ. Daarom hebben we in **hoofdstuk 2** een systematische review uitgevoerd om te onderzoeken of STE een geschikte methode is om verschillen in cardiale functie te kunnen vaststellen tussen

zwangere vrouwen met HAZ en normotensieve zwangere vrouwen. In deze review wordt ook onderzocht of er verschillen in STE resultaten gezien worden tussen vrouwen met een voorgeschiedenis van HAZ en vrouwen met een voorgeschiedenis van een ongecompliceerde zwangerschap. De databases van Medline (Pubmed), EMBASE en Central werden systematisch doorzocht waarbij we aanvankelijk 200 artikelen vonden. Deze artikelen werden gescreend waarna er 16 geschikte artikelen overbleven met daarin in totaal 870 vrouwen met (een voorgeschiedenis van) HAZ en 693 normotensieve vrouwen.

Van de drie richtingen waarin strain gemeten wordt was linker ventrikel (LV) globale longitudinale strain het meest verminderd bij zwangere vrouwen met HAZ, met een significante afname bij zowel pre-eclampsie als zwangerschapshypertensie. Verder bleken LV globale radiale strain en LV globale circumferentiele strain verlaagd bij vrouwen met vroege en/of ernstige pre-eclampsie in vergelijking met normotensieve zwangere vrouwen, terwijl bij vrouwen met late en/of milde pre-eclampsie geen verschillen werden gevonden in deze twee parameters.

Bij vrouwen met een vroege pre-eclampsie in de voorgeschiedenis werden blijvende veranderingen in het myocard weefsel gezien, met een significante afname van alle drie de richtingen van LV strain. Daarentegen werden geen significante verschillen gezien in strain tussen vrouwen met een voorgeschiedenis van late pre-eclampsie en vrouwen met een ongecompliceerde zwangerschap in de voorgeschiedenis.

Naast STE is hartslagvariabiliteit een andere veelbelovende methode om de verschillen tussen normotensieve en hypertensieve zwangerschappen te onderzoeken. Verschillende studies stellen dat HAZ mogelijk geassocieerd is met disfunctie van het autonome zenuwstelsel,^{4,5} en dat hartslagvariabiliteit kan gebruikt worden als maatstaf voor het functioneren van het autonome zenuwstelsel.

Om te onderzoeken in welke mate hartslagvariabiliteit geschikt is om HAZ op te sporen werd er een systematische review uitgevoerd waarvan de resultaten beschreven zijn in **hoofdstuk 3**. Studies die hartslagvariabiliteit vergeleken tussen vrouwen met (een voorgeschiedenis van) HAZ en normotensieve vrouwen werden onderzocht. De databases van Medline (Pubmed), EMBASE en Central werden systematisch doorzocht, waarbij aanvankelijk 523 artikelen werden verkregen. Na een verdere selectie werden 24 studies opgenomen in deze review die voldeden aan de inclusiecriteria met daarin in totaal 850 vrouwen met (een voorgeschiedenis van) HAZ en 1205 normotensieve vrouwen ter controle.

De resultaten van de geïncludeerde studies lieten grote variatie zien, mogelijk door de heterogeniteit van de data. Bij de frequentie-domein parameters werd een toename gezien in de ratio tussen de laagfrequente en hoogfrequente component van hartslagvariabiliteit (LF/HF-ratio) bij vrouwen met (matig ernstige) pre-eclampsie vergeleken met normotensieve zwangere vrouwen (vier van de zes studies). In geen enkele studie werd een afname van LF/HF-ratio gezien bij vrouwen met pre-eclampsie. Vergelijkbare resultaten werden gezien voor zwangerschapshypertensie en chronische hypertensie.

Wat betreft de tijd-domein parameters waren beide parameters die de totale mate van hartslagvariabiliteit weergeven, SDNN en hartslagvariabiliteit index, verlaagd bij vrouwen met pre-eclampsie vergeleken met normotensieve controles. Deze resultaten wijzen op een overactiviteit van het sympathische zenuwstelsel, wat mogelijk geassocieerd is met een verlaging van de parasympatische activiteit. Echter is voorzichtigheid geboden bij het trekken van deze conclusies door de grote spreiding in de resultaten van de onderzochte studies.

Helaas werd er in de onderzochte studies weinig beschreven over de medicatie die gebruikt werd in de studiepopulatie. Dit kan mogelijk een verstorend effect hebben op de studieresultaten. Het is gebruikelijk dat vrouwen met vroege HAZ bepaalde medicatie krijgen zoals antenataal toegediende corticosteroïden. Er is echter weinig bekend over het effect van dit type medicatie op maternale hartslagvariabiliteit. Daarom hebben we een longitudinale cohort studie opgezet, de MAMA-hart studie, om het effect van verschillende routinematige verloskundige medicamenten op maternale hartslagvariabiliteit te onderzoeken. Het studieprotocol is beschreven in **hoofdstuk 4**. Door gebruik te maken van een apparaat dat om de pols wordt gedragen zullen continue, langdurige fotoplethysmografie (PPG) metingen verricht worden waarmee hartslagvariabiliteit parameters van afgeleid kunnen worden.

De studie bestaat uit twee fases. In de *primaire fase* zullen patiënten worden geïncludeerd met een zwangerschapsduur tussen de 23+5 weken en 33+6 weken die een indicatie hebben voor de toediening van antenatale corticosteroïden. Gedurende de gehele ziekenhuisopname zullen PPG metingen verricht worden. De studie richt zich voornamelijk op het effect van antenataal toegediende corticosteroïden, die veelvuldig gebruikt worden en waarvan bekend is dat ze de foetale hartslagvariabiliteit beïnvloeden.^{6,7} Toch is er een duidelijke kenniskloof over het effect van deze medicatie op maternale hartslagvariabiliteit. Daarnaast zullen we ook het effect onderzoeken van magnesiumsulfaat, nifedipine, labetalol en methyldopa, aangezien deze medicijnen ook routinematig gebruikt worden in de verloskundige zorg.

De parameters van hartslagvariabiliteit zullen vergeleken worden tussen de periode voorafgaand aan de medicatie toediening en de periode nadat de medicatie is toegediend. Door het effect te kwantificeren dat routinematig gebruikte verloskundige medicatie heeft op maternale hartslagvariabiliteit, kan dit voortaan worden meegenomen in de analyses van toekomstige studies over maternale hartslagvariabiliteit. Hierdoor wordt het voor toekomstige studies mogelijk om onderscheid te maken tussen veranderingen in hartslagvariabiliteit die geassocieerd zijn met zwangerschapscomplicaties zoals HAZ en vroeggeboorte en veranderingen die veroorzaakt worden door het gebruik van medicatie. Dit kan bijdragen aan betere kennis over hoe disfunctie van het autonome zenuwstelsel kan leiden tot zwangerschapscomplicaties waardoor mogelijk eerdere diagnose en behandeling van deze complicaties bereikt kan worden.

Tijdens de *secundaire fase* van het onderzoek zullen continue PPG-metingen verkregen worden bij dezelfde patiënten uit de *primaire fase*, welke gedurende een 24-uurs durende monitoringsperiode zullen plaatsvinden zes weken na de bevalling. Hetzelfde om de pols gedragen apparaat dat wordt gebruikt in de *primaire fase* zal tijdens de *secundaire fase* in vrije leefomstandigheden thuis worden gedragen. Deze aanvullende meting biedt de unieke mogelijkheid om cardiovasculaire parameters te vergelijken tussen de periode voor en na de bevalling, waarover nog maar weinig bekend is.

Deel I presenteert de resultaten van ons onderzoek naar monitoring en behandeling van moeders die risico lopen tijdens de zwangerschap. Hoewel we aantoonden dat twee veelbelovende methoden aanvullende informatie kunnen verschaffen over zwangerschappen die gecompliceerd zijn door HAZ, hebben we ook vastgesteld dat er behoefte is aan extra onderzoek naar de effecten van routinematig gebruikte verloskundige medicatie op maternale hartslagvariabiliteit.

Deel II van dit proefschrift richt zich op het monitoren en behandelen van foetale nood tijdens de bevalling aangezien het risico op verlaagde foetale oxygenatie toeneemt tijdens de bevalling ten gevolge van verminderde zuurstoftoevoer wat veroorzaakt wordt door samentrekkingen van de baarmoeder.⁸ Om de foetale conditie en oxygenatie tijdens de bevalling te kunnen monitoren kunnen verschillende methodes gebruikt worden waaronder een gecombineerde registratie van de foetale hartslag in relatie tot de contracties van de baarmoeder (cardiotocogram; CTG), de analyse van het ST-segment van het foetale elektrocardiogram (ST-analyse), microbloedonderzoek (MBO) en foetale pulsoxymetrie.

Wanneer er een verdenking is op foetale nood kunnen verschillende intra-uteriene resuscitatietechnieken worden gebruikt om de foetale conditie te verbeteren, waaronder houdingsveranderingen van de moeder, het stoppen van toediening van oxytocine, het toedienen van tocolytica, amnioninfusie, intraveneuze vloeistofbolus en maternale hyperoxygenatie. Het bewijs over het effect van deze intra-uteriene resuscitatietechnieken is echter beperkt en soms ook tegenstrijdig.⁹ Het is aannemelijk dat dit gebrek aan robuust bewijs zal leiden tot variatie in klinische richtlijnen. Daarom werden in **hoofdstuk 5** de nationale richtlijnen van verschillende westerse landen vergeleken ten aanzien van hun aanbevelingen over foetale monitoring en behandeling bij verdenking op foetale nood tijdens de bevalling. Er werden acht richtlijnen verkregen

met betrekking tot het monitoren van de foetale conditie tijdens de bevalling. Al deze richtlijnen adviseerden om MBO's te verrichten in aanvulling op het CTG. Het gebruik van ST-analyse werd aanbevolen in drie richtlijnen, terwijl het in drie andere richtlijnen juist werd afgeraden. Twee richtlijnen raadden het gebruik van foetale pulsoximetrie af, terwijl de andere deze techniek niet benoemden in hun richtlijn.

Vijf richtlijnen bevatten aanbevelingen ten aanzien van het gebruik van intra-uteriene resuscitatietechnieken. Alle richtlijnen raadden het stopzetten van toediening van oxytocine aan, net als houdingsveranderingen van de moeder en het toedienen van tocolytica. Met betrekking tot maternale hyperoxygenatie waren de richtlijnen tegenstrijdig. In de Verenigde Staten en Canada wordt maternale hyperoxygenatie aanbevolen in geval van foetale nood, terwijl de nationale richtlijn uit het Verenigd Koninkrijk deze behandeling afraadt vanwege de mogelijke schadelijke bijwerkingen en het ontbreken van wetenschappelijk bewijs. De aanbevelingen over het gebruik van amnioninfusie en intraveneuze vloeistofbolus varieerden ook tussen de verschillende richtlijnen.

Het leek voor de hand liggend dat deze verschillen in aanbevelingen zouden leiden tot verschillen in het klinisch handelen op de werkvloer. Om deze hypothese te testen werd in alle 86 Nederlandse ziekenhuizen een enquête gehouden over de wijze van monitoring en behandeling bij verdenking op foetale nood. Alle ziekenhuizen gebruiken CTG om de foetale conditie te monitoren en in 98% van de Nederlandse ziekenhuizen is tevens MBO beschikbaar indien nodig. In 23% van de ziekenhuizen wordt ST-analyse gebruikt om de foetale conditie te monitoren. Bijna 100% van de Nederlandse ziekenhuizen past de volgende methoden toe in geval van verdenking op foetale nood: het stopzetten van oxytocine-infusie, toediening van tocolytica en houdingsveranderingen van de moeder. Er blijkt echter een grote spreiding te zijn in het toepassen van maternale hyperoxygenatie en amnioninfusie, wat in respectievelijk 58% en 33% van de Nederlandse ziekenhuizen wordt gebruikt.

Deze verschillen in de klinische praktijk en internationale richtlijnen zou verklaard kunnen worden door het gebrek aan goede klinische onderzoeken. Met betrekking tot maternale hyperoxygenatie is het effect van deze behandeling bij verdenking op foetale nood tijdens de uitdrijvingsfase van de bevalling nog nooit onderzocht in een gerandomiseerde studie (RCT). Verschillende kleine, niet-gerandomiseerde onderzoeken en een wiskundig computermodel lieten een verbetering zien in het foetale hartslagpatroon en/of MBO uitslag na toepassing van deze behandeling in het geval van foetale nood.¹⁰⁻¹⁴ Naar aanleiding van de resultaten van deze studies hebben we een RCT opgezet en uitgevoerd om het effect van deze intra-uteriene resuscitatietechniek tijdens de uitdrijvingsfase van de bevalling te onderzoeken. Het studieprotocol hiervan wordt beschreven in **hoofdstuk 6**. In deze studie, genaamd de INTEREST O2 studie, zullen patiënten met verdenking op foetale nood tijdens de uitdrijvingsfase van de bevalling worden geïncludeerd. Zij

zullen via loting worden toegewezen voor het ontvangen van maternale hyperoxygenatie met 100% zuurstof op 10L/min, toegediend via een mond-neus masker, dan wel voor gebruikelijke zorg zonder additionele zuurstof. De diagnose 'verdenking foetale nood' werd gesteld o.b.v. een suboptimaal of abnormaal CTG volgens de gemodificeerde FIGOcriteria.¹⁵

De primaire uitkomstmaat van deze studie is de verandering in het foetale hartslagpatroon na aanvang van de studie, gemeten als de verandering in de diepte en duur van de deceleraties. Dit betreft een surrogaat eindpunt voor de foetale conditie, gezien een afwijkend hartslagpatroon wordt beschouwd als een indicatie voor potentiële foetale hypoxie.^{16,17} Bovendien zou voor een onderzoek dat is gebaseerd op neonatale morbiditeit een studiepopulatie van meer dan 10.000 vrouwen nodig zijn.¹⁸ Aangezien sommige studies bezorgdheid uitten over de mogelijk schadelijke bijwerkingen van maternale hyperoxygenatie, werd gekozen om zo'n grote groep vrouwen en hun baby's niet bloot te stellen aan deze interventie voordat de veiligheid ervan verder onderzocht was.

De secundaire uitkomstmaten van de INTEREST O2 studie bevatten verschillende perinatale en maternale uitkomsten: veranderingen in FIGO-classificatie, Apgar Score, navelstreng bloedgasanalyses, activiteit van vrije zuurstofradicalen, opnames op de neonatale intensive care unit (NICU), neonatale sterfte, wijze van bevallen en maternale bijwerkingen. Verder zullen gedetailleerde analyses van het foetale hartslagpatroon worden uitgevoerd. Deze analyses zijn gebaseerd op instabiele basislijn, PRSAgebaseerde parameters, deceleraties met verlies van variabiliteit en deceleraties in combinatie met foetale bradycardie of tachycardie. Al deze parameters zijn gerelateerd aan een slechtere neonatale uitkomst.¹⁹⁻²²

De veranderingen in het foetale hartslagpatroon worden eerst per patiënt geanalyseerd en vervolgens vergeleken tussen de interventie en de controlegroep. Voor elke patiënt wordt de periode van 10 minuten vóór aanvang van het onderzoek vergeleken met de periode 5 tot 15 minuten na aanvang van het onderzoek.

Naast de intention-to-treat-analyses zullen per-protocol analyses en subgroep analyses worden uitgevoerd. De eerste twee subgroepen bestaan uit patiënten met suboptimaal CTG danwel abnormaal CTG (volgens de FIGO-classificatie). De derde subgroep bestaat uit neonaten met een te laag geboortegewicht (geboortegewicht <p10).

De perinatale en maternale uitkomsten van de INTEREST O2-studie worden beschreven in **hoofdstuk 7**. Er namen 117 patiënten deel aan dit onderzoek. Om de mogelijk schadelijke effecten van vrije zuurstofradicalen te onderzoeken werd Malondialdehyde in navelstrengbloed gemeten. Dit toonde geen significante verschillen tussen de interventie en controlegroep. De volgende secundaire uitkomstmaten toonden geen significante

verschillen tussen beide groepen; Apgar Score, navelstreng bloedgasanalyses en wijze van bevallen. In de subgroep analyse van abnormaal CTG werden significant minder episiotomieën op foetale indicatie geplaatst in de interventie groep (25%) vergeleken met de controlegroep (65%, p<0.01). Deze afname in de interventiegroep wordt als een gunstig resultaat beschouwd, aangezien episiotomieën een potentieel schadelijke interventie zijn.²³ Bovendien zou deze vermindering kunnen wijzen op een betere foetale conditie in de interventiegroep, aangezien er minder vaak sprake was van dusdanige foetale nood dat een episiotomie geïndiceerd was om directe geboorte van het kind na te streven.

Verandering in het foetale hartslagpatroon kon voor 71 vrouwen (61%) worden geanalyseerd, aangezien de andere 46 vrouwen (39%) reeds bevallen was voor het einde van de onderzoeksperiode waardoor er onvoldoende hartslagpatroon bestond in de "post-studie" periode om het goed te kunnen analyseren. In dit artikel werd het effect op foetale hartslagpatroon onderzocht als de veranderingen in FIGO-classificatie, beoordeeld door clinici. Verbetering van de FIGO-classificatie (bijvoorbeeld van abnormaal naar suboptimaal) werd meer dan vier keer zo vaak gezien in de interventiegroep in vergelijking met de controlegroep (13.9% versus 2.9%). Bovendien werd verslechtering van de FIGO-classificatie minder vaak gezien in de interventiegroep dan in de controlegroep (8.3% versus 34.4%). Het verschil in FIGO-classificatie in het voordeel van maternale hyperoxygenatie was statistisch significant (OR 5.7, 95% BI 1,7-19,1). Er werd een gelijke trend gezien in de subgroep analyses, zij het niet significant.

De populatiegrootte van de INTEREST O2 studie was berekend op veranderingen in het foetale hartslagpatroon in plaats van neonatale uitkomst. Het positieve effect van maternale hyperoxygenatie op verandering in FIGO-classificatie zou kunnen wijzen op een positief effect op neonatale uitkomst. Er zijn echter grotere studies nodig om deze bevinding te bevestigen.

De primaire uitkomstmaat van de INTEREST O2 is verandering in foetaal hartslagpatroon. Hoewel visuele analyse van de foetale hartslag, zoals toegepast wordt voor de FIGO-classificatie in hoofdstuk 7, een van de meest gebruikte methodes is om het foetale hartslag patroon te beoordelen, heeft het een grote intra- en interobservatie variabiliteit.^{24,25} Daarom hebben we tevens een meer gedetailleerde computer analyse van het foetale hartslagpatroon in de INTEREST O2 uitgevoerd, waarvan de resultaten staan weergegeven in **hoofdstuk 8**. Deze gedetailleerde analyse geeft aanvullende informatie over de fysiologische veranderingen die veroorzaakt worden door maternale hyperoxygenatie.

Er werden verschillen gezien in duur en diepte van deceleraties tussen maternale hyperoxygenatie (-2.3%) en de controlegroep (+10%). Deze verschillen waren niet

significant (p=0.24), wat te wijten kan zijn aan het hogere percentage van ontbrekende gegevens dan waar vooraf van werd uitgegaan.

De verandering in PRSA- acceleratie capaciteit (PRSA-AAC) en PRSA-deceleratie capaciteit (PRSA-ADC) na de start van de studie was statistisch significant verschillend in het voordeel van maternale hyperoxygenatie (p=0,03 en p=0,02, respectievelijk). Deze significante verschillen in op PRSA gebaseerde parameters in het voordeel van maternale hyperoxygenatie duiden op minder compensatoire activering van het autonome zenuwstelsel na maternale hyperoxygenatie. Dit zou kunnen wijzen op een positief effect op neonatale uitkomst.

Er werden geen significante verschillen gevonden tussen de interventie- en controlegroep voor de volgende foetale hartslagkenmerken: perioden met instabiele basislijn, deceleraties met verlies van variabiliteit en deceleraties in combinatie met tachycardie of bradycardie.

De resultaten van de INTEREST O2 studie laten een significant positief effect zien van maternale hyperoxygenatie op het foetale hartslagpatroon wat betreft veranderingen in FIGO-classificatie en op PRSA gebaseerde parameters. Het verschil in duur en diepte van deceleraties was echter niet significant. Er werden geen schadelijke effecten van maternale hyperoxygenatie gevonden.

In **hoofdstuk 9** worden de belangrijkste bevindingen van dit proefschrift besproken waarna aanbevelingen worden gedaan voor toekomstig onderzoek.

De belangrijkste conclusies van dit proefschrift zijn:

- 1. Speckle tracking echografie is een geschikte methode om preklinische veranderingen in hartfunctie op te sporen bij zwangere vrouwen met HAZ in vergelijking met normotensieve zwangere vrouwen, met name bij gebruik van LV globale longitudinale strain. Bij vrouwen met een voorgeschiedenis van HAZ werden alleen in de subgroep van vroege pre-eclampsie blijvende veranderingen in hartfunctie gevonden in vergelijking met vrouwen met een normotensieve zwangerschap in de voorgeschiedenis.
- 2. Bepaalde verschillen in het functioneren van het autonome zenuwstelsel kunnen worden vastgesteld met behulp van hartslagvariabiliteit in vrouwen met HAZ, zoals een verlaagde hartslagvariabiliteit bij pre-eclampsie vergeleken met normotensieve zwangere vrouwen. Er werd een trend gezien richting een verhoogde LF/HF-ratio bij vrouwen met pre-eclampsie vergeleken met normotensieve zwangere vrouwen.
- 3. De grote heterogeniciteit in de hartslagvariabiliteit studies en het gebrek aan kennis over het mogelijk verstorende effect van routinematige verloskundige medicatie zorgen ervoor dat de resultaten van hartslagvariabiliteit studies niet

goed geïnterpreteerd kunnen worden. Er is daarom behoefte aan een prospectieve studie die het effect van routinematige verloskundige medicatie op maternale hartslagvariabiliteit onderzoekt.

- 4. Er zijn grote verschillen in internationale richtlijnen wat betreft hun aanbevelingen over de monitoring en behandeling van foetale nood tijdens de bevalling.
- 5. Bij verdenking op foetale nood wordt in alle Nederlandse ziekenhuizen gebruik gemaakt van het stoppen van oxytocine en toediening van tocolytica. Het herpositioneren van de moeder wordt toegepast in 98% van de Nederlandse ziekenhuizen. Maternale hyperoxygenatie en amnioninfusie worden toegepast in respectievelijk 58% en 33% van de ziekenhuizen.
- 6. Het toepassen van maternale hyperoxygenatie bij verdenking op foetale nood tijdens de uitdrijvingsfase van de bevalling heeft een positief effect op het foetale hartslagpatroon, met een significant positief effect op verandering in FIGO-classificatie en op PRSA gebaseerde parameters. De duur en diepte van deceleraties nam ook af door maternale hyperoxygenatie, zij het niet significant.
- 7. Maternale hyperoxygenatie heeft geen schadelijke bijwerkingen op neonatale of maternale uitkomst, met name geen toename van vrije zuurstofradicalen activiteit of afname van de pH van navelstrengbloed.

REFERENTIES

- 1. Poon LCY, Kametas NA, Chelemen T, Leal A, Nicolaides KH. Maternal risk factors for hypertensive disorders in pregnancy: a multivariate approach. *J Hum Hypertens*. 2010;24(2):104-110. doi:10.1038/jhh.2009.45
- 2. Verghese L, Alam S, Beski S, Thuraisingham R, Barnes I, MacCallum P. Antenatal screening for pre-eclampsia: evaluation of the NICE and pre-eclampsia community guidelines. *J Obstet Gynaecol*. 2012;32(2):128-131. doi:10.3109/01443615.2011.63 5224
- 3. Townsend R, Khalil A, Premakumar Y, et al. Prediction of pre-eclampsia: review of reviews. *Ultrasound Obstet Gynecol.* 2019;54(1):16-27. doi:10.1002/uog.20117 [doi]
- 4. Reyes LM, Usselman CW, Davenport MH, Steinback CD. Sympathetic Nervous System Regulation in Human Normotensive and Hypertensive Pregnancies. *Hypertens (Dallas, Tex 1979)*. 2018;71(5):793-803. doi:10.1161/HYPERTENSIONAHA.117.10766
- 5. Julius S, Majahalme S. The changing face of sympathetic overactivity in hypertension. Ann Med. 2000;32(5):365-370. doi:10.3109/07853890008995939
- Verdurmen KM, Renckens J, van Laar JO, Oei SG. The influence of corticosteroids on fetal heart rate variability: a systematic review of the literature. *Obstet Gynecol Surv.* 2013;68(12):811-824. doi:10.1097/OGX.0000000000000009 [doi]
- 7. Noben L, Verdurmen KMJ, Warmerdam GJJ, Vullings R, Oei SG, van Laar JOEH. The fetal electrocardiogram to detect the effects of betamethasone on fetal heart rate variability. *Early Hum Dev.* 2019;130:57-64. doi:10.1016/j.earlhumdev.2019.01.011
- 8. Yli BM, Kjellmer I. Pathophysiology of foetal oxygenation and cell damage during labour. Best Pract Res Clin Obstet Gynaecol. 2016;30:9-21. doi:10.1016/j.bpobgyn.2015.05.004
- Bullens LM, van Runnard Heimel PJ, van der Hout-van der Jagt MB, Oei SG. Interventions for Intrauterine Resuscitation in Suspected Fetal Distress During Term Labor: A Systematic Review. *Obstet Gynecol Surv.* 2015;70(8):524-539. doi:10.1097/ OGX.00000000000215 [doi]
- 10. Bullens LM, van der Hout-van der Jagt MB, Van Runnard Heimel PJ, Oei G. A simulation model to study maternal hyperoxygenation during labor. *Acta Obstet Gynecol Scand*. 2014;93(12):1268-1275. doi:10.1111/aogs.12486 [doi]
- 11. Althabe Jr O, Schwarcz RL, Pose S V, Escarcena L, Caldeyro-Barcia R. Effects on fetal heart rate and fetal pO2 of oxygen administration to the mother. *Am J Obstet Gynecol.* 1967;98(6):858-870. doi:0002-9378(67)90205-0 [pii]
- 12. Gare DJ, Shime J, Paul WM, Hoskins M. Oxygen administration during labor. *Am J Obstet* Gynecol. 1969;105(6):954-961. doi:0002-9378(69)90104-5 [pii]
- 13. Haydon ML, Gorenberg DM, Nageotte MP, et al. The effect of maternal oxygen administration on fetal pulse oximetry during labor in fetuses with nonreassuring fetal heart rate patterns. *Am J Obstet Gynecol.* 2006;195(3):735-738. doi:S0002-9378(06)00867-2 [pii]
- 14. Hidaka A, Komatani M, Ikeda H, Kitanaka T, Okada K, Sugawa T. A comparative study of intrauterine fetal resuscitation by beta-stimulant and O2 inhalation. *Asia-Oceania J Obstet Gynaecol.* 1987;13(2):195-200.
- 15. Ayres-de-Campos D, Spong CY, Chandraharan E, Panel FIFMEC. FIGO consensus guidelines on intrapartum fetal monitoring: Cardiotocography. *Int J Gynaecol Obstet*. 2015;131(1):13-24. doi:10.1016/j.ijgo.2015.06.020 [doi]

- 16. Alfirevic Z, Devane D, Gyte GM. Continuous cardiotocography (CTG) as a form of electronic fetal monitoring (EFM) for fetal assessment during labour. *Cochrane database Syst Rev.* 2013;(5):CD0060(5):CD006066. doi:10.1002/14651858.CD0060666.pub2 [doi]
- 17. Elliott C, Warrick PA, Graham E, Hamilton EF. Graded classification of fetal heart rate tracings: association with neonatal metabolic acidosis and neurologic morbidity. *Am J Obstet Gynecol.* 2010;202(3):258.e1-8. doi:10.1016/j.ajog.2009.06.026
- 18. Hamel MS, Anderson BL, Rouse DJ. Oxygen for intrauterine resuscitation: of unproved benefit and potentially harmful. *Am J Obstet Gynecol.* 2014;211(2):124-127. doi:10.1016/j.ajog.2014.01.004 [doi]
- 19. Georgieva A, Payne SJ, Moulden M, Redman CW. Relation of fetal heart rate signals with unassignable baseline to poor neonatal state at birth. *Med Biol Eng Comput.* 2012;50(7):717-725. doi:10.1007/s11517-012-0923-7 [doi]
- 20. Georgieva A, Papageorghiou AT, Payne SJ, Moulden M, Redman CW. Phase-rectified signal averaging for intrapartum electronic fetal heart rate monitoring is related to acidaemia at birth. *BJOG*. 2014;121(7):889-894. doi:10.1111/1471-0528.12568 [doi]
- 21. Kazandi M, Sendag F, Akercan F, Terek MC, Gundem G. Different types of variable decelerations and their effects to neonatal outcome. *Singapore Med J.* 2003;44(5):243-247.
- 22. Holzmann M, Wretler S, Cnattingius S, Nordstrom L. Cardiotocography patterns and risk of intrapartum fetal acidemia. *J Perinat Med.* 2015;43(4):473-479. doi:10.1515/jpm-2014-0105
- 23. Jiang H, Qian X, Carroli G, Garner P. Selective versus routine use of episiotomy for vaginal birth. *Cochrane database Syst Rev.* 2017;2(2):CD000081. doi:10.1002/14651858. CD000081.pub3
- 24. Bernardes J, Ayres-de-Campos D. Poor reliability of visual analysis of fetal heart rate tracings: what should be done about it? *Am J Obstet Gynecol*. 2012;206(6):e6. doi:10.1016/j.ajog.2012.02.027
- 25. Ayres-de-Campos D, Bernardes J, Costa-Pereira A, Pereira-Leite L. Inconsistencies in classification by experts of cardiotocograms and subsequent clinical decision. *Br J Obstet Gynaecol*. 1999;106(12):1307-1310. doi:10.1111/j.1471-0528.1999.tb08187.x





APPENDICES

List of abbreviations List of publications Dankwoord Curriculum Vitae

LIST OF ABBREVIATIONS

ACOG	American College of Obstetricians and Gynecologists	NICU
AGA	Appropriate for gestational age	NN
ANS	Autonomic nervous system	NOS
BMI	Body-mass index	N.u.
BPM	Beats per minute	NVO
СН	Chronic Hypertension	
CI	Confidence interval	OCU
CONSORT	Consolidated Standards of Reporting Trials	OR
CS	Cesarean section	pCO ₂
CTG	Cardiotocography	pO ₂
CVD	Cardiovascular disease	PDL
DNPH	2,4-Dinitrophenylhydrazine;	PE
ECG	Electrocardiography	рНа
EO-PE	Early-onset preeclampsia	pO ₂
FHR	Fetal heart rate	PPG
fHRV	Fetal heart rate variability	PRIS
FIGO	International Federation of Gynecology and Obstetrics	
FSBS	Fetal scalp blood sampling	PRSA
GDPR	General Data Protection Regulation	PRSA
GH	Gestational hypertension	PRSA
HF	High Frequency (power)	PTB
HPD	Hypertensive pregnancy disorders	RCOO
HPLC-MS/MS	High-performance liquid chromatography-tandem mass spectrometry	RCT
HR	Heart rate	RMSS
HRV	Heart rate variability	SAE
LF	Low frequency (power)	SDAN
Ln	Natural logarithm	SDN
LO-PE	Late-onset preeclampsia	SGA
LV	Left ventricle	SOGO
LVEF	Left ventricular ejection fraction	SPIRI
LV-GCS	Left ventricular global circumferential strain	SpO ₂
LV-GLS	Left ventricular global longitudinal strain	SK
LV-GRS	Left ventricular global radial strain	SIAN
MDA	Malondialdehyde	SIE
MgSO ₄	Magnesium sulfate	SUSA
mHR	Maternal heart rate	UCBO
mHRV	Maternal heart rate variability	USA
m/z	Mass-to-charge ratio	

:U	Neonatal intensive care unit	
	Normal-to-normal	
S	Newcastle-Ottawa scale Normalized units	
OG	Dutch Society of Obstetricians and Gynecologists (Nederlandse Vereniging voor Obstetrie en Gynaecologie)	
U	Obstetric care unit	
	Odds ratio	
D_2	Partial carbon dioxide pressure	
2	Partial oxygen pressure	
L	Philips Data Logger	
	Preeclampsia	
a	pH in arterial blood gas	
2	Partial oxygen pressure	
3	Photoplethysmography	
SMA	Preferred Reporting Items for Systematic Reviews and Meta-Analyses checklist	
5A	Phase-rectified signal averaging	
SA-AAC	Phase-rectified signal averaging accelerative capacity	
SA-ADC	Phase-rectified signal averaging decelerative capacity	
3	Preterm birth	
DG	Royal College of Obstetricians and Gynecologists	
Г	Randomized controlled trial	
SSD	Root mean square of successive differences	
	Serious adverse event	
ANN	Standard deviation of the averages of NN intervals	
NN	Standard deviation of all NN intervals	
4	Small for gestational age	
GC	Society of Obstetricians and Gynaecologists of Canada	
RIT	Standard Protocol Items: Recommendations for Interventional Trials	
) ₂	Oxygen saturation	
	Strain rate	
N	ST-analysis	
	Speckle tracking echocardiography	
SAR	Suspected unexpected serious adverse reaction	
BG	Umbilical cord blood gas	
4	United States of America	

LIST OF PUBLICATIONS

SMoors, NHM van Oostrum, C Rabotti, X Long, MEMH Westerhuis, HMC Kemps, SG Oei, JOEH van Laar. Speckle tracking echocardiography in hypertensive pregnancy disorders: a systematic review. *Obstetrical & Gynecological Survey 2020 Aug*;75(8):497-509.

S Moors, KJJ Staaks, MEMH Westerhuis, LRC Dekker, KMJ Verdurmen, SG Oei, JOEH van Laar. Heart rate variability in hypertensive pregnancy disorders: a systematic review. *Pregnancy Hypertension: An International Journal of Women's Cardiovascular Health 20C* (2020) pp. 56-68

S Moors, LM Bullens, PJ van Runnard Heimel, JP Dieleman, W Kulik, DL Bakkeren, ER vd Heuvel, MB van der Hout- van der Jagt, SG Oei. The effect of intrauterine resuscitation by maternal hyperoxygenation; a randomized controlled trial. *American Journal of Obstetrics & Gynecology MFM*; 2020;2:100102.

S Moors, R Joshi, LM Bullens, NHM van Oostrum, M Regis, ER van den Heuvel, JOEH van Laar, SG Oei, MB van der Hout-van der Jagt. A randomized controlled trial studying the effect of maternal hyperoxygenation on fetal heart rate in case of fetal distress. *Physiol Meas.* 2020 Oct 13. doi: 10.1088/1361-6579/abcOb6. Online ahead of print.

M Bester*, **S Moors***, R Joshi, MB van der Hout-van der Jagt, SG Oei, M Mischi, R Vullings, JOEH van Laar. Changes in maternal heart rate variability in response to the administration of routine obstetric medications in hospitalized patients; study protocol for a cohort study (MAMA-heart study). *Submitted for publication in BMJ open.* *equal contribution

J Burd, J Quist-Nelson, **S Moors**, N Raghuraman, H Aly, V Berghella. Effect of intrapartum oxygen on the rate of cesarean delivery: a meta-analysis. *Submitted for publication in American Journal of Obstetrics & Gynecology*

LM Bullens, ADJ Hulsenboom, **S Moors**, R Joshi, PJ van Runnard Heimel, MB van der Hout- van der Jagt, ER van den Heuvel, SG Oei. Intrauterine resuscitation during the second stage of term labour by maternal hyperoxygenation versus conventional care: study protocol for a randomised controlled trial (INTEREST O2). *Trials. 2018 Mar 23*;19(1):195.

LM Bullens, **S Moors**, PJ van Runnard Heimel, MB van der Hout- van der Jagt, SG Oei. Practice variation in the management of intrapartum fetal distress in The Netherlands and the Western world. *Eur J Obstet Gynecol Reprod Biol*. 2016 Oct;205:48-53. MR Hazebroek, **S Moors,** R Dennert, A van den Wijngaard, I Krapels, M Hoos, J Verdonschot, JJ Merken, B de Vries, P Wolffs, HJGM Crijns, HP Brunner-La Rocca, S Heymans. Prognostic relevance of gene-environment interactions in dilated cardiomyopathy patients: applying the MOGES classification. *Journal of the American College of Cardiology 2015 Sep 22;66(12):1313-23.*

CONFERENCE PRESENTATIONS

S Moors, Extra zuurstof tijdens de bevalling, gunstig of gevaarlijk? Refereeravond Gynaecology and Obstetrics, Máxima MC, 12 November 2019, Veldhoven, the Netherlands,

S Moors. Extra zuurstof tijdens de bevalling, gunstig of gevaarlijk? Wetenschapsavond Máxima MC, 28 March 2019, Veldhoven, the Netherlands

S Moors. Intrauterine resuscitation during term labor by maternal hyperoxygenation. European Congress on Intrapartum Care, 14-16 March 2019, Turin, Italy.

S Moors, Summary of fetal monitoring research in Máxima Medical Center. Pre-congress Intrapartum Fetal Monitoring Networking Meeting, European Congress on Intrapartum Care, 14-16 March 2019, Turin, Italy.

CONFERENCE POSTERS

S Moors, NHM van Oostrum, MEMH Westerhuis, SG Oei, JOEH van Laar. Speckle tracking echocardiography in hypertensive pregnancy disorders: a systematic review. e/MTIC Symposium 'Technology meets Value-Based Health Care', 11 October 2019, Eindhoven, the Netherlands

S Moors, LM Bullens, PJ van Runnard Heimel, JP Dieleman, MB van der Hout- van der Jagt, SG Oei. Effect of intrauterine resuscitation by maternal hyperoxygenation during the second stage of term labour on free oxygen radicals; a randomised controlled trial. Birth symposium, 15-17 November 2018, Venice, Italy

S Moors, LM Bullens, PJ van Runnard Heimel, JP Dieleman, MB van der Hout- van der Jagt, SG Oei. Effect of intrauterine resuscitation by maternal hyperoxygenation during the second stage of term labour on neonatal outcome and mode of delivery; a randomised controlled trial. Birth symposium, 15-17 November 2018, Venice, Italy

DANKWOORD

Daar ligt het dan; mijn proefschrift. Er zijn momenten geweest tijdens de totstandkoming ervan dat ik niet voor mogelijk hield dat het zou gaan lukken. Daarom wil ik een aantal mensen heel graag bedanken die mij geholpen hebben deze klus te klaren.

Allereerst wil ik alle patiënten bedanken die hebben deelgenomen aan de INTEREST O2 studie. Bedankt dat jullie tijdens één van de meest bijzondere momenten in jullie leven wilden bijdrage aan dit onderzoek.

Prof. dr. S.G. Oei, beste Guid. Wat heb jij een onuitputtelijke stroom aan ideeën en een enthousiasme om dit alles uit te voeren. Bedankt voor alle kansen en mogelijkheden die je me gegeven hebt. Toen ik als co-assistent bij je kwam aankloppen met de vraag of er nog onderzoeken waren waar ik aan kon meewerken had ik niet kunnen bedenken dat een aantal jaren later dit proefschrift er zou liggen. En of het nou op een terrasje was in Venetië of Turijn, of juist op jouw werkkamer in het ziekenhuis, altijd waren daar weer jouw interesse in de voortgang van mijn promotie en de suggesties hoe problemen opgelost konden worden. Bedankt hiervoor!

Dr. Ir. M.B. van der Hout- van der Jagt, beste Beatrijs. Bedankt dat je me wegwijs hebt gemaakt in de wereld van de wetenschap en uitleg hebt gegeven over alle geschreven en met name ook over de ongeschreven regels hierin (vooral de regel van drie bleek belangrijk te zijn). Op de momenten dat ik niet meer zag hoe het ooit nog goed moest komen met mijn promotie was jij daar om het vertrouwen te herstellen. En ik moet zeggen, je hebt inderdaad gelijk gehad; eind goed, al goed.

Dr. J.O.E.H. van Laar, beste Judith, je bent als laatste toegevoegd aan mijn promotieteam en wat heb ik veel gehad aan jouw adagium schrappen, schrappen, schrappen. Zowel in de opzet van de reviews als in de manuscripten zelf heeft dit het geheel een stuk leesbaarder gemaakt. Bedankt ook voor je interesse in mijn verdere ontwikkeling als arts en je steun hierin.

Beste leden van de beoordelingscommissie, Prof. dr. M.E.A. Spaanderman, Prof. dr. W.P. de Boode, Prof. dr. R.M. Aarts en Prof. dr. L.R.C. Dekker, hartelijk dank voor de bereidheid dit proefschrift kritisch te beoordelen.

Dank voor iedereen die heeft meegeholpen aan de studies, met name ook bedankt voor alle **(semi-)artsen, verloskundigen en verpleegkundigen** die de INTEREST O2 studie hebben mogelijk gemaakt! **Beste Lauren**, bedankt voor al je hulp bij mijn promotie, de fijne samenwerking en dat ik jouw project mocht voortzetten.

Ook bedankt aan alle **co-auteurs** voor alle constructieve feedback op de manuscripten. **Beste Michelle**, bedankt voor je begeleiding aan het begin van mijn promotie en je hulp bij het opzetten van de reviews.

Dear Rohan, a special thanks for the detailed computerized analyses presented in chapter 8. **Dear Maretha**, thank you for the nice collaboration that let to the MAMA-hart trial. Good luck with your own PhD!

Beste Bart de Vries, bedankt voor je hulp met het opzetten van de systematic reviews van hoofdstuk 2 en 3 en het opvragen van de vele artikelen.

Beste gynaecologen van het Máxima MC en het Elisabeth Tweesteden ziekenhuis, bedankt voor de ruimte die ik gekregen heb om mijn klinische werkzaamheden te combineren met een promotietraject. Het is fijn dat ik in jullie team ben opgenomen en hierin de mogelijkheden kreeg om te groeien als mens en als dokter.

Beste FUN-groep, bedankt voor de gezellige en interessante FUN avonden. Nanette en Guid bedankt voor jullie gastvrijheid hierbij. Daarnaast natuurlijk de leuke, gezellige en inspirerende congressen in Venetië en Turijn. Dankjewel allemaal!

Dank aan alle mede-promovendi van het Máxima MC. Wat begon als het claimen van een onderzoekshokje in de MMC academie is ondertussen uitgegroeid tot een volwaardige kippenbalzaal. De nauwe samenwerking tussen de FUN'ers en de moordvrouwen was een fijne toevoeging waarbij laagdrempelig tips&tricks gedeeld konden worden. **Lieve Veerle, Lore, Noortje, Marion, Tamara, Inez, Bettine en Thomas**, succes met het vervolg van jullie promotie-trajecten en bedankt voor de gezelligheid, of het nou op de werkvloer was of bij de (zoom)borrels, escape room of bij de etentjes. Hou vol bij tegenslagen, jullie zijn allemaal toppers!

Lieve pinguïngroep, bedankt voor alle gezelligheid buiten het werk. Bijzonder dat ik nu de tweede Pinguïn doctor zal worden. Komende carnaval zal voor het eerst in jaren geen Pinguïn zondag hebben, maar daar vinden we vast een goed alternatief voor.

Lieve studievriendinnetjes, **lieve Marjolein, Lisa en Bao-Oanh**, wat fijn dat we ondanks de afstand elkaar toch blijven zien. Altijd leuk om te horen hoe het bij andere specialismen eraan toe gaat, maar vooral heerlijk om samen te kunnen borrelen en loempia's te eten!

Lieve buurtjes, bedankt voor jullie interesse in werk en alle eieren/melk/marshmallows die we de afgelopen jaren hebben mogen lenen. Beter een goede buur dan en verre vriend, op nog vele barbecues en ijsjes samen.

Lieve Susan, wat is het toch altijd fijn als we elkaar zien. Hoewel onze werelden enorm van elkaar verschillen en het met corona soms lastig is om 'live' af te spreken vinden we hier elke keer weer een weg in. Ik bewonder je creativiteit, doorzettingsvermogen en interesse in je medemens.

Lieve Vivian. Al sinds de basisschool mijn vriendinnetje. Wat hebben we veel meegemaakt door de jaren heen en wat ben ik trots op wie jij bent. Je steunt me met alles, of het nu mijn promotie is of het organiseren van mijn bruiloft, en daar ben ik je heel dankbaar voor. Wat fijn dat we in drukke tijden elkaar toch altijd even een update sturen hoe het gaat en altijd tijd vinden voor een kop thee samen. Nu dit project af is lijkt het me tijd voor een volgend project; New York 202020 met de mannen!

Lieve paranimfen, lieve Laureen. Zo zette je als getuige je handtekening voor mijn huwelijk en nu sta je naast me als paranimf. Van iedere week ons hele leven bespreken op de fiets naar de dansles toe tot vakanties samen met de mannen naar Italië. Wat ben ik trots dat je het hebt aangedurfd om een nieuwe stap te zetten in je carrière!

Lieve Veerle, mijn workwife. Vanaf het moment dat we van links naar rechts door de MMC academie schoven was het duidelijk dat je veel gezelligheid zou gaan brengen in mijn promotie. Bedankt dat ik altijd bij je mocht spuien, bedankt voor al je feedback en het meedenken in oplossingen. Maar met name ook bedankt voor de gezelligheid, de superfout feestjes en het Snollebollekes concert zullen me zeker bijblijven!

Lieve schoonfamilie, bedankt voor jullie interesse in mijn 'werkstuk' [©]. Het is nooit saai als we met z'n alle samen zijn en er is altijd ruimte voor een grap, een drankje en lekker eten. Sinds bijna 1,5 jaar mag ik me officieel ook een Kaanders noemen, en het voelt heerlijk om onderdeel te zijn van deze gezellige familie.

Lieve familie, bedankt voor jullie belangstelling in mijn werk en dat jullie mijn vreugde deelden als er een artikel (eindelijk) gepubliceerd was. Ondanks dat we allemaal heel verschillend zijn, is het fijn dat we allemaal van elkaars leven op de hoogte blijven. Lieve papa, bedankt voor je luisterend oor; wat leuk om met je te kunnen sparren over de dingen die ik in het ziekenhuis meemaak en om jouw nuchtere huisartsenblik hierop te horen. Natuurlijk ook bedankt voor alle kennis qua lekker eten en goede wijnen, het wordt zeer gewaardeerd!

Lieve mama, bedankt dat ik altijd bij je terecht kan met alles wat mij bezig houdt. De eindeloze telefoontjes in de auto op weg naar huis zijn altijd een fijne afsluiting van mijn werkdag. Het medische wereldje is zeker niet jouw wereld, wat een frisse blik op bepaalde zaken kan bieden. Jouw onvoorwaardelijke steun zorgt ervoor dat ik weer met goede moed verder kan blijven gaan. Bedankt ook voor alle kopjes thee, alle lekkere koekjes en taartjes daarbij en alle gezellige spelletjes die zorgen voor de nodige ontspanning. Lieve Bart, lieve schattie. Wat ben ik dankbaar voor jouw steun, positieve energie en onvoorwaardelijke liefde. De wereld van gynaecologie en verloskunde is soms een ver van jouw bed show en wat is dat toch heerlijk verhelderend. Jouw optimisme, humor en aanmoediging hebben een grote bijdrage gehad dat dit proefschrift hier nu ligt. Jouw brede interesse maakte je een perfect oefenpubliek voor al mijn presentaties. We zijn een goed team dat het beste in elkaar naar boven haalt. Ik ben trots op wat je doet en hoe je jezelf ontwikkeld hebt. Dankjewel dat je bent wie je bent, ik hou van je. Op naar nog meer uitdagingen, samen kunnen we de wereld aan!

CURRICULUM VITAE

Suzanne Moors werd geboren op 25 maart 1992 in Geldrop. Zij groeide op in Eindhoven en in 2010 behaalde zij haar VWO-diploma aan het Augustinianum college te Eindhoven. Hierna is zij Geneeskunde gaan studeren aan de Universiteit van Maastricht. Tijdens haar buitenlandse keuze co-schap heeft zij stage gelopen bij de afdeling General Medicine van het Whakatane hospital in Whakatane, Nieuw-Zeeland. Tijdens haar reguliere Gynaecologie co-schap in het Máxima MC te Veldhoven werd



haar interesse voor dit vakgebied 'geboren' en is zij begonnen met wetenschappelijk onderzoek onder leiding van prof. dr. Oei in de FUNdamentele Perinatologie onderzoeksgroep. Het laatste jaar van de opleiding Geneeskunde was zij semi-arts op de afdeling Gynaecologie en Obstetrie in het Máxima MC, waarbij ze meehielp met het opzetten van de INTEREST O2 studie. Na het behalen van haar artsenexamen in juli 2016, startte ze met werken als fertiliteitsarts en later als arts-assistent niet in opleiding op de afdeling Gynaecologie en Obstetrie van het Zuyderland MC. Per oktober 2017 is ze gaan werken als arts-assistent niet in opleiding op de afdeling Gynaecologie en Obstetrie van het Máxima MC. Tevens bleef zij naast haar werkzaamheden in de kliniek betrokken bij wetenschappelijk onderzoek. Van januari 2019 tot en met maart 2020 heeft zij voltijd aan haar promotietraject gewerkt. Sinds april 2020 is zij werkzaam als arts-assistent niet in opleiding tot specialist op de afdeling Gynaecologie en Obstetrie van het Elisabeth Tweesteden Ziekenhuis te Tilburg.

In 2007 ontmoette zij haar partner Bart Kaanders met wie zij in 2019 getrouwd is.